

Exploring Muscle Structure, Function, and Gait Patterns in People with Distal Hereditary Motor Neuropathy: Natural History and the Effect of Rehabilitation Interventions

By: Aljwhara Ahmed N. Alangary

Supervised by: Dr. Gita Ramdharry, Dr. Jasper Morrow, Dr. Matilde Laura

PhD Thesis

Submitted in partial fulfilment of the requirements for the Doctor of Philosophy in Neuromuscular Disease

School of Life and Medical Sciences Faculty of Brain Sciences UCL Queen Square Institute of Neurology Department of Neuromuscular Diseases University College London London. September 2024



Declaration

I, Aljwhara Alangary, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Abstract

Background Distal Hereditary Motor Neuropathy (DHMN) is a rare heterogenous inherited neuromuscular disorder. It is characterised by distal progressive weakness.

Objectives This thesis provides preliminary longitudinal data to describe the natural history of DHMN in terms of muscle structure, muscle strength, and gait parameters, also to investigate the effect of commonly used rehabilitation interventions.

Methods DHMN adult participants underwent the following measures: MRI scans of the foot, calf, and thigh muscles, isokinetic and isometric strength measures of the lower limb using dynamometer, 3D motion analysis to capture kinetic and kinematic data of walking gait. For direct comparison, matched health controls underwent the same measures. Measures were repeated after 6 and 12 months to explore the natural history of the disease. DHMN participants underwent additional gait analysis wearing bilateral carbon fibre ankle foot orthoses to explore the effect on gait. Eligible DHMN participants were prescribed a home based resistance training program, and the response to training was analysed by the same measures after 6 months of training.

Results The study identified significant progressive muscle atrophy and increased intramuscular fat accumulation at the calf in DHMN participants, with a notable decline in muscle strength over time and altered gait mechanics. The use of ankle-foot orthoses showed improvements in gait stability, while the resistance training program indicated potential benefits in maintaining muscle function, but adherence was a key challenge.

Conclusion The preliminary data from this study provide valuable insights into the natural history of DHMN, highlighting the progressive nature of muscle degeneration and functional decline. These findings offer useful guidance for health practitioners in managing DHMN and emphasize the need for targeted rehabilitation interventions to improve patient outcomes. Future research should focus on longer-term studies with larger cohorts to validate these findings and further explore effective management strategies.

Impact Statement

Distal hereditary motor neuropathy (DHMN) is one of the rarest inherited diseases. It causes gradual muscle weakness and wasting starting from the feet, lower leg, then the thigh. This weakness can negatively influence walking abilities in the affected people, having an impact on their activities of daily living and their quality of life. The causes of DHMN are still under investigation, and it is incurable. Despite lacking enough supporting evidence, rehabilitation options including exercises and orthotics are commonly used to improve walking in people with DHMN.

This study is the first to explore DHMN effect on muscle tissue over 12 months, to provide a detailed understanding on the pattern of muscle involvement and how this affects function in terms of muscle strength and walking. In addition, the effect of strengthening exercises and ankle foot orthoses, as a commonly prescribed rehabilitation options, were explored.

This thesis used the following: MRI scans, dynamometry, and 3D motion analysis, and identified muscle involvement patterns and alteration in walking gait in people with DHMN compared to controls. Changes over 12 months were best detectable by MRI scans. These findings can guide the diagnosis process clinically and inform future research on outcome measures suitable for use in DHMN research.

The outcomes from intervening with exercise and ankle foot orthoses can inform clinical decision making, and appropriate prescription, according to individual needs. It can also inform future exercises trials in designing an adherable exercise protocol.

The study design and findings were presented at a number of scientific events and discussed with experts in this field. The work took place in a single site in the UK, and dissemination of this study will allow future, multisite research collaboration in DHMN as a rare and under researched condition.

Acknowledgments

My sincere acknowledgment goes to my supervisors Dr. Gita Ramdharry, Dr. Jasper Morrow, and Dr. Matilde Laura for their invaluable guidance and advice throughout this project. Their expertise and insights have been crucial in shaping my research. I extend my gratitude to the Royal Embassy of Saudi Arabia Cultural Bureau in the UK & Ireland and King Saud University, Riyadh, Saudi Arabia for the generous scholarship that funded my PhD journey, enabling me to pursue my academic goals with dedication and ambition.

I would also like to express my deep appreciation to all the study participants who showed remarkable interest and commitment. Their willingness to participate and contribute to this research has been instrumental in the success of this project.

To my colleagues at the Institute of Neurology and at the National Hospital for Neurology and Neurosurgery who have been part of my journey, your support has shaped this experience into not only an academic endeavour but a period of personal growth and discovery. Thank you all for being my inspiration and for believing in me.

A heartfelt thank you to my parents, Dr. Ahmed Alangary and Mrs. Hayat Almangour, my family and friends, for their unwavering support and encouragement. Their belief in my abilities and their constant reassurance have been a source of strength and motivation during the challenging moments of my PhD journey.

A special thanks to my husband, Abdulrahman Bin Muammar, whose support has been my anchor through this journey. His belief in me, endless patience, and encouragement have been the light guiding me through the demanding times. His presence and support have been a reminder of the strength we can find in our loved ones, making this achievement not just mine but ours.

IV

Table of Contents

Declaration	I
Abstract	II
Impact Statement	.III
Acknowledgments	.IV
List of Tables	X
List of FiguresX	IVI
List of Abbreviations	XII
Chapter 1: Introduction	1
Chapter 2: Literature Review	7
2.1. Distal Hereditary Motor Neuropathy (DHMN) Overview	7
2.2. Muscle MRI in Hereditary Peripheral Neuropathy	.10
2.2.1. Muscle MRI Physics	.10
2.2.2. Intramuscular Fat Quantification	.13
2.2.3. Muscle MRI As an Outcome Measure in Hereditary Peripheral Neuropathy	.15
2.3. Dynamometry and Muscle Strength Assessment in Hereditary Peripheral Neuropath	у 23
2.4 Gait and Walking in Hereditary Peripheral Neuropathy	28
2.4.1 Definitions and Terminology	.28
2.4.2. The Gait Cycle	.29
2.4.3. Gait Mechanics and Locomotor Functions	.30
2.4.4. Methods of Gait Assessment	.36
2.4.5. The Normal Gait in Three-Dimensional Motion Analysis	.38
2.4.6. Altered Gait in Hereditary Peripheral Neuropathy	.46
2.5. Current Management and Rehabilitation Methods in Hereditary Peripheral Neuropat	hy .55
2.5.1. Ankle Foot Orthoses (AFOs)	.57
2.5.2. Therapeutic Exercise	.63
Chapter 3: Study Objectives	.71
Chapter 4: Methods	.72
4.1. MRI Methods	.73
4.1.1. MRI Protocol	.73
4.1.2. Hardware	.73
4.1.3. Positioning	.73
4.1.4. Choice of Anatomical Coverage	.74

4.1.5. MRI Sequences	75
4.1.6. Qualitative Image Analysis	77
4.1.7. Quantitative Image Analysis	
4.2. Dynamometry Methods	81
4.2.1. Hardware	81
4.2.2. Parameters Measured	
4.2.3. Dynamometry Protocol	
4.3. Three-Dimensional Motion Analysis	85
4.3.1. Kinematics	
4.3.2. Kinetics	
4.3.3. Lab Settings and Equipment	
4.3.4. Gait Parameters and Analysis	
4.4. Clinical Assessment	
4.4.1. CMT Examination Score Version 2 (CMTESv2)	
4.4.2. Foot Posture Index (FPI-6)	93
4.4.3. Manual Muscle Testing (MMT)	93
4.4.4. The Visual Analog Scale (VAS) of Pain	
4.4.5. Walk-12 Questionnaire	94
4.5. Participants	94
4.6. Ethical Framework	96
4.7. Trial Design and Procedure	
4.8. Overall Statistical Considerations	96
4.8.1. Sample	96
4.8.2. Statistical Analysis	
4.8.3. Difference Between Right and Left Sides	
4.8.4. Data Normality Test	
4.8.5. Strategies to Minimise Statistical Errors	
Chapter 5: Pattern of Involvement and Deviation from "Normal" in Mu	ascle Structure
and Function in Distal Hereditary Motor Neuropathy (DHMIN)	
5.1. Introduction	
5.2. Methods	
5.2.1. Data Analysis	
5.3. Kesults	
5.3.1. Subjects and Clinical Assessment	
5.3.2. Pattern of Muscle Structural Involvement on MRI	

5.3.3. Difference in Muscles Isokinetic and Isometric Strength	112
5.3.4. Deviations From Normal in Walking Gait	113
5.4. Discussion	119
5.4.1. Pattern of Muscle Involvement	119
5.4.2. Deviation from Normal in Walking Gait	121
5.5. Summary	
Chapter 6: The Natural History of Muscle Structure and Function in Distal He Motor Neuropathy (DHMN)	ereditary 125
6.1. Introduction	125
6.2. Methods	127
6.2.1. Data Analysis	127
6.3. Results	129
6.3.1. Subjects and Clinical Assessment	129
6.3.2. Longitudinal Change in Intramuscular Fat on MRI	130
6.3.3. Longitudinal Change in Muscle Area on MRI	133
6.3.4. Longitudinal Change in Intramuscular Water on MRI	134
6.3.5. Longitudinal Change in Strength Measurements	135
6.3.6. Longitudinal Change in Gait Parameters	138
6.3.7. Responsiveness Over 12 Months	143
6.4. Discussion	144
6.5. Summary	148
Chapter 7: The Relationships Between Intramuscular Fat Fraction, Muscle Vo Muscle Strength, and Moments/Power Generation in Distal Hereditary Motor Neuropathy (DHMN)	lume, 149
7.1. Introduction	
7.2. Methods	
7.2.1. Data Analysis	
7.3. Results	
7.3.1. Subjects and Clinical Assessment	
7.3.2. Relationships Between MRI Parameters	
7.3.3. Relationships Between MRI and Dynamometry Parameters	
7.3.4. Relationships Between MRI and Gait Parameters	
7.3.5. Relationships Between MRI Parameters and Clinical Measurements	171
7.3.6. Relationships Between Dynamometry and Gait Parameters	172
7.3.7. Relationships Between Dynamometry Parameters and Clinical Measurem	nents176
7.3.8. Relationships Between Gait Parameters and Clinical Measurements	

7.4. Discussion	
7.4.1. Relationships Between Intramuscular Fat Fraction, Muscle Volum Isometric Muscle Strength, and Kinetics of Gait	ne, Isokinetic and
7.4.2. Correlations with Clinical Measures	
7.4.3. Quantitative and Qualitative Measurements	
7.5. Summary	
Chapter 8: The Effect of Bilateral Carbon Fibre Ankle Foot Orthoses (A Kinetics and Kinematics of Gait of People with Distal Hereditary Motor (DHMN)	FOs) on the Neuropathy 185
8.1. Introduction	
8.2. Methods	
8.2.1. Data analysis	
8.3. Results	
8.3.1. Subjects and Clinical Assessment	
8.3.2. The Effect of Carbon Fibre Ankle Foot Orthoses	
8.4. Discussion	
8.5. Summary	
Chapter 9: The Effect of Resistance Training on Muscle Structure, Fund Patterns in Distal Hereditary Motor Neuropathy (DHMN)	ction, and Gait
9.1. Introduction	
9.2. Methods	
9.2.1. Data Analysis	
9.3. Results	
9.3.1. Adherence to Exercises	
9.3.2. Study Participants and Clinical Assessment	
9.3.3. Effect of Exercises on Muscle Structure (MRI Parameters)	
9.3.4. Effect of Exercise on Function	
9.4. Discussion	
9.4.1. Effectivness and Safity of Resistive Exercises	
9.4.2. Factors to Improve Adherence for Successful Exercise Trials	
9.5. Summary	
Chapter 10: Overview	
10.1. Muscle Structure and MRI as an Outcome Measure in DHMN	238
10.2. Muscle Function and Weakness Patterns in DHMN	
10.3. Walking Gait in DHMN	

10.4. Relationships Between Intramuscular Fat Fraction, Muscle Volume, Isokinetic Isometric Muscle Strength, and Kinetics of Gait	and 241
10.5. The Effect of Rehabilitation Strategies in Management of DHMN	243
10.5.1. Effect of Carbon Fibre AFOs on Gait Parameters in DHMN	243
10.5.2. Effectiveness and Safety of Resistive Exercises	244
10.6. Study Limitations and Recommendations for Future Work	245
10.6.1. Impact of COVID-19 Pandemic on the Study	246
10.6.2. Methodological Challenges	247
10.6.3. Research and Clinical Implications	248
Chapter 11: Conclusion	252
References	254
Appendix I: Difference Between Right and Left Side in The Primary Parameters and Application of the Modified Bonferroni Correction	267
Appendix II: Friedman Test Manual Calculation	272
Appendix III: Correlograms Showing the Correlation and Significance Level for All Parameters Comparisons in Chapter 7	277
Appendix IV: Exercise Induced Pain Management	282
Appendix V: Exercise Program	285
Appendix VI : Kinematic and Kinetic Gait Measurements of the Exercising Participant	s289
Appendix VII: Participants Study Allocation and Visits Activities	293
Appendix VIII: Published Abstracts Related to This PhD	294

List of Tables

Table 1: Types and clinical features of Distal Hereditary Motor Neuropathies (DHMN)9
Table 2: Tissue appearance in T1, T2, and STIR MRI scans. 12
Table 3: Pattern of muscle involvement in DHMN cohort
Table 4: Normal gait cycle, Right side. 29
Table 5: Summary of the objectives and joints positions in each gait period
Table 6: Time parameters to describe the gait cycle
Table 7: Linear parameters to describe the foot placement on the ground during a full gait
cycle
Table 8: Sample axial images of unprocessed Dixon sequence (echo time= 3.45 ms) with
overlaid muscle regions of interest of the left side for the same image in a healthy control,
mild DHMN case, and severe DHMN case
Table 9: Dynamometry protocol. 84
Table 10: Derivation of embedded vector basis (EVBs) for each segment
Table 11: Relation of distal segment Euler angles to a reference frame. 88
Table 12: Gait parameters used for primary and secondary analysis.
Table 13: Inclusion and exclusion criteria. 95
Table 14: Visit activity for each participant. 96
Table 15: Summary of demographics and clinical assessment of DHMN and control groups
Table 16: Muscle cross sectional area at the thigh, calf, and foot level of DHMN and control
groups
Table 17: Muscle Fat Fraction at the thigh, calf, and foot level of DHMN and control groups

Table 18: Remaining Muscle Area at the thigh, calf, and foot level of DHMN and control
groups108
Table 19: MRI Muscle Fat qualitative score using Modified Mercuri's Scale at the calf and
foot level of DHMN and control groups109
Table 20: MRI Muscle water quantitative analysis using T2 mapping at the calf and foot level
of DHMN and control groups
Table 21: MRI Muscle water qualitative analysis using STIR image at the calf and foot level
of DHMN and control groups
Table 22: Isometric and isokinetic dynamometry of DHMN and control groups 112
Table 23: Gait spatiotemporal parameters of DHMN and control groups
Table 24: Gait kinematics parameters of DHMN and control groups
Table 25: Gait kinetics parameters of DHMN and control groups 116
Table 26: Summary of pattern in gait and muscle involvement in DHMN
Table 27: : Summary of demographics and clinical assessment of DHMN at baseline, 6
months, and 12 months
Table 28: Longitudinal MRI mean muscle fat fraction of DHMN and repeated measures
analysis results
Table 29: Longitudinal MRI median muscle fat qualitative score using Modified Mercuri's
Scale of DHMN and repeated measures analysis results
Table 30: Longitudinal MRI muscle cross sectional area of DHMN and repeated measures
analysis results
Table 31: Longitudinal MRI remaining muscle area of DHMN and repeated measures
analysis results
Table 32: Longitudinal quantitative T_{2m} MRI muscle water of DHMN and repeated measures
analysis results

Table 33: Longitudinal qualitative STIR MRI muscle water of DHMN and repeated measures
analysis results
Table 34:Longitudinal manual muscle testing of DHMN and repeated measures analysis
results
Table 35: Longitudinal dynamometry of DHMN and repeated measures analysis results136
Table 36: Longitudinal spatiotemporal parameters of DHMN and repeated measures analysis
results
Table 37: Longitudinal kinematics parameters of DHMN and repeated measures analysis
results
Table 38: Longitudinal kinetics parameters of DHMN and repeated measures analysis results.
Table 39: Responsiveness of the study outcome measures with significant difference over 12
months143
Table 40: Summary of longitudinal changes in DHMN cohort
Table 41: Primary and secondary correlation tests. 152
Table 42:Summary of demographics and clinical assessment of DHMN participants at
baseline and 12 months
Table 43: Correlation between baseline fat fraction (FF) and baseline T _{2m} 156
Table 44: Correlation between baseline fat fraction (FF) and Mercuri score159
Table 45: Correlation between baseline intramuscular water T_{2m} and STIR hyperintensity
score
Table 46: Correlation between baseline clinical measurements and baseline fat fraction171
Table 47: Correlation between baseline clinical measurements and baseline dynamometry
parameters176

Table 48: Correlation between isometric dynamometry (X) and manual muscle testing (Y).
Table 49: Correlation between isokinetic dynamometry (X) and manual muscle testing (Y).
Table 50: Correlation between baseline clinical measurements and baseline spatiotemporal
parameters178
Table 51:Summary of positive and negative correlation between study parameters
Table 52: Summary of demographics and clinical assessment of DHMN participants189
Table 53: Strength assessment of DHMN participants
Table 54: Spatiotemporal data, difference, and effect size of walking with and without AFOs
in DHMN191
Table 55: Gait kinematic data, difference, and effect size of walking with and without AFOs
in DHMN191
Table 56: Gait kinetics data, difference, and effect size of walking with and without AFOs in
DHMN
Table 57: Summary of the effect of AFO on DHMN gait. 198
Table 58: Factors that influenced adherence to exercise. 204
Table 59: Summary of the subjects' demographics and clinical assessment at baseline by
group, exercises (n=5) and no exercise (n=5)205
Table 60: Summary of the subjects' demographics and clinical assessment at baseline for
exercise case 1, case 2, and no exercises group (n=8)206
Table 61: Longitudinal fat fraction and differences between measurement points in exercise
group $(n=5)$ and no exercise group $(n=5)$
Table 62: Longitudinal cross sectional area and differences between measurement points in
exercise group (n=5) and no exercise group (n=5)

Table 63: Longitudinal remaining muscle area and differences between measurement points
in exercise group (n=5) and no exercise group (n=5)210
Table 64: Longitudinal muscle oedema and differences between measurement points in
exercise group (n=5) and no exercise group (n=5)
Table 65: Longitudinal fat fraction and differences between measurement points in case 1,
case 2, and no exercise group (n=8)
Table 66: Longitudinal cross sectional area and differences between measurement points in
case 1, case 2, and no exercise group (n=8)214
Table 67: Longitudinal remaining muscle area and differences between measurement points
in case 1, case 2, and no exercise group (n=8)215
Table 68: Longitudinal muscle oedema and differences between measurement points in case
1, case 2, and no exercise group (n=8)215
Table 69: Longitudinal clinical assessment in exercise group (n=5) and no exercise group
(n=5)
Table 70: Longitudinal clinical assessment in exercise case 1, case 2, and no exercises group
(n=8)
Table 71: Longitudinal isomeric dynamometry and differences between measurement points
in exercise group (n=5) and no exercise group (n=5)
Table 72: Longitudinal isokinetic dynamometry and differences between measurement points
in exercise group (n=5) and no exercise group (n=5)
Table 73: Longitudinal isomeric dynamometry and differences between measurement points
in case 1, case 2, and no exercise group (n=8)224
Table 74: Longitudinal isokinetic dynamometry and differences between measurement points
in case 1, case 2, and no exercise group (n=8)

Table 75: Longitudinal spatiotemporal gait parameters and differences between
measurement points in exercise group $(n=5)$ and no exercise group $(n=5)$
Table 76: Longitudinal spatiotemporal gait parameters and differences between measurement
points in case 1, case 2, and no exercise group (n=8)
Table 77: Applying the Behaviour Change Wheel Model to DHMN Exercise Adherence236
Table 78:Correlation between baseline plantar flexors measurements
Table 79:Correlation between baseline dorsiflexors measurements
Table 80:Correlation between baseline knee extensors measurements. 279
Table 81: Correlation between baseline knee flexors measurements. 280
Table 82:Correlation between baseline clinical measurements. 281
Table 83: Correlation between baseline strength measurements and spatiotemporal
parameters
Table 84: Longitudinal kinematic gait parameters and differences between measurement
points in exercise group (n=5) and no exercise group (n=5)
Table 85: Longitudinal kinematic gait parameters and differences between measurement
points in case 1, case 2, and no exercise group
Table 86: Longitudinal kinetic gait parameters and differences between measurement points
in exercise group (n=5) and no exercise group (n=5)
Table 87: Longitudinal kinetic gait parameters and differences between measurement points
in case 1, case 2, and no exercise group

List of Figures

Figure 1: Thesis structure and chapters content
Figure 2: Typical activity of major muscle groups during the gait cycle
Figure 3: Sagittal plane joint angles (degrees) during a single gait cycle of right hip (flexion
positive), knee (flexion positive) and ankle (dorsiflexion positive)41
Figure 4: Ground reaction forces. Lateral, Anteroposterior (fore-aft), and vertical
components of the ground reaction force, in newtons, for right foot (solid line) and left foot
(dashed line)43
Figure 5: Sagittal plane internal joint moments (newton-meters per kilogram body mass)
during a single gait cycle of right hip (extensor moment positive), knee (extensor moment
positive) and ankle (plantar flexor moment positive)44
Figure 6: Sagittal plane joint powers (watts per kilogram body mass) during a single gait
cycle of right hip, knee, and ankle. Power generation is positive, absorption is negative46
Figure 7: Flowchart outlining the thesis objectives and research questions
Figure 8: Summary of the measurements' methods used in each chapter. AFOs, Ankle Foot
Orthoses72
Figure 9: HUMAC NORM Testing & Rehabilitation System (CSMi, Massachusetts, USA).82
Figure 10: Euler angles are quantified as ordered rotations about 3 axes and represent joint
motion displacements in the anatomical body planes
Figure 11:Segment Embedded Coordinates in sagittal (A) and frontal (B) view. Adapted from
CODAmotion ODIN user guide
Figure 12: Helen Hayes model for marker placement. Adapted from CODAmotion ODIN
user guide
Figure 13: Box and whisker plot of the total calf fat fraction (%) in DHMN and control
group

Figure 14:Pattern of fat infiltration using Modified Mercuri's Scale at the calf and foot level
of DHMN group109
Figure 15:Grand average angular displacement in the sagittal plane over one gait cycle for the
ankle, knee, and hip (average of left and right sides). Comparison of people with DHMN and
controls
Figure 16:Grand average joint moments in the sagittal plane over one gait cycle for the ankle,
knee and hip (average of left and right sides)117
Figure 17:Grand average joint powers in the sagittal plane over one gait cycle for the ankle,
knee and hip (average of left and right sides)118
Figure 18: Longitudinal change of intramuscular fat in DHMN
Figure 19: Longitudinal change of muscle strength in DHMN
Figure 20:Grand average angular displacement in the sagittal plane over one gait cycle for the
ankle, knee, and hip (average of left and right sides). Comparison of longitudinal
measurements of people with DHMN
Figure 21: Grand average joint moments in the sagittal plane over one gait cycle for the
ankle, knee and hip (average of left and right sides). Comparison of longitudinal
measurements of people with DHMN
Figure 22:Grand average joint powers in the sagittal plane over one gait cycle for the ankle,
knee and hip (average of left and right sides). Comparison of longitudinal measurements of
people with DHMN142
Figure 23: Correlation between plantar flexors fat fraction (%) and remaining muscle area
(mm ²)
Figure 24: Correlation between plantar flexors remaining muscle area (mm ²) and cross
sectional area (mm ²)

Figure 25: Correlation between dorsiflexors (DF) fat fraction (%) and remaining muscle area
(RMA) and cross sectional area (CSA) (mm ²)155
Figure 26: Correlation between dorsiflexors remaining muscle area (RMA) and cross
sectional area (CSA) (mm ²)
Figure 27: Correlation between total calf fat fraction (%) and T_{2m} intramuscular water (ms).
Figure 28: Correlation between knee extensors (EX) fat fraction (%) and remaining muscle
area (RMA) and cross sectional area (CSA) (mm ²)157
Figure 29: Correlation between knee extensors (EX) remaining muscle area (RMA) and cross
sectional area (CSA) (mm ²)157
Figure 30: Correlation between knee flexors (FLX) fat fraction (%) and remaining muscle
area (RMA) and cross sectional area (CSA) (mm ²)158
Figure 31: Correlation between knee flexors (FLX) remaining muscle area (RMA) and cross
sectional area (CSA) (mm ²)158
Figure 32: Correlation between calf fat fraction (%) and Mercuri score
Figure 33: Correlation between plantar flexors fat fraction (%) and Mercuri score
Figure 34: Correlation between dorsiflexors fat fraction (%) and Mercuri score160
Figure 35: Correlation between dorsiflexors STIR score and T_{2m} intramuscular water (ms).
Figure 36: Correlation between Plantar flexion (PF) isometric dynamometry (Nm) and
remaining muscle area (RMA) and cross sectional area (CSA) (mm ²)162
Figure 37: Correlation between dorsiflexion isometric dynamometry (Nm) and fat fraction
(%)163
Figure 38: Correlation between dorsiflexion (DF) isometric dynamometry (Nm) and
remaining muscle area (RMA) and cross sectional area (CSA) (mm ²)163

Figure 39: Correlation between knee extensors (EX) cross sectional area (CSA)(mm ²) and
isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm)164
Figure 40: Correlation between knee extensors (EX) remaining muscle area (RMA)(mm ²)
and isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm)164
Figure 41: Correlation between knee flexion (FLX) isometric dynamometry (Nm) and
remaining muscle area (RMA) (mm ²), and cross sectional area (CSA) (mm ²)165
Figure 42: Correlation between knee flexion (FLX) isometric dynamometry (Nm) and fat
fraction (FF)(%)
Figure 43: Correlation between knee flexion (FLX) isokinetic dynamometry (Nm) and
remaining muscle area (RMA) (mm ²), and cross sectional area (CSA) (mm ²)166
Figure 44: Correlation between knee flexion (FLX) isokinetic dynamometry (Nm) and fat
fraction (FF)(%)
Figure 45: Correlation between dorsiflexors fat fraction (%) and ankle power generation
during swing (W/Kg)167
Figure 46: Correlation between knee extensor cross sectional area (mm ²) and knee extensor
moment (Nm/Kg)and knee power generation and absorption (W/Kg)168
Figure 47: Correlation between knee extensor remaining muscle area (mm ²) and knee
extensor moment (Nm/Kg)and knee power generation and absorption (W/Kg)168
Figure 48: Correlation between knee power absorption (W/Kg) and knee flexors (FLX) cross
sectional area (CSA) (mm ²), and remaining muscle area (RMA)(mm ²)169
Figure 49: Correlation between knee extensors moment (Nm/Kg) and knee flexors (FLX)
cross sectional area (CSA)(mm ²), and remaining muscle area (RMA)(mm ²)170
Figure 50: Correlation between knee extensors moment (Nm/Kg) and knee flexors (FLX) fat
fraction (%)170

Figure 51: Correlation between the Charcot Marie Tooth Examination Score (CMTES) and
fat fraction (FF) (%) at the thigh and calf level
Figure 52: Correlation between plantar flexion moments (Nm/Kg)and plantar flexion
isokinetic dynamometry (Nm)
Figure 53: Correlation between dorsiflexors isometric dynamometry (Nm) and plantar flexion
moments (Nm/Kg)172
Figure 54: Correlation between knee extensors (EX) moment (Nm/Kg) and isometric (ISOM)
and isokinetic (ISOK) dynamometry (Nm)173
Figure 55: Correlation between knee power absorption in stance (W/Kg) and knee extensors
(EX) isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm)173
Figure 56: Correlation between knee power absorption (W/Kg) and knee flexors (FL)
isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm)174
Figure 57: Correlation between knee extensors moment (Nm/Kg) and knee flexors (FL)
isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm)174
Figure 58: Correlation between isometric dorsiflexion (Nm) and gait spatiotemporal
parameters175
Figure 59: Correlation between the Charcot Marie Tooth Examination Score (CMTES) and
dynamometry parameters176
Figure 60: Correlation between the Charcot Marie Tooth Examination Score (CMTES) and
gait spatiotemporal parameters
Figure 61: Correlation between Walk-12 questionnaire and gait spatiotemporal parameters.
Figure 62: Matrix Max Carbon fibre Ankle foot Orthosis (Trulife, UK)

Figure 63:Grand average angular displacement in the sagittal plane over one gait cycle for the
ankle, knee, and hip (average of left and right sides). Comparison of people with DHMN
walking with and without AFO193
Figure 64:Grand average joint moments in the sagittal plane over one gait cycle for the ankle,
knee and hip (average of left and right sides). Comparison of people with DHMN walking
with and without AFO
Figure 65: Grand average joint powers in the sagittal plane over one gait cycle for the ankle,
knee and hip (average of left and right sides). Comparison of people with DHMN walking
with and without AFO195
Figure 66: Longitudinal MRI parameters in case 1, case 2, and no exercise group
Figure 67: Longitudinal isometric and isokinetic strength in case 1, case 2, and no exercises
group
Figure 68: Longitudinal spatiotemporal parameters in case 1, case 2, and no exercises group.
Figure 69: The Behaviour Change Wheel (BCW)

List of Abbreviations

μm	Micrometres
3D	Three Dimensional
AD	Autosomal Dominant
AFO	Ankle Foot Orthoses
AR	Autosomal Recessive
ATP7A	Copper-Transporting Atpase 1
B0	Static main field
B1	Radiofrequency magnetic fields
BAG3	Bcl2-Associated Athanogene 3
BICD2	BICD cargo adaptor 2
BSCL2	Berardinellie-Seip Congenital Lipodystrophy Type 2
cm/s	centimetre per second
CMT	Charcot Marie Tooth
CMTESv2	CMT Examination Score Version 2
COM	Centre of mass
COQ7	Coenzyme Q7
CSA	Cross sectional area
DCTN1	P150 Subunit of Dynactin
DHMN	Distal Hereditary Motor Neuropathy
DYNC1H1	Cytoplasmic Dynein Heavy Chain 1
EMG	Electromyography
EVB	Embedded victor bases
F	Fat
FA	Feet adjacent
FF	Fat Fraction
FOV	Field of view
FPI-6	Foot Posture Index
FSE	Fast spin-echo
GARS	Glycyl-tRNA synthetase
GCS	Global coordinate system
GDI1	GDP dissociation inhibitor 1
GRE	Gradient-echo
GRF	Ground Reaction Force
HARS1	Histidyl-tRNA synthetase 1
HR	Heel rise
HRA	Health Research Authority
HSPB1	Heat-Shock Protein B1
HSPB3	Heat-Shock Protein B3
HSPB8	Heat-Shock Protein B8
Hz	Hertz
IBM	Inclusion body myositis
IC	Initial contact
IGHMBP2	Immunoglobulin M Binding Protein 2
iPat	Integrated Parallel Acquisition Techniques
ITE	Inter-echo time
J/kg	Joule Per Kilogram
m	meter

m/s	meter per second
MAX GET	Maximum Gravity Effect Torque
mm	millimetre
MMT	Manual Muscle Testing
mNCV	Motor Nerve Conduction Velocity
MRC	Medical Research Council
MRI	Magnetic Resonance Imaging
MTR	Magnetization transfer ratio
MUAP	Motor Unit Action Potentials
n	Number of cases
Ν	Newton
NEX	Number of excitations
NHNN	National Hospital for Neurology and Neurosurgery
Nm	Newton meter
Nm/Kg	Newton Meter per Kilogram
NMD	Neuromuscular diseases
OI	Opposite initial contact
OT	Opposite toe off
RNA	Ribonucleic Acid
ROI	Region of interest
ROM	Range Of Motion
S	second
SE	Spin-echo
SETX	Senataxin
SIGMAR1	Sigma Non-Opioid Intracellular Receptor 1
SORD	Sorbitol Dehydrogenase
STIR	Short-TI inversion recovery
T1	Longitudinal relaxation time
T2	Transverse relaxation time
TE	Echo Time
ТО	Toe off
TR	Repetition Time
TRPV4	Transient Receptor Vallanoid 4 Gene
TRPV4	Transient receptor vallanoid 4 gene
TSE	Turbo-spin-echo
TV	Tibia vertical
UCL	University College London
UCLH	University College London Hospitals
UMN	Upper Motor Neuron
VAS	The Visual Analog Scale
VO2	Oxygen uptake
VRK1	VRK serine/threonine kinase 1
W	Water
W/kg	Watts Per Kilogram

Chapter 1: Introduction

Distal Hereditary Motor Neuropathies (DHMN), also known as distal spinal muscular atrophies, are a heterogeneous group of neural diseases that predominantly affect the lower motor neuron (Rossor et al., 2017, Beetz et al., 2012). This group of diseases is characterised by length dependent, slowly progressive weakness and atrophy of distal muscles in the upper and lower limbs (Greenbaum et al., 2020, Blumen et al., 2012). Distal Hereditary Motor Neuropathies (DHMN) lay under the Charcot-Marie-Tooth disease (CMT) umbrella due to the clinical and genetic features overlap. However, unlike CMT, sensory involvement and foot deformities are infrequent with DHMN. On neurophysiology, the sensory responses are usually preserved or mildly affected, while electromyography (EMG) shows signs of muscle denervation in the affected muscle groups (Barwick et al., 2012). It usually starts in the first two to four decades of life (Rossor et al., 2017) and symptoms are initially associated with poor motor functioning. People report poor performance in sports at school (Rossor et al., 2012b), and early signs of plantar flexor weakness including difficulties with standing balance and standing on toes (Rossor et al., 2017).

Distal Hereditary Motor Neuropathies (DHMN) is a clinically and genetically heterogeneous disorder associated with mutations in a number of genes (Sumner et al., 2013). The prevalence of DHMN is 2.14 affected individuals per 100,000 inhabitants (95% confidence interval 1.62–2.66) in the North of England (Bansagi et al., 2017). Recent evolution in genetic testing improved diagnostic rates, and a genetic diagnosis is usually reached in 48.9% of DHMN cases (Record et al., 2024). These mutations affect cellular functions, for example: protein misfolding (HSPB1, HSPB8, BSCL2), RNA metabolism (IGHMBP2, SETX, GARS), axonal transport (HSPB1, DYNC1H1, DCTN1) and cation-channel dysfunction (ATP7A and TRPV4) (Rossor et al., 2012a).

Previous exploration of muscle function and structure in Charcot Marie Tooth disease (CMT) gives us the tools to understand more about the presentation and impact of DHMN. MRI imaging has revealed the type and extent of muscle atrophy in CMT and has been proven to be sensitive to change in CMT1A (Morrow et al., 2016). Muscle function has been explored in CMT type 1A using dynamometer and can be applied to DHMN (Morrow et al., 2016). So, there is potential to use MRI and dynamometer in the same way to also help us understand how muscles are affected in people with DHMN. Alongside muscle function, muscle performance during walking can be better understood using threedimensional (3D) motion analysis (Lee et al., 2013). Qualitative and quantitative methods have been used to describe different gait patterns in similar neurological conditions (Ounpuu et al., 2013, Don et al., 2007, Newman et al., 2007). Threedimensional (3D) motion analysis can be applied to DHMN to observe joint motion and force during walking.

Various rehabilitation interventions have been considered to improve gait and muscle function among people with peripheral neuropathies, including orthotics (Õunpuu et al., 2021, Dufek et al., 2014), and therapeutic exercises (Djordjevic et al., 2017, Burns et al., 2017). Previous studies, however, do not specifically focus on specific intervention outcomes for DHMN. As such, the current study seeks to consider observation and intervention outcomes for DHMN, specifically in the context of improving muscle function and gait for people living with this condition.

The current study will contribute to our understanding of the presentation and the natural history of the disease, providing information for health practitioners and clinical experts to guide management of DHMN. The outcomes of this research can potentially help improve patient care and outcomes in terms of mobility and management of muscle weakness. DHMN is a rarer condition than CMT and has a different pattern of involvement (Rossor et al., 2012b). This work has potential to ascertain if there is indeed a difference in muscle structure and progression rate. Also, it will investigate the effect of two common rehabilitation interventions. This will have implications for more focused disease specific rehabilitation in this type of neuropathy.

This PhD project provides an evaluation of the relationships between muscle structure, muscle function, and function within an activity for people with Distal Hereditary Motor Neuropathy. Over 12 months, muscle changes in DHMN were observed in terms of structure and function using three observational methods; MRI; dynamometry; and 3D motion analysis. In addition, the effect of a 24-weeks (6 months) exercise program on muscle structure and function in DHMN was measured by the same observational methods. To address walking gait directly in DHMN, gait patterns were compared with and without carbon fibre ankle foot orthoses (AFO) using 3D motion analysis.

An overview of DHMN and published literature related to those measurement techniques and rehabilitation methods are reviewed in chapter 2. Chapter 3 provides the main objectives of the PhD research project. Methods used to fulfil those objectives including assessments, participants, and overall statistical methods are provided in chapter 4. Chapter 5 explores the Pattern of Involvement and deviation from "normal" in muscle structure and function in Distal Hereditary Motor Neuropathy (DHMN). Chapter 6 explore the natural history of muscle structure and function in Distal Hereditary Motor Neuropathy (DHMN) over one year. Chapter 7 is dedicated to ascertaining relationships between intramuscular fat fraction and muscle volume (measured by MRI), isokinetic and isometric muscle strength (measured by dynamometry) and moments/power generation (measured by 3D motion analysis). Chapter 8 explore the effect of bilateral carbon fibre ankle foot orthoses (AFO) on the kinetics and kinematics data of gait of people with DHMN (measured by 3D motion analysis). Chapter 9 explore the effect of resistance training on muscle structure, function, and gait patterns.

Chapter 10 provides an overall discussion of the research project methods and outcomes. Chapter 11 concludes this thesis with a brief summary of the project and its findings and recommendations for future research. An illustration of the thesis structure and chapters content is shown in Figure 1.



Figure 1: Thesis structure and chapters content.

Chapter 2: Literature Review

2.1. Distal Hereditary Motor Neuropathy (DHMN) Overview

Distal Hereditary Motor Neuropathies (DHMN) is a clinically and genetically heterogenous hereditary neuropathy characterized by predominant or exclusive motor involvement and degeneration of lower motor neurons (Kang et al., 2020, Rossor et al., 2017). Affected people usually present with slowly progressive, adult-onset distal muscle weakness and wasting without sensory involvement and infrequently with skeletal deformity due to its late onset (Wu et al., 2022, Tanabe et al., 2018). Although DHMN mainly affects lower motor neurons, the clinical and genetical heterogenicity overlaps with axonal forms of Charcot Marie Tooth disease type 2 (CMT2). With more than 30 genes identified, DHMN can be due to inheritance of autosomal dominant, autosomal recessive or X-linked gene mutations (Wu et al., 2022, Kang et al., 2020, Greenbaum et al., 2020, Rossor et al., 2017).

Neurophysiology and neuropathology of sensory nerves in DHMN are usually normal. Peripheral nerve biopsy does not show abnormalities in hereditary motor neuropathy patients, muscle biopsy shows mainly denervation changes and autopsy studies show atrophy of spinal roots, loss of anterior horn cells and chromatolysis of the motor neurons (Ishihara et al., 2020, Rossor et al., 2017, Kuhlenbäumer et al., 2005). Neurophysiological tests usually show normal motor nerve conduction velocity (mNCV) with reduced motor responses, indicating axonal neuropathy. In some cases, mNCVs might be slightly reduced (35±60 m/s) due to the loss of large fibres. Sensory responses are usually normal (Ishihara et al., 2020, Kuhlenbäumer et al., 2005). Electromyography (EMG) reveals typical signs of denervation including high motor unit action potentials (MUAP), so called "neurogenic" potentials with an amplitude of over 10 mV, pathologic spontaneous activity, consisting of fibrillations, complex repetitive discharges, and positive sharp waves, are often observed which are most prominent in distal muscles (Ishihara et al., 2020, Kuhlenbäumer et al., 2005).

Distal Hereditary Motor Neuropathy (DHMN) has been classified into different subtypes (DHMN I – DHMN VII) (Harding, 1993). However, the more recent identification of genes causing DHMN has helped with the classification and differentiation from other neuropathies. Types and clinical features described in previous studies of DHMN are summarized in Table 1 (de Fuenmayor-Fernández de la Hoz et al., 2024, Jacquier et al., 2023, Pons et al., 2023, Wu et al., 2022, Frasquet et al., 2021, Liu et al., 2020, Ververis et al., 2019, Rossor et al., 2012a, Rossor et al., 2012b, Kuhlenbäumer et al., 2005).

DHMN type	Phenotype	Inheritance	Known gene	Age unable to walk	Life expectancy
DHMN I	Juvenile onset with distal wasting and weakness	AD	HSPB1, HSPB8, GARS, DYNC1H1	Rare	Normal
DHMN II	Adult onset with distal wasting and weakness	AD	HSPB1, HSPB8, BSCL2, HSPB3, BAG3	Rare	Normal
DHMN III	Slowly progressive wasting and weakness	AR	SORD	Rare	Normal
DHMN IV	Slowly progressive wasting and weakness with diaphragmatic paralysis	AR	Unknown	About 30 years	Unknown
DHMN V	Upper-limb predominance	AD	GARS BSCL2	Rare	Normal
DHMN VI	Spinal muscular atrophy with respiratory distress type 1	AR	IGHMBP2	Never	1 year
DHMN VII	Adult onset with vocal-cord paralysis	AD	DCTN1 TRPV4	Rare	Probably normal
X-linked	Distal onset wasting and weakness	X-linked	ATP7A	Rare	Probably normal
DHMN with UMN signs	DHMN and upper motor neuron signs	AD	SETX, BSCL2, COQ7	Rare	Probably normal
DHMN Jerash	DHMN and pyramidal signs originating in the Jerash region of Jordan	AR	SIGMAR1	Rare	Probably normal
Congenital distal spinal muscular atrophy	Distal weakness at birth and arthrogryposis	AD	TRPV4	Rare	Probably normal
AD- autosomal dominant AR- autosomal recessive ATP7A- conner transporting ATPase 1 BAG3- Bcl2 associated athanogene 3 BSCI 2-					

AD= autosomal dominant, AR= autosomal recessive, ATP7A= copper-transporting ATPase 1, BAG3= Bcl2-associated athanogene 3, BSCL2= Berardinellie-Seip congenital lipodystrophy type 2, COQ7= Coenzyme Q7, DCTN1= P150 subunit of dynactin, DYNC1H1= cytoplasmic dynein heavy chain 1, GARS= glycyl-tRNA synthetase, HSPB1= heat-shock protein B1, HSPB3= heat-shock protein B3, HSPB8= heat-shock protein B8, IGHMBP2= immunoglobulin m binding protein 2, TRPV4= transient receptor vallanoid 4 gene, SORD= Sorbitol dehydrogenase, SIGMAR1= sigma non-opioid intracellular receptor 1.

Table 1: Types and clinical features of Distal Hereditary Motor Neuropathies (DHMN).

2.2. Muscle MRI in Hereditary Peripheral Neuropathy

Muscle tissue is adversely affected by denervation in the same way despite the underlying cause of neuropathy. Intramuscular fat accumulation is observed with chronic muscle denervation and has been used as an indirect outcome measure in neuropathies to diagnose and monitor the progression of the disease, as well as to identify patterns of muscles denervation in different neuropathies. MRI is one of the imaging methods widely used in the medical field that allows safe and frequent use without exposure to ionizing radiation.

2.2.1. Muscle MRI Physics

Signals from the nuclei of hydrogen atoms in water and lipid molecules in human body originate in response to a radiofrequency pulse applied during MRI scan. Magnetic resonance causes changes in the physical properties of those nuclei. Understanding variant signal magnitude detected, due to altered physical properties of proton density, resonant frequency, and relaxation time, allows different MRI sequences to be applied, and acquisition of images with source of contrasts to differentiate tissues (Grimm et al., 2018, McRobbie et al., 2017).

T1 weighted sequence gives an image generated from signals during the Repetition Time (TR); the amount of time in between excitations by radiofrequency pulse. Tissues with shorter T1 value, for example the fat or blood, appears brighter than tissues with longer T1 values for example bones or the lungs (table 2) (Grimm et al., 2018, McRobbie et al., 2017). When the tissue is affected

by a pathology the T1 value can be affected as well. Therefore, it can be expected that conditions associated with oedema, or an increased number of capillaries will appear darker compared to the surrounding healthy tissue in MRI images, whereas lesions containing a high amount of fat will appear brighter due to their higher signal intensities (Table 2) (McRobbie et al., 2017).

T2 weighted sequences give an image generated from signals during the Echo Time (TE); the time following the spin excitation by radiofrequency pulse. Tissues with longer T2 value, for example blood, appears brighter than tissues with shorter T2 values such as bones (Grimm et al., 2018, McRobbie et al., 2017). T2 scans are commonly used in pathology as the collection of abnormal fluid appear bright (increased signal) against the darker normal surrounding tissue. For example, the accumulation of intramuscular water (muscle oedema) can be used as a sign of active denervation in peripheral neuropathy (Table 2) (McRobbie et al., 2017).

Fat signal magnitudes are very high in most of the MRI sequences due to its short relaxation time (T) and appears bright in T1 and T2 weighted images. Fat signal suppression found to be useful for diagnostic purposes. Examples of fatsuppression techniques include frequency-selective saturation pulse sequences, inversion recovery, and hybrid chemical shift-based techniques, such as the Dixon technique (Grimm et al., 2018). The most common fat suppression techniques are fat selective saturation and short-TI inversion recovery (STIR), both of which are based on the difference between the behaviour of water and that of fat in the MRI environment (Table 2). Since water and fat molecules have different resonant frequencies, this chemical shift allows the use of fat saturation where the saturation radiofrequency pulse has selective frequency centred on the main fat peak, meaning that the total signal will have a minimal fat contribution. Thus, the fat saturation technique can be applied to T1, T2, or proton density weighted images in spin-echo (SE), fast spin-echo (FSE), or gradient-echo (GRE) and is reliable for intravenous contrast-enhanced T1 imaging. However, it is prone to incomplete fat suppression, because of inhomogeneity of the calibrations of main field (B0) and radiofrequency (B1) magnetic fields, resulting in imperfect fat saturation (Grimm et al., 2018).

Tissue	T1 weighted	T2 weighted	STIR
Healthy muscle	Dark	Dark	Dark
Muscle oedema	Dark	Bright	Bright
Fat	Bright	Bright	Very Dark
Nerve	Dark	Dark	Dark
Injured nerve	Dark	Bright	Bright
Bone	Dark	Dark	Dark
Bone marrow	Bright	Bright	Dark

Table 2: Tissue appearance in T1, T2, and STIR MRI scans.

The Dixon technique, unlike other fat suppression features, allows the contribution of the fat signal to be suppressed in post-processing rather than during acquisition, as well as providing water and fat distribution maps (Lins et
al., 2020, Ma, 2008). Based on the chemical shift, the Dixon method requires knowledge of the calibrations of main field (B0) variations to properly separate water from fat signals. It allows four types of images to be obtained: fat-only, water-only, in-phase, and out-of-phase. Water-only images represent total fat suppression while fat-only images, visibly similar to T1-weighted images, allow the study of fat, and it is important to remember that in this case only the fat will have a high signal intensity (Lins et al., 2020, Ma, 2008). Fat Fraction (FF) can be derived from these water-only and fat-only images where FF = F/(W+F) (Chen et al., 2019), often expressed as a percentage: % FF = 100 * F/(W+F). Increased muscular FF reflecting intramuscular fat accumulation which is a common effect of denervation in peripheral neuropathies and has been shown to be a valuable biomarker in neuromuscular diseases (Morrow et al., 2018, Morrow et al., 2016, O'Donnell et al., 2022, Esteller et al., 2023).

2.2.2. Intramuscular Fat Quantification

Different semi-quantification scales to classify intramuscular fat infiltration were used in the literature. Goutallier scale is one of the most common scales used with different conditions involving muscle degeneration. It has 5 stages to describe areas of high signals on T1-weighted images (Kim et al., 2019).

Modified Mercuri's scale is another example of semi-quantification scale. It also describe areas of high signals on T1-weighted images but using 6 grading scores (Mercuri et al., 2003).

However, classifications with limited number of grades do not consider the floor effect found in early stages of slowly progressive disease like hereditary peripheral neuropathy. Alternatively, the 3-point Dixon method was found to be a more sensitive outcome measure to quantify intramuscular fat infiltration. In a study by Kim et al. (2019), they evaluated the potential value of 3D multiple gradient echo Dixon-based MRI sequence as a tool for thigh intramuscular fat quantification in people with Charcot Marie Tooth disease (CMT). They performed a prospective comparison study including 18 CMT participants and 18 matched controls. MRI using 3D multiple gradient echo Dixon-based imaging was performed for each subject. Region of interest analyses were performed at the upper and lower third of both thighs. The two-sample t-test or Wilcoxon rank sum test was used for intergroup comparison of the mean muscle fat fraction. Intraclass correlation coefficients were used to evaluate the interobserver agreement and test-retest reproducibility. Semi- quantitative analysis using the Goutallier classification (Grades 0–4) was done on T1-weighted images in upper thigh muscles. For Goutallier Grade 0 muscles, comparison of the mean intramuscular fat fraction between volunteers and CMT patients was performed. The interobserver agreements were excellent for all measurements (intraclass correlation coefficients > 0.8). Mean muscle fat fractions were significantly higher in all the measured muscles of CMT participants (P < 0.05) except in the adductor magnus in the upper thigh (P = 0.109). Goutallier Grade 0 muscles of the CMT participants showed a significantly higher mean fat fraction compared

with that of the volunteers (P < 0.05). These finding suggest that the 3D multiple gradient echo Dixon-based MRI is a reproducible and sensitive technique which can reveal a significant difference in the fat fraction of thigh muscle, including muscles with Goutallier Grade 0 (Kim et al., 2019). These results came consistent with other studies aimed to explore the validity and responsiveness of fat fraction as a biomarker measured using Dixon-based quantitative MRI in Hereditary sensory neuropathy type 1 (Kugathasan et al., 2019), CMT1A, and inclusion body myositis (IBM) (Morrow et al., 2018, Morrow et al., 2016) which may imply that intramuscular fat measurement using a Dixon-based quantitative MRI sequence can be a more sensitive and objective tool for screening of muscular degeneration.

2.2.3. Muscle MRI As an Outcome Measure in Hereditary Peripheral

Neuropathy

Muscle MRI correlation, as an outcome measure in peripheral neuropathy, with other validated measures for muscle strength and general function was explored. It has been shown that there is a significant correlation between MRI and manual muscle testing with DHMN and CMT (Del Porto et al., 2010). While manual muscle testing (MMT) lacks sensitivity and validity for mild weakness; higher MMT grade, MRI had the ability to detect subclinical muscle pathology (Del Porto et al., 2010). Morrow et al. (2016) investigated the sensitivity of MRI to detect changes in 20 patients with Charcot-Marie-Tooth disease 1A (CMT1A) and 20 patients with inclusion body myositis (IBM). The validity of the magnetic

resonance measures was supported by strong correlations with clinical functional measures, and the responsiveness to disease progression over 1 year was shown to be better with MRI than with the clinical functional tests. Even though Charcot Marie Tooth disease 1A progresses slowly, magnetic resonance measures detected substantial increases in disease pathological changes in 1 year. They also found that MRI-measured T2 and MTR are abnormal in these diseases even when fat fraction values are within normal limits (Morrow et al., 2016). Suggesting that MRI can detect early signs of active intramuscular denervation before fatty infiltration. In an earlier study using the same MRI protocol, they explored the dependence upon age, gender and weight (body mass index) (Morrow et al., 2014). They found that quantitative MRI measurements show small but significant inter-subject age and weight dependency. However, these demographically driven differences are smaller than the expected pathological changes in NMDs, and thus too small to pose a significant finding in longitudinal studies (Morrow et al., 2014).

The three-point Dixon technique is an MRI sequence that has been applied to skeletal muscles in neuropathy to quantify intramuscular fat accumulation in different events. This sequence has been found to be highly responsive as presented by Kim et al. (2019) and correlating with other validated outcome measures (Kim et al., 2019). The London MRC Centre for Neuromuscular Diseases has developed a quantitative muscle MRI protocol (using the three-point

Dixon technique) as a highly responsive biomarker for CMT1A in adult patients. Effort to validate its responsiveness has been undertaken by Morrow and colleagues (2018). The results of this study confirmed the reliability, validity, and responsiveness of the MRC centre MRI quantified calf muscle FF protocol as an outcome measure in CMT1A. Selection of study participants with increased baseline calf muscle FF provided a highly responsive biomarker in this patient group, suitable for utilization in multicentre international clinical trials (Morrow et al., 2018). The MRC centre MRI calf muscle fat fraction protocol has also been validated in hereditary sensory neuropathy type 1 (HSN1) (Kugathasan et al., 2019).

Patterns of intramuscular fat and muscle atrophy were also explored using MRI in more recent studies. Esteller et al. (2023) explored MRI imaging in a large cohort of DHMN (Esteller et al., 2023). They carried out a retrospective analysis of clinical, genetic, and muscle imaging data of 84 people with DHMN. The MRI examinations included T1-weighted and T2-weighted Short Tau Inversion Recovery (STIR-T2w) images. The extent of muscle fat infiltration was measured using the Mercuri score. They used hierarchical clustering to detect specific patterns of muscle involvement (Esteller et al., 2023). Although the cohort showed proximal lower limbs and distal upper limbs involvement, the most common phenotype was distal lower limb weakness. Among the 84 cases, only 38 (45.2%) with a causative pathogenic variant identified, most commonly in HSPB1 in 12 cases (14.2%) (Esteller et al., 2023). Analysis of MRI images in general was symmetric in most cases (64/84 cases) and showed distal to proximal gradient of increasing fat replacement in all muscles of the lower legs while in the thigh, it was commonly seen in the vastus lateralis, intermedius, and the semimembranosus. They described muscle involvement in 4 different patterns shown in Table 3 (Esteller et al., 2023).

In another study by O'Donnell et al. (2022), they used T1-weighted and Short Tau Inversion Recovery (STIR) to qualitatively analyse a group of 6 people with SORD pathogen (5 DHMN, 1CMT2) (O'Donnell et al., 2022). Mean fat accumulation, atrophy and STIR grades were higher in calf than thigh muscles on all scans. However, some fat accumulation was seen in thigh muscles of all patients, most commonly distally within vastus lateralis and intermedius. There was calf muscle atrophy in all patients, and the overall pattern was of greater involvement of posterior and lateral compartments. STIR hyperintensity, representing active denervation, was present on all scans, most markedly in tibialis anterior (Table 3). Analysis showed a highly significant correlation between calf anterior compartment fat accumulation with ankle dorsiflexion MRC grade (r -0.827, p 0.001) (O'Donnell et al., 2022).

The findings of these studies (Table 3) (Esteller et al., 2023, O'Donnell et al., 2022) revealed detailed characteristics and patterns of muscle involvement using MRI in DHMN cases with known pathogen that can be used for diagnostic

purposes. They showed a predominant involvement of the posterior calf compartment, with lateral compartment in some cases. However, these studies are limited by its retrospective nature, lack of control group, and the reliability of the MRI analysis could be affected by the choice of qualitative Mercuri score and by multiple observers.

Study	Pattern (n)	Known Pathogen (n)	MRI Muscle involvement		
(Esteller et al., 2023)	1 (9)	HSPB1(3), HSPB8(2), TRPV4 (1) , DCTN1 (1).	Severe involvement of all muscles on the lower leg.		
	2 (6)	BICD2(3), DYNC1H1(3)	Severe fat replacement of the lower leg with preservation of the toe extensor muscles		
	3 (12)	GARS (1), HARS1(1), VRK1(1)	Predominant involvement of the superficial posterior compartment of the legs.		
	4 (11)	HARS1(1), DYNC1H1(1), GDI1(1)	More severe involvement of the peroneus group in comparison with the rest of the compartments		
(O'Donnell et al., 2022) SORD (5)		SORD (5)	Greater involvement of posterior and lateral compartments of the lower leg with some fat accumulation in vastus lateralis and intermediu		
(n)= number of cases, BICD2= BICD cargo adaptor 2, DCTN1= P150 subunit of dynactin,					
DYNC1H1= cytoplasmic dynein heavy chain 1, GARS= glycyl-tRNA synthetase, GDI1=					
GDP dissociation inhibitor 1, HARS1= histidyl-tRNA synthetase 1, HSPB1= heat-shock					
protein B1, HSPB8= heat-shock protein B8, TRPV4= transient receptor vallanoid 4 gene,					
VRK1= VRK serine/threonine kinase 1, SORD= Sorbitol dehydrogenase.					

Table 3: Pattern of muscle involvement in DHMN cohort.

MRI studies in CMT revealed a different pattern of fatty infiltration. In a study aimed to quantitatively describe the MRI fat infiltration pattern of muscle degeneration in CMT1A disease and to look for correlations with clinical variables, Bas et al. (2020) assessed MRI fat fraction in lower-limb muscles of patients with CMT1A and healthy controls. In particular, 14 muscle

compartments were selected at leg and thigh levels and for proximal, distal, and medial slices. Muscle fat infiltration profile was determined quantitatively in each muscle compartment and along the entire volume of acquisition to determine a length-dependent gradient of fat infiltration (Bas et al., 2020). Clinical impairment was evaluated with muscle strength measurements and CMT Examination Score (CMTES). Based on quantitative MRI measurements combined with a dedicated segmentation method, muscle fat infiltration quantified in patients with CMT1A revealed a length dependent pattern of infiltration with the largest fat infiltration was quantified in the anterior and lateral compartments of the lower leg (17.7% and 21.8%, respectively, in comparison to controls). Muscle fat infiltration was correlated to main clinical variables including muscle strength measurements and CMTES (Bas et al., 2020). Despite the genetic overlap between CMT and DHMN, they exhibit distinct patterns of muscle involvement. Both conditions are length dependent with a predominant calf involvement, However, in CMT, the anterior compartment is more affected, while in DHMN, the posterior compartment is primarily involved.

Muscle oedema, the accumulation of water within the muscle as a sign of active denervation, can also be detected and quantified using MRI. Locher et al. (2020) assessed quantitative water T2 relaxometry for the early detection of neuromuscular diseases (NMD) in comparison to standard qualitative MRI imaging in a clinical setting (Locher et al., 2022). They retrospectively analysed

20

83 patients with suspected NMD who underwent MRI with a subsequent muscle biopsy between 2015 and 2019. Qualitative T1-weighted and fat suppressed T2weighted images were graded to be either pathological or normal. Mean and median water T2 relaxation times were obtained from manually drawn region of interests in biopsied muscle from multi-echo sequence. Histopathologic pattern of corresponding muscle biopsies was used as a reference. Analysis in cases prior to late-stage fatty infiltration signal alternations in T1-weighted images showed that quantitative water T2 relaxometry had a significantly higher sensitivity in detecting muscle abnormalities than subjective grading of T2- weighted images. In 49 patients without late-stage changes, T2- weighted grading achieved a sensitivity of 56.4%, while mean and median water T2 had a sensitivity of 87.2% and 97.4% to detect early-stage neuromuscular diseases. Median water T2 ranged between 36 and 42 ms depending on histopathologic pattern (Locher et al., 2022). The author recommended that quantitative water T2 relaxometry should be considered complementary to subjectively rated fat-suppressed T2-weighted images in clinical practice to detect early changes in neuromuscular diseases (Locher et al., 2022). The ability to detect early changes by quantifying intramuscular water can be utilised in clinical trials to assess the safety and efficacy of therapies such as resistance exercises.

In summary, quantitative muscle MRI analysis is a valid and responsive tool in longitudinal studies to explore intramuscular fat using Dixon technique and to

21

explore muscle oedema using T2-relaxometry to identify early signs of active denervation. Qualitative MRI analysis to explore intramuscular fat using T1-weighted image and to explore muscle oedema using STIR imaging, was shown to be sufficient to describe the pattern of muscle involvement and can be used clinically for diagnostic purposes.

2.3. Dynamometry and Muscle Strength Assessment in Hereditary Peripheral Neuropathy

There are a number of techniques used to assess muscle strength. Some of these methods are relevant in the clinical setting and others are more applicable for research purposes. These include manual muscle testing using the MRC scale, hand-held dynamometry (with or without an immobilizing device), and isometric and isokinetic dynamometry.

The concept of isokinetic dynamometry was developed by Perrine and Hislop in 1967 (Hislop and Perrine, 1967). Since then, there has been an increase in its use for rehabilitation and research. Isokinetic dynamometers are considered the gold standard in evaluating dynamic muscular performance (Le-Ngoc and Jansse, 2012, Lund et al., 2005, Drouin et al., 2004, Kannus, 1994, Baltzopoulos and Brodie, 1989, Osternig, 1986). It is used for isometric and isokinetic strength measurements, allowing valid and reliable measurement of maximal voluntary muscle contraction (isometric), as well as maximal muscle contraction through a specific range of movement with controlled speed and position (isokinetic). Strength profiles are created to illustrate the relationship between instant torque and joint angle, allowing the identification of various characteristics including dynamic peak torque, the angle at which peak torque occurs, torque specific to certain angles, power, and the energy expended. These dynamic strength profiles also enable the identification of minor weaknesses within particular ranges of motion (ROM) of a joint. These features made it useful clinically as it provides sufficient sensitivity to detect muscle strength progress during rehabilitation (Webber and Porter, 2010, Tiffreau et al., 2007, Andersen, 1996).

Additionally, the isokinetic dynamometer offers several benefits over alternative methods for assessing muscle strength. It eliminates the need for the assessor's strength, ensures consistent stabilization of the subject during tests, and allows simultaneous measurement of joint angle and strength (Le-Ngoc and Jansse, 2012, Martin et al., 2006, Lund et al., 2005, Harlaar et al., 1996). Disadvantages of these devices are their size and cost, which make them impractical for routine clinical examinations (Le-Ngoc and Jansse, 2012, Li et al., 2006, Mital et al., 1995).

One of the primary parameters measured using dynamometry is muscle torque. Torque is the moment of forces applied about an axis of rotation. The formula of torque is:

Torque (Nm) = Force (N) × Distance (m)

where "distance" indicates the perpendicular distance from the input of force to the centre of rotation (Dvir, 2004, Dvir, 2000, Perrin, 1994). In isokinetic testing, Peak torque is the maximum torque production during movement throughout the range of motion. Peak torque is an indicative of maximum muscular tension capability taking into account changes due to biomechanical leverage and the muscular length and tension relationship that occurs throughout the range of motion. Torque can also be measured under isometric conditions against immobile lever of the dynamometer, which is an important indicator of explosive force production of the tested muscle or the ability of a muscle to exert maximal force promptly (Dvir, 2004, Dvir, 2000, Perrin, 1994).

Muscle force varies at different joint angles due to numerous biomechanical elements of the musculoskeletal system (Baltzopoulos and Brodie, 1989). In the isokinetic method, resistance in the dynamometer would be equal to the muscular capacity in numerous joint angles, providing efficient loading in the muscles at dynamic points (Baltzopoulos and Brodie, 1989). Moreover, isokinetic dynamometers in comparison to gravity-loaded systems do not store potential energy and therefore the return movement does not require eccentric contraction to control the return of the limb-lever arm system to the initial position (Baltzopoulos and Brodie, 1989, Thistle et al., 1967).

The reliability of isokinetic dynamometer has been shown to be high in healthy subjects (Webber and Porter, 2010), and in people affected with neuromuscular diseases even with significant weakness (\leq 3- on MRC scale) (Tiffreau et al., 2007). In the MRI study by Morrow et al. (2016), isokinetic dynamometer correlated with MRI measures in Charcot-Marie-Tooth disease type 1A (CMT1A) and inclusion body myositis (IBM), and was able to detect difference in strength over 12 months (Morrow et al., 2016).

25

In a further study, that also included people with CMT1A (n=33) aimed to determine correlations between the isokinetic muscular strength of knee flexors and knee extensors and walk parameters, People with CMT1A showed excellent correlation between isokinetic knee extension and walking speed, and moderate correlation between isometric knee flexion and walking speed (Reynaud et al., 2019). These correlations were higher in participants younger than 50 years old. The lack of correlation in older patients may be explained by other factors limiting the quality of walking in these patients such as degenerative phenomena (osteoarthritis, the natural evolution of neuropathy, or the static deformation of the feet) and orthopaedic surgical history of the feet and/or ankle, which is frequent in older patients and was not considered in the study design (Reynaud et al., 2019).

A trial explored the feasibility and effect of community-based aerobic exercise training for people with two of the most common neuromuscular diseases: Charcot-Marie-Tooth disease type 1A (CMT) and inclusion body myositis (IBM) (Wallace et al., 2019). Isokinetic dynamometer was used as a secondary outcome measures to evaluate the effect of aerobic exercises on muscle strength, in addition to peak oxygen uptake (VO2 peak) as a primary outcome measure. Both disease groups demonstrated improvements in VO2 peak, with a moderate effect size in the CMT participants (Cohen's d = 0.53) and a strong effect size in the IBM group (Cohen's d = 1.72). However, there was no major changes in the

isokinetic dynamometer (Wallace et al., 2019). There was a notable challenge when using the isokinetic dynamometer for very weak muscles in generating sufficient torque to trigger the motor. This may have affected the reliability of the dynamometry data (Wallace et al., 2019). Moreover, using VO2 peak is more suitable outcome measure in training trials targeting aerobic capacity, rather than using isokinetic dynamometer which is more suitable for measuring peak torque in muscle resistance training trials (Carter et al., 1995).

The reliability of isokinetic dynamometer in neuromuscular conditions has been explored, and the use of isokinetic dynamometer in CMT for longitudinal studies or for clinical trials as an outcome measure has been described. However, none of these studies explore DHMN specifically. When utilizing isokinetic dynamometers in clinical trials, several factors may influence the reliability of measurements, that need careful consideration. These include reduced range of joint motion, which may be associated with aging or orthopaedic surgeries, and the focus of the exercise program, with isokinetic dynamometry most relevant for training targeting muscle strength.

2.4. Gait and Walking in Hereditary Peripheral Neuropathy

Quantifying human gait has utility for research, diagnostic or clinical purposes. Subsequently, gait analysis is an important assessment tool that uses physical measurements and models, to identify functional capabilities, limitations, and gait pattern in different neuropathies with the purpose of evaluating the effectiveness of rehabilitation therapies.

2.4.1. Definitions and Terminology

Locomotion can be described as the motion to move from one point to another (Rose and Gamble, 2006). Walking is one method of locomotion. In humans, bipedal gait uses cyclical motion to progress the body forward in a linear path. Forward progression occurs by placing one foot in front of the other repeatedly but never having both feet off the ground at the same time. While running could be considered faster walking with shorter gait cycle and never having both feet on the ground at the same time, the difference is there is a period where both feet are off the ground (flight phase) (Rose and Gamble, 2006, Whittle, 2007).

Angular motion of a joint that connects adjacent body segments is caused by force acting at a distance from the joint axis resulting in a torque or moment (Valero-Cuevas, 2016, Robertson DGE et al., 2014a, Knudson, 2007, Dvir, 2000). The direction of joint motion is determined by combining all moments acting on body segments. Angular moments during walking are caused by a combination of externally and internally generated forces and moments. External forces include gravity force that acts through the centre of mass of the segment in downward direction, and ground reaction force that acts at the point where the body contacts the ground. Internally generated moments are mainly due to the actions of muscles (Valero-Cuevas, 2016, Robertson DGE et al., 2014a, Knudson, 2007, Dvir, 2000). The contributions of ligaments and surrounding soft tissues to the torque are minimal because their points of action are near the joint axis, and these structures are usually lax unless the joints are fully extended or flexed (Valero-Cuevas, 2016, Robertson DGE et al., 2007, Dvir, 2000).

2.4.2. The Gait Cycle

The gait cycle can be divided into two major phases, the stance phase, and the swing phase, which make up approximately 60% and 40% of the cycle, respectively. Phases, events, periods, and support during a full gait cycle are explained in Table 4 (Vu et al., 2018, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

					Å				
% of cycle	0%	0-10%	10-30%	30-50%	50-60%	60-70%	70-85%	85-100%	
Phase	Stance phase					Swing phase			
Event	Initial contact	Foot Opposit	flat, e toe off	Heel off, Opposite initial contact		Toe off	Tibia vertical, Feet adjacent	Initial contact (next cycle)	
Period	Loading response	Mic	Mid-stance		Pre- swing	Initial swing	Mid-swing	Terminal swing	
Support	Double support	Single leg support			Double support	Single leg support			

Table 4: Normal gait cycle, Right side.

The sequential combination of the phases enables the limb to accomplish three basic tasks. These are weight acceptance, single limb support, and limb advancement. Weight acceptance initiates the stance phase and uses initial contact and loading response. Single limb support continues stance with the mid stance and terminal stance. Limb advancement begins in the final period of stance (pre-swing) and then continues through the three phases of swing (initial swing, mid-swing, and terminal swing). Table 5 is a summary of the objectives and joints positions in each gait period (Vu et al., 2018, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Period	Initial Contact	Loading Response	Mid Stance	Terminal Stance	Pre-Swing	Initial Swing	Mid Swing	Terminal Swing
Hip	Flexed to 30°	Flexed	Extending	Extending 15°-30°	Flexing	Flexing	Flexing to 30°	Flexed to 30°
Knee	Extended	Flexing 5°- 10°	Extending but not to 0°	Extend, then Flex	Flexing 30°- 40°	Flexing up to 65°	Extending	Extending
Ankle	Neutral	Plantarflexin g to 20°	Dorsiflexing	15° Dorsiflexing to Neutral	Plantarflexin g 20°-30°	Dorsiflexing to 0°	Dorsiflexing to 0°	Neutral
Objective	Begin Stance	Shock Absorption, Advance body over Heel Rocker	Advance body over stationary foot, ankle rocker	Advance body over forefoot rocker	Prepare for Swing, transfer load to contralateral limb	Clear foot and advance limb	Advance limb and clear foot	Advance limb

Table 5: Summary of the objectives and joints positions in each gait period.

2.4.3. Gait Mechanics and Locomotor Functions

2.4.3.1. Role of Major Joints and Muscles

Propulsion and Shock Absorption are the major functions of the joints and muscles during walking. Below is a summary of the main functions of the locomotion intersegmental joints and the muscles acting upon them during gait cycle (Figure 2), emphasising the role of the ankle joint and muscles that are generally affected in peripheral neuropathy (Vaughan, 1999, Chambers and Sutherland, 2002).



Figure 2: Typical activity of major muscle groups during the gait cycle. $IC = initial \ contact, \ OT = opposite \ toe \ off, \ HR = heel \ rise, \ OI = opposite \ initial \ contact, \ TO$ $= toe \ off, \ FA = feet \ adjacent, \ TV = tibia \ vertical.$ Figure adapted from (Whittle, 2007).

Advancement of the body depends on stance limb mobility. As body weight is dropped onto the limb, the force is primarily directed toward the floor. Advancement of the body depends on redirecting some of this force in a manner that combines progression and stability. The essential element for progression over the stance limb is rocker action by the foot and ankle. Full ranges of extension at the knee and hip are the other critical factors for progression (Whittle, 2007, Rose and Gamble, 2006, Perry and Burnfield, 1992).

Heel Rocker: Using the heel as the fulcrum, the foot rolls into plantar flexion as the stance phase begins. Limb progression is preserved as the Pretibial muscles decelerate the foot drop and draw the tibia forward. Quadriceps action extends the progression of the tibia initiated by the heel rocker to advancing the thigh (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Perry and Burnfield, 1992).

Ankle Rocker: With the ankle as the fulcrum, the tibia (and whole limb) rolls forward in response to the body momentum. The rate of tibial progression is decelerated by the soleus muscle (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Perry and Burnfield, 1992).

Forefoot Rocker: Tibial progression is continued over the forefoot rocker. Both gastrocnemius and soleus act vigorously to decelerate the rate of tibial advancement (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Perry and Burnfield, 1992).

The knee Joint: Plays a vital role in shock absorption through flexion immediately after heel strike during the early stance phase and aids in propulsion by extending before toe-off in the late stance phase. In the swing phase, it flexes to allow foot clearance from the ground (Whittle, 2007, Rose and Gamble, 2006, Perry and Burnfield, 1992).

The hip Joint: Plays a critical role in providing stability and movement. During the stance phase, the hip extensors work to support the body, while during the swing phase, the hip flexors facilitate leg advancement (Whittle, 2007, Rose and Gamble, 2006, Perry and Burnfield, 1992).

2.4.3.2. Energy Conservation

Walking utilizes energy-saving mechanisms that reduce the metabolic cost of walking by storing and releasing energy such as the inverted pendulum effect, where the body vaults over the leg in stance phase, conserving energy through gravitational potential energy and kinetic energy conversion. Another mechanism is the elastic recoil of tendons. Tendons store energy during certain phases and release it to aid movement, reducing overall energy expenditure. For example, the calf store energy when it is stretched in terminal stance, and recoil in preswing to assist progressing the leg forward (Perry and Burnfield, 1992, Kuo and Donelan, 2010, Chambers and Sutherland, 2002).

2.4.3.3. Stance Stability and Balance

Functionally, the body is divided into two units, the passenger unit, which includes the head, neck, trunk and arms, and the locomotor unit, which include the two lower limbs and pelvis (Perry and Burnfield, 1992). Stability in the upright position is determined by the functional balance between the alignment of the body and muscle activity at each joint. The falling body weight, ligamentous tension, and muscular activity are forces that can act on the joints in

upright position (Perry and Burnfield, 1992). Standing stability is challenged by three anatomical features. First is the mass distribution between the passenger unit (70%) and the locomotor system (30%). Second is the multisegmented nature of the supporting limbs. The third factor is the contours of the lower limb joints. The centre of gravity is a hypothetical point around which the force of gravity appears to act, located approximately anterior to the second sacral vertebra and is the origin of the gravity line (Perry and Burnfield, 1992). The goal of static balance in standing is to keep the gravity line; a line falling from the centre of gravity down to the ground, within the base of support; or the distance between the centre of both heels sideways (Perry and Burnfield, 1992, Kuo and Donelan, 2010).

Balance during walking is maintained by minimum lateral and vertical displacement of the centre of gravity. It changes constantly with body or limbs movements. (Rose and Gamble, 2006). Two actions are essential to preserve balance over a single limb during walking, including lateral shift of the body mass and local muscular stabilization of the hip joint to keep the pelvis and trunk erect (Kuo and Donelan, 2010).

Central and peripheral nervous systems orchestrate the timing and strength of muscle contractions. The central pattern generators in the spinal cord produce basic rhythmic patterns, while the brain adjusts these patterns for dynamic balance, obstacle avoidance, and direction changes (Winter, 2009, Stergiou, 2020).

The six optimizations for body adjustments used to minimize the excursions of the centre of gravity during walking are called the 'determinants of gait' by Saunders et al. (1953) and they are; pelvic rotation, pelvic obliquity, knee flexion in stance phase, ankle mechanism when the heel sticks out beyond the ankle joint and effectively lengthens the leg during the loading response, foot mechanism when the ankle moves from dorsiflexion into plantarflexion the forefoot lengthens the leg at the end of stance, and lateral displacement of body (Whittle, 2007, Stergiou, 2020, Kuo and Donelan, 2010, Chambers and Sutherland, 2002).

2.4.3.4. Biomechanical Adaptations and Variability

Biomechanical variability arises from individual differences in anatomy, age, and health status. For instance, children and elderly individuals have distinct gait patterns due to different levels of muscle strength, flexibility, and balance. People with disabilities or injuries may develop compensatory gait patterns, altering their stride length, cadence, or joint use to accommodate pain or reduced mobility (Chambers and Sutherland, 2002, Kuo and Donelan, 2010, Vaughan, 1999). Moreover, obesity showed an impact on gait kinetics and kinematics with reduced ankle angle and decreased plantar flexor moment and power generation at terminal stance (Capodaglio et al., 2021). Understanding these detailed aspects of walking gait biomechanics helps in identifying gait abnormalities, designing effective rehabilitation strategies, and improving mobility aids such as prosthetics and orthotics. This knowledge is crucial for professionals in biomechanics, physical therapy, and related fields to enhance patient care and support mobility and health.

2.4.4. Methods of Gait Assessment

A comprehensive gait assessment usually includes the study of body movement and position in space (kinematics), and the study of forces, moments, and powers associated with this movement (Kinetics). In normal walking, there are five major attributes which are frequently lost in pathological gait: stability in stance, foot clearance in swing, pre-positioning of the foot for initial contact, adequate step length, and energy conservation (Chambers and Sutherland, 2002). Observing these five attributes and any deficits which might cause them provides a particularly good functional assessment of the walking patterns (Chambers and Sutherland, 2002, Kuo and Donelan, 2010). Gait can also be described in relation to time (Table 6) or linear parameters (Table 7) (Whittle, 2007, Perry and Burnfield, 1992).

Cadence	The number of steps in a standard time frame.				
Cycle Time	The duration to complete a full gait cycle.				
Speed	The distance covered by the whole body in a given time.				

Table 6: Time parameters to describe the gait cycle.



Table 7: Linear parameters to describe the foot placement on the ground during a full gait cycle.

There are a number of different assessment tools used widely for clinical and research purposes for kinematic analysis of gait. Kinematics describes the motion of joint angular displacement, velocity, and accelerations. Assessment of movement kinematics can be obtained by electro goniometers, video recording or a simple visual gait analysis. However, the data collected is two dimensional and subject to inaccuracies and bias (Winter, 2009, Stergiou, 2020).

Three-dimensional motion analysis systems digitally reconstruct the individual's body as a multisegmented system. After optoelectronic infrared markers are placed at specific anatomic landmarks, their position is triangulated by cameras to calibrate the individual into the system. Construction of the coordinates and orientation of the rigid body segments allow calculation of joint angles of the proximal and distal segment, joint angular velocity, and joint acceleration. Measurements are collected for each joint in all three cardinal planes of motion (Whittle, 2007, Winter, 2009, Stergiou, 2020).

Kinetics reflect the cause of movement, and therefore the forces, power, and energy that affect the manner in which an individual moves. Ground reaction forces are measured using force plates on the ground to refer to the forces that act on the body throughout the stance phase. Analysis of the ground reaction force acting on the centre of gravity is typically broken down into vertical, mediolateral, and anteroposterior force plots (Whittle, 2007, Winter, 2009, Stergiou, 2020).

2.4.5. The Normal Gait in Three-Dimensional Motion Analysis

2.4.5.1. Kinematics

The description in this section is focused on the sagittal plane measurements as they are probably the most studied, best understood, and most accurately reproduced (Figure 3) (Rose and Gamble, 2006, Whittle, 2007).

Anterior/Posterior Pelvic Tilt: This aspect of the graph shows the forward and backward tilting motion of the pelvis. A normal walking pattern displays a rhythmic oscillation of anterior and posterior tilt, which contributes to leg extension and absorption of impact during the stance phase. The pelvis is most horizontal (least amount of tilt) at foot-off and opposite foot-off, with maximum flexion occurring in mid- to late stance and terminal swing (Figure 3) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Hip Flexion/Extension: Increase in hip flexion at initial contact at the beginning of the stance phase, followed by a gradual extension as the body moves over the supporting leg until opposite foot contact. As soon as the opposite foot strikes the ground, weight is transferred to the forward limb, and the behind leg begins to flex at the knee and hip, while pivoting on the forefoot to initiate swing phase. As the foot leaves the ground, the hip continues to flex throughout the swing phase. The hip extensor muscles decelerate the thigh and diminish the hip flexion in preparation for weight acceptance (Figure 3) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Knee Flexion/Extension: The knee typically undergoes flexion immediately after heel strike to absorb impact and then extends during mid-stance as the ground reaction force line progresses anterior to the knee to support body weight. It should be noted that this passive extension cannot occur without the strong eccentric contraction of the plantar flexors restraining the shank from progressive forward rotation. During the swing phase, the knee flexes again to facilitate foot clearance from the ground. It reaches maximum flexion as the swinging foot passes the opposite limb. The knee joint is then rapidly extended nearly to full extension just prior to foot strike (Figure 3) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Ankle Plantar Flexion/Dorsiflexion: The most complex of the sagittal curves and can be broken down into four separate functional segments:

This first segment occurs between foot strike and opposite foot-off. The ankle is positioned at approximately neutral when the foot strikes with heel first. The position of the ground reaction force is posterior to the ankle centre causes plantar flexion until the foot is flat prior to opposite foot-off. This part of the ankle motion curve is also known as the first rocker (Figure 3) (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

The second segment occurs during single limb stance and opposite limb swing. It is typically convex superiorly and reflects the body passing over the fixed flat foot (second rocker). At approximately 40% of the cycle, the heel begins to rise as the plantar flexors increase their force of contraction and act concentrically (third rocker) (Figure 3) (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

The third segment continues with rapid plantar flexion to a maximum of 20 to 25 degrees just as the foot is lifted off the ground. The transfer of weight to the opposite limb occurs very rapidly, and the plantar flexion movement occurring after opposite foot strike is passive. This movement may be entirely due to gravity

and inertia or the passive tension in the plantar flexors (Figure 3) (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

The fourth segment is rapid ankle dorsiflexion throughout the swing phase to ensure maximum foot clearance. The ankle is maintained in this neutral position by isometric contraction of the anterior compartment muscles until foot strike when these same muscles are again needed to eccentrically restrain the plantar flexion that repeats the first segment of the cycle (Figure 3) (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).



Figure 3: Sagittal plane joint angles (degrees) during a single gait cycle of right hip (flexion positive), knee (flexion positive) and ankle (dorsiflexion positive).

 $IC = initial \ contact, \ OT = opposite \ toe \ off, \ HR$ = heel rise, $OI = opposite \ initial \ contact, \ TO =$ toe off, $FA = feet \ adjacent, \ TV = tibia \ vertical.$ Figure adapted from (Whittle, 2007).

2.4.5.2. Kinetics

Ground Reaction Force (GRF)

Vertical Component: This is often the most significant part of the GRF and illustrates the forces acting perpendicular to the ground. It shows two main peaks during a walking cycle. The first peak corresponds to the initial contact and weight acceptance phase (heel strike), indicating the impact force. The second peak reflects the push-off phase (toe-off), demonstrating the force used to propel the body forward. The valley between these peaks represents the midstance phase, where the body's weight is fully supported by the limb (Figure 4) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Anteroposterior Component: This component shows the forces acting in the forward and backward direction. Initially, there's a decelerating force as the foot makes contact with the ground and decelerates the body, followed by a propulsive force during the push-off phase, accelerating the body forward (Figure 4) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Mediolateral Component: Representing the forces acting from side to side, this component is usually smaller compared to the vertical and anteroposterior forces. It illustrates the body's need to stabilize and balance during the gait cycle, with shifts in weight from one foot to the other (Figure 4) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).



Figure 4: Ground reaction forces. Lateral, Anteroposterior (fore–aft), and vertical components of the ground reaction force, in newtons, for right foot (solid line) and left foot (dashed line).

IC = initial contact, OT = opposite toe off, HR = heel rise, OI = opposite initial contact, TO = toe off, FA = feet adjacent, TV = tibia vertical. Figure adapted from (Whittle, 2007).

Moment

Hip Flexion/Extension Moments: During the initial stance phase, an extension moment is observed as the hip works to support the body's weight and stabilize the pelvis. This shifts to a flexion moment as the leg moves forward during the swing phase, preparing for the next step (Figure 5) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Knee Flexion/Extension Moments: An extension moment is noted in the early stance phase, aiding in supporting the body's weight and stabilizing the knee as it bears the load. As the gait cycle progresses, flexion moments might be observed, particularly as the foot prepares to leave the ground, facilitating the transition to

the swing phase (Figure 5) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Ankle Plantar flexion/Dorsiflexion Moments: In the early stance phase, dorsiflexion moments prevail as the foot adjusts to ground contact, facilitating shock absorption. The transition to plantarflexion moments marks the push-off phase, where the foot exerts force against the ground to propel the body forward (Figure 5) (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).



Figure 5: Sagittal plane internal joint moments (newton-meters per kilogram body mass) during a single gait cycle of right hip (extensor moment positive), knee (extensor moment positive) and ankle (plantar flexor moment positive).

 $IC = initial \ contact, \ OT = opposite \ toe \ off,$ $HR = heel \ rise, \ OI = opposite \ initial$ $contact, \ TO = toe \ off, \ FA = feet \ adjacent,$ $TV = tibia \ vertical.$ Figure adapted from (Whittle, 2007).

Power

Hip Power Generation and Absorption: Power absorption mainly occur during the initial contact and early stance phase as the hip adjusts to load bearing, and power generation is mostly notable during the transition from stance to swing phase. These phases are indicative of the hip's role in stabilizing the body and propelling it forward (Figure 6) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Knee Power Absorption and Generation: The knee absorbs power primarily during initial contact and early stance when the limb is loaded, and phases of power generation, notably in the latter part of the stance phase leading into the swing phase. These phases reflect the knee's role in damping the impact with the ground and then contributing to propulsion (Figure 6) (Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

Ankle Power Generation and Absorption: An initial phase of power absorption occurs during heel strike and early stance, where the muscles and tendons store energy as the foot makes contact with the ground. Followed by a significant phase of power generation during push-off, where the ankle muscles actively produce force to propel the body forward (Figure 6) (Brockett and Chapman, 2016, Whittle, 2007, Rose and Gamble, 2006, Vaughan, 1999, Perry and Burnfield, 1992).

45



Figure 6: Sagittal plane joint powers (watts per kilogram body mass) during a single gait cycle of right hip, knee, and ankle. Power generation is positive, absorption is negative.

 $IC = initial \ contact, \ OT = opposite \ toe \ off,$ $HR = heel \ rise, \ OI = opposite \ initial$ $contact, \ TO = toe \ off, \ FA = feet \ adjacent,$ $TV = tibia \ vertical.$ Figure adapted from (Whittle, 2007).

2.4.6. Altered Gait in Hereditary Peripheral Neuropathy

Early signs of gait alteration in hereditary peripheral neuropathy may start in childhood or adolescence as the weakness progresses slowly. The change in gait pattern is associated with muscle weakness due to denervation muscle atrophy (Wang et al., 2019, Hong et al., 2019). Other possible causes of altered gait could include limited joint range of motion, foot deformities, pain (Beckmann et al., 2015), sensory impairment (Nardone et al., 2014), and high body mass index (Hackett et al., 2019). Limited walking abilities could affect the activity level, energy consumption, and overall wellbeing (Menotti et al., 2011). Although the underlying genetic cause and clinical characteristics vary between affected people, their gait starts to deviate initially at the point when the symptoms around

the foot and ankle are present. Gait abnormalities in most DHMN cases are expected to start in late adulthood when the symptoms are progressed since the condition is a slowly progressive adult-onset disease (Forrester et al., 2020, Wang et al., 2019, Tanabe et al., 2018). Because of the limited existing literature in gait abnormalities in DHMN, herein a review of gait abnormalities in adult population affected by CMT, as this is a related condition and often overlaps with DHMN clinically and genetically (Kang et al., 2020, Forrester et al., 2020, Rossor et al., 2017).

Newman et al. (2007) was the first to use 3D motion analysis to observe gait in people with CMT. They analysed 16 CMT subjects aged 8–52 years old (11 with type I, 5 with type II) and 40 matched control subjects with no gait impairment or chronic health condition. The CMT group showed 15% slower gait and a combination of tight tendo-Achilles, footdrop, failure of plantar flexion and increased foot supination. They also presented with excessive internal rotation of the knee and/or tibia, knee hyperextension in stance phase, excessive external rotation at the hips and decreased hip adduction in stance. More proximal deviations may be adaptations to or consequences of the disrupted ankle and foot biomechanics. A direct relation to the neuropathy is also possible since subnormal muscle power was observed in the proximal joints for most subjects on both manual testing and kinetic analysis (Newman et al., 2007). Nevertheless, due to the heterogeneity of the biomechanical impairments in that study they were

not able to identify a specific gait strategy as a compensatory mechanism. On the contrary, Don et al. (2007) considered functional differences within the CMT group and how they are related to different gait patterns. They evaluated a sample of 21 CMT subjects and 21 matched controls to describe the temporal, kinetic, kinematic, electromyographic and the mechanical energy expenditure of the body in terms of energy recovery and energy consumption in relation to the wholebody centre of mass (COM) in CMT subjects with foot drop and plantar flexion failure. According to Vinci and Perelli (2002), foot drop is the inability of the foot to dorsiflex up to 90° during the swing phase of gait making the limb functionally longer, and plantar flexor failure is the inability to stabilize the ankle joint in the sagittal plane while wearing a shoe with a 2cm heel, which is the most common heel height for ready-made footwear (Vinci and Perelli, 2002, Don et al., 2007). Don et al. (2007) classified the CMT group as having isolated foot drop (group 1) and combination of foot drop and plantar flexion failure (group 2). They observed two distinctive gait patterns when CMT group were subdivided into group 1 or 2. Group 1 showed a gait pattern with some characteristics of the 'steppage pattern'. The complex motor strategy adopted by group 1 assists foot clearance and preserves step length despite high energy consumption and slowed the swing velocity. Patients in Group 1 typically land with an excessively plantarflexed ankle in loading response. Most of the individuals exhibit a flatfoot landing due to a lack of dorsiflexor angular impulse, attributed to the weakness of dorsiflexor muscles which fail to control foot landing properly. To compensate
for the weak dorsiflexor muscles and preserve body progression, patients increase passive ankle dorsiflexion in mid-stance by delaying plantar flexor muscle activation, as evidenced by electromyographic (EMG) data. This results in reduced plantar flexor angular impulse. There is an increase in hip extension and higher knee angular impulse during the stance phase. This coordination is necessary to maintain body progression and balance, necessitating additional effort from knee extensors, indicated by prolonged EMG activity in the vastus medialis and rectus femoris muscles. Despite delayed plantar flexor activation, an increased plantar flexor angular impulse is observed in push-off, which contributes to propulsion by engaging muscles less affected by the disease. Hip flexors also play a crucial role in propulsion, lifting the lower limb and enhancing hip flexion. During swing phase, patients exhibit foot drop, compensated by increased knee and hip flexion. This adaptation leads to a 'steppage' gait pattern, characterized by reduced gait speed due to increased swing phase duration, yet maintaining normal step length. Compared to healthy individuals, these patients exhibit increased energy consumption and recovery during gait, attributed to the mechanical effort required for their unique gait strategy, especially in hip flexor, knee extensor, and ankle plantar flexor muscles (Don et al., 2007). Group 2 displayed a 'clumsy pattern' characterised by very slow gait with reduced step length, a broader support area and great reduction in the cadence. This group of patients is showed a low energy consumption and greater energy recovery (Don et al., 2007). Similar to Group 1, patients in Group 2 land with an excessively

plantar flexed ankle, though to a lesser degree than group 1. They also exhibit a lack of dorsiflexion angular impulse during the Loading-Response. However, they show an increased hip extensor angular impulse and decreased knee flexion range of motion (ROM) and knee extensor angular impulse, suggesting reliance on hip extensors over knee extensors for load acceptance. In Mid-Stance, Group 2 display greater passive ankle dorsiflexion and decreased plantar flexor angular impulse compared to Group 1, indicative of a compensatory mechanism for flatfoot landing and a mechanical effect of plantar flexor failure. This failure also leads to decreased hip extension, aimed at maintaining anterior-posterior balance. In Push-Off, plantar flexor failure, characterised by decreased angular impulse and plantar flexion ROM, was partially compensated by increased knee extensor angular impulse. Unlike Group 1, hip flexor angular impulse does not increase, reflecting different compensation mechanisms during swinging. In Swing-Phase there was a reduction in hip and knee flexion, with increased hip abduction and pelvic elevation on the swinging side through prolonged Gluteus Medius activation. Despite similar degrees of foot drop as Group 1, Group 2 has a lower degree of flatfoot landing at initial contact, requiring less lower limb elevation to prevent tripping. Group 2 shows greater energy recovery and lower energy consumption compared to Group 1, attributed to broader muscle weakness requiring greater energy recovery and compensatory mechanisms possibly related to frontal plane movements (Don et al., 2007).

It has been suggested that hip flexors play a role in compensation for plantar flexors weakness to initiate the swing phase during walking (Ramdharry et al., 2009). This has been studied in a cohort of 18 subjects with CMT who were compared with 14 matched controls while they walked on a treadmill to a predetermined point of perceived effort. A significant reduction was observed in peak hip flexor velocity during walking and hip flexor maximal voluntary contraction. In a second session following selective fatigue of the hip flexors by performing isometric contractions, hip flexor velocity decreased immediately on walking, and walking duration was greatly reduced. This study suggests that hip flexors compensate for distal weakness and that fatigue in the hip flexors can limit walking duration (Ramdharry et al., 2009).

These findings highlight the importance of considering functional differences and the level of weakness in motion analysis studies in peripheral neuropathies to identify patterns of deviation and compensation accurately. Understanding compensatory mechanisms can inform future research in designing therapeutic exercises to target associated muscle in order to improve gait and function.

The effect of muscle strength on gait has been studied by Guillebastre et al. (2013). They examined the relationships between lower limb muscular weakness and postural and gait capacities of 26 CMT subjects and 19 matched controls. Barefoot gait and postural control were analysed using a walking mat and a force platform, respectively. Muscular strength of the plantar and dorsal ankle flexors

was assessed using the Medical Research Council scale (MRC). The CMT group was subcategorized into Group 1 and Group 2 according to their MRC score for plantar and dorsal ankle flexors muscles (up to 5 MRC and up to 3 MRC, respectively) (Guillebastre et al., 2013). Postural parameters correlated only with plantar flexor strength, whereas gait parameters correlated with both dorsiflexors and plantar flexors strength. In comparison to control, Group 2, the weaker group, showed less postural control along the medio-lateral and antero-posterior axes, whereas Group 1, the stronger group, showed less postural control along the antero-posterior axes only. Gait velocity, cadence, and step length were impaired in Group 2 more than Group 1 (Guillebastre et al., 2013). These findings highlight the role of plantar flexor muscle strength in postural control and the role of both dorsiflexor and plantar flexor muscle strength in maintaining gait performance in individuals with CMT. Reyn et. al (2019) looked specifically at correlations between the isokinetic muscular strength of knee flexors and knee extensors and walking parameters for 33 subjects with CMT1A. The isokinetic muscular strength of the knee was assessed on an isokinetic dynamometer (Cybex) and the gait was assessed by instrumented walkway analysis (GaitRite). Results showed a moderate correlation between walking speed and isokinetic muscular strength of knee extensors for the entire cohort and stronger correlations between walking speed and isokinetic muscular strength of knee extensors and knee flexors for patients younger than 50 years. This correlation highlights the importance of knee muscle strength in maintaining walking speed. Knee extensors are crucial for controlling knee flexion during the loading response of walking. While knee flexors are involved in decelerating the leg during the swing phase and preparing for foot placement. However, the study findings showed no correlation with cadence or step/stride length (Reynaud et al., 2019). The lack of correlation between isokinetic knee muscles strength and walking cadence or stride length suggests that other factors beyond knee muscle strength are more influential in determining these specific gait parameters. This finding highlights the importance of comprehensive gait analysis in peripheral neuropathy, considering the contributions of muscle groups at the ankle and the hip levels.

Mildly affected people with CMT may not show altered gait while walking on a level surface with relatively low speed. Mild weakness can result in altered gait kinetics, kinematics, and economics during more complex ambulatory tasks. Lencioni et. al (2018) investigated biomechanics of step negotiation during walking. They compared gait during walking on level ground and steps ascending and steps descending on two-step stair of 21 CMT subjects with mild-to-moderate impairment and normalized ROM during swing and work produced at the ankle during push-off comparable to 31 healthy subjects. To assess muscle activity, each EMG profile was integrated over 100% of task duration and the activation percentage was computed in four phases that constituted the step negotiation tasks (Lencioni et al., 2018). The CMT group showed distal muscle hypoactivation with proximal adaptive motor strategies to overcome the challenge of stair

ascending and descending, even though the enrolled CMT subjects had no apparent level walking abnormalities. In both tasks they showed a dorsiflexion deficit and an increased hip flexion during the swing phase allowing for a safe clearance of the foot moving above the step edge. In addition, during ascending stance phase CMT subjects displayed less activation of vastus medialis and rectus femoris and a greater activation of hip extensor in comparison to healthy subjects. This pattern was associated to reduced moment and power at the knee joint and a considerable larger hip extensor moment and power production (Lencioni et al., 2018).

Mild impairment was found to have an effect on walking energy as well. Menotti et al. (2011) quantified the walking energy cost of a group of CMT1A subjects with low severity of walking impairment in comparison with matched controls. Oxygen uptake was measured in 8 CMT subjects and 8 healthy individuals when walking on a circuit for 5-min at their self-selected speeds (slow, comfortable, and fast). Both comfortable and fast speeds were lower in CMT group than in the control group, whereas walking energy cost per unit of distance was higher in patients than in the control group. CMT1A subjects, therefore, chose to walk slower but with higher metabolic cost compared to controls, despite no clinically evident walking impairment, which is likely due to altered walking patterns (Menotti et al., 2011).

These findings confirm the importance of comprehensive motion analysis in identifying coping strategies in individuals mildly affected by peripheral neuropathies. Cases with mild weakness may show early signs of adaptation in their energy expenditure or in ascending and descending stepping, which are more physically demanding tasks than walking on a level surface. Early detection of these signs can inform patient care and rehabilitation research, addressing these functional challenges effectively.

Besides muscle weakness, sensory loss is an important factor contributing to the challenges faced by CMT patients with walking balance. People with CMT fall frequently (Ramdharry et al., 2018). Afferent input loss was found to contribute to a delayed response to unexpected uneven surfaces while walking (Nardone et al., 2014, Van der Linden et al., 2009). However, DHMN is a condition of motor neuron loss and muscle weakness so the impact of sensory impairment will not be a significant feature in this cohort.

2.5. Current Management and Rehabilitation Methods in Hereditary Peripheral Neuropathy

Currently, there are no medical treatment for DHMN, and management is based on symptomatic treatments which include physiotherapy, surgery, and pain management. The effect of rehabilitative treatment in DHMN has not been systematically investigated.

55

Recently, advances in molecular genetics and molecular biology, and the development of various animal models of different neuropathies, have led to a better understanding of disease pathomechanisms. This knowledge represents a prerequisite for the development of future therapies. Although the efficacy of various molecules has been shown in vitro and in animal models, no significant positive effect has yet been showed in clinical trials in CMT and related disorders, however some clinical trials are still ongoing.

Rehabilitation is currently the only conservative symptomatic treatment available for this type of neuropathies. However, in cases with complex presentation and rigid deformities surgery found to be more beneficial to restore foot posture (Laurá et al., 2024). Rehabilitation includes different types of exercises, orthotics and devices prescribed by physiotherapists, orthotists, and occupational therapists. Interventions aim to address issues affecting locomotion and the function of daily activities of neuropathy patients including weakness, loss of balance, fatigue, and pain. Although rehabilitation does not affect the natural progression of the disease, it provides some improvement in functional status and walking stability (Dimitrova et al., 2016). Working in partnership with patients to include their preferences helps to optimize their rehabilitation program (Padua et al., 2014).

2.5.1. Ankle Foot Orthoses (AFOs)

Ankle foot orthoses (AFOs) are commonly prescribed for patients with peripheral neuropathy to support ankle and foot posture and to address foot drop. Different designs of AFOs are made from different materials; plastic, carbon fibre, or elastic fabric, working in different ways to support the ankle and foot. Size, weight, comfort, and acceptability can vary with different devices (Landfeldt et al., 2017, McCaughan et al., 2019). They give stability and support during walking (Phillips et al., 2011, McCaughan et al., 2019), can improve walking economy (Bean et al., 2001, Menotti et al., 2014b), and minimizes the load on the proximal muscles as that compensate for distal weakness (Ramdharry et al., 2012b). They are prescribed for people with CMT who are present with mild to moderate plantar flexors weakness and/or foot drop to assist the progression of the foot into swing phase and support the ankle during standing and stance phase in gait (Laurá et al., 2024).

However, AFOs may not always meet cosmetic or practical standards for the people expected to wear them (Phillips et al., 2011, McCaughan et al., 2019). Studies showed poor compliance with AFO use. Some people with CMT discard AFOs because they highlight their disability, if they are not essential for daily walking and/or are uncomfortable (Vinci and Gargiulo, 2008). Studies also indicated that CMT patients with more severe symptoms tend to use AFOs more often to improve function (Ramdharry et al., 2012b).

57

Recent studies evaluated the user experience of AFO as a commonly prescribed orthotics for individuals with CMT (Bertini et al., 2024, Zuccarino et al., 2021). They used an online based survey with questions related to the device and the service provided. Bertini et al. (2024) included responses from 266 CMT participants. Overall, 70% of subjects were prescribed lower limb orthoses, but 19% did not use them. AFOs was prescribed for 23% of subjects including Codivilla spring (41%), Toe- off (28%) and Peromed (24%) as the most frequently used, followed by Dyna- ankle (6%) and Foot- up (1%) (Bertini et al., 2024). Frequency of abandonment was 31% for AFOs after a mean time of use of 6.3±9.1 years. Complications were more frequently related to AFOs including skin reddening, moderate to severe pain, foot ulcerations, calluses, and emotional distress which contributed to AFO abandonments (Bertini et al., 2024). Interestingly, distal weakness and CMT severity influenced AFO acceptance. Users with moderate to severe weakness (MRC \leq 3) and high disease burden (CMTES>8) reported overall higher tolerability, compliance, and perceived benefit than those with mild or no weakness (MRC \geq 4-) and low disease burden (CMTES ≤ 8) (Bertini et al., 2024). Zuccarino et al (2021) reported a high rate of satisfaction in a 314 CMT participant, Ratings of AFO showed that greater than 50% of individuals responded in the affirmative (Strongly Agree or Agree) to most questions. However, over a third of individuals provided negative responses to multiple questions regarding their AFOs. This included 42% who indicated they dislike the appearance of their AFO, 32% who experienced discomfort, 35%

who experienced abrasions or irritations, and 36% who experienced pain with AFO use (Zuccarino et al., 2021). Both studies showed a high satisfactory rate, however they identified clear areas of improvements regarding prescription and AFO designs. A rational, patient oriented and multidisciplinary approach to orthoses prescription must be encouraged to ensure appropriate prescription. Improvement in AFO's material and customization techniques including 3D printing have potentials in improving overall satisfaction and minimize discomfort and emotional distress related to function or appearance.

Carbon fibre AFOs showed a high satisfactory rate (89%) in neuromuscular diseases population (Mnatsakanian et al., 2017). Functionally, the carbon fibre material allows the orthoses to store energy and release at pre-swing and therefore benefit patients with plantarflexion weakness (Zou et al., 2014, Bartonek et al., 2007). The effect of custom-made carbon fibre AFOs on gait has been studied by Dufek et al. (2014). They compared walking with and without AFOs in 8 CMT patients. Participants walked faster (89.4 \pm 13.3 vs 115.6 \pm 18.0 cm/s), with longer steps and greater frequency in the braced versus unbraced condition. As velocity increased, maximum joint moments during loading response shifted from the hip joint to the ankle and knee joints. During propulsion, the hip joint moment dominated, and most subjects loaded the brace tibial shell with an average load of 54.6% at the highest velocity and 16.6% with the lowest velocity. Energy storage in the brace averaged 9.6 \pm 6.6 J/kg (Dufek et al., 2014). These

results may imply that ankle weakness can be compensated by the recoil action of the carbon fibre AFO instead of using proximal muscles at the knee and hip as described by Ramdharry et al. (2012,2009).

Customization using 3D printing was explored in children with CMT (Wojciechowski et al., 2022). They included 14 CMT children and comprehensively assessed their gait with 4 different conditions including traditional AFOs, 3D printed replica AFOs (same design as traditional AFOs), 3D printed redesigned AFOs, and a shoes only control condition. The 3D printed replica had the same effect on gait as the traditional AFO. However, the redesigned 3D printed AFO significantly lighter (mean -35.2, SD 13.3%), and normalized maximum ankle dorsiflexor moment in loading response compared to shoes only and traditional AFOs (Wojciechowski et al., 2022). The study showed overall improvement functionally with redesigned 3D printed AFOs, however, CMT is known for its variability and analysing the effect of AFO would be more useful if the study participants were sub-grouped according to their functional variations. Similar to the study by Õunpuu et al. (2021), they explored the effect of 4 types of AFO on gait in CMT children. Due to the gait variability between cases, the cohort was sub-grouped for analysis into 2 groups. Group A included patients who had increased peak dorsiflexion in terminal stance or plantar flexion failure, and group B included patient who had increased equinus in mid-swing or foot drop (Ounpuu et al., 2021). Across the cohort there was a significant improvement in most of the gait parameters, spatiotemporal, kinematics and kinetics. But when looking at gait based sub-grouping, only group B, who had foot drop, showed a significant improvement in ankle dorsiflexion angle, ankle moments and power. Improvement was in parameter where a sufficient dorsiflexion angle is needed. children with foot drop, without AFOs, showed compensations including increased peak hip power generation at toe off and increased hip flexion in swing to clear the foot. But these compensations were no longer needed when ankle angle in swing was corrected. With the AFO there was a reduction in the peak hip power generation at toe off and mean peak hip flexion in swing.(Õunpuu et al., 2021). On the other hand, in group A, who had plantar flexion failure, the only change in stance was a decrease in peak ankle power generation. This indicates that AFOs reduce the ankle plantar flexion abilities at toe-off. In this group, the main gait disfunction was plantar flexion failure which caused increased ankle dorsiflexion with wight bearing, delayed heal raise, and decrease in ankle moment and power in terminal stance (Ounpuu et al., 2021). The AFO design and material used did not allow enough plantar flexion range of motion and it was not flexible enough to store energy with weight bearing, therefore, inhibited the use of the existing plantar flexor strength at toeoff (Ounput et al., 2021). With mild to moderate plantar flexors weakness, less stiff and less restrictive AFO would be more appropriate. Carbon fibre spring splints, for example, can increase ankle power generation in pre-swing. The carbon fibre material allows the orthoses to store energy and release on push off and therefore benefit patients with plantarflexion weakness (Bartonek et al., 2007).

In another study by Van Der Wilk et al. (2018), they introduced adjustable mechanical hinge AFOs to optimize ankle range of motion and AFO stiffness needed throughout the gait cycle (Van Der Wilk et al., 2018). However, it was applied on one case of flaccid paralysis and not specifically designed for CMT cases that could be presented with limited range of motion and foot deformities.

In the long term, AFOs combined with exercises can optimize a patient's function. Bensoussan et al. (2016) reported a 10-year follow-up case where bracing with orthopaedic shoes along with physical therapy twice a week was effective in treating pain, improving the gait, and enhancing the walking distance (>500 m) without assistive device in a 55 year old woman with Charcot-Marie-Tooth disease. Moreover, during the 10 years of follow-up, the physical examination parameters had stabilized since 2001; falls, sprains and walking distance had improved as compared to 2000; pain had alleviated since 2001 and gait parameters had improved up to 2007 and stabilized between 2007 and 2011 (Bensoussan et al., 2016). However, the results of this study cannot be generalized as it presents a single case, and the level of therapy received may not be transferrable to other health systems.

62

2.5.2. Therapeutic Exercise

Therapeutic exercise is one of the few available treatments for peripheral neuropathies (Montes and Garber, 2017). Exercise has the potential to improve balance, fatigue, physical performance, psychological condition and therefore the quality of life (Roberts-Clarke et al., 2016b, Vita et al., 2016, Ramdharry et al., 2012c, Burns et al., 2017). However, the long term effect and safety of exercises on peripheral neuropathy remains unclear (Sman et al., 2015). Currently, there is a paucity of evidence-based exercise protocols for peripheral neuropathy and designing exercise plans depends mainly on the therapist experience (Djordjevic et al., 2017, Montes and Garber, 2017, Chetlin et al., 2004a, Chetlin et al., 2004b). Different types of exercises can include strengthening, aerobic, and stretching exercises. Combinations of these types have also been proposed (Maggi et al., 2011).

As muscle weakness is the major clinical manifestation of CMT, studies in the literature mainly explored strengthening exercises as a treatment. A study by Djordjevic and colleagues (2017) aimed to examine differences in strength among exercising and non-exercising CMT individuals. This was the first study that explored the association between self-directed full body exercises and muscle strength. Two hundred and ninety seven participants with CMT who exercised were significantly stronger in elbow flexion and dorsiflexion than those who did not exercise. They suggested that self-directed exercise may be a convenient,

sustainable, and effective method of improving strength and decreasing disability in this population (Djordjevic et al., 2017). However, due to its retrospective nature, the study lacked a standardized measurement or definition of exercise. It also lacked longitudinal strength measurements.

An exercise trial by Chetlin and colleagues (2004) explored the effect of a 3 month home based resistance training program combined with creatine supplement (group 1) or with placebo (group 2) in 20 CMT participants. Among both groups, results showed that strength and function improved with increased type I muscle fibre diameter (48.2 \pm 14.2 μ m vs. 55.4 \pm 14.8 μ m) in response to resistive exercises (Chetlin et al., 2004b, Smith et al., 2006). Comparison between groups showed changes in myosin heavy chain composition in the creatine group (group 1) which was associated with improvement in function (faster chair risetimes). However, difference in muscle strength between groups was not detected (Chetlin et al., 2004b, Smith et al., 2006). Nine of the 20 participants agreed to undergo a follow up assessment after 20 to 34 months after the completion of the program, and only 3 of the 9 participants reported continuing the resistance exercises program. Findings showed that both those who continued training and those who discontinued training lost strength, but functional ability was lost only in those who discontinued training (Chetlin et al., 2010). This trial provided a protocol for exercises prescription which can inform clinical trials and exercise progression (Chetlin et al., 2004a). However, they failed to ascertain the effect of creatine supplement on muscle strength possibly due to the lack of appropriate muscle strength measurement as their protocol evaluated strength only at the strongest angle, where moment arms of effort were maximized (90°). In exercise protocols with low to medium intensity, it is possible that patients experienced improvement in strength at the weaker angles, therefore, measuring isometric strength in different angles throughout the range or using isokinetic testing might detect strength gain (Knapik et al., 1983, Marginson and Eston, 2001, Yang et al., 2014, Ha and Han, 2017).

Considering the role of the proximal muscles in compensatory gait patterns, as discussed in section 2.4.6, it has been hypothesized that targeting proximal muscles in strengthening exercises has a potential in improving gait. This hypothesis was explored by a pilot study in CMT (Ramdharry et al., 2014). They investigated the effect of a 16-week home-based programme of resistance training on hip flexor muscle strength in 26 people with CMT. Results showed no negative effects and a high level of adherence (93%). Despite a small change in strength on the left side, no changes were observed in walking speed and endurance measures. Variability of cases, the exercises protocol, and the outcome measures used may contributed to the lack of more significant improvement as the authors discussed (Ramdharry et al., 2014). Moreover, the condition of the foot and ankle, and the appropriate foot and ankle support, was not considered in the assessment or treatment protocol. Vinci (2001) suggests that if distal

alterations causing postural changes in stance are not first corrected, proximal strengthening may not be effective (Vinci, 2001).

Proximal exercises, however, were found to be more helpful in improving balance in CMT. Individualized multi-sensory balance training combined with proximal lower limb and trunk strength training showed moderate improvements in postural stability in CMT1A people (Dudziec et al., 2024). Sensory impairment and foot deformity are commonly present in CMT which may contribute to balance impairments in this cohort more than DHMN.

Overwork weakness has been a concern in therapeutic exercises for people with CMT. Studies that have investigated the role of overwork weakness in CMT have typically compared the muscle strength of dominant with nondominant sides of the body, mainly because overwork weakness occurs in muscles that are used more frequently. However, these studies gave contradictory and inconclusive results (Roberts-Clarke et al., 2016a). Studies using manual muscle testing as an outcome measure, which is somewhat subjective and dependent on the strength of the examiner, are unlikely to be sensitive enough to detect smaller differences in strength between dominant and nondominant sides (Piscosquito et al., 2014). Using hand dynamometry to measure grip and pinch strength might be a more objective measure of muscle strength. However, the fact that grip and pinch strength is generated not only by the intrinsic hand muscles, but also by relatively unaffected forearm muscles that can undergo hypertrophy with overuse as in

healthy subjects (Prada et al., 2018). It has been argued that as normal control showed stronger dominant side, an equal strength between the dominant and non-dominant sides in CMT could possibly be a sign of overwork weakness (Vinci et al., 2009).

A study by Burns et al. (2017) explored the safety and efficacy of 6 months progressive resistance exercise using adjustable exercise cuffs for foot dorsiflexion weakness in children with Charcot-Marie-Tooth disease (Burns et al., 2017). They randomly assigned 60 children to receive progressive resistance exercise (n=30) or sham training (n=30). The primary efficacy outcome was the isometric dorsiflexion strength assessed by hand-held dynamometry. The primary safety outcome was MRI analysis to quantify muscle volume, intramuscular fat fraction, and any sign of acute denervation. The difference between the two groups was measured at baseline, 6, 12, and 24 months (Burns et al., 2017). Analysis showed no immediate differences in strength between groups at 6 months, however, there was an overall gain of 5% in dorsiflexion strength after 24 months in the resistance exercise group compared with a deterioration of 23% in the sham group (Burns et al., 2017). Moreover, there was no effect on muscle volume, intramuscular fat fraction, and no signs of acute denervation or overwork on MRI (Burns et al., 2017). These finding suggest that resistance training has an effect in preserving muscle tissue over time safely rather than improving muscle strength by increasing muscle volume. MRI signal intensity as an indicator for acute denervation was utilized to assess safety of the exercise programme used in this study (Burns et al., 2017), which supports the accuracy of the study results as discussed in section 2.2.3. The gain in muscle strength in the resistance exercises group without change in the MRI muscle volume supports the hypothesis that strength gain induced by resistance training can be due to neural enhancements rather than changes in the contractile components of skeletal muscle. Resistance training enhance corticospinal excitability, improving the transmission of neural signals from the brain to muscles. This results in better motor unit recruitment and increased firing rates leading to greater force production. (Herda, 2022, Deschenes and Kraemer, 2002).

Impaired walking capacity and aerobic function due to muscle weakness is often reported in people with CMT. Interval exercises cycling training (El Mhandi et al., 2011, El Mhandi et al., 2008) and anti-gravity treadmill training (Knak et al., 2017) showed improvement in cardiorespiratory capacities and ambulatory functions in small CMT cohort (8 and 5 participants respectively). However, these studies were undertaken in hospital environments and supervised by health professionals. It has been suggested that transitioning exercise from hospitals, as an intervention, to community gyms, as a leisure activity, could facilitate physical activity and encourage self-management (Wallace et al., 2019). This hypothesis was tested in 23 people with CMT1A and 17 people with Inclusion Body Myositis (IBM), they were included in a 12-week aerobic training program using exercise bicycles compared to a control period. The training occurred 3 times per week in community gyms local to the participants. Support was available from trained gym staff and a research physiotherapist (Wallace et al., 2019). The study findings provided evidence that aerobic training programs increase aerobic capacity for people with CMT1A and IBM. They showed that supervised aerobic training in community leisure facilities is feasible and safe. However, there was a low rate of continuation following the study related to time, expenses, and selfconfidence as reported by study participants (Wallace et al., 2019). Therefore, developing an approach to self-management support can improve adherence to physical activity, help overcome challenges, and improve the quality of life in people with neuromuscular conditions.

People with CMT carry out a low level of activity in comparison to healthy controls, which correlated with muscle strength in adults and children (Menotti et al., 2014a). Physical activity and quality of life are influenced by the CMT severity as well as mental status. The mental domain of SF-36 (the Short Form 36 Health Status Survey for assessment of quality of life) strongly correlate with the CMT exam score (r = 0.75, p = 0.03) (Roberts-Clarke et al., 2016b). This raises the possibility that the negative effects of CMT on quality of life were more related to how the individual coped with the condition psychologically, than with the specific clinical realities of how severely they were affected. People with CMT reported that fatigue has a major impact on physical activity (Ramdharry et

al., 2012c). A later exploratory study revealed that personal factors such as fatigue, poor balance, muscle weakness, and pain were important barriers for physical activity (Anens et al., 2015). While self-efficacy for physical activity, activity-related factors, and assistive devices were facilitators of physical activity (Anens et al., 2015). Using guidance, modelling, and feedback to increase self-efficacy can be utilized for behavioural modification. Health systems and professionals need to be aware of the barriers to physical activity, such as fatigue, to support patients with CMT disease to overcome these barriers, and designing interventions to successfully support engagement in physical activity need to be multi-level to achieve meaningful change, e.g. education and supported self-management programs (Buscemi et al., 2023).

Chapter 3: Study Objectives

Based on the gaps in knowledge explored in chapter 2, I am presenting in Figure 7 the main objectives for the thesis and associated research questions stemming from each objective.



Figure 7: Flowchart outlining the thesis objectives and research questions.

Chapter 4: Methods

This thesis discusses five questions each using similar methods, all of which are described in this chapter. Any methodological details specific to one or another study are indicated in Figure 8.



Figure 8: Summary of the measurements' methods used in each chapter. AFOs, Ankle Foot Orthoses.

4.1. MRI Methods

4.1.1. MRI Protocol

The MRI protocol was primarily developed by the Neuromuscular MRI team from the Queen Square Centre for Neuromuscular Diseases, UCL: Dr Stephen Wastling, Dr Jasper Morrow, and Professor John Thornton. This protocol has been used in a number of exploratory and natural history studies in neuromuscular diseases (Morrow et al., 2016, Kugathasan et al., 2019, Morrow et al., 2018), found to correlate with validated clinical measurements (CMTES) and Dynamometry (Morrow et al., 2016), and have a minimal association with age, gender, and body wight (body mass index) (Morrow et al., 2014)

4.1.2. Hardware

Imaging was performed with a whole-body MR PrismaFit system (Siemens Healthineers) operating at 3 Tesla (T) in body-coil transmit mode with a 36-channel radio frequency (RF) receive coil designed to image the legs (PS matrix coil), which includes a dedicated leg rest to support both calf. A 16-channel foot coil was used to scan one foot. The subjects were imaged feet-first and supine, with thigh and calf imaged separately.

4.1.3. Positioning

A scout-image-based slice prescription was used. The right knee joint space was identified on a coronal scout image and an axial slice through the knee joint space. The main imaging sections was then prescribed such that the central slice was a fixed distance of 20 cm above, and 15 cm below the right knee joint space for thigh and calf, respectively (Fischmann et al., 2014).

4.1.4. Choice of Anatomical Coverage

DHMN is a length-dependent neuropathy, and the pathology is expected to start distally, progressing proximally over time (Rossor et al., 2012b). Lower limb muscles including thigh, calf, and foot were chosen as the region for study as they are a key site of pathology in DHMN and weakness in these areas is a key cause of gait alteration in this patient group. Lower limb imaging has practical advantages since both limbs may be imaged simultaneously, which reduces the scan time (compared to scanning the arms for example). Unlike calf and thigh, both feet cannot be scanned simultaneously hence a single foot was scanned at all visits. Data from a single foot is adequate for the analysis considering the symmetrical pathological nature of peripheral neuropathies (Rossor et al., 2012b). Both limbs were scanned within the field of view (FOV). Axial-slice matrices and FOV is 256x128 and 41x200mm for thigh and calf level images, except for FF acquisitions where matrices were 448x224. A single foot was scanned with

the minimal FOV (>12mm).

4.1.5. MRI Sequences

4.1.5.1. Primary Sequence

Dixon fat fraction measurement

A 3D 3-point Dixon acquisition was performed for a single foot, and both calf and thigh from the upper thigh to ankles. This provides accurate measures of muscle cross sectional area and percentage intramuscular fat accumulation.

Three 3D gradient-echo acquisitions were performed with parameters TE1/TE2/TE3=3.45/4.60/5.75ms, TR=23 ms, flip angle= 5°, bandwidth 450 Hz/pixel, NEX=1, 96 contiguous 5mm slices, 448x224 matrix (160x160 for foot), iPat=2. Phase unwrapping is performed using PRELUDE (FSL, FMRIB, Oxford) (Smith et al., 2004) and after fat (F) and water (W) image decomposition (Glover and Schneider, 1991), FF calculated as FF = 100% x F/(F+W). The TE1=3.45ms image is used for the ROI placement.

4.1.5.2. Secondary Sequences

T1 weighted imaging

Standard T1-weighted images were acquired for one foot and both calf to allow qualitative assessment of intramuscular fat accumulation. Images were acquired with a turbo-spin-echo readout prior to commencing the quantitative protocol (TR/TE=700/8.1ms, 10 slices, 10mm thickness, 10mm slice gap, 256x128

matrix, iPat acceleration of 2, 444 Hz/pixel bandwidth (BW), TSE factor=3, refocusing flip angle (fa) 120°, NEX=1, acquisition time (TA) = 63s).

Short tau inversion recovery imaging (STIR)

STIR imaging was performed for the single foot, both calf, as a qualitative measure of acute muscle denervation or oedema.

For the calf, STIR was acquired with a (TR/TE/Inversion Time = 5500/57/220ms, NEX=1, flip angle 180°, parallel imaging factor =2) imaging performed with (10x10mm slices, 10mm gap, 256x128 matrix).

For the foot, STIR was acquired with a (TR/TE/Inversion Time = 5700/64/220ms, NEX=2, flip angle 180°, parallel imaging factor =2) imaging performed with (30x6mm slices, 1.5mm gap, 192x144 matrix).

<u>T2-Relaxometry (T_{2m}) </u>

Multi-echo T2-relaxometry was performed of a single foot and through both calf. This was analysed to quantify the T2 relaxation time of the water within the muscle, as a marker of acute denervation or muscle injury. Multi-echo turbo spin echo with 22 echoes (TE 10 to 220 ms in 10 ms steps), TR=3500 ms, 1 average, flip angle 180°, 128x128 matrix, iPat 2. T_{2m} was calculated as a multi-component slice profile-corrected EPG model [s(TE) = $(1 - ff_a) \cdot sEPG(B_1f, T_{2m}, \alpha, \sigma_N, TE) + ff_a \cdot [0.33 \cdot sEPG(B_1f, T_2=40ms, \alpha, \sigma_N, TE) + 0.67 \cdot sEPG(B_1f, T_2=198ms, \alpha, \sigma_N, TE)]$ was fitted pixel-wise to the data using maximum likelihood estimation (MLE) in a custom-written MATLAB tool, to estimate T_{2m} , the B_1 field error factor (B_1f), apparent fat fraction (ff_a), overall amplitude (α) and Rician noise SD (σ_N). The fixed 2-component fat signal model parameters were determined in a preliminary calibration as mean values estimated from 4 subcutaneous fat ROIs in 8 representative subjects. The TE1=3.45ms image is used for the ROI placement.

4.1.6. Qualitative Image Analysis

4.1.6.1. Slice Selection

The axial slices were identified at 15cm distal to the right knee joint space for calf scan (Fischmann et al., 2014). Coronal foot slice was identified using the first tarsal metatarsal joint from sagittal view.

4.1.6.2 Qualitative Image Grading

Time and group blinded qualitative analysis of apparent fatty infiltration on the T1 weighted images was performed by grading using the 6-point Mercuri scale (Mercuri et al., 2003) (0=normal,1=mild fatty streaks, 2a=early confluence, 2b=fatty infiltration 30-60%, 3=fatty infiltration >60%, 4=complete fat replacement). For STIR hyperintensity a three point scale (0=none, 1=mild, 2=marked) was used to grade the intensity and a four point scale (0=none, 1=< 30%, 2=30%-60%, 3=>60%) was used to grade the extent. Repeat scans were graded without reference to baseline scans. The same muscles were assessed as in the quantitative analysis below.

4.1.7. Quantitative Image Analysis

4.1.7.1. Slice Selection

The axial slices were identified at 20cm proximal and 15cm distal to the right knee joint space for thigh and calf scan, respectively (Fischmann et al., 2014). Coronal foot slice was identified using the first tarsal metatarsal joint from sagittal view.

4.1.7.2. Region of Interest Definition

I undertook the established training programme for defining regions of interest (ROI) on lower limb muscles and assessed as competent in this before starting analysis in this dataset (Morrow et al., 2024). ROI was defined for each subject on the single slice as described above, at mid-thigh, mid-calf, and mid-foot level of an unprocessed Dixon acquisition (TE=3.45ms) using ITK-SNAP software (Yushkevich et al., 2006). ROIs were defined to encompass the entire muscle cross-sectional area to the fascia (Table 8).

Left and right limb ROIs were defined for the rectus femoris, vastus lateralis, vastus intermedius, vastus medialis, semimembranosus, semitendinosus, biceps femoris, adductor magnus, sartorius, gracilis, tibialis anterior, peroneus longus, lateral gastrocnemius, medial gastrocnemius, soleus, tibialis posterior, and foot muscles (Table 8).

4.1.7.3. Transfer of Region of Interest to Maps

The ROIs were transferred to the co-registered parameter maps (FF, T2) and quantitative parameters extracted. All generated maps were inspected visually for correct ROI placement and presence of artefact. ROI values originating from areas of gross artefact were excluded from the analysis.

4.1.7.4. Data Analysis

For each muscle custom written software extract mean fat fraction (FF), crosssectional area (CSA), and muscle water (T_{2m}) from whole ROI. After extraction, all data was cross-checked for outliers and errors identified were corrected. In addition to individual muscle values, summary measures for each parameter were created for all muscles at thigh level, calf levels, and foot level separately. Total CSA and the remaining muscle area were also calculated. Longitudinal changes were quantified on a muscle-by-muscle, parameter-by-parameter basis, and combined as detailed above to create separate all-muscle summary variables at thigh, calf, and foot levels.



RF= Rectus Femoris, VL= Vastus Lateralis, VM= Vastus Medialis, VI= Vastus Intermedius, Sa= Sartorius, G= Gracilis, AM= Adductor Magnus, SM= Semimembranosus, ST= Semitendinosus, BF= Biceps Femoris, TA= Tibialis Anterior Group, MG= Medial Head Of Gastrocnemius, So= Soleus, TP= Tibialis Posterior, PL= Peroneus Longus, LG= Lateral Head Of Gastrocnemius, M= Foot Muscles, F= Femur, T= Tibia. MT= Metatarsals.

Table 8: Sample axial images of unprocessed Dixon sequence (echo time=3.45 ms) with overlaid muscle regions of interest of the left side for

the same image in a healthy control, mild DHMN case, and severe DHMN case.

4.2. Dynamometry Methods

4.2.1. Hardware

The HUMAC NORM Testing & Rehabilitation System (CSMi, Massachusetts, USA) (Figure 8) was used to explore muscle function. Its components and capabilities are designed to conduct isometric, isotonic, isokinetic, and passive range of motion tests for all major muscle groups. Key components of the HUMAC NORM Dynamometer system include:

<u>Dynamometer Unit:</u> The core of the system equipped is with a torque sensor to measure force across various movements and exercises.

<u>Computer and Software:</u> The system comes with the HUMAC application program (CSMi, Massachusetts, USA) to control the dynamometer, collect data, and analyse performance. The software provides real-time feedback, detailed reports, and allows for customization of testing protocols.

<u>Adjustable Chair or Bench</u>: This ensures that the participant can be positioned correctly for a wide range of tests, accommodating different exercises and body sizes.

<u>Attachments and Accessories:</u> Various grips, straps, and attachments allow for a broad spectrum of muscle testing and rehabilitative exercises, targeting specific muscle groups accurately.

<u>Safety Features:</u> Includes emergency stop mechanisms and adjustable limits on range of motion to ensure the safety of users during testing and rehabilitation sessions.



Figure 9: HUMAC NORM Testing & Rehabilitation System (CSMi, Massachusetts, USA).

4.2.2. Parameters Measured

4.2.2.1. Isometric Torque

The NORM system measures isometric Peak torque (Newton-Meter) while the participant is exerting maximum force against a stationary arm at a specific joint angle.

4.2.2.2. Isokinetic Torque

The NORM system measures isokinetic torque (Newton-Meter) instantaneously every half degree in the range of motion. Peak torque is the maximum torque production during movement throughout the range of motion.

4.2.2.3. Gravity Effect Torque

When testing with the NORM System, the gravity effect torque of the limb and the input adapter was determined; the computer can then correct for the effect of gravity. This is called Maximum Gravity Effect Torque (MAX GET). The computer uses this value and multiplies it by the cosine of the angle at each point in the participant's range of motion. This is the amount of torque that is being contributed or taken away due to the weight of the limb and input accessory. For example, during knee extension, when gravity resists the motion, the appropriate gravity effect torque value is added, and the reported torque will be calculated as:

*Reported Torque = Measured Torque + (MaxGET * Cosine (Angle))*

During knee flexion, when gravity assists the patient, the appropriate gravity effect torque value is subtracted, and the reported torque will be calculated as:

*Reported Torque = Measured Torque - (MaxGET * Cosine (Angle))*

4.2.3. Dynamometry Protocol

Patients and controls had detailed lower limb dynamometry on a HUMAC NORM dynamometer (CSMi, Massachusetts, USA). Dynamometry was

performed according to Table 9. Hip extension, hip flexion, knee extension, knee flexion, ankle dorsiflexion and ankle plantarflexion were assessed bilaterally using both isometric and isokinetic protocols and the maximum torque in Newton-Meter (Nm) was recorded for analysis. Isometric assessments consisted of three attempts following a practice run of 3 seconds duration with 10 seconds interval of which the best attempt was selected. For the isokinetic assessments, following a practice run and 10 seconds interval, three successive movements through full range were performed and the highest value obtained was selected. The machine setup was recorded at first visit using the included software, which was then retrieved to allow identical set-up on repeat testing. Isometric and Isokinetic data were exported as a report in PDF format. This protocol has been used in previous research in similar neuromuscular conditions (Wallace et al., 2019) (Morrow et al., 2016).

Joint	Muscle	Movement	Angle(°)/Speed(s)
	Function		
Hip	Isometric	Extension	45°
		Flexion	45°
	Isokinetic	Extension/ Flexion	60°/60s
Knee	Isometric	Extension	45°
		Extension	90°
		Flexion	45°
		Flexion	90°
	Isokinetic	Extension/Flexion	60°/60s
		Extension/Flexion	120°/120s
Ankle	Isometric	Plantarflexion	10°
		Dorsiflexion	10°
		Dorsiflexion	30°
	Isokinetic	Plantarflexion/Dorsiflexion	60°/60s

Table 9: Dynamometry protocol.
4.3. Three-Dimensional Motion Analysis

4.3.1. Kinematics

To describe 3D kinematics, the rigid link segment model is used. In such method, the lower part of the body is modelled as a rigid chain of segments interconnected by joints, assuming that the length of each defined segment is a constant distance between the centre of the proximal and distal joint and the centre of mass for each segment is located in the middle of the mass. In gait analysis, segments including foot, leg, thigh, and pelvic are used. These segments are connected with ankle, knee, and hip joints respectively (Robertson DGE et al., 2014b). The model measures Euler rotations (Figure 10) of segments in relation to the pelvis.





Euler sequence XYZ of rotations first about (a) the X-axis of the stationary coordinate system (α); then about (b) the new y¹-axis (β); and finally, about (c) the z²-axis (γ). Figure adapted from (Robertson DGE et al., 2014).

4.3.1.1. Segmental Angle Kinematics

The model measures the amount of Euler displacement of a segment in relation to the local coordinate system during gait. A local coordinate system or embedded vector bases (EVB) is consisted of three orthogonal axis (Table 10, Figure 11) and defined for each segment using the joint centre and markers including virtual and actual markers. The centre of each joint, and relevant virtual markers was automatically computed using markers placed on specific anatomical points and measurements using a calliper. Markers extended away from the body, on the pelvic frame, thigh wand, and tibial wand, were used to improve visibility and to capture segmental rotation around the uz axis at the local coordinate system (Table 10, Figure 11) (CharnwoodDynamics, 2004, Robertson DGE et al., 2014a).

EVB	Principal axis	2 nd axis	3 rd axis
	Line connecting the heel	Line running from the	Orthogonal to 1 st and 2 nd
Foot	and toe markers that is	heel marker to ankle	axes (u _y)
FUUL	offset by 1/2 inter-	marker and orthogonal to	
	malleolar distance (u _x)	the principal axis (uz)	
	Ankle joint centre to knee	Tibial wand orientation	Orthogonal to 1 st and 2 nd
Shank	joint centre (uz)	orthogonal to principal	axes (u _y)
		axis (u _x)	
	Knee joint centre to hip	Thigh wand orientation	Orthogonal to 1 st and 2 nd
Thigh	joint centre (uz)	orthogonal to principal	axes (u _y)
		axis (u _x)	
	Line between right and	Line connecting mid PSIS	Orthogonal to 1 st and 2 nd
Dolvis	left ASIS markers (uy)	to mid ASIS and	axes (u _z)
1 CIVIS		orthogonal to the	
		principal axis (ux)	

Table 10: Derivation of embedded vector basis (EVBs) for each segment.

u; unit vector, x; axis x, y; axis y, z; axis z.



Figure 11:Segment Embedded Coordinates in sagittal (A) and frontal (B) view. Adapted from CODAmotion ODIN user guide.

4.3.1.2. Inter-Segmental (Joint) Angle

The intersegmental joint angle is the Euler angle between two segments on either side of the joint. The rotation of the distal segment about the proximal segment occurred in the following sequence: Z (internal/external axial rotation) then X (medio-lateral bending) then Y (flexion/ extension). Distal segmental orientation (distal EVB) was described in relation to the proximal segment (proximal EVB) except pelvis and foot segments were described according to the global coordinate system (GCS) (Table 11). Once the angles were known, the time derivatives were calculated to give angular velocity and acceleration (CharnwoodDynamics, 2004).

Joint Angle	Distal Segment	Proximal Segment
Hip Joint Angle	Thigh EVB	Pelvis EVB
Knee Joint Angle	Shank EVB	Thigh EVB
Ankle Joint Angle	Foot EVB	Shank EVB
Pelvic Rotation	Pelvis EVB	GCS
Foot Rotation	GCS	Foot EVB

Table 11: Relation of distal segment Euler angles to a reference frame.EVB; Embedded vector basis, GCS; global coordinate system.

4.3.2. Kinetics

Inverse dynamics is a biomechanical approach where Equations of motion are used to calculate Joint forces, moments, and power from Segmental inertial characteristics, Kinematics (positions, velocities, accelerations), and External forces (ground reaction force) (Robertson DGE et al., 2014c).

Kinematics data are measured as described in the previous section. External forces (ground reaction force) are measured using force plates impeded in the walkway. Segmental inertial characteristics are estimated from anthropometric data based on the subjects' height, weight, and gender as they are important in calculating the segment length, segment mass, and centre of mass. To simplify inertial properties, each segment is assumed to have fixed and uniform distribution of mass around a longitudinal axis connecting the joint centres. Once kinematics, external force, and inertial characteristics are known, automatically computed scripts are used to solve equation of motions and identify forces, moments, and power around the lower limb joints (CharnwoodDynamics, 2004). The net moment or torque at the major joints during walking, is the sum of all the active moments generated by muscle contraction, and passive moments generated

by soft tissue, ligaments, and bone to bone force. Joint power represents the rate of change of energy. Regions of positive power represent power generation through concentric muscle activity. And regions of negative power represent power absorption through eccentric muscle action.

4.3.3. Lab Settings and Equipment

4.3.3.1. Hardware

The CODA motion CX-1 3-D Motion tracking system (Coda cx-1; Charnwood Dynamics, Leicestershire, UK) was used for calculations of 3-dimensional segmental gait analysis. The measurement unit contains three pre-aligned solid-state cameras which track the position of a number of active markers (infra-red LEDs) in real-time with 200Hz sampling rate. The interfacing for synchronous analogue and digital data acquisition enables measuring the ground reaction force and torque using two force plates impeded in the walk walkway (AMTI's AccuGait, Watertown, MA, USA).

4.3.3.2. Marker Placement

The Helen Hayes model with 24 markers for the lower body was used as suggested by the CODA motion system company. The different position of wands, marker drive boxes, and markers are described on Figure 12 (CODAmotion, 2016).



Figure 12: Helen Hayes model for marker placement. Adapted from CODAmotion ODIN user guide.

4.3.4. Gait Parameters and Analysis

ODIN software (Codamotion Ltd., Rothley, UK) was used to analyse and export reports of 3 gait cycles of left and right side for each participant each visit. A single gait cycle was identified from the heel marker acceleration traces and vertical forces. After exporting to Microsoft Excel, the three cycles were averaged and the data from the left and right legs were averaged. Exported variables are listed in Table 12.

	Polyic z Maximum (Rotation forwards)	
Horizontal Angle (°)	Pelvic z Minimum (Rotation backwords)	
	Pelvic x Maximum (Obliquity, lateral raise)	
Coronal Angle (°)	Pelvic x Minimum (Obliquity, lateral drop)	
	Pelvic v Maximum (Anterior tilt)	
	Pelvic v Minimum (Posterior tilt)	
	Hip v Maximum (Flexion)	
	Hip v Minimum (Extension)	
Sagittal Angle (°)	Knee y Maximum (Flexion)	
	Knee v Minimum (Extension)	
	Ankle v Maximum (Dorsiflexion)	
	Ankle y Minimum (Plantarflexion)	
	Hip Maximum (Extension)	
	Hip Minimum (Flexion)	
Sagittal Moments	Knee Maximum (Extension)	
(Nm/Kg)	Knee Minimum (Flexion)	
	Ankle Maximum (Plantarflexion)	
	Ankle Minimum (Dorsiflexion)	
	Hip Maximum (generation in swing phase)	
	Hip Minimum (absorption in stance phase)	
	Hip Maximum (generation in stance phase)	
	Knee Maximum (generation in stance phase)	
Sagillar Power (W/I_{rec})	Knee Maximum (generation in swing phase)	
$(\mathbf{v}\mathbf{v}/\mathbf{k}\mathbf{g})$	Knee Minimum (absorption in stance phase)	
	Ankle Maximum (generation in stance phase)	
	Ankle Minimum (absorption in stance phase)	
	Ankle Maximum (generation in swing phase)	
	Speed (m/s)	
	Stride Length (m)	
	Stride Time (s)	
	Strides / Minute	
	Step Length (m)	
Spatiatamporal	Step Time (s)	
Spanotemporal	Steps / Minute	
	Percent Stance (%)	
	Single Support (s)	
	Double Support (s)	
	Opposite Toe Off %	
	Opposite Foot Contact (%)	

Table 12: Gait parameters used for primary and secondary analysis.

4.4. Clinical Assessment

Along with primary measures, clinical assessment was undertaken each visit for descriptive purposes of the cohort and to correlate with the primary measurements, including CMT Examination Score version 2 (CMTESv2), Foot Posture Index (FPI-6), Manual Muscle Testing (MMT), The Visual Analog Scale (VAS) of Pain, and Walk-12 Questionnaire.

4.4.1. CMT Examination Score Version 2 (CMTESv2)

The Charcot-Marie-Tooth neuropathy score (CMTNS) is a reliable and validated composite scale of nine assessments including symptoms (three items), signs (four items), and neurophysiology (two items) (Shy et al., 2005). CMT Examination Score (CMTES) is a sub score of the CMTNS, calculated by the sum of the symptoms plus the signs. It is designed to measure length-dependent motor and sensory impairment in genetic neuropathies (Shy et al., 2005). Each assessment is scored on a 0 to 4 point scale, reflecting severity of impairment. In version 2, CMTNS was modified to reduce floor and ceiling effects and to standardize patient assessment, aiming to improve its sensitivity for detecting change over time and the effect of an intervention (Murphy et al., 2011). CMTNS2 is a reliable scale in patients with CMT, with high inter- and intra-rater reliability for its clinical symptoms and signs components (Murphy et al., 2011).

4.4.2. Foot Posture Index (FPI-6)

The Foot Posture Index-6 (FPI-6) is a simple and reliable quantification tool to assess static foot alignment in all three planes (Redmond et al., 2006, Keenan et al., 2007). The Foot Posture Index comprises six individual parameters, to which a score from -2 to +2 can be allocated. Negative numbers indicate supination, and positive numbers indicate pronation. The total result defines the foot posture and ranges between -12 and +12 (Redmond et al., 2006).

4.4.3. Manual Muscle Testing (MMT)

Muscle strength was assessed using a modified Medical Research Council Scale (MRC) with 5: normal strength, 5-: barely detectable weakness, 4+: gravity and moderate to maximal resistance, 4: Gravity and moderate resistance, 4-: Gravity and minimal resistance, 3: Full range of motion against gravity only, 2: Movement when gravity is eliminated, 1: Flicker of movement seen or felt, 0: No movement (O'Brien, 2023).

4.4.4. The Visual Analog Scale (VAS) of Pain

The Visual Analog Scale (VAS) is a validated, subjective measure of a person's pain intensity. It is a simple and commonly used tool in clinical and research settings (Delgado et al., 2018). The VAS is typically a horizontal line, 10 centimetres in length, anchored by two verbal descriptors, one for each end. "No pain" at 0 and "worst imaginable pain" at 10. Patients mark on the line the point

that they feel represents their perception of their current state of pain (Huskisson, 1974, Delgado et al., 2018).

4.4.5. Walk-12 Questionnaire

Patient reported outcome measure designed to assess walking ability and impairment in individuals, particularly those with peripheral neuropathy. The questionnaire evaluates three key aspects of walking: pain severity, distance, and speed. It consists of 12 items that ask respondents to rate their difficulty in walking various distances, walking speeds, and the degree of pain or discomfort experienced while walking. A higher score indicates a greater limitation in perceived walking ability (Graham and Hughes, 2006).

4.5. Participants

The disease investigated was DHMN. Our target was to include 20 people living with DHMN and 20 matched controls. The key inclusion and exclusion criteria are in Table 13. However, due to the restrictions resulting from the COVID-19 pandemic, we were only able to recruit a total of 12 DHMN participants and 9 matched controls.

Recruitment of participants living with DHMN was through the appropriate NHS consultant responsible for their care at the National Hospital for Neurology and Neurosurgery in London. Candidate participants with DHMN were informed about the study primarily when they attended a routine clinical appointment (remotely or in person) and their interest in the trial was assessed.

Healthy controls participants were recruited among relatives, spouses, and carers of the participating patients. controls were also recruited through staff in UCL departments and at the National Hospital for Neurology and Neurosurgery or their family members.

Written informed consent was obtained from each participant prior to participation in the trial, following adequate explanation of the aims, methods, anticipated benefits, and potential hazards of the trial.

		DHMN		Control
	1.	Adult patients aged 18 or above.	1.	Adult aged 18 or above.
	2.	Clinical diagnosis and electrophysiological	2.	Participants who have no known
		features of DHMN.		neuromuscular disease.
	3.	Participant is able to give informed consent	3.	Participant is able to give informed
Ι		and has signed the informed consent form.		consent and has signed the informed
nc	4.	Patients who are able to complete calf and		consent form.
lus		thigh dynamometry and 3D gait analysis.	4.	Participants who are able to
ion	5.	Patients who have no contraindication to		complete calf and thigh
-		MRI scanning.		dynamometry and 3D gait analysis.
	6.	Participant for the exercises trial, at least one	5.	Participants who have no
		of the major muscle groups of the ankle,		contraindication to MRI scanning.
		dorsiflexors or plantar flexors, score over 4/5		
		on the MRC scale.		
	1.	Known neuromuscular disorder other than	1.	The participant has a
		DHMN.		contraindication for MRI.
	2.	Lower limbs surgery planned during the	2.	Participants who cannot complete
Ę		study period or performed within the		calf and thigh dynamometry and 3D
KC		preceding 12 months.		gait analysis.
usi	3.	Bilateral ankle arthrodesis.	3.	Participant has a history, signs or
on	4.	The participant has a contraindication for		symptoms of a neuromuscular
		MRI scan.		disorder.
	5.	Females who are pregnant, planning	4.	Females who are pregnant, planning
		pregnancy or breastfeeding.		pregnancy or breastfeeding.

Table 13: Inclusion and exclusion criteria.

4.6. Ethical Framework

This study was reviewed and given favourable opinion by the London – Camden and Kings Cross Research Ethics Committee and the Health Research Authority (HRA). People from Charcot Marie Tooth UK charity was involved in reviewing the participants information sheet.

4.7. Trial Design and Procedure

The study was three main visits. In each visit, study measurements were collected as explained in previous sections. Visit activity for each participant are illustrated in Table 14.

	Visit I (Baseline)			Visit II (6 Months)	Visit III (12 Months)
Activity	A) MRI, HUMAC, 3D Gait Analysis	B) 3D Gait AnalysisShoesCarbon fibre AFOs	C) Test eligibility and recruit for the exercises trial	MRI, HUMAC, 3D Gait Analysis	MRI, HUMAC, 3D Gait Analysis
Subjects	DHMN, Controls	DHMN	DHMN	DHMN-Exercise, DHMN-No exercise	DHMN-Exercise, DHMN-No exercise
Objectives	Comparison to identify deviations	Effect of carbon fibre AFOs on walking	Randomise DHMN participants into: -Group A:Exercises -Group B: No exercise	Effect of resistance training on muscle.	-Explore natural history. -Explore muscles structure and function after Exercises washout period

Table 14: Visit activity for each participant.

4.8. Overall Statistical Considerations

4.8.1. Sample

Rather than a formal sample size calculation, convenience sampling was undertaken as one of the objectives of this study was initial exploration to calculate the effect size to conduct a sample size calculation for a larger trial. Recruiting a large sample was challenging due to the low prevalence of the disease. Distal Hereditary Motor Neuropathies (DHMN) affects 2.14 individuals per 100,000 (Bansagi et al., 2017). The sample was recruited from the National Hospital for Neurology and Neurosurgery since it is a specialist hospital accepting referrals of neurological inherited diseases from most of the country. However, convenience sampling is often criticized for its lack of representativeness and higher risk of sampling bias, potentially limiting the generalizability of the findings to the broader population (Etikan et al., 2016).

4.8.2. Statistical Analysis

Demographic data (age, gender, weight, height) were collected. These measures are expressed as means (for continuous data) or medians (categorical data). Comparison was done using unpaired t-tests for continuous data, and Mann Whitney U tests for nominal or categorical data.

4.8.3. Difference Between Right and Left Sides

Due to the symmetrical nature of the condition, the decision was made to analyse the primary and secondary parameters for each objective using the average between right and left sides. Moreover, averaging right and left sides could eliminate type 1 error caused by multiple comparisons if both sides analysed separately. To ascertain symmetry in our sample, difference between right and left side in the primary parameters was tested using unpaired T.Test for normally distributed data and Mann-Whitney test for non-normally distributed data. The test showed no significant difference in all primary parameters as shown in appendix I.

4.8.4. Data Normality Test

An assessment of the normality of data is a prerequisite for many statistical tests because normal data is an underlying assumption in parametric testing. Since the data collected for this study was from a small sample, less than 50 participants, the Shapiro-Wilk test was used to assess normality. Where the data was normal, parametric tests were used if the Significance value of the Shapiro-Wilk Test was greater than 0.05. If below 0.05, the data significantly deviates from a normal distribution and non-parametric tests were used for analysis.

4.8.5. Strategies to Minimise Statistical Errors

Multiple comparisons have higher probability of type I (false positive) error, defined as a statistically significant difference by chance when there is no real finding. To reduce the chance of type I error, first, the number of comparisons was decreased by using the average between right and left, rather than separately for each side as explained in section 4.8.3. The second approach used to account for multiple comparisons was adjusting the alpha value using modified Bonferroni procedure. The classical Bonferroni correction is considered too conservative as the correction does not account for dependency among the data (Simes, 1986). In addition, there is also the potential to increase type II (false

negative) error as the test shows no statistically significant difference when there is. A modified Bonferroni procedure was used as it is less conservative and based on the ordered P-values of individual tests which decreases type II error probability (Simes, 1986). The procedure is demonstrated in appendix I.

Chapter 5: Pattern of Involvement and Deviation from "Normal" in Muscle Structure and Function in Distal Hereditary Motor Neuropathy (DHMN)

5.1. Introduction

To address the first objective for this study, which is to explore the effect of Distal Hereditary Motor Neuropathy (DHMN) on muscle structure and function, we need to establish an understanding of the morphological differences and functional alterations associated with the condition. Previous MRI studies described DHMN groups with a predominant involvement of the lower leg, mainly in the posterior and lateral compartments. Which was associated with distal lower limb weakness (Esteller et al., 2023, O'Donnell et al., 2022). Other studies in CMT described gait alterations relevant to distal weakness as foot drop and/or plantarflexes failure with the possibility of proximal compensation (Don et al., 2007, Newman et al., 2007, Ramdharry et al., 2009). However, gait in DHMN has not been described before. This study is aimed to identify patterns of muscle involvement and walking gait deviations in people with DHMN in comparison to healthy controls. Based on previous findings in CMT, we expect our DHMN cohort to show:

• Higher rates of fatty infiltration and signs of active denervation distally more proximally, and in the posterior compartment more than anterior compartment of the calf.

- Predominant distal weakness more than proximal, and in plantar flexors more than dorsiflexors.
- Decreased ankle plantarflexion angle, moment, and power generation at preswing.
- Decreased ankle dorsiflexion angle in swing phase.
- Increased hip flexion angle, moment, and power generation at pre-swing and swing phase.
- Slow gait with shorter stride length.

5.2. Methods

Participants with DHMN were recruited to undergo clinical assessment, an MRI scan, isokinetic and isometric dynamometry of the lower limb, and 3D motion analysis to capture kinetic and kinematic data of walking gait as described in chapter 4. For direct comparison of gait deviations and muscle involvement, age and gender matched healthy controls were recruited to undergo the same measurements.

5.2.1. Data Analysis

To identify pattern of involvement and deviations from normal, comparisons of DHMN and control parameters of MRI, isokinetic/isometric dynamometry, and motion analysis using unpaired t-test, if both data sets were normally distributed or Mann-Whitney U test, if they are not normally distributed. A modified Bonferroni procedure was used to account for multiple comparisons. However, due to the large number of comparisons and small sample, it was expected that few of the group differences will show significant. Therefore, it was decided to look for non-significant trends in the data using p<0.05 as a guide.

5.3. Results

5.3.1. Subjects and Clinical Assessment

Twelve DHMN participants and 9 age and gender matched healthy controls were recruited. There were no significant differences observed in the groups matching criteria. One DHMN female participant used walking sticks. Two male DHMN participants with HSPB1genetic diagnosis did not complete all measurements, one aged 53 completed MRI scan only, and one aged 41 completed all assessment except for the MRI scan (appendix VII). Assessment showed limitation in plantarflexion and dorsiflexion passive and active range of motion. Comparison between DHMN and control groups showed significant difference in manual muscle testing in dorsiflexion and plantarflexion strength (P=0.0001). A summary of the subjects' demographics and clinical assessment is presented in Table 15.

Demographics and clinical assessment	DHMN	CONTROL	Р		
Numbers	12	9	-		
Gender (M/F)	(7/5)	(4/5)	-		
Age; mean years; range	56 (41/75)	50 (29/76)	-		
Differences in groups matching criteria (M/F)	-	-	(0.47/0.23)		
Genetic diagnosis (HSPB1/unknown)	(8/4)	-	-		
CMTES; mean (SD)	6.2 (3.6)	-	-		
Foot Posture Index-6 (Pronated/Normal/Supinated)	(1/8/1)	-	-		
Fall Frequency (weekly/Monthly/Yearly)	(1/1/2)	-	-		
Walk-12 Questionnaire; mean (SD)	36 (12.4)	-	-		
Range of Motion	DHMN	CONTROL	Р		
Dorsiflexion; count (Limited/Normal)	(12/0)	(0/9)	-		
Plantarflexion; count (Limited/Normal)	(11/1)	(0/9)	-		
Manual Muscle Testing	DHMN	CONTROL	Р		
Hip Flexion; median (IQR)	5 (5-4+)	5 (5-5)	0.0827		
Hip Extension; median (IQR)	5 (5-4-)	5 (5-5)	0.0824		
Knee Flexion; median (IQR)	5 (5-4-)	5 (5-5)	0.0824		
Knee Extension; median (IQR)	5 (5-5)	5 (5-5)	0.1674		
Dorsiflexion; median (IQR)	3+ (4-1+)	5 (5-5)	0.0001*		
Plantarflexion; median (IQR)	4 (4+-3)	5 (5-5)	0.0001*		
M- Mala E- Eamala USDD1- Hast shock 27 KD Protoin 1 SD- Standard Deviation CMTES					

M= Male, F= Female, HSPB1= Heat-shock 27-KD Protein 1, SD= Standard Deviation, CMTES= Charcot-Marie-Tooth Disease Examination Score, IQR= interquartile range, *= significant after modified Bonferroni correction.

Table 15: Summary of demographics and clinical assessment of DHMN and control groups

5.3.2. Pattern of Muscle Structural Involvement on MRI

Three point Dixon images were processed to generate fat fraction (FF) maps.

Region of interest at the thigh, calf, and foot levels were applied to the FF maps

to quantify the cross sectional area, intramuscular fat fraction, and the remaining

muscle area.

5.3.2.1. Muscle Cross Sectional Area

Quantitative analysis showed no significant difference in muscle cross sectional area between DHMN and control group, neither at the thigh, calf, or foot level (Table 16).

Region of interest	DHMN	CONTROL	Mean diff	P*			
mm ²	mean (SD)	mean (SD)	Witcan uni	-			
Total Thigh	21877.7(7024.5)	20184.2(5093.4)	-1693.53	0.55			
Knee Extension	10748.8(3283)	10053.7(2985.9)	-695.07	0.60			
Rectus Femoris	474.9(187.6)	436.5(160.9)	-38.40	0.63			
Vastus Intermedius	1601.5(524.3)	1778.3(505.1)	176.75	0.46			
Vastus Lateralis	1824.5(467.7)	1687.7(468.8)	-136.78	0.26			
Vastus Medialis	1473.5(616.7)	1124.4(490.7)	-349.11	0.29			
Knee Flexion	7975.2(2956.7)	6959.8(2407.9)	-1015.39	0.41			
Semimembranosus	894.5(381.9)	838.2(437.4)	-56.26	0.46			
Semitendinosus	791.6(350.7)	657.1(264.6)	-134.53	0.36			
Biceps Femoris	1478.7(566.6)	1312.9(440.6)	-165.75	0.48			
Sartorius	374.2(132.3)	326.5(112.7)	-47.65	0.40			
Gracilis	448.7(190.4)	345.2(128.1)	-103.50	0.18			
Adductor Magnus	1576.9(721.1)	1585.3(604.5)	8.46	0.98			
Total Calf	10565.6(3102.8)	11572.9(1904.7)	1007.28	0.41			
Dorsiflexion	2005.7(783.7)	2063.5(261.7)	57.83	0.84			
Tibialis Anterior	1002.9(391.8)	1031.8(130.9)	28.92	0.84			
Peroneus Longus	475.0(144.1)	524.1(135.8)	49.11	0.45			
Plantarflexion	6838.8(2286.0)	7610.5(1434.4)	771.62	0.39			
Lateral Gastrocnemius	665.1(439.5)	603.1(133.5)	-62.01	0.82			
Medial Gastrocnemius	800.5(358.1)	1046.0(301.0)	245.49	0.12			
Soleus	1953.9(607.1)	2156.2(374.0)	202.33	0.40			
Tibialis Posterior	385.6(166.5)	425.4(120.6)	39.80	0.56			
Foot	485.4(123.6)	434.9(82.7)	-50.46	0.31			
SD= Standard Deviation, mm ²	e square millimetres, dif	f= difference, *= signi	ificance level w	vith			
modified Bonferroni correctio	modified Bonferroni correction.						

Table 16: Muscle cross sectional area at the thigh, calf, and foot level of DHMN and control

groups

5.3.2.2. Quantitative Analysis of Intramuscular Fat Fraction

Muscle fat fraction analysis showed significant differences at the foot (P<0.0001), and the total calf level (P=0.0007) (Table 17) (Figure 13). At the thigh level, however, significant difference was in the Biceps Femoris muscle only (P= 0.0459) (Table 17).

Region of interest %	DHMN mean (SD)	CONTROL mean (SD)	Mean diff	Р
Total Thigh	12.4(12.6)	5(1.8)	-7.4	0.13
Knee Extension	11.4(13.4)	4.5(1.8)	-7	0.23
Rectus Femoris	9.2(9.3)	7.1(2.8)	-2.1	0.55
Vastus Intermedius	12.5(14.4)	3.5(1.7)	-8.9	0.07
Vastus Lateralis	13.0(16.5)	4.6(1.9)	-8.3	0.37
Vastus Medialis	8.5(8.3)	4.8(2.1)	-3.7	0.41
Knee Flexion	13.6(12.5)	6.3(2.4)	-7.3	0.08
Semimembranosus	14.3(13.0)	6.2(2.9)	-8.2	0.06
Semitendinosus	12.4(14.9)	5.2(2.2)	-7.2	0.23
Biceps Femoris	15.7(13.5)	5.9(2.1)	-9.8	0.0459*
Sartorius	16.4(17.1)	9.9(4.6)	-6.5	0.88
Gracilis	8.7(9.8)	7.0(3.6)	-1.7	0.88
Adductor Magnus	14.5(18.2)	4.3(1.3)	-10.6	0.03
Total Calf	32.2(21)	3.2(1)	-28.9	0.0007*
Dorsiflexion	18.(18.8)	2.7(0.60)	-35.2	0.0031*
Tibialis Anterior	18.0(18)	2.3(0)	-15.8	0.0031*
Peroneus Longus	29.8(28)	4.5(1)	-25.4	0.0145*
Plantarflexion	38.7(23)	3.5(1.3)	-15.8	0.0002*
Lateral Gastrocnemius	28.6(28)	4.1(1)	-24.5	0.0002*
Medial Gastrocnemius	41.0(22)	3.6(1)	-37.4	0.0001*
Soleus	42.2(24)	3.2(1)	-39	0.0001*
Tibialis Posterior	22.9(22)	2.1(0)	-20.8	<0.0001*
Foot	49(12.3)	6.2(2.2)	-42.7	<0.0001*

SD= Standard Deviation, diff= difference, *= significant with modified Bonferroni correction. *Table 17: Muscle Fat Fraction at the thigh, calf, and foot level of DHMN and control groups*



Figure 13: Box and whisker plot of the total calf fat fraction (%) in DHMN and control group.

5.3.2.3. Remaining Muscle Area

Analysis of the remaining muscle area showed significant difference at the foot (P=0.0038) and the calf level (P=0.0126). The difference at the calf level caused by a significant reduction of muscle area in the total plantar flexors (P=0.0082). in particular the Medial Gastrocnemius (P=0.0057), and Soleus (P=0.0029) muscles (Table 18).

Region of interest	DHMN mean (SD)	CONTROL mean (SD)	Mean diff	Р
Total Thigh	19735.1(7540.5)	19199.7(5048.8)	-535.5	0.82
Knee Extension	9804.6(3766.4)	9622.4(2968.9)	-182.2	0.88
Rectus Femoris	439.7(197.7)	407.4(159.6)	-32.2	0.94
Vastus Intermedius	1447.1(600.3)	1716.9(496.5)	269.8	0.30
Vastus Lateralis	1631.3(592.4)	1612.9(470.1)	-18.4	0.77
Vastus Medialis	1384.2(632.6)	1073.5(482.4)	-310.8	0.46
Knee Flexion	7139.7(3045.3)	6545.6(2401.6)	-594.2	0.50
Semimembranosus	797.9(376.6)	791.7(434.3)	-6.2	0.66
Semitendinosus	722.3(354.5)	623.2(256.1)	-99.1	0.49
Biceps Femoris	1302.6(606.1)	1239.9(435.9)	-62.7	0.80
Sartorius	326.3(150.9)	295.1(109.2)	-31.2	0.77
Gracilis	420.8(192.0)	323.1(128.2)	-97.6	0.21
Adductor Magnus	1395.4(702.3)	1515.9(576.1)	120.5	0.68
Total Calf	7454.9(3677.1)	11201.5(1860.1)	3746.6	0.0126*
Dorsiflexion	1744.3(910.6)	2016.8(254.1)	272.5	0.40
Tibialis Anterior	872.1(455.3)	1008.4(127.1)	136.3	0.40
Peroneus Longus	337.0(165.7)	500.7(131.3)	163.8	0.027^{\dagger}
Plantarflexion	4420.2(2661.5)	7350.2(1413.9)	2930.0	0.0082*
Lateral Gastrocnemius	529.1(478.6)	578.9(130.9)	49.8	0.46
Medial Gastrocnemius	503.8(371.1)	1009.3(295.8)	505.5	0.0057*
Soleus	1177.2(718.1)	2086.9(364.3)	909.8	0.0029*
Tibialis Posterior	308.2(183.6)	416.5(117.9)	108.3	0.14
Foot	256.8(113.7)	408.7(84.2)	152	0.0038*
SD= Standard Deviation, mn	n ² = square millimetres,	$\dagger = significant with P < 0$	0.05, diff= di	fference, *=

significant with modified Bonferroni correction. *Table 18: Remaining Muscle Area at the thigh, calf, and foot level of DHMN and control*

groups

5.3.2.4. Qualitative Analysis of Intramuscular Fat

Qualitative analysis of intramuscular fat using Modified Mercuri's Scale showed consistent results as the intramuscular fat quantification at the calf and foot level (Table 19) (Figure 14). The difference was significant at the foot (P<0.0001) and all calf muscles (P=0.0003-0.0034).

Region of interest	DHMN median (IQR)	CONTROL median (IQR)	Р
Tibialis Anterior	2.5 (2.0 to 3.375)	1 (1.0 to 1.0)	0.0034*
Tibialis Posterior	3(2.0 to 4.875)	1 (1.0 to 1.0)	0.0034*
Peroneus Longus	3.5(2.0 to 4.875)	1 (1.0 to 1.0)	0.0035*
Medial Gastrocnemius	4 (3.125 to 5.0)	1 (1.0 to 1.0)	0.0003*
Lateral Gastrocnemius	3 (2.0 to 5.0)	1 (1.0 to 1.0)	0.0012*
Soleus	4.5 (3.0 to 5.375)	1 (1.0 to 1.0)	0.0003*
Foot	6 (6.0 to 6.0)	1 (1.0 to 1.0)	<0.0001*

Modified Mercuri's Scale: 1=Stage 0= Normal appearance; 2=Stage 1= Early Moth-eaten appearance, with scattered small areas of increased signal; 3=Stage 2a= Late Moth-eaten appearance, with numerous discrete areas of increased signal with beginning confluence, comprising less than 30% of the volume of the individual muscle; 4=Stage 2b= Late Moth-eaten appearance, with numerous discrete areas of increased signal with beginning confluence, comprising 30 – 60% of the volume of the individual muscle; 5=Stage 3= Washed-out appearance, fuzzy appearance due to confluent areas of increased signal; 6=Stage 4= End stage appearance, muscle replaced increased density connective tissue and fat, with only A rim of fascia and neurovascular structures distinguishable, IQR= interquartile range, *= significant with modified Bonferroni correction. *Table 19: MRI Muscle Fat qualitative score using Modified Mercuri's Scale at the calf and*

foot level of DHMN and control groups



Figure 14:Pattern of fat infiltration using Modified Mercuri's Scale at the calf and foot level of DHMN group.

5.3.2.5. Quantitative Analysis of Intramuscular Oedema

Region of interest was applied to the T2 map at the calf and foot level to quantify intramuscular oedema as a sign of active denervation. Analysis showed significant differences between DHMN and control group at the calf (P=0.0028) and foot (P<0.0001) (Table 20).

Region of interest ms	DHMN mean (SD)	CONTROL mean (SD)	Mean diff	Р		
Total Calf	38.52(7.09)	30.07(1.83)	-8.45	0.0028*		
Dorsiflexion	37.66(7.08)	30.02(2.32)	-7.64	0.0063*		
Tibialis Anterior	37.66(7.08)	30.02(2.32)	-7.64	0.0063*		
Peroneus Longus	38.73(8.58)	30.41(1.97)	-8.32	0.0109*		
Plantarflexion	39.03(7.50)	30.09(1.76)	-8.93	0.0001*		
Lateral Gastrocnemius	38.61(8.86)	30.62(1.85)	-7.98	0.0165*		
Medial Gastrocnemius	39.81(8.10)	30.28(2.40)	-9.53	0.0032*		
Soleus	38.52(7.34)	29.92(1.50)	-8.60	<0.0001*		
Tibialis Posterior	37.40(6.74)	29.59(1.71)	-7.81	0.0034*		
Foot	45.61(5.35)	30.38(2.37)	-15.23	<0.0001*		
SD= Standard Deviation, ms= Bonferroni correction.	SD= Standard Deviation, ms= millisecond, diff= difference, *= significant with modified Bonferroni correction					

 Table 20: MRI Muscle water quantitative analysis using T2 mapping at the calf and foot level

of DHMN and control groups

5.3.2.6. Qualitative Analysis of Intramuscular Oedema

Qualitative analysis of short tau inversion recovery imaging (STIR) hyperintensity showed consistent results as the quantitative T2 mapping analysis. There was a significant difference at the foot (P=0.0001) and all calf muscles (P=0.0002-0.0074) (Table 21).

Destant of the family	DHMN median (IQR)		CONTROL median (IQR)		D
Region of interest	Intensity	Extent	Intensity	Extent	P
Tibialis Anterior	1.5(1.5 to 2)	2.5 (2.5 to 3)	0 (0 to 0)	0 (0 to 0)	0.0002*
Tibialis Posterior	1(1.6 to 1.4)	2 (2.6 to 3)	0 (0 to 0)	0 (0 to 0)	0.0008*
Peroneus Longus	1(1.6 to 1.4)	1.5 (1.5 to 2.7)	0 (0 to 0)	0 (0 to 0)	0.0003*
Medial Gastrocnemius	1(1. to 1.4)	2 (2 to 2.4)	0.5 (0.5 to 1)	0.5 (0.5 to 1)	0.0074*
Lateral Gastrocnemius	1(1. to 1.500)	1 (1 to 3)	0 (0 to 1)	0(0 to 1)	0.0046*
Soleus	1.5(1.5 to 1.9)	1.5 (1.5 to 2.9)	0 (0 to 0)	0 (0 to 0)	0.0002*
Foot	2(2 to 2)	2 (2 to 2)	0 (0 to 0)	0 (0 to 0)	0.0001*
Intensity: 0= none, 1=mild, 2=marked, Extent: 0= none, 1= < 30%, 2=30%-60%, 3= >60%, IQR= interguartile range *= significant with modified Bonferroni correction					

 Table 21: MRI Muscle water qualitative analysis using STIR image at the calf and foot level

of DHMN and control groups

5.3.3. Difference in Muscles Isokinetic and Isometric Strength

Dynamometric muscle strength was reduced in distal muscle groups compared with their matched controls (Table 22). Differences were significant only in isometric dorsiflexion 10° (P=0.0134), isometric dorsiflexion 30° (P=0.0121), and isokinetic plantarflexion $60^{\circ}/60^{\circ}$ (P=0.0062). However, no difference was seen in isokinetic dorsiflexion (P=0.0729) (Table 22).

Isometric Dynamometry	DHMN	CONTROL	Mean	р
Nm	mean (SD)	mean (SD)	diff	1
Hip Extension 45°	137.2(68.5)	142.9(56.1)	5.66	0.84
Hip Flexion 45°	107.7(47.5)	99.6(34.4)	-8.13	0.88
Knee Extension 45°	108.4(59.4)	114.6(59.9)	6.20	1.00
Knee Extension 90°	110.2(58.2)	116.3(49.9)	6.15	0.81
Knee Flexion 45°	60.4(36.7)	65.9(32.4)	5.54	0.73
Knee Flexion 90°	40.0(25.7)	46.6(20.8)	6.61	0.54
Ankle Plantarflexion 10°	27.3(18.7)	45.9(5.7)	18.62	0.0102^{\dagger}
Ankle Dorsiflexion 10°	14.0(17.2)	27.9(5.6)	13.93	0.0134*
Ankle Dorsiflexion 30°	18.8(17.3)	33.1(9.0)	14.28	0.0121*
Isokinetic Dynamometry	DHMN	CONTROL	Mean	D
Nm	mean (SD)	mean (SD)	diff	I
Hip Extension 60°/60s	113.5(71.6)	105.8(70.2)	-7.72	0.81
Hip Flexion 60°/60s	103.3(51.5)	95.4(36.5)	-7.83	0.71
Knee Extension 60°/60s	84.1(52.3)	89.9(49.4)	5.85	0.80
Knee Flexion 60°/60s	42.3(27.2)	51.4(40.7)	9.13	0.82
Knee Extension 120°/120s	60.8(44.6)	74.7(55.9)	13.90	0.54
Knee Flexion 120°/120s	32.9(24.4)	41.6(33.3)	8.70	0.51
Ankle Plantarflexion 60°/60°	16.8(16.7)	42.2(24.0)	25.35	0.0062*
Ankle Dorsiflexion 60°/60°	14.7(8.2)	21.6(7.8)	6.88	0.07
SD= Standard Deviation, Nm= Newton meter, \dagger = significant with P<0.05, diff= difference, *= significant after modified Bonferroni correction.				

Table 22: Isometric and isokinetic dynamometry of DHMN and control groups

5.3.4. Deviations From Normal in Walking Gait

5.3.4.1. Spatiotemporal Analysis

Three dimensional gait analysis showed significant difference in parameter related to time rather than distance (Table 23). DHMN group walked with slower speed (P=0.0322) with more stride time (P=0.0279) and step time (P=0.0279), and therefore, less strides and steps per minute (P=0.0279). DHMN group spent more time in single and double support (P=0.032, 0.0015, respectively) which contributed to a higher stance phase rates (P=0.0002) and delayed opposite toe off (P=0.003) (Table 23).

Gait Spatiotemporal Parameters	DHMN mean (SD)	CONTROL mean (SD)	Mean diff	Р
Speed; m/s	0.98(0.29)	1.23(0.14)	0.24	0.0322*
Stride Length; m	1.18(0.26)	1.37(0.15)	0.19	0.07
Stride Time; s	1.23(0.16)	1.12(0.08)	-0.11	0.0279*
Strides Per Minute	49.48(5.08)	53.87(3.79)	4.39	0.0279*
Step Length; m	0.59(0.13)	0.69(0.07)	0.10	0.07
Step Time; s	0.62(0.08)	0.56(0.04)	-0.06	0.0279*
Steps Per Minute	98.96(10.17)	107.74(7.58)	8.78	0.0279*
Percent Stance	0.64(0.02)	0.59(0.03)	-0.05	0.0002*
Single Support Time; s	0.61(0.05)	0.56(0.04)	-0.05	0.032*
Double Support Time; s	0.17(0.08)	0.10(0.03)	-0.08	0.0015*
Percent Opposite Toe Off	13.90(4.51)	8.84(2.74)	-5.06	0.003*
Percent Opposite Foot Contact	49.65(0.80)	48.48(3.52)	-1.16	0.45
SD= Standard Deviation, m= meters, s= seconds, diff= difference, *= significant with modified Bonferroni correction.				

Table 23: Gait spatiotemporal parameters of DHMN and control groups

5.3.4.2. Kinematics

People with DHMN walked with significantly higher dorsiflexion angle (P=0.0435) and lower plantar flexion angle (P=0.0259) than healthy controls. Higher dorsiflexion angle occurred in stance phase and lower plantar flexion angle at pre-swing to early swing (Figure 15). Significantly higher hip flexion angle (P=0.0076) was also observed in swing and lower hip extension angle in stance (P=0.0413) (Table 24) (Figure 15).

Gait Kinematics	DHMN mean (SD)	CONTROL mean (SD)	Mean diff	Р
Pelvis X Max; °	4.49(1.32)	4.35(1.63)	-0.14	0.84
Pelvis X Min; °	-4.79(1.74)	-4.39(1.30)	0.40	0.58
Pelvis Y Max; °	15.82(5.84)	10.29(7.59)	-5.53	0.09
Pelvis Y Min; °	10.63(4.51)	5.42(8.29)	-5.22	0.10
Pelvis Z Max; °	6.75(2.55)	5.25(1.93)	-1.50	0.17
Pelvis Z Min; °	-6.96(2.98)	-5.52(1.93)	1.44	0.23
Hip Max; °	47.33(5.70)	36.93(11.78)	-10.39	0.0076*
Hip Min; °	3.32(4.26)	-4.82(10.80)	-8.14	0.0413*
Knee Max; °	74.00(4.74)	69.66(6.50)	-4.34	0.07
Knee Min; °	6.51(5.06)	4.48(7.81)	-2.03	0.36
Ankle Max; °	25.07(5.81)	23.42(20.73)	-1.66	0.0435*
Ankle Min; °	-10.39(4.42)	-15.53(4.76)	-5.14	0.0259*

SD= Standard Deviation, diff= difference, *= significant with modified Bonferroni correction= Pelvic z Maximum= Rotation forwards, Pelvic z Minimum= Rotation backwords, Pelvic x Maximum, lateral raise, Pelvic x Minimum= lateral drop, Pelvic y Maximum= Anterior tilt, Pelvic y Minimum = Posterior tilt, Hip y Maximum= Flexion, Hip y Minimum= Extension, Knee y Maximum= Flexion, Knee y Minimum= Extension, Ankle y Maximum= Dorsiflexion, Ankle y Minimum= Plantarflexion.

Table 24: Gait kinematics parameters of DHMN and control groups



Figure 15:Grand average angular displacement in the sagittal plane over one gait cycle for the ankle, knee, and hip (average of left and right sides). Comparison of people with DHMN and controls.

OTO, Opposite side toe off; OIC, Opposite side initial contact; TO, Toe off; A, B, foot drop; C, plantar flexion failure.

5.3.4.3. Kinetics

People with DHMN produce smaller ankle moments and power generation during walking. Ankle plantar flexion moment at terminal stance was significantly lower than controls (P=0.0002) and significantly lower ankle power generation was observed at the same period (P=0.0011). There was also a significant reduction in ankle power generation in swing phase (p=0.0279) (Table 25) (Figure 16,Figure 17).

Gait Kinetics	DHMN mean (SD)	CONTROL mean (SD)	Mean diff	Р
Hip Moment Y Max; Nm/Kg	1.49(0.56)	1.19(0.30)	-0.31	0.1614
Hip Moment Y Min; Nm/Kg	-0.49(0.17)	-0.51(0.11)	-0.01	0.8656
Knee Moment Y Max; Nm/Kg	0.69(0.32)	0.59(0.27)	-0.10	0.4945
Knee Moment Y Min; Nm/Kg	-0.68(0.29)	-0.50(0.09)	0.19	0.0837
Ankle Moment Y Max; Nm/Kg	1.03(0.26)	1.51(0.26)	0.48	0.0002*
Ankle Moment Y Min; Nm/Kg	-0.08(0.07)	-0.12(0.05)	-0.04	0.0535
Hip Power Max During Swing; W/kg	1.02(0.32)	1.10(0.36)	0.08	0.6201
Hip Power Min During Stance; W/kg	-0.84(0.65)	-0.71(0.32)	0.14	0.5793
Hip Power Max During Stance; W/kg	1.84(0.75)	1.52(0.53)	-0.32	0.3054
Knee Power Max During Swing; W/kg	0.36(0.13)	0.45(0.20)	0.09	0.2779
Knee Power Min During Stance; W/kg	-1.53(0.91)	-1.26(0.84)	0.27	0.5168
Knee Power Max During Stance; W/kg	1.67(1.33)	1.03(0.50)	-0.63	0.198
Ankle Power Max During Swing; W/kg	0.04(0.02)	0.63(0.88)	0.59	0.0279*
Ankle Power Min During Stance; W/kg	-1.16(0.38)	-1.23(0.50)	-0.07	0.7338
Ankle Power Max During Stance; W/kg	1.03(0.43)	2.15(0.77)	1.11	0.0011*

SD= Standard Deviation, Nm/Kg= Newton Meter per Kilogram, W/kg= Watts Per Kilogram, diff= difference, *= significant with modified Bonferroni correction= Hip Moments Maximum= Extension, Hip Moments Minimum= Flexion, Knee Moments Maximum= Extension, Knee Moments Minimum= Flexion, Ankle Moments Maximum= Plantarflexion, Ankle Moments Minimum= Dorsiflexion, Hip Power Maximum = generation in swing phase, Hip Power Minimum= absorption in stance phase, Hip Power Maximum= generation in stance phase, Knee Power Maximum= generation in stance phase, Knee Power Maximum= generation in swing phase, Knee Power Minimum= absorption in stance phase, Ankle Power Maximum= generation in stance phase, Ankle Power Minimum= absorption in stance phase, Ankle Power Maximum= generation in swing phase.

Table 25: Gait kinetics parameters of DHMN and control groups



Figure 16: Grand average joint moments in the sagittal plane over one gait cycle for the ankle, knee and hip (average of left and right sides).

Comparison of people with DHMN and controls. OTO, Opposite side tore off; OIC, Opposite side initial contact; TO, Toe off.



Figure 17:Grand average joint powers in the sagittal plane over one gait cycle for the ankle, knee and hip (average of left and right sides).

Comparison of people with DHMN and controls. OTO, Opposite side tore off; OIC, Opposite side initial contact; TO, Toe off.

5.4. Discussion

5.4.1. Pattern of Muscle Involvement

This study described the pattern of muscle involvement in DHMN cohort primarily using dynamometry and quantitative MRI analysis of muscle fat fraction and oedema. MRI parameters confirmed the study hypothesis. Results showed higher rates of fatty infiltration and signs of active denervation distally, in the foot and calf, more than proximally, at the thigh level, and in the posterior compartment more than anterior compartment of the calf. Results were consistent in both quantitative and qualitative analysis. Interestingly, the remaining muscle tissue variable was significantly lower in plantar flexors, in the Medial Gastrocnemius and Soleus specifically. This difference between DHMN and control was not seen in dorsiflexors or proximal muscle groups.

Proximally, Biceps femoris muscle is the only muscle that showed significantly higher fat fraction rate. However, this muscle involvement was not reflected on other MRI parameters or muscle dynamometry. Biceps femoris might be a candidate muscle to describe the pattern of involvement for future longitudinal studies and can be supported by observing the levels of active denervation measured using T2 mapping which, unfortunately, was not done in this study to shorten the scanning time.

As hypothesised, the cohort showed predominant distal weakness more than proximal, and in plantar flexors more than dorsiflexors. These findings are consistent with the pattern of involvement shown in MRI. Isokinetic dynamometry showed weakness in plantar flexors. However, isometric dynamometry showed dorsiflexion weakness that was not reflected in isokinetic testing. Suggesting that the reduced isometric strength observed for ankle dorsiflexion may be partly due to the angle chosen for assessment combined with reduced range of ankle dorsiflexion. Moreover, the difference in isometric plantar flexors torque was not significant after modified Bonferroni correction, unlike isokinetic plantar flexion. It was noted that participants tend to engage their upper body to push downwards into the dynamometer footplate which might contributed to the overall torque measured. Changing the testing position from supine to seated might eliminate this factor. Beside positioning and test application, the sample size was small with a high standard deviation due to high variability in strength within the cohort. A larger sample size will allow us to see true differences.

Regardless of the genetic diagnosis, our DHMN cohort showed similar patterns of muscle involvement as previously reported samples (Esteller et al., 2023, O'Donnell et al., 2022). However, if we consider that most of our cohort are with HSPB1 genetic diagnosis (7/12), Esteller et al.(2023) reported a different pattern of involvement in their HSPB1 cohort (2 participants). They showed Severe involvement of all muscles on the lower leg (Esteller et al., 2023). Although the number of participants is limited in our study, it still presents the largest DHMN
cohort with most of them genetically diagnosed, and the first to utilise quantitative MRI analysis to identify the pattern of involvement in this particular group.

5.4.2. Deviation from Normal in Walking Gait

The DHMN group took more time to complete a gait cycle over an equivalent distance compared to the control group. The increase in step and stride time led to the decreased number of steps and strides per minute, and therefore, a slower walking speed.

A delay in opposite toe off contributed to the increase of the double support time and therefore a longer stance phase. Delayed opposite toe off and no difference on opposite foot contact, led to increased double support time in stance phase, and shorter single support time. Therefore, any increase in the single support time is most likely to be relevant to the swing phase which is commonly affected by foot drop in peripheral neuropathy.

Kinematic data showed less ankle dorsiflexion angle in swing phase (Figure 15, A, B) that could relate to the increased hip flexion angle observed, and decreased ankle dorsiflexors power generation. This pattern suggests the presence of foot drop combined with compensatory strategy using hip flexion to improve foot clearance of the swinging side. This pattern has been seen in CMT cohorts (Ramdharry et al., 2009, Don et al., 2007, Vinci and Perelli, 2002).

In stance phase, kinematic variables showed reduced ankle control with increased passive dorsiflexion angle in terminal stance and decreased ankle plantar flexion in pre-swing (Figure 15, C). Kinetic analysis showed a decrease in plantar flexion moment and power generation in terminal stance, which confirms the presence of plantar flexion failure. This pattern has been seen in CMT cohorts (Don et al., 2007, Vinci and Perelli, 2002).

The differences between the healthy controls and people with DHMN when walking could partly be explained by their differences in walking speed, as gait kinetic and kinematic parameters are decreased at slower speeds and increased at faster speeds (Fukuchi et al., 2019). This highlights the importance of taking into account the effects of walking speed when comparing gait data of pathological individuals with normal or control individuals. Future studies involving such type of comparisons must control for the effects of different gait speeds, for example by having walking speed as a covariate in the analysis or getting the control group to walk at matched speeds.

Although MRI studies showed different patterns of muscle involvement between CMT and DHMN (Esteller et al., 2023, Bas et al., 2020), 3D motion analysis showed more similar patterns. Plantar flexion failure and foot drop with proximal compensation are patterns described previously in CMT cohorts with predominant distal muscle weakness (Ramdharry et al., 2009, Don et al., 2007, Vinci and Perelli, 2002). These studies included mixed types of CMT and severe

cases with plantar flexion weakness, which explain the presence of the plantar flexion failure pattern. Unlike CMT, the DHMN cohort of the current study did not show significant dorsiflexors involvement, however, gait analysis showed pattern of foot drop, suggesting that even with mild dorsiflexors weakness, foot clearance of the swinging limb can be affected.

The findings of the current study suggest that although 3D motion analysis is a valuable tool in describing function and detecting deviation from normal, quantitative MRI is more sensitive to differences between neuropathy groups to describe muscular involvement.

5.5. Summary

The current study described pattern of involvement and deviation from normal in muscle structure and function. The study findings are summarised in Table 26.

		DH	MN Pat	tern i	n Muscl	le Iı	nvolvement	and Gait		
Muscle Structure	Higher more th	Higher rates of fatty infiltration and signs of active denervation distally at the foot and calf more than proximally, and in the posterior compartment (Plantar flexors) more than anterior compartment (Dorsiflexors) of the calf.								
Muscle Function	Plantar flexion (PF) weakness affected the stability of the ankle joint in stance phaseDorsiflexion (DF) weakness aff foot clearance in swing phase				ss affected phase					
% of cycle	0%	0-10%	10-30	%	30-509	%	50-60%	60-70% 70-85% 85-100%		
Phase			Stance	e phas	se			·	Swing phase	
Spatio- Temporal	Increased time due	l double s to delaye	support ed OTO	In due	to increased sto increased stride	star asec len	nce time d step and gth	Increase	d single suppo	rt time
Variables				S	low gait	due	e to the ever	nts timing		
Kinematics Variables	Inc	Increased ankle passive dorsiflexion dorsiflexion dorsifl			xion and on					
Kinetics Variables	Decre	eased pla	ntar flex	ion m	oment a	ınd j	power	Decreased	ankle power g	eneration
Gait Pattern		Pl	antar flez	xion f	ailure				Foot drop	
				The	Normal	l Ga	it Cycle			
% of cycle	0%	0-10%	10-30	%	30-50%	%	50-60%	60-70%	70-85%	85-100%
Phase			Stance	e phas	se				Swing phase	
Event	Initial contact	Foo Opposi (O	t flat, te toe off TO)	f O	He pposite i	el o initi	ff, al contact	Toe off	Tibia vertical, Feet adjacent	Initial contact (next cycle)
Period	Loading response	Mid-	stance	Т	erminal stance		Pre- swing	Initial swing	Mid-swing	Termina 1 swing
Support	Double support	S	Single leg	g supp	port		Double support	Sir	ngle leg suppor	t

Table 26: Summary of pattern in gait and muscle involvement in DHMN

Chapter 6: The Natural History of Muscle Structure and Function in Distal Hereditary Motor Neuropathy (DHMN)

6.1. Introduction

Distal Hereditary Motor Neuropathy (DHMN) is a slowly progressive condition that is incurable. There is a need to understand the condition trajectory in order to improve patients' outcomes in managing symptoms and prevent complications. Until therapeutic options are available for clinical trials, there is a need to confirm the responsiveness of outcome measures in DHMN for potential use in future investigation of effect.

Previous natural history studies in CMT showed longitudinal change at the calf level and high responsiveness of quantitative muscle fat fraction MRI over one year (Morrow et al., 2016). Quantitative MRI, to measure intramuscular oedema, was sensitive enough to detect subclinical muscle abnormality as well (Locher et al., 2022, Morrow et al., 2016).

The current longitudinal study is the first to explore the progression of DHMN over one year and to assess the responsiveness of the quantitative methods used (MRI, dynamometry, and 3D motion analysis) in order to determine predictors of change in muscle structure, function, and gait patterns. Based on the findings in chapter 5 and previous studies in CMT we expect our DHMN cohort to show:

- Significant change in MRI quantitative parameters in the distal lower limb muscles, alongside deterioration in isokinetic strength, plus moment and power generation during gait.
- Responsiveness in quantitative MRI parameters at the calf level will be higher in comparison to the thigh level and other methods of quantification, dynamometry and motion analysis.

6.2. Methods

Eight DHMN participants were recruited in the natural history group to undergo an MRI scan, isokinetic and isometric dynamometry of the lower limbs, and 3D motion analysis to capture kinetic and kinematic data of walking gait as described in chapter 4. The same measurements were repeated after 6 months and 12 months to observe the natural history of the disease.

6.2.1. Data Analysis

To explore the natural history of muscle structure and function in Distal Hereditary Motor Neuropathy (DHMN) over one year, baseline, 6 months, and 12 months follow up values were compared on a muscle-by-muscle, parameter-by-parameter basis using repeated measures analysis of variances (ANOVA). A post hoc analysis compared the different measurement to each other using paired t-test. If data sets were not normally distributed, a Friedman test was used. If the Friedman test was positive (P<0.05), the different measurement points were compared for the post hoc analysis using a paired Wilcoxson- Signed rank test. Modified Bonferroni correction for multiple comparisons was applied for P-values and confidence intervals (Simes, 1986). Where the data type was ordinal, such as manual muscle testing and qualitative MRI scores, the Friedman test was used.

To ascertain the sensitivity of the MRI, Dynamometry, and 3D motion analysis variables, a standardised response mean (SRM) was calculated as the ratio of

mean change to standard deviation of change. SRM was categorised by magnitude according to Cohen's suggestion: < 0.2 minimal responsiveness; 0.2-0.5 small responsiveness; 0.5-0.8 moderate responsiveness; >0.8 large responsiveness. This statistical method has been used in previous research to examine the sensitivity of MRI and dynamometry longitudinally in slowly progressive in conditions (Morrow, et al., 2016).

6.3. Results

6.3.1. Subjects and Clinical Assessment

Eight DHMN participants were recruited. They underwent MRI scan, Dynamometry, and 3D motion analysis at baseline, 6 months, and 12 months. One DHMN female participant completed only baseline and 6 months measurements before she withdrew from the study (appendix VII), therefore, the last observation at 6 months was carried forward to 12 months for analysis purposes. There were no significant changes in clinical assessment over 12 month in terms of disease severity (CMTES), foot posture, ankle ROM, falls, and perceived walking ability (walk-12). A summary of the subjects' demographics and clinical assessment over 12 months is presented in Table 27.

Demographics and clinical assessment	BASELINE	6 MONTHS	12 MONTHS
Numbers	8	8	7
Gender (M/F)	(3/5)	(3/5)	(3/5)
Age; mean years; range	57 (42/75)	57 (42/75)	57 (42/75)
Genetic diagnosis (HSPB1/unknown)	(6/2)	(6/2)	(6/2)
CMTES; mean (SD)	7.25(3.23)	6.62(3.31)	6.87(3.5)
Foot Posture Index-6 (Pronated/Normal/Supinated)	(1/6/1)	(1/3/4)	(1/3/4)
Fall Frequency (weekly/Monthly/Yearly)	(1/1/2)	(1/1/2)	(1/1/2)
Walk-12; mean (SD)	36(12)	35(13)	36(13)
Range of Motion	BASELINE	6 MONTHS	12 MONTHS
Dorsiflexion; count (Limited/Normal)	(8/0)	(8/0)	(8/0)
Plantarflexion; count (Limited/Normal)	(8/0)	(8/0)	(8/0)

M= Male, F= Female, HSPB1= Heat-shock 27-KD Protein 1, SD= Standard Deviation, CMTES= Charcot-Marie-Tooth Disease Examination Score, *= significant after modified Bonferroni correction.

Table 27: : Summary of demographics and clinical assessment of DHMN at baseline, 6

months, and 12 months.

6.3.2. Longitudinal Change in Intramuscular Fat on MRI

Quantitative analysis of MRI muscle fat fraction showed significant longitudinal difference at the calf level (P=0.003) in most of the calf muscles except the medial and lateral gastrocnemius. No change was shown at the thigh and foot (Table 28) (Figure 18). Qualitative Modified Mercuri's Scale analysis showed significant longitudinal difference at the foot (P<0.00001) and calf muscles (P=0.00781-0.03725) except the lateral gastrocnemius (Table 29) (Figure 18). The Friedman test was calculated manually to confirm the results (appendix II). Calculating the test statistics (χ 2) yield a value of 0 and P-value =1. Thus, the null hypothesis is accepted and there is no difference between measurements.

Region of interest	BASELINE	6 MONTHS	12 MONTHS	F	р
%	mean (SD)	mean (SD)	mean (SD)	Ľ	L
Total Thigh	15.95(13.22)	16.58(15.22)	17.15(15.18)	2.28	0.14
Knee Extension	14.83(14.34)	15.26(15.21)	15.41(15.56)	0.67	0.53
Rectus Femoris	11.45(10.04)	11.23(10.45)	11.21(10.18)	0.21	0.81
Vastus Intermedius	16.43(15.12)	16.77(15.37)	17.18(15.50)	0.60	0.56
Vastus Lateralis	16.85(18.01)	17.21(18.45)	16.94(18.97)	0.89	0.43
Vastus Medialis	10.81(8.66)	11.78(11.76)	12.18(12.04)	0.09	0.92
Knee Flexion	17.40(12.74)	18.91(16.35)	19.14(16.07)	3.03	0.08
Semimembranosus	18.19(13.38)	19.28(15.91)	20.00(15.23)	1.75	0.21
Semitendinosus	16.18(16.09)	19.64(22.23)	20.45(22.50)	2.09	0.16
Biceps Femoris	20.30(12.98)	20.85(15.11)	21.59(14.78)	1.41	0.28
Sartorius	20.40(18.70)	21.41(19.57)	21.23(21.12)	0.03	0.97
Gracilis	10.72(11.02)	12.04(12.85)	11.37(13.18)	0.92	0.42
Adductor Magnus	18.37(20.25)	19.13(20.30)	19.53(21.05)	0.37	0.37
Total Calf	36.61(22.35)	39.17(22.87)	39.83(22.49)	13.27	0.003*
Dorsiflexion	23.89(19.01)	25.83(19.60)	26.57(19.84)	13.42	0.00056*
Tibialis Anterior	23.89(19.01)	25.83(19.60)	26.57(19.84)	12.50	0.001*
Peroneus Longus	39.30(27.19)	40.41(27.26)	41.61(26.19)	7.05	0.009*
Plantarflexion	40.16(23.61)	43.50(24.70)	44.04(24.11)	9.53	0.009*
Lateral Gastrocnemius	32.88(31.04)	34.50(30.33)	35.71(30.63)	3.27	0.07

Medial Gastrocnemius	41.46(22.89)	43.44(21.59)	43.89(22.15)	1.91	0.20
Soleus	43.22(23.70)	47.67(25.74)	47.74(24.53)	8.57	0.013*
Tibialis Posterior	30.03(22.23)	33.82(22.12)	34.05(22.18)	9.31	0.008*
Foot	54.99(7.12)	56.00(7.86)	57.49(7.25)	3.05	0.09
N=8, SD= Standard Dev	iation, *= signi	ficant (P<0.05)=	post-hoc test sho	wed differenc	e between
baseline and 6 months.					

Table 28: Longitudinal MRI mean muscle fat fraction of DHMN and repeated measures

analysis results.

Region of interest	BASELINE median (IQR)	6 MONTHS median (IQR)	12 MONTHS median (IQR)	Р
Tibialis Anterior	2.75(2.5 to 4.25)	2.75(2.5 to 5)	2.75(2.5 to 5)	0.02602*
Tibialis Posterior	3.75(2.75 to 5)	3.75(2.75 to 5.75)	3.75(2.75 to 5.75)	0.02602*
Peroneus Longus	4.25(3.25 to 5.25)	4.25(3.25 to 6)	4.25(3.25 to 6)	0.03725*
Medial Gastrocnemius	4(3.25 to 5)	4(3.25 to 6)	4(3.25 to 6)	0.00781*
Lateral Gastrocnemius	2.5(2 to 5)	2.5(2 to 6)	2.5(2 to 6)	0.4
Soleus	4.75(3 to 5.25)	4.75(3 to 6)	4.75(3 to 6)	0.00781*
Foot	6(6 to 6)	6(6 to 6)	6(6 to 6)	<0.00001*

N=8, Modified Mercuri's Scale: 1=Stage 0= Normal appearance, 2=Stage 1= Early Moth-eaten appearance= with scattered small areas of increased signal, 3=Stage 2a= Late Moth-eaten appearance= with numerous discrete areas of increased signal with beginning confluence= comprising less than 30% of the volume of the individual muscle, 4=Stage 2b= Late Moth-eaten appearance= with numerous discrete areas of increased signal with beginning confluence= comprising 30 - 60% of the volume of the individual muscle, 5=Stage 3= Washed-out appearance= fuzzy appearance due to confluent areas of increased signal, 6=Stage 4= End stage appearance= muscle replaced increased density connective tissue and fat= with only A rim of fascia and neurovascular structures distinguishable, IQR= interquartile range, *= significant (P<0.05)= post-hoc test showed difference between baseline and 6 months.

Table 29: Longitudinal MRI median muscle fat qualitative score using Modified Mercuri's

Scale of DHMN and repeated measures analysis results.



Quantitative Intramuscular Fat (FF %)



Figure 18: Longitudinal change of intramuscular fat in DHMN. N=8, *M*= months.

6.3.3. Longitudinal Change in Muscle Area on MRI

Analysis of the cross sectional area did not show any change at all levels (Table 30). However, when the remaining muscle area was calculated significant longitudinal difference was seen at the gracilis muscle (P=0.046) and soleus muscle (P=0.015) which affected the total plantar flexion change over one year (P=0.048) (Table 31).

Region of interest	BASELINE	6 MONTHS	12 MONTHS	н	Р
mm ²	mean (SD)	mean (SD)	mean (SD)	Ľ	1
Total Thigh	10009.8(3716.7)	9967.8(3544.5)	10066.9(3586.5)	0.46	0.63
Knee Extension	4758.2(1487.3)	4686.4(1431.1)	4718.3(1495.4)	0.59	0.52
Rectus Femoris	437.7(173.8)	426.6(175.9)	430.4(187.7)	1.21	0.33
Vastus Intermedius	1416.2(428.5)	1381.8(394.4)	1414.7(423.0)	0.99	0.40
Vastus Lateralis	1630.7(383.3)	1606.2(372.3)	1623.3(405.3)	0.47	0.57
Vastus Medialis	1273.5(601.6)	1271.8(601.3)	1249.9(594.1)	0.97	0.40
Knee Flexion	3691.6(1658.6)	3785.4(1649.2)	3772.0(1553.1)	1.41	0.28
Semimembranosus	820.3(426.4)	849.1(445.6)	812.3(402.6)	1.29	0.30
Semitendinosus	765.4(405.7)	805.3(401.1)	792.8(339.2)	0.90	0.43
Biceps Femoris	1306.5(577.2)	1330.6(570.0)	1341.1(566.2)	2.63	0.11
Sartorius	350.8(146.7)	352.2(141.7)	364.1(152.9)	0.58	0.53
Gracilis	448.6(224.7)	448.2(225.4)	461.6(235.0)	2.84	0.09
Adductor Magnus	1560.0(830.4)	1496.0(724.2)	1576.6(796.2)	2.00	0.19
Total Calf	4925.0(1636.0)	4813.6(1802.5)	4819.5(1860.3)	0.82	0.43
Dorsiflexion	828.6(277.4)	825.4(293.9)	830.6(301.8)	0.09	0.92
Tibialis Anterior	828.6(277.4)	825.4(293.9)	830.6(301.8)	0.09	0.92
Peroneus Longus	454.3(165.8)	461.8(192.1)	448.3(183.1)	0.38	0.69
Plantarflexion	3327.5(1235.4)	3199.9(1346.8)	3230.6(1387.3)	1.72	0.23
Lateral Gastrocnemius	663.3(512.5)	612.5(497.6)	635.5(489.1)	1.23	0.32
Medial Gastrocnemius	784.9(355.2)	761.2(385.4)	772.9(382.7)	0.24	0.69
Soleus	1879.3(656.5)	1826.1(693.9)	1822.3(703.8)	2.47	0.14
Tibialis Posterior	314.5(133.2)	326.5(125.4)	309.9(140.4)	0.31	0.64
Foot	465.9(127.7)	446.6(116.4)	442.5(102.8)	1.06	0.35
N=8, SD= Standard Devia	ation, mm ² = square	millimetres.	1		

Table 30: Longitudinal MRI muscle cross sectional area of DHMN and repeated measures

analysis results.

Region of interest	BASELINE	6 MONTHS	12 MONTHS	F	Р
mm ²	mean (SD)	mean (SD)	mean (SD)		-
Total Thigh	17364.0(7525.7)	17204.2(7504.1)	17227.3(7442.8)	0.66	0.46
Knee Extension	8337.7(3319.3)	8251.5(3342.8)	8206.7(3344.1)	1.46	0.27
Rectus Femoris	395.8(185.3)	389.4(191.0)	390.5(197.7)	0.75	0.48
Vastus Intermedius	1217.6(473.5)	1190.6(448.7)	1200.3(461.6)	1.57	0.24
Vastus Lateralis	1387.7(493.2)	1370.4(500.6)	1371.8(497.2)	0.59	0.50
Vastus Medialis	1167.7(602.2)	1175.4(624.7)	1140.8(607.5)	2.87	0.09
Knee Flexion	6360.4(3268.6)	6446.6(3406.6)	6384.5(3174.1)	0.29	0.68
Semimembranosus	704.5(405.1)	725.0(440.9)	689.6(380.8)	0.83	0.41
Semitendinosus	677.0(401.9)	682.7(419.5)	669.7(379.7)	0.25	0.78
Biceps Femoris	1088.4(575.1)	1105.3(583.4)	1101.0(568.7)	0.59	0.57
Sartorius	294.2(165.2)	299.3(173.6)	308.3(177.1)	0.77	0.43
Gracilis	416.0(226.9)	410.9(226.9)	432.2(238.6)	3.86	0.046*
Adductor Magnus	1332.9(795.5)	1253.1(705.5)	1329.1(800.1)	1.97	0.18
Total Calf	6469.0(3538.9)	6159.4(3559.1)	6110.2(3644.6)	3.78	0.08
Dorsiflexion	1322.9(616.8)	1293.1(631.0)	1288.4(645.1)	1.15	0.35
Tibialis Anterior	661.5(308.4)	646.5(315.5)	644.2(322.5)	1.15	0.35
Peroneus Longus	273.6(147.7)	272.7(156.4)	262.9(154.6)	1.07	0.37
Plantarflexion	4166.3(2608.4)	3879.6(2605.5)	3879.8(2644.0)	4.98	0.048*
Lateral Gastrocnemius	512.2(553.5)	486.0(520.1)	486.3(521.6)	1.72	0.21
Medial Gastrocnemius	476.6(335.1)	455.2(339.5)	459.7(336.1)	1.02	0.36
Soleus	1094.3(642.4)	998.5(652.2)	993.9(638.1)	8.32	0.015*
Tibialis Posterior	216.3(109.1)	220.7(125.0)	208.1(126.8)	0.32	0.63
Foot	215.9(89.6)	201.0(80.1)	193.0(74.7)	2.00	0.19
N=8, SD= Standard Devi	ation, mm ² = square	millimetres,*= sigr	nificant (P<0.05).		

Table 31: Longitudinal MRI remaining muscle area of DHMN and repeated measures analysis results.

6.3.4. Longitudinal Change in Intramuscular Water on MRI

Quantitative analysis of intramuscular water on MRI showed significant longitudinal difference only at the tibialis posterior muscle (P=0.03725). No change was shown at the rest of the calf muscles and foot (Table 32). Qualitative analysis showed significant longitudinal difference only at the medial gastrocnemius muscle (P=0.01912). No change was shown at the rest of the calf muscles and foot (Table 33).

Region of interest ms	BASELIN E mean (SD)	6 MONTHS mean (SD)	12 MONTHS mean (SD)	F	Р
Total Calf	40.76(7.0)	38.94(7.2)	41.54(6.3)	2.59	0.11
Dorsiflexion	40.11(6.8)	39.51(6.5)	42.09(6.7)	3.66	0.05
Tibialis Anterior	40.11(6.8)	39.51(6.5)	42.09(6.7)	3.66	0.05
Peroneus Longus	41.50(8.5)	37.85(5.3)	41.27(6.3)	2.03	0.17
Plantarflexion	41.12(7.8)	39.07(8.2)	41.47(6.8)	1.72	0.21
Lateral Gastrocnemius	41.22(8.5)	37.58(7.9)	40.86(6.3)	2.16	0.15
Medial Gastrocnemius	42.25(8.0)	39.09(5.2)	41.11(4.1)	0.58	0.57
Soleus	40.36(7.8)	38.97(9.9)	41.27(8.5)	0.26	0.77
Tibialis Posterior	39.48(6.7)	39.59(9.2)	41.73(7.2)	4.20	0.03725*
Foot	46.63(5.9)	46.79(7.3)	50.55(4.1)	2.44	0.15
N-8 SD- Standard Day	istion me-mil	lisecond *- sign	if i cant $(\mathbf{P}_{<}0.05)$ -	nost has test	showed

N=8, SD= Standard Deviation, ms= millisecond, *= significant (P<0.05)= post-hoc test showed difference in 12 months in comparison to baseline and 6 months.

Table 32: Longitudinal quantitative T_{2m} MRI muscle water of DHMN and repeated measures

analysis results.

Region of	BASE mediar	ELINE n (IQR)	6 MON median	THS (IQR)	12 MC media	DNTHS n (IQR)	Р
interest	Intensity	Extent	Intensity	Extent	Intensity	Extent	
Tibialis Anterior	2(1.5-2)	3(2.5-)	2(1.75-2)	3(2-3)	2(2-2)	3(2 to 3)	0.55
Tibialis Posterior	1(1-1.75)	2.75(1.75-3)	1.25(0.5-1.75)	2.25(0.5-3)	1.25(1-2)	2.25(1.25-3)	0.76
Peroneus Longus	1(1-1.5)	1.75(1-3)	1(0.75-1.5)	1.75(1-2.5)	1.25(0.75-1.5)	1.75(1-2.75)	0.72
Medial Gastrocnemius	1(1-1.25)	2(1.25-2.5)	1(1-1.5)	2(1.25-2)	1.5(1-1.75)	2.25(1.75-2.75)	0.019 12*
Lateral Gastrocnemius	1(1-1.5)	1.5(1-3)	1.5(1-1.5)	2(1-3)	1.5(1-1.5)	2.25(1.5-3)	0.16
Soleus	1.25(1-1.5)	2.5(1.25-3)	1.5(1-1.75)	2(1-2.75)	1.25(1-1.75)	2(1.25-2.5)	0.53
Foot	1(1 to 2)	1.25(1 to 2)	1(1 to 2)	1.75(1-2.5)	1(0.75-1.5)	1.25(1-2.5)	0.61
N=8, Intensity: 0	= none, 1=m	ild, 2=marke	d, Extent: 0= r	none, $1 = < 3$	0%, 2=30%-6	0%, 3 = > 60%,	IQR=
interquartile ran	ge, *= sign	ificant (P<0.	.05)= post-hoo	c test show	ved difference	e in 12 mon	ths in
comparison to ba	aseline and 6	months.					

Table 33: Longitudinal qualitative STIR MRI muscle water of DHMN and repeated measures analysis results.

6.3.5. Longitudinal Change in Strength Measurements

Friedman repeated measures analysis showed significant longitudinal difference only in the plantar flexion strength (P=0.046), measured using manual muscle testing (Table 34) (Figure 19). Post-hoc testing showed a decrease in median strength between baseline and 6 months. Repeated measures ANOVA showed longitudinal differences in muscle strength measured using dynamometry for isometric dorsiflexion and isokinetic plantar flexion (Table 35) (Figure 19). Posthoc testing showed differences were between baseline and 6 months, with an

Manual Muscle Testing	BASELINE median (IQR)	6 MONTHS median (IQR)	12 MONTHS median (IQR)	Р
Hip Flexion	5 (4 to 5)	5 (4 to 5)	5 (4 to 5)	0.13
Hip Extension	5 (3+ to 5)	5 (4- to 5)	5 (4- to 5)	0.39
Knee Flexion	5 (4- to 5)	5 (5 to 5)	5 (5 to 5)	0.13
Knee Extension	5 (4+ to 5)	5 (5 to 5)	5 (5 to 5)	0.39
Dorsiflexion	3- (1+ to 4)	2+ (1 to 4+)	2+ (1 to 4+)	0.84
Plantar flexion	4 (2 to 4+)	3+ (2- to 4+)	3+ (2- to 4+)	0.04606*
N=8, IQR= interquartile rang	ge, *= significant (P	<0.05)= post-hoc te	est showed difference	ce between

increase in mean ankle strength.

baseline and 6 months.

Table 34:Longitudinal manual muscle testing of DHMN and repeated measures analysis

results.

Isometric Dynamometry	BASELINE	6 MONTHS	12 MONTHS	Б	р
Nm	mean (SD)	mean (SD)	mean (SD)	Г	r
Hip Extension 45°	126.2(68.9)	122.8(67.5)	107.1(59.7)	1.85	0.20
Hip Flexion 45°	93.9(46.9)	98.8(50.2)	99.7(44.3)	0.66	0.53
Knee Extension 45°	86.1(44.5)	88.9(47.6)	80.6(42.3)	1.82	0.20
Knee Extension 90°	86.9(43.3)	85.4(43.8)	77.9(44.2)	1.90	0.20
Knee Flexion 45°	48.9(30.7)	50.8(33.2)	46.4(29.2)	0.66	0.47
Knee Flexion 90°	31.3(22.7)	37.6(23.6)	33.3(20.0)	1.52	0.25
Ankle Plantar flexion 10°	25.8(20.8)	28.6(23.8)	25.4(19.1)	0.42	0.67
Ankle Dorsiflexion 10°	5.6(6.5)	9.8(8.9)	9.1(6.5)	4.57	0.03*
Ankle Dorsiflexion 30°	10.8(8.6)	17.8(12.1)	18.4(10.3)	6.70	0.009*
Isokinetic Dynamometry	BASELINE	6 MONTHS	12 MONTHS	Б	р
Isokinetic Dynamometry Nm	BASELINE mean (SD)	6 MONTHS mean (SD)	12 MONTHS mean (SD)	F	Р
Isokinetic Dynamometry Nm Hip Extension 60°/60s	BASELINE mean (SD) 96.5(70.2)	6 MONTHS mean (SD) 101.6(71.6)	12 MONTHS mean (SD) 99.2(95.6)	F 0.39	P 0.69
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s	BASELINE mean (SD) 96.5(70.2) 86.1(45.0)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5)	F 0.39 3.88	P 0.69 0.07
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s Knee Extension 60°/60s	BASELINE mean (SD) 96.5(70.2) 86.1(45.0) 63.4(39.1)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8) 67.6(39.5)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5) 64.7(95.7)	F 0.39 3.88 0.73	P 0.69 0.07 0.46
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s Knee Extension 60°/60s Knee Flexion 60°/60s	BASELINE mean (SD) 96.5(70.2) 86.1(45.0) 63.4(39.1) 34.1(23.8)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8) 67.6(39.5) 36.9(26.2)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5) 64.7(95.7) 35.3(95.0)	F 0.39 3.88 0.73 0.61	P 0.69 0.07 0.46 0.56
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s Knee Extension 60°/60s Knee Flexion 60°/60s Knee Extension 120°/120s	BASELINE mean (SD) 96.5(70.2) 86.1(45.0) 63.4(39.1) 34.1(23.8) 46.9(31.9)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8) 67.6(39.5) 36.9(26.2) 52.0(31.3)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5) 64.7(95.7) 35.3(95.0) 47.4(95.1)	F 0.39 3.88 0.73 0.61 1.68	P 0.69 0.07 0.46 0.56 0.23
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s Knee Extension 60°/60s Knee Flexion 60°/60s Knee Flexion 120°/120s Knee Flexion 120°/120s	BASELINE mean (SD) 96.5(70.2) 86.1(45.0) 63.4(39.1) 34.1(23.8) 46.9(31.9) 25.4(18.3)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8) 67.6(39.5) 36.9(26.2) 52.0(31.3) 29.4(22.6)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5) 64.7(95.7) 35.3(95.0) 47.4(95.1) 26.2(95.3)	F 0.39 3.88 0.73 0.61 1.68 1.05	P 0.69 0.07 0.46 0.56 0.23 0.38
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s Knee Extension 60°/60s Knee Flexion 60°/60s Knee Flexion 120°/120s Knee Flexion 120°/120s Ankle Plantar flexion 60°/60°	BASELINE mean (SD) 96.5(70.2) 86.1(45.0) 63.4(39.1) 34.1(23.8) 46.9(31.9) 25.4(18.3) 11.0(7.4)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8) 67.6(39.5) 36.9(26.2) 52.0(31.3) 29.4(22.6) 18.9(13.8)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5) 64.7(95.7) 35.3(95.0) 47.4(95.1) 26.2(95.3) 15.6(95.5)	F 0.39 3.88 0.73 0.61 1.68 1.05 5.78	P 0.69 0.07 0.46 0.56 0.23 0.38 0.031*
Isokinetic Dynamometry Nm Hip Extension 60°/60s Hip Flexion 60°/60s Knee Extension 60°/60s Knee Flexion 60°/60s Knee Flexion 120°/120s Knee Flexion 120°/120s Ankle Plantar flexion 60°/60°	BASELINE mean (SD) 96.5(70.2) 86.1(45.0) 63.4(39.1) 34.1(23.8) 46.9(31.9) 25.4(18.3) 11.0(7.4) 11.8(7.0)	6 MONTHS mean (SD) 101.6(71.6) 100.5(53.8) 67.6(39.5) 36.9(26.2) 52.0(31.3) 29.4(22.6) 18.9(13.8) 17.4(10.6)	12 MONTHS mean (SD) 99.2(95.6) 103.9(95.5) 64.7(95.7) 35.3(95.0) 47.4(95.1) 26.2(95.3) 15.6(95.5) 20.4(95.7)	F 0.39 3.88 0.73 0.61 1.68 1.05 5.78 3.53	P 0.69 0.07 0.46 0.56 0.23 0.38 0.031* 0.06

showed difference between baseline and 6 months.

Table 35: Longitudinal dynamometry of DHMN and repeated measures analysis results.



Isometric Dynamometry (ISOM)





Figure 19: Longitudinal change of muscle strength in DHMN. N=8, M= months, Nm= Newton meter. Hip and knee on MMT graph are at the same level of strength.

6.3.6. Longitudinal Change in Gait Parameters

Longitudinal analysis of 3D gait parameters, spatiotemporal (Table 36), kinematics (Table 37) (Figure 20), and kinetics (Table 38) (Figure 21, Figure 22), did not show significant change across the three measurement points. The only parameter that showed significant difference on Friedman test was the minimum hip moment (P= 0.00461). Post-hoc test showed difference in baseline in comparison to 6 months and 12 months. From baseline to 6 months, analysis showed an increase in hip flexor moment, however, the 12 months measurement showed a decrease in hip flexor moment.

Gait Spatiotemporal	BASELINE mean (SD)	6 MONTHS mean (SD)	12 MONTHS mean (SD)	F	Р
Speed; m/s	0.90(0.25)	0.95(0.23)	0.90(0.27)	1.91	0.18
Stride Length; m	1.10(0.22)	1.12(0.19)	1.10(0.21)	1.19	0.33
Stride Time; s	1.25(0.18)	1.21(0.14)	1.26(0.17)	1.47	0.26
Strides Per Minute	48.79(5.47)	50.14(5.25)	48.5(6.27)	1.93	0.20
Step Length; m	0.55(0.11)	0.56(0.10)	0.55(0.10)	1.18	0.34
Step Time; s	0.63(0.09)	0.61(0.07)	0.63(0.09)	1.47	0.26
Steps Per Minute	97.58(10.93)	100.27(10.50)	97.1(12.55)	1.93	0.20
Percent Stance	0.64(0.02)	0.64(0.02)	0.65(0.02)	1.52	0.25
Single Support Time; s	0.61(0.05)	0.61(0.07)	0.64(0.10)	0.58	0.57
Double Support Time; s	0.19(0.08)	0.17(0.04)	0.17(0.04)	1.00	0.39
Percent Opposite Toe Off	15.09(4.16)	13.99(2.29)	13.5(1.78)	0.39	0.69
Percent Opposite Foot Contact	49.48(0.69)	49.96(0.67)	50.0(0.78)	2.75	0.10
N=8, SD= Standard Deviation, r	n= meters, s= se	conds.			

 Table 36: Longitudinal spatiotemporal parameters of DHMN and repeated measures

analysis results.

Gait Kinematics	BASELINE mean (SD)	6 MONTHS mean (SD)	12 MONTHS mean (SD)	F	Р
Pelvis X Max; °	4.08(0.85)	3.89(1.11)	3.94(0.79)	0.65	0.54
Pelvis X Min; °	-4.26(1.16)	-3.88(1.22)	-4.3(0.82)	0.26	0.77
Pelvis Y Max; °	15.75(5.97)	14.99(4.72)	16.58(3.58)	0.37	0.70
Pelvis Y Min; °	10.72(4.80)	9.80(3.87)	11.5(3.11)	0.48	0.63
Pelvis Z Max; °	7.04(2.78)	7.60(3.67)	7.23(4.49)	0.29	0.75
Pelvis Z Min; °	-7.33(3.15)	-7.11(3.84)	-7.8(3.76)	0.03	0.97

Hip Max; °	46.81(6.01)	44.49(5.48)	47.8(3.15)	0.20	0.82
Hip Min; °	4.18(4.38)	0.61(5.84)	3.18(3.54)	1.38	0.28
Knee Max; °	73.24(4.97)	73.94(3.33)	73.8(3.26)	0.12	0.82
Knee Min; °	6.47(5.35)	5.47(5.78)	5.41(8.09)	0.22	0.70
Ankle Max; °	24.09(6.15)	24.52(4.90)	24.0(4.55)	0.07	0.84
Ankle Min; °	-11.69(3.86)	-12.35(6.03)	-13.(5.28)	0.64	0.49

N=8, SD= Standard Deviation= Pelvic z Maximum= Rotation forwards, Pelvic z Minimum= Rotation backwords, Pelvic x Maximum, lateral raise, Pelvic x Minimum= lateral drop, Pelvic y Maximum= Anterior tilt, Pelvic y Minimum = Posterior tilt, Hip y Maximum= Flexion, Hip y Minimum= Extension, Knee y Maximum= Flexion, Knee y Minimum= Extension, Ankle y Maximum= Dorsiflexion, Ankle y Minimum= Plantarflexion.

Table 37: Longitudinal kinematics parameters of DHMN and repeated measures analysis results.

12 6 BASELINE F **Gait Kinetics** MONTHS MONTHS Р mean (SD) mean (SD) mean (SD) Hip Moment Y Max; Nm/Kg 1.29(0.41) 1.28(0.48) 1.28(0.61) 0.00 0.97 Hip Moment Y Min; Nm/Kg -0.81(0.84)-0.5(0.26)0.00461* -0.45(0.16)8.10 Knee Moment Y Max; Nm/Kg 0.73(0.32) 0.58(0.26) 1.06(1.34) 1.47 0.26 Knee Moment Y Min; Nm/Kg -0.6(0.35) -0.60(0.26)-0.62(0.30) 0.37 0.70 Ankle Moment Y Max; Nm/Kg 1.00(0.26) 0.97(0.26) 0.86(0.42) 0.58 0.49 Ankle Moment Y Min; Nm/Kg -0.05(0.04)-0.47(1.22)-0.0(0.05)0.45 0.86 Hip Power Max During Swing; W/kg 0.97(0.27) 0.95(0.33) 0.90(0.20) 0.61 0.56 Hip Power Min During Stance; W/kg -0.9(0.85)-0.74(0.67)-1.44(1.56)0.54 0.65 Hip Power Max During Stance; W/kg 1.61(0.63) 1.80(0.80) 1.84(1.05)0.51 0.56 Knee Power Max During Swing; W/kg 0.34(0.11) 0.31(0.06) 0.41(0.33) 0.26 0.77 Knee Power Min During Stance; W/kg -1.33(0.91)-2.77(4.22) -1.8(1.11) 2.92 0.09 Knee Power Max During Stance; W/kg 1.64(1.24) 1.41(1.10) 1.14(0.66) 1.40 0.28 Ankle Power Max During Swing; W/kg 0.04(0.02)0.06(0.04) 0.09(0.10) 1.72 0.21 Ankle Power Min During Stance; W/kg -1.07(0.36)-1.58(1.45)-1.1(0.35)0.86 0.45 Ankle Power Max During Stance; W/kg 1.02(0.46) 2.11(2.72) 1.11(0.35) 1.47 0.26

N=8, SD= Standard Deviation, Nm/Kg= Newton Meter per Kilogram, W/kg= Watts Per Kilogram, *= significant (P<0.05)= post-hoc test showed difference in baseline in comparison to 6 months and 12 months= Hip Moments Maximum= Extension, Hip Moments Minimum= Flexion, Knee Moments Maximum= Extension, Knee Moments Minimum= Flexion, Ankle Moments Maximum= Plantarflexion, Ankle Moments Minimum= Dorsiflexion, Hip Power Maximum= generation in swing phase, Hip Power Minimum= absorption in stance phase, Knee Power Maximum= generation in stance phase, Knee Power Maximum= generation in stance phase, Ankle Power Maximum= generation in

Table 38: Longitudinal kinetics parameters of DHMN and repeated measures analysis

results.



Figure 20:Grand average angular displacement in the sagittal plane over one gait cycle for the ankle, knee, and hip (average of left and right sides). Comparison of longitudinal measurements of people with DHMN.

OTO, *Opposite side tore off; OIC*, *Opposite side initial contact; TO*, *Toe off, N=8, M= months*.



Figure 21: Grand average joint moments in the sagittal plane over one gait cycle for the ankle, knee and hip (average of left and right sides). Comparison of longitudinal measurements of people with DHMN.

OTO, *Opposite side tore off; OIC*, *Opposite side initial contact; TO*, *Toe off, N=8, M= months*.



Figure 22:Grand average joint powers in the sagittal plane over one gait cycle for the ankle, knee and hip (average of left and right sides). Comparison of longitudinal measurements of people with DHMN.

OTO, *Opposite side tore off; OIC*, *Opposite side initial contact; TO*, *Toe off, N=8, M= months*.

6.3.7. Responsiveness Over 12 Months

Standardise response mean (SRM) was calculated as the ratio of mean change to standard deviation of change to measure the responsiveness of the outcome measures used (Table 39). Overall, in parameters with significant longitudinal difference, they were responsive distally more than proximally (Table 39). Calf muscles MRI fat fraction was the most responsive with SRM ranging from 1.1 to 1.78 (Table 39). Calf remaining muscle area and muscle water T_{2m} showed large responsiveness as well with SRM ranging from -1.08 to 1.06 (Table 39). Moderate responsiveness was shown in Gracilis muscle remaining muscle area (SRM= 0.78) and hip flexors moment (SRM= -0.72) (Table 39).

Muscle Fat Fraction	BASELINE	12 MONTHS	Difference	D	SDM	
(%)	mean (SD)	mean (SD)	mean (SD)	L	SINI	
Total Calf	36.61(22.35)	39.83(22.49)	3.22(2.18)	0.003	1.48	
Dorsiflexion	23.89(19.01)	26.57(19.84)	2.68(1.68)	0.001	1.60	
Tibialis Anterior	23.89(19.01)	26.57(19.84)	2.68(1.68)	0.001	1.60	
Peroneus Longus	39.30(27.19)	41.61(26.19)	2.31(1.47)	0.009	1.57	
Plantarflexion	40.16(23.61)	44.04(24.11)	3.88(3.24)	0.009	1.20	
Soleus	43.22(23.70)	47.74(24.53)	4.52(4.12)	0.013	1.10	
Tibialis Posterior	30.03(22.23)	34.05(22.18)	4.02(2.26)	0.008	1.78	
Remaining Muscle	BASELINE	12 MONTHS	Difference	Р	SPM	
Area (mm ²)	mean (SD)	mean (SD)	mean (SD)	1	JIMI	
Gracilis	416.0(226.9)	432.2(238.6)	16.14(20.6)	0.05	0.78	
Plantarflexion	4166.3(2608.4)	3879.8(2644)	-286.6(306.3)	0.05	-0.94	
Soleus	1094.3(642.4)	993.9(638.1)	-100.44(93)	0.02	-1.08	
Muscle Water T _{2m}	BASELINE	12 MONTHS	Difference	р	SDM	
(ms)	mean (SD)	mean (SD)	mean (SD)	r	SKIVI	
Tibialis Posterior	39.48(6.7)	41.73(7.2)	2.25(2.1)	0.04	1.06	
Coit Vinctics (Nor /Va)	BASELINE	12 MONTHS	Difference	р	SRM	
Gait Kinetics (Nm/Kg)	mean (SD)	mean (SD)	mean (SD)	r		
Hip Moment Y Min	-0.45(0.16)	-0.5(0.26)	-0.11(0.2)	0.005	-0.72	
N=8, SRM= Standardise response mean, SD= standard deviation, mm ² = square millimetres, ms=						
millisecond, Nm/Kg= Newton Meter per Kilogram, SRM < 0.2 minimal responsiveness, SRM =0.2-						
0.5 small responsiveness, SRM =0.5-0.8 moderate responsiveness, SRM >0.8 large responsiveness.						

Table 39: Responsiveness of the study outcome measures with significant difference over 12months

6.4. Discussion

This study explored the longitudinal changes in a DHMN cohort primarily using MRI, dynamometry, and 3D motion analysis. Over 12 months, quantitative and qualitative MRI parameters were used to assess the effect of DHMN on muscle fatty infiltration, muscle oedema, and muscle cross sectional area. Manual muscle testing and dynamometry were used to assess the effect on muscle strength. Three-D motion analysis was used to understand functional deterioration in walking gait kinematics, kinetics, and spatiotemporal parameters.

To summarise, MRI fat fraction at the calf level was the most sensitive and responsive measure to change. Results showed significant deterioration with an increase in fatty infiltration over 12 months. This deterioration was not detected using qualitative measures. Analysis T1w image, using Modified Mercuri's Scale, did not show change over time. While cross sectional area did not change over 12 months, calculation of the actual remaining muscle area showed deterioration in the Soleus muscle which affected the overall plantar flexors muscle area. These findings are consistent with the established knowledge regarding the quantitative fat fraction in peripheral neuropathy from previous CMT studies (Morrow et al., 2016), and regarding plantar flexors as a targeted group of muscles in DHMN (Esteller et al., 2023, O'Donnell et al., 2023).

Qualitative analysis results were consistent with the quantitative analysis and the expected manifestation of peripheral neuropathy, however, the significant

difference shown at the foot level could be a type I error since the pairwise analysis did not show difference across measurement time points. The Friedman test was calculated manually to confirm the results (appendix II). Calculating the test statistics (χ^2) yield a value of 0 and P-value =1. Thus, the null hypothesis is accepted and there is no difference between measurements. This confirms that there should be no significant result when all values are the same and indicates that in such scenarios the test statistic calculation might leads to an invalid result (appendix II).

Signs of active denervation were shown in the posterior compartment of the lower leg superficially (Medial Gastrocnemius) and deeply (Tibialis posterior) in the qualitative (STIR) and quantitative (T_{2m}) analysis of muscle oedema, respectively. Qualitative Medial Gastrocnemius hyperintensity was not associated with a significant increase in fat fraction levels as in Tibialis posterior. Moreover, quantitative assessment is generally superior to qualitative visual inspection, as quantitative methods are observer-independent and can report changes on a continuous scale versus an ordinal scale for visual inspection (Willis et al., 2013, Dahlqvist et al., 2020).

Deterioration in muscle strength over 12 months was detectable using manual muscle testing. Although this method has been critiqued (Escolar et al., 2001), it showed change in strength over time that was not seen in dynamometry where there is distal weakness and limitation of range of motion that limit the testing

ability and alter ideal positioning and movement isolation. Besides the learning effect, it was noted that participants tend to engage their upper body to pull upwards and push downwards into the dynamometer footplate which might contributed to the overall torque measured and result in the increase in Ankle isokinetic and isometric measurements.

Earlier research highlighted the challenges of using dynamometry in the presence of significant weakness and limited joint mobility (Guillebastre et al., 2013, Reynaud et al., 2019) . In contrast, proximal testing of hip and knee movement where joints are not affected by the neuropathy weakness and limited range of motion, dynamometry was able to detect change over 12 months. Although repeated measure analysis was not powerful enough to show significant difference over one year due to the small sample size and subject variability, the standardised response mean showed high responsiveness in isokinetic hip flexion. However, a larger longitudinal study that include a DHMN group and a control group would help to understand the proximal changes overtime.

Proximal muscles often compensate for distal weaknesses in peripheral neuropathies (Ramdharry et al., 2009). Minimum hip moment represents the activation of hip flexors at pre-swing which is expected to increase as a compensatory strategy for plantar flexors weakness as discussed in chapter 3. The proximal changes including an increase in the Gracilis muscle area and an increase in hip flexion moment shows a possible pattern of proximal

compensation in DHMN. The Gracilis is a long, slender muscle located in the inner thigh. It is one of the muscles in the adductor group, yet it aids in hip and knee flexion (Neumann, 2010, Goldberg et al., 2004). Biomechanical studies showed that maximum knee flexion velocity in swing phase is influenced by the activation of Gracilis and plantar flexors muscles at pre-swing (Goldberg et al., 2004, Anderson et al., 2004, Arnold et al., 2005, Akalan et al., 2017). Therefore, with plantar flexors reduced moments and power generation, the Gracilis is expected to compensate to facilitate knee flexion at swing phase besides its secondary role in assessing hip flexion. This repeated overuse mechanism can cause muscle hypertrophy. However, this hypothesis is based on MRI and kinetic data from a small and variable sample of DHMN. The change in hip moment need to be treated with some caution, however, it is increased at 6 months and decreased at 12 months. This fluctuation would not be expected in a slowly progressive condition. The finding could be a type I error, more likely in this situation of a small sample with high between subject variability. Future research with a larger sample and electromyography studies in addition to MRI and 3d motion analysis can confirm this hypothesis.

6.5. Summary

The current study explored the natural history of DHMN over one year (Table 40), and the responsiveness of the measures used. Among measurements, distal parameters were more responsive than proximal parameters, and MRI fat fraction of the calf was the most responsive.

	Fat Fraction	Muscle Oedema	Muscle Area	Muscle Strength	Gait
Proximal	0	0	Gracilis muscle area	0	↑ Hip flexors moment
Calf/ Distal	1	0	↓	0	Ο
Dorsiflexion	1	0	0	0	0
Tibialis Anterior	↑	0	0	0	0
Peroneus Longus	←	0	0	0	0
Plantarflexion	1	0	0	→	0
Lateral Gastrocnemius	0	0	0	0	0
Medial Gastrocnemius	0	0	0	0	0
Soleus	1	0	↓	0	0
Tibialis Posterior	←	←	0	0	0
Foot	0	0	0	0	0
O, No change; ↑, Increase; ↓, Decrease					

Table 40: Summary of longitudinal changes in DHMN cohort

Chapter 7: The Relationships Between Intramuscular Fat Fraction, Muscle Volume, Muscle Strength, and Moments/Power Generation in Distal Hereditary Motor Neuropathy (DHMN)

7.1. Introduction

In the previous chapters we established an idea of the clinical presentation of DHMN, and we identified a pattern of involvement showing weakness and fatty infiltration predominantly in the distal muscles. Results also showed trends of change in gait parameters at the ankle, knee, and hip level. Previous studies in CMT showed that intramuscular fat fraction correlate cross sectionally as well as longitudinally with strength and disease specific measurements. Understanding the relationship between measurements will inform future research and design of clinical trials. For example, understanding the relation between muscle strength and gait parameters can inform intervention trials in designing strengthening exercises programmes to improve walking.

This study aimed to ascertain relationships between intramuscular fat fraction, muscle volume, isokinetic and isometric muscle strength, and kinetics of gait. Based on the finding in the previous chapters, relationships were expected to be found between:

• Fat fraction with strength, moment, and power generation during gait, for distal lower limb muscles.

- Quantitative and qualitative measurements of intramuscular fat and water, and muscles strength measurements.
- Clinical severity measured with fat fraction, strength, and gait parameters.

7.2. Methods

Parameters associated directly with muscle performance were used primary to explore the relationships. Relations between the rest of the parameters and correlations between quantitative and qualitative measurements were explored cross sectionally. For cross sectional correlation analysis, baseline measurements from 10 DHMN participants recruited for study 1 (chapter 5) who completed MRI, dynamometry, and 3D motion analysis are used.

7.2.1. Data Analysis

To explore relationships between parameters, Pearson correlation was used, if the data set was normally distributed, or Spearman rank correlation, where the data set was not normally distributed. Correlation can be positive or negative and categorised by magnitude (Mukaka, 2012): 0.0-0.29, Minimal/Negligible; 0.30-0.49, Low; 0.50-0.69, Moderate; 0.70-0.89, High; 0.90-1.0, Very High.

It is worth noting that gait parameters with constant, absolute negative signs, such as minimum moments (dorsiflexion, knee and hip flexion), and minimum power (absorption), were interpreted by what they present and not by the absolute negative value. The negative value was used to differentiate the graphs, e.g. the muscle group, as in moment graphs, or power absorption or generation, in power graphs. For example, a value of -2.0 minimum power presents a higher power absorption levels than a value of -1.0 minimum power although -1.0 absolute value is more than -2.0. This could affect the correlation direction if not accounted

for when comparing against parameters with absolute constant positive value such as isometric strength. Therefore, minimum moment and minimum power variables were included in the analysis using their absolute value of the magnitude without the (-) sign. the Parameters for correlations test were paired according to Table 41.

Primary	Distal	 Baseline measurements of: Calf MRI quantitative intramuscular FF and the calculated remaining muscle area. Ankle isokinetic and isometric dynamometry. Ankle moment and power generation during gait.
Comparisons	Proximal	 Baseline measurements of: Thigh MRI quantitative intramuscular FF and the calculated remaining muscle area Knee isokinetic and isometric dynamometry. Knee moment and power generation during gait.
Secondary Comparisons	Cross Sectional at Baseline	 Between MRI quantitative measurements (FF, RMA, CSA,T_{2m}). Between MRI FF and other clinical measurements (CMTES,Walk-12,FPI-6). Between quantitative and qualitative measurements. Between strength and spatiotemporal gait parameters.

Table 41: Primary and secondary correlation tests.

FF, fat fraction; *RMA*, remaining muscle area; *CSA*, cross sectional area; T_{2m} , quantitative intramuscular water; *CMTES*, *CMT* examination score; *FPI*, foot posture index.

7.3. Results

7.3.1. Subjects and Clinical Assessment

Data from 10 DHMN participants who underwent MRI scan, Dynamometry, and 3D motion analysis at baseline was used for cross sectional correlation analysis (appendix VII). A summary of the subjects' demographics and clinical assessment is presented in Table 42.

Demographics and clinical assessment	BASELINE			
Numbers	10			
Gender (M/F)	(5/5)			
Age; mean years; range	57 (42/75)			
Genetic diagnosis (HSPB1/unknown)	(7/4)			
CMTES; mean (SD)	6.2(3.6)			
Foot Posture Index-6 (Pronated/Normal/Supinated)	(1/8/1)			
Fall Frequency (weekly/Monthly/Yearly)	(1/1/2)			
Walk-12; mean (SD)	36(12.4)			
Range of Motion	BASELINE			
Dorsiflexion; count, (Limited/Normal)	(11/0)			
Plantarflexion; count, (Limited/Normal) (10/1)				
M= Male, F= Female, HSPB1= Heat-shock 27-KD Protein 1, SD= Standard Deviation, CMTES= Charcot-Marie-Tooth Disease Examination Score, *= significant after modified Bonferroni correction.				

Table 42:Summary of demographics and clinical assessment of DHMN participants at baseline and 12 months

Correlation analysis was completed using Pearson correlation or Spearman rank correlation. Significant positive and negative correlations are reported below by muscle group; ankle plantar flexors, ankle dorsiflexors, knee extensors, and knee flexors. Correlograms showing the correlation and significance level for all parameters comparisons are shown in appendix III.

7.3.2. Relationships Between MRI Parameters

Plantar Flexors

Between MRI parameters, correlation analysis showed a significant high negative correlation between plantar flexion muscle fat fraction and the remaining muscle area (-0.791, P=0.0065) (Figure 23). The relationship was significant and highly positive between the remaining muscle area and cross sectional area (0.843, P=0.0022) (Figure 24) (Appendix III, Table 78). Quantitative calf intramuscular water measured using (T_{2m}) showed a moderate correlation with plantar flexion muscle fat fraction, however, it was not significant (0.58, P=0.0790) (Table 43).



Figure 23: Correlation between plantar flexors fat fraction (%) and remaining muscle area (mm^2) .



Figure 24: Correlation between plantar flexors remaining muscle area (mm^2) and cross sectional area (mm^2) .

Dorsiflexors

Between MRI parameters, correlation analysis showed a significant high negative correlation between dorsiflexors muscle fat fraction with the remaining muscle area (-0.867, P=0.0012) and moderate negative with the cross sectional area (-0.673, P=0.0330) (Figure 25). The relationship was significant and very highly positive between the remaining muscle area and cross sectional area (0.981, P<0.0001) (Figure 26) (Appendix III, Table 79). Quantitative calf intramuscular water measured using T_{2m} showed a highly significant, positive correlation with dorsiflexors fat fraction (0.865, P=0.0012) (Table 43).



Figure 25: Correlation between dorsiflexors (DF) fat fraction (%) and remaining muscle area (RMA) and cross sectional area (CSA) (mm²).



Figure 26: Correlation between dorsiflexors remaining muscle area (RMA) and cross sectional area (CSA) (mm²).

Total Calf

Quantitative calf intramuscular water measured using (T_{2m}) showed a highly significant, positive correlation with the total calf fat fraction at baseline (0.736, P=0.0153) (Table 43) (Figure 27).

	Calf T _{2m}	Plantar flexion T _{2m}	Dorsiflexion T _{2m}	Foot T _{2m}
	0.736			
Calf FF	P=0.0153			
	n=10			
		0.58		
Plantar flexion FF		P=0.0790		
		n=10		
			0.865	
Dorsiflexion FF			P=0.0012	
			n=10	
				0.535
Foot FF				P=0.1112
				n=10
	1	0	-1	

Table 43: Correlation between baseline fat fraction (FF) and baseline T_{2m} .



Figure 27: Correlation between total calf fat fraction (%) and T_{2m} intramuscular water (ms).
Knee Extensors

Between MRI parameters, correlation analysis showed a significant high negative correlation between knee extensors muscle fat fraction and the remaining muscle area (-0.812, P=0.0044) (Figure 28) and a significant moderate negative correlation with the cross sectional area (-0.695, P=0.0257) (Figure 28). The relationship was significant and very highly positive between the remaining muscle area and cross sectional area (0.977, P<0.0001) (Figure 29) (Appendix III, Table 80).



Figure 28: Correlation between knee extensors (EX) fat fraction (%) and remaining muscle area (RMA) and cross sectional area (CSA) (mm²).



Figure 29: Correlation between knee extensors (EX) remaining muscle area (RMA) and cross sectional area (CSA) (mm²).

Knee Flexors

Correlations between MRI parameters showed a significant high negative relationship between knee flexors muscle fat fraction with the remaining muscle area (-0.798, P=0.0056) (Figure 30) and cross sectional area (-0.735, P=0.0155) (Figure 30). the correlation was significant and very high positive between the remaining muscle area and cross sectional area (0.99, P<0.0001) (Figure 31) (Appendix III, Table 81).



Figure 30: Correlation between knee flexors (FLX) fat fraction (%) and remaining muscle area (RMA) and cross sectional area (CSA) (mm²).



Figure 31: Correlation between knee flexors (FLX) remaining muscle area (RMA) and cross sectional area (CSA) (mm²).

Relationship Between Quantitative and Qualitative MRI Measurements

Intramuscular fat measured qualitatively using modified Mercuri scale correlated significantly and positively with quantitative fat fraction. High correlations were observed with the calf (0.806, P=0.0048) (Table 44) (Figure 32) and the plantar flexors (0.85, P=0.0019) (Table 44) (Figure 33). A very high correlation was seen with the dorsiflexors (0.914, P=0.0002) (Table 44) (Figure 34).

	Calf	Plantar flexion	Dorsiflexion	Foot
	0.806			
Calf FF	P=0.0048			
	n=10		_	
		0.85		
Plantar flexion FF		P=0.0019		
		n=10		
			0.914	
Dorsiflexion FF			P=0.0002	
			n=10	
				0.406
Foot FF				P=0.2441
				n=10
	1	0 .	-1	

Table 44: Correlation between baseline fat fraction (FF) and Mercuri score.



Figure 32: Correlation between calf fat fraction (%) and Mercuri score.



Figure 33: Correlation between plantar flexors fat fraction (%) and Mercuri score.



Figure 34: Correlation between dorsiflexors fat fraction (%) and Mercuri score.

Intramuscular water measured qualitatively using STIR hyperintensity score showed over all negligible correlation with quantitative T_{2m} , except at the dorsiflexors, which showed a significant moderate positive correlation (0.646, P=0.0437) (Table 45) (Figure 35).

	Calf	Plantar flexion	Dorsiflexion	Foot
	0.177			
Calf T _{2m}	P=0.6240			
	n=10			
		0.073		
Plantar flexion T _{2m}		P=0.8422		
		n=10		
			0.646	
Dorsiflexion T _{2m}			P=0.0437	
			n=10	
				-0.319
Foot T _{2m}				P=0.3686
				n=10
	1	0 -	1	

Table 45: Correlation between baseline intramuscular water T_{2m} *and STIR hyperintensity score.*



Figure 35: Correlation between dorsiflexors STIR score and T_{2m} *intramuscular water (ms).*

7.3.3. Relationships Between MRI and Dynamometry Parameters

Plantar Flexors

Plantar flexion isometric dynamometry showed a significant high positive correlations with MRI plantar flexors remaining muscle area (0.754, P=0.0118) (Figure 36) and cross sectional area (0.827, P=0.0031) (Figure 36) (Appendix III, Table 78).



Figure 36: Correlation between Plantar flexion (PF) isometric dynamometry (Nm) and remaining muscle area (RMA) and cross sectional area (CSA) (mm²).

Dorsiflexors

Dorsiflexor fat fraction showed a significant high negative correlation with dorsiflexion isometric dynamometry (-0.867, P=0.0012) (Figure 37) (Appendix III, Table 79).



Figure 37: Correlation between dorsiflexion isometric dynamometry (Nm) and fat fraction (%).

Dorsiflexion isometric dynamometry showed a significant high positive correlation with the MRI dorsiflexors remaining muscle area (0.891, P=0.0005) (Figure 38) and with the cross sectional area (0.703, P=0.0235) (Figure 38) (Appendix III, Table 79).



Figure 38: Correlation between dorsiflexion (DF) isometric dynamometry (Nm) and remaining muscle area (RMA) and cross sectional area (CSA) (mm²).

Knee Extensors

Knee extensor cross sectional area showed a significant high correlation with isometric dynamometry (0.855, P=0.0016) (Figure 39) and with isokinetic dynamometry (0.785, P=0.0071) (Figure 39). Similarly, knee extensor remaining muscle area showed a significant, high correlation with isometric dynamometry (0.844, P=0.0021) (Figure 40) and with isokinetic dynamometry (0.761, P=0.0105) (Figure 40) (Appendix III, Table 80).



Figure 39: Correlation between knee extensors (EX) cross sectional area (CSA)(mm²) and isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm).



Figure 40: Correlation between knee extensors (EX) remaining muscle area (RMA)(mm²) and isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm).

Knee Flexors

Knee flexor isometric dynamometry showed a significant very high positive correlation with the remaining muscle area (0.918, P=0.0002) (Figure 41) and a significant high positive correlation with the cross sectional area (0.897, P=0.0004) (Figure 41). A strong negative correlation was observed with knee flexors fat fraction was significant and negatively high (-0.71, P=0.0213) (Figure 42). Knee flexor isokinetic dynamometry showed a significant high positive correlation with flexors remaining muscle area (0.831, P=0.0029) (Figure 43) and cross sectional area (0.806, P=0.0049) (Figure 43). However, correlation with knee flexors fat fraction was significant and negatively moderate (-0.649, P=0.0421) (Figure 44) (Appendix III, Table 81).



Figure 41: Correlation between knee flexion (FLX) isometric dynamometry (Nm) and remaining muscle area (RMA) (mm^2), and cross sectional area (CSA) (mm^2).



Figure 42: Correlation between knee flexion (FLX) isometric dynamometry (Nm) and fat fraction (FF)(%).



Figure 43: Correlation between knee flexion (FLX) isokinetic dynamometry (Nm) and remaining muscle area (RMA) (mm^2), and cross sectional area (CSA) (mm^2).



Figure 44: Correlation between knee flexion (FLX) isokinetic dynamometry (Nm) and fat fraction (FF)(%).

7.3.4. Relationships Between MRI and Gait Parameters

Dorsiflexors

Dorsiflexors fat fraction showed a significant moderate negative correlation with ankle power generation (maximum power) during swing (-0.673, P=0.0330) (Figure 45) (Appendix III, Table 79).



Figure 45: Correlation between dorsiflexors fat fraction (%) and ankle power generation during swing (W/Kg).

Knee Extensors

Knee extensor cross sectional area showed a significant positive high correlation with the knee extensor (maximum) moment during stance (0.876, P=0.0009) (Figure 46), a significant positive moderate correlation with knee power generation (maximum) during stance (0.659, P=0.0384) (Figure 46), and a significant high positive correlation with knee power absorption (minimum) during stance (0.865, P=0.0012) (Figure 46). Similarly, knee extensor remaining muscle area showed a significant positive high correlation with knee extensors (maximum) moment during stance (0.818, P=0.0038) (Figure 47), a significant positive moderate correlation with knee power generation (maximum) during stance (0.657, P=0.0390) (Figure 47), and a significant high positive correlation with knee power absorption (minimum) during stance (-0.754, P=0.0117) (Figure 47) (Appendix III, Table 80).



Figure 46: Correlation between knee extensor cross sectional area (mm^2) and knee extensor moment (Nm/Kg) and knee power generation and absorption (W/Kg).



Figure 47: Correlation between knee extensor remaining muscle area (mm^2) and knee extensor moment (Nm/Kg) and knee power generation and absorption (W/Kg).

Knee Flexors

Knee power absorption (minimum) during stance showed a strong, significant positive correlation with knee flexor remaining muscle area (0.818, P=0.0039) (Figure 48) and cross sectional area (0.829, P=0.0031) (Figure 48). Knee extensors moment (maximum) in stance showed a strong, significant, positive correlation with knee flexor remaining muscle area (0.846, P=0.0021) (Figure 49) and cross sectional area (0.809, P=0.0046) (Figure 49). Knee flexors fat fraction showed a strong significant, negative correlation with knee extensor moment (maximum) in stance (-0.783, P=0.0074) (Figure 50) (Appendix III, Table 81).



Figure 48: Correlation between knee power absorption (W/Kg) and knee flexors (FLX) cross sectional area (CSA) (mm²), and remaining muscle area (RMA)(mm²).



Figure 49: Correlation between knee extensors moment (Nm/Kg) and knee flexors (FLX) cross sectional area (CSA)(mm²), and remaining muscle area (RMA)(mm²).



Figure 50: Correlation between knee extensors moment (Nm/Kg) and knee flexors (FLX) fat fraction (%).

7.3.5. Relationships Between MRI Parameters and Clinical Measurements

The Charcot Marie Tooth Examination Score (CMTES) showed significant positive correlations with MRI fat fraction. At the total thigh level, the correlation was high (0.891, P=0.0005) (Table 46) (Figure 51), with high correlation with knee flexors (0.89, P=0.0006), and moderate with knee extensors (0.741, P=0.0143). At the total calf level, the correlation was moderate (0.648, P=0.0427) (Table 46) (Figure 51), with ankle plantar flexors showing a low correlation (0.432, P=0.2128), and ankle dorsiflexors a strong correlation (0.782, P=0.0076) (Table 46).

	0.891	0.741	0.89	0.648	0.432	0.782	0.537
CMTES	P=0.0005	P=0.0143	P=0.0006	P=0.0427	P=0.2128	P=0.0076	P=0.1092
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.412	0.145	0.538	0.128	0.081	0.231	0.185
WALK-12	P=0.2365	P=0.6890	P=0.1085	P=0.7251	P=0.8230	P=0.5201	P=0.6083
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	-0.232	-0.204	-0.176	-0.396	-0.507	-0.272	0.063
FPI-6	P=0.5183	P=0.5717	P=0.6267	P=0.2579	P=0.1346	P=0.4473	P=0.8638
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	Thigh FF	Knee	Knee	Calf FF	Plantar	Dorsiflexion	Foot FF
	r ingli FF	extension FF	flexion FF		flexion FF	FF	root rr
			1	0 -1			

Table 46: Correlation between baseline clinical measurements and baseline fat fraction.



Figure 51: Correlation between the Charcot Marie Tooth Examination Score (CMTES) and fat fraction (FF) (%) at the thigh and calf level.

7.3.6. Relationships Between Dynamometry and Gait Parameters

Plantar Flexors

Plantar flexion isokinetic dynamometry showed a significant high positive correlations with gait plantar flexion moments (maximum) in stance (0.778, P=0.0080) (Figure 52) (Appendix III, Table 78).



Figure 52: Correlation between plantar flexion moments (Nm/Kg)and plantar flexion isokinetic dynamometry (Nm).

Dorsiflexors

Dorsiflexion isometric dynamometry showed a significant moderate positive correlations with gait plantar flexion moments (maximum) in stance (0.666, P=0.0354) (Figure 53) (Appendix III, Table 79).



Figure 53: Correlation between dorsiflexors isometric dynamometry (Nm) and plantar flexion moments (Nm/Kg).

Knee Extensors

Knee extensors moment (maximum) in stance showed a significant very high positive correlation with isometric dynamometry (0.902, P=0.0004) (Figure 54) and a significant high positive correlation with isokinetic dynamometry (0.867, P=0.0012) (Figure 54). Knee power absorption (minimum) in stance showed a significant high positive correlation with isometric dynamometry (0.752, P=0.0121) (Figure 55) and with isokinetic dynamometry (0.754, P=0.0118) (Figure 55) (Appendix III, Table 80).



Figure 54: Correlation between knee extensors (EX) moment (Nm/Kg) and isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm).



Figure 55: Correlation between knee power absorption in stance (W/Kg) and knee extensors (EX) isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm).

Knee Flexors

Knee power absorption (minimum) in stance showed a significant high positive correlation with knee flexors isometric dynamometry (0.81, P=0.0045) (Figure 56) and with knee flexors isokinetic dynamometry (0.792, P=0.0064) (Figure 56). Knee extensors moment (maximum) in stance showed a significant high positive correlation with knee flexors isometric dynamometry (0.867, P=0.0012) (Figure 57) and knee flexors isokinetic dynamometry (0.856, P=0.0016) (Figure 57) (Appendix III, Table 81).



Figure 56: Correlation between knee power absorption (W/Kg) and knee flexors (FL) isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm).



Figure 57: Correlation between knee extensors moment (Nm/Kg) and knee flexors (FL) isometric (ISOM) and isokinetic (ISOK) dynamometry (Nm).

<u>Relationships Between Dynamometry and Gait Spatiotemporal Parameters</u> Isometric and isokinetic dynamometry correlated with speed, step length, and stride length significantly and positively. Proximally, correlations were high to moderate (0.765 - 0.647, P= 0.0433 - 0.0100) (appendix III, Table 83). Distally, isometric dorsiflexion correlated highly with speed (0.794, P=0.0061) (Figure 58), step length (0.818, P=0.0038) (Figure 58), and stride length (0.818, P=0.0038) (Figure 58). Speed correlated moderately with isometric plantar flexion (0.645, P=0.0439) and isokinetic plantar flexion (0.65, P=0.0417) (Appendix III, Table 83).



Figure 58: Correlation between isometric dorsiflexion (Nm) and gait spatiotemporal parameters.

7.3.7. Relationships Between Dynamometry Parameters and Clinical Measurements

The Charcot Marie Tooth Examination Score (CMTES) showed significant negative correlations with dynamometry parameters (Table 47) (Figure 59). Negative correlations were observed with isometric dorsiflexion (-0.781, P=0.0077), isokinetic plantar flexion (-0.817, P=0.0039), and isokinetic dorsiflexion (-0.825, P=0.0033). Correlation was moderately negative between walk-12 questionnaire and isokinetic plantar flexion (-0.634, P=0.0491).

			Ison	netric		0		1	Isok	inetic		
	Hip Flexio n	Hip Extens ion	Knee Flexio n	Knee Extens ion	Planta r flexion	Dorsifle xion	Hip Flexio n	Hip Extens ion	Knee Flexio n	Knee Extens ion	Planta r flexion	Dorsifle xion
0	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
TE	307	622	054	842	030	77	976	904	330	435	039	33
Ň	P=0.1	P=0.3	P=0.1	P=0.0	P=0.1	P=0.00	P=0.0	P=0.1	P=0.1	P=0.1	P=0.0	P=0.00
0	-0.511	-0.323	-0.542	-0.572	-0.545	-0.781	-0.553	-0.451	-0.509	-0.497	-0.817	-0.825
1	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
ALK	P=0.3 246	P=0.1 091	P=0.3 155	P=0.2 824	P=0.2 320	P=0.08 50	P=0.2 147	P=0.1 188	P=0.1 987	P=0.2 352	P=0.0 491	P=0.22 70
≤	-0.348	-0.537	-0.354	-0.377	-0.416	-0.571	-0.43	-0.525	-0.444	-0.413	-0.634	-0.42
	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
-6	235	948	687	058	049	84	923	713	372	536	710	36
FPI	P=0.7	P=0.7	P=0.8	P=0.6	P=0.5	P=0.49	P=0.4	P=0.4	P=0.5	P=0.4	P=0.1	P=0.34
	0.128	0.095	0.06	0.187	0.24	0.243	0.247	0.258	0.222	0.268	0.47	0.335

Table 47: Correlation between baseline clinical measurements and baseline dynamometry parameters.



Figure 59: Correlation between the Charcot Marie Tooth Examination Score (CMTES) and dynamometry parameters.

Relationship Between Quantitative and Qualitative Strength Measurements

Muscle strength measured using manual muscle testing showed significant high positive correlation with isometric dynamometry in hip extension (0.764, P=0.0101), knee flexion (0.742, P=0.0141), and ankle dorsiflexion (0.75, P=0.0125) (Table 48). Manual muscle testing showed significant high to moderate positive correlation with isokinetic dynamometry in hip extension (0.809, P=0.0046), knee flexion (0.809, P=0.0046), and knee extension (0.696, P=0.0253) (Table 49).



Table 48: Correlation between isometric dynamometry (X) and manual muscle testing (Y).



Table 49: Correlation between isokinetic dynamometry (X) and manual muscle testing (Y).

7.3.8. Relationships Between Gait Parameters and Clinical Measurements

The Charcot Marie Tooth Examination Score (CMTES) showed significant correlations with spatiotemporal gait parameters. Negative correlations were observed with speed (-0.763, P=0.0103), step length (-0.71, P=0.0214), and stride length (-0.71, P=0.0214) (Table 50) (Figure 60). Correlations were moderately positive with stride time (0.561, P=0.0916), and step time (0.561, P=0.0916) (Table 50) (Figure 60). The Walk-12 questionnaire showed significant negative correlations with spatiotemporal gait parameters; highly correlated with speed (-0.773, P=0.0088), step length (-0.772, P=0.0089), and stride length (-0.772, P=0.0089) (Table 50) (Figure 61).

			1	0 -1			
	Speed	Stride Length	Step Length	Stride Time	Step Time	Steps Per Minute	Strides Per Minute
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
FPI-6	P=0.7104	P=0.6830	P=0.6827	P=0.7126	P=0.7126	P=0.6505	P=0.6505
	0.135	0.148	0.148	0.134	0.134	-0.164	-0.164
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
WALK-12	P=0.0088	P=0.0089	P=0.0089	P=0.2536	P=0.2536	P=0.3403	P=0.3403
	-0.773	-0.772	-0.772	0.399	0.399	-0.337	-0.337
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
CMTES	P=0.0103	P=0.0214	P=0.0214	P=0.0916		P=0.1710	P=0.1710
	-0.763	-0.71	-0.71	0.561	0.561	-0.47	-0.47

Table 50: Correlation between baseline clinical measurements and baseline spatiotemporal parameters.



Figure 60: Correlation between the Charcot Marie Tooth Examination Score (CMTES) and gait spatiotemporal parameters.



Figure 61: Correlation between Walk-12 questionnaire and gait spatiotemporal parameters.

7.4. Discussion

This study is aimed to explore relations between measurements at baseline. To the best of my knowledge, this study is the first to explore relations between MRI, dynamometry, and gait parameters in DHMN. Primarily to ascertain the effect of the neuropathy on muscle performance and function. Since the calf fat fraction was the most responsive in this cohort (chapter 6), it was used to guide the correlation analysis.

7.4.1. Relationships Between Intramuscular Fat Fraction, Muscle Volume, Isokinetic and Isometric Muscle Strength, and Kinetics of Gait

At the distal and proximal levels, muscle fat fraction correlated negatively with the remaining muscle area and cross sectional area. This relation is expected as the increase in the intramuscular fat and progressive muscle atrophy are symptoms of peripheral motor neuropathy. And since the remaining muscle area is part of the total cross sectional area, they showed positive high correlations at all levels.

At the proximal level, relationships were observed between MRI, dynamometry, and kinetics parameters, with knee extensors and flexors. These muscle groups are relatively less affected than the distal muscles. With increased intramuscular fat fraction, there was a decrease in muscular capacity; decreased isometric and isokinetic strength, and a decrease in moment and power generation and absorption during gait. on the contrary, muscle capacities were increased in relation to higher muscle area which make MRI, dynamometry, and gait kinetics parameters candidate outcome measures in clinical trials that aim to target to muscle volume and performance.

At the distal level, where muscles are more affected by the neuropathy, as shown in chapter 5, relations were seen between MRI and isometric dynamometry. With greater muscle area in the plantar flexors and dorsiflexors, there was higher isometric strength, as there is more contractile tissue to generate force. On the other hand, with increased intramuscular fat fraction there was lower isometric strength as shown in dorsiflexion. We expect to see a deterioration in muscle strength as the neuropathy advances. Similar relationships were shown in people with hereditary peripheral neuropathies (O'Donnell et al., 2023, Morrow et al., 2016).

For the plantar flexors, relationships were observed between isokinetic strength and plantar flexors moments in gait. In stance phase, the greater the strength in plantar flexors, the more capable the muscle group to produce moments to control the ankle rocker movement in mid stance to generate a propulsive action in preswing (Whittle, 2007, Brockett and Chapman, 2016).

For the dorsiflexors, relationships were observed between fat fraction and power generation in swing phase. With more intramuscular fat fraction, there was a decrease in ankle power generation in swing phase. Power is generated in swing mostly from the dorsiflexors to assess in foot clearance (Whittle, 2007).

181

Therefore, identifying gait deviations and the underlying muscle involvement is informative for rehabilitation assessment and treatment. For instance, the effect of rehabilitation, including therapeutic exercises and orthotics, that focuses on the plantar flexors as a targeted group of muscles in DHMN can be assessed using plantar flexors moments in stance phase. And when the targeted muscle group is dorsiflexion, power is generation in swing will be more useful.

However, due to the high variability within a small sample, some significant correlations identified in this study may lack functional significance. For instance, the relationship between dorsiflexor strength and plantar flexor moments. Future research with a larger sample size and a control group is recommended to confirm these relationships.

7.4.2. Correlations with Clinical Measures

Clinical measurements showed correlations with MRI, dynamometry, and gait spatiotemporal parameters. With higher disease severity measured by the Charcot Marie Tooth Examination Score (CMTES), there was higher intramuscular fat fraction in the knee flexors and ankle dorsiflexors. Also, with higher severity and worse mobility measured by the CMTES and Walk-12 questionnaire, there was weaker ankle plantar flexor and dorsiflexor, slower speed, and shorter step length and stride length. Moreover, speed, step length, and stride length showed positive correlations with dorsiflexion isometric strength more than plantar flexion isometric strength. The neuropathy affected the speed and spatial parameters but not the temporal, suggesting that they walk with slower gait, not because of the timing , but because of shorter strides. This may be due to the knee flexor and ankle dorsiflexor weakness that contributes to problems with foot clearance in swing phase.

7.4.3. Quantitative and Qualitative Measurements

Qualitative and quantitative measurements used in this study showed a range of results. MRI intramuscular fat showed high correlations at the calf level between the Mercuri scale and fat fraction, but intramuscular water correlations were mostly negligible. This can be explained by the lack of sensitivity in the hyperintensity scale with limited numbers of scores, 3 levels for intensity and 3 levels for extent. Isometric and isokinetic dynamometry showed mostly low correlations with manual muscle testing. This could be explained by challenges with ankle dynamometry testing, where there is limited joint range of motion. Proximal variables at the knee and hip showed higher correlations where there are fewer joint restrictions. It is worth noting that for the dorsiflexors, very high to high correlations were observed between quantitative and qualitative intramuscular fat, intramuscular water, and strength measurements. The dorsiflexors may be a candidate for future clinical trials in developing outcome measures in peripheral neuropathy.

183

7.5. Summary

The current study explored the relationships between MRI, dynamometry, and 3D motion analysis measures in DHMN. The study findings are summarised in Table 51.

	Positive correlations	Negative correlation
	RMA vs. CSA	FF vs. RMA
DF	ISOM vs. RMA	
ГГ	SOM vs. CSA	
	ISOK vs. gait PF moments	
	RMA vs. CSA	FF vs. RMA
	ISOM vs. RMA	FF vs. CSA
DE	ISOM vs. gait PF moments	FF vs. ISOM
Dr	ISOM DF vs. speed	FF vs. ankle power generation
	ISOM DF vs. step length	during swing
	ISOM DF vs. stride length	
Calf	T_{2m} vs. FF	-
	RMA vs. CSA	FF vs. RMA
	CSA vs. ISOM	FF vs. CSA
Knoo	CSA vs. ISOK	
FV	CSA vs. knee extensor moment.	
LA	RMA vs. knee extensor moment.	
	ISOM & ISOK vs. Knee extensor moment.	
	ISOM & ISOK vs. Knee power absorption.	
	RMA vs. CSA	FF vs. RMA
Knoo	ISOM vs. RMA	FF vs. CSA
FIX	CSA vs. ISOM	FF vs. ISOM
I LA	RMA & CSA vs. Knee power absorption	
	CMTES vs. FF in thigh	CMTES vs. speed
CMTES	CMTES vs. FF in calf	CMTES vs. ISOM
		CMTES vs. ISOK
MMT	MMT vs. ISOM in hip extension	-
PF= Plant	ar Flexors, DF= Dorsiflexors, EX= Knee Exte	nsion, FLX= Knee Flexion, RMA=
Remainin	g Muscle Area, CSA= Cross Sectional Area,	ISOM= Isometric Dynamometry,
ISOK= Is	sokinetic Dynamometry, FF= Fat Fraction,	CMTES= Charcot Marie Tooth

Examination Score, MMT= Manual Muscle Testing.

Table 51:Summary of positive and negative correlation between study parameters.

Chapter 8: The Effect of Bilateral Carbon Fibre Ankle Foot Orthoses (AFOs) on the Kinetics and Kinematics of Gait of People with Distal Hereditary Motor Neuropathy (DHMN)

8.1. Introduction

Ankle foot orthoses (AFO) are medical devices usually prescribed for people presenting with foot drop, and commonly issued to people with neuropathy (Landfeldt et al., 2017, McCaughan et al., 2019). Studies showed low adherence in using AFO, as users are concerned that they highlight their disability(Bertini et al., 2024), but they have been shown to help with ankle control (Phillips et al., 2011), and compensate for distal weakness(Ramdharry et al., 2012a).

Different materials used for ankle foot orthoses (AFOs) include plastic, carbon fibre, and elastic fabric. Each material supports the ankle and foot in different ways and varies in size, weight, comfort, and acceptability (Landfeldt et al., 2017, McCaughan et al., 2019). Carbon fibre is particularly notable for its high satisfaction rate among users with neuromuscular diseases(Mnatsakanian et al., 2017). This material is thought to assist walking by allowing the orthoses to store energy and release it at pre-swing, which benefits patients with plantar flexion weakness(Zou et al., 2014, Bartonek et al., 2007). In a study by Dufek et al. (2014), custom-made carbon fibre AFOs were found to improve walking speed, step length, and frequency in patients with Charcot-Marie-Tooth disease. The carbon fibre AFOs also helped shift the maximum joint moments during loading from the hip to the ankle and knee joints, reducing the reliance on proximal muscles. This energy storage and release mechanism, averaging 9.6 J/kg, allows for better propulsion and compensates for ankle weakness (Dufek et al., 2014).

In this study we are aiming to explore the effect of carbon fibre AFO on gait kinetics and kinematics in our DHMN cohort using 3D motion analysis. The hypothesis is that the carbon fibre AFO material can store energy in stance phase and release it at toe off to compensate for ankle muscle weakness. Also, to explore any assist of foot clearance in swing phase.

8.2. Methods

To explore the effect of bilateral carbon fibre ankle foot orthoses on the kinetics and kinematics of gait of people with DHMN, 10 DHMN participants were recruited to undergo isokinetic and isometric dynamometry of the lower limbs and 3D motion analysis to capture spatiotemporal, kinetic, and kinematic data of walking gait, as described in chapter 4. There were two conditions: wearing shoes only, and wearing bilateral Matrix Max carbon fibre AFOs (Trulife, UK) (Figure 62). "Shoes only" was the control condition. The order of wearing or not wearing the AFOs was randomised to account for learning and/or fatigue effects. Small, medium, and large size AFOs were used to ensure fit. In cases where the participant's footwear was unsuitable, lace up shoes were provided to wear.



Figure 62: Matrix Max Carbon fibre Ankle foot Orthosis (Trulife, UK)

8.2.1. Data analysis

To explore the effect of bilateral carbon fibre ankle foot orthoses (AFO) on the kinetics and kinematics of gait of people with DHMN, gait variables were compared with and without AFOs using a paired t-test, if all data sets were normally distributed, or a paired Wilcoxson- Signed rank test, if at least one of the data sets were not normally distributed. A modified Bonferroni procedure was used to account for multiple comparisons. However, due to the large number of comparisons, it was anticipated that with small sample, significant differences may be few. Therefore, it was decided to also explore trends in the data using p<0.05 as a guide.

To measure the effect size, Hedge's G was calculated as the difference between the two means divided by the pooled standard deviation, with a correction for small sample bias. The effect size was categorised by magnitude according to Cohen's suggestion: < 0.2 minimal effect; 0.2-0.5 small effect; 0.5-0.8 moderate effect; >0.8 large effect.

8.3. Results

8.3.1. Subjects and Clinical Assessment

Ten DHMN participants were recruited. Participants were 6 male and 4 female ranging in age from 41 to 75. Clinical assessment showed distal muscle weakness and limitation in plantarflexion and dorsiflexion active and passive range of motion. A summary of the subjects' demographics and clinical assessment is presented in Table 52 and Table 53.

Demographics and clinical assessment						
Numbers	10					
Gender (M/F)	(6/4)					
Age; mean years; range	57.5 (41/75)					
Genetic diagnosis (HSPB1/unknown)	(6/4)					
CMTES; mean (SD)	5.2(2.6)					
Foot Posture Index-6 (Pronated/Normal/Supinated)	(1/8/1)					
Fall Frequency (weekly/Monthly/Yearly)	(1/1/2)					
Walk-12; mean (SD)	34(10.72)					
Range of Motion						
Dorsiflexion; count (Limited/Normal)	(10/0)					
Plantarflexion; count (Limited/Normal)	(9/1)					
M= Male, F= Female, HSPB1= Heat-shock 27-KD Protein 1, SD= Standard Deviation, CMTES= Charcot-Marie-Tooth Disease Examination Score, *= significant after modified Bonferroni						

correction. Table 52: Summary of demographics and clinical assessment of DHMN participants.

Manual Muscle Testing	
Hip Flexion; mode; (range)	5 (5-3)
Hip Extension; mode; (range)	5 (5-3)
Knee Flexion; mode; (range)	5 (5-3)
Knee Extension; mode; (range)	5 (5-4)
Dorsiflexion; mode; (range)	4+ (5-0)
Plantarflexion; mode; (range)	4+ (4+-0)
Isometric Dynamometry	
Hip Extension 45°; mean (SD), Nm	144.6(67.4)
Hip Flexion 45°; mean (SD), Nm	114.25(44.5)
Knee Extension 45°; mean (SD), Nm	117.55(53.9)
Knee Extension 90°; mean (SD), Nm	119.45(52.0)

Knee Flexion 45°; mean (SD), Nm	65.5(34.3)
Knee Flexion 90°; mean (SD), Nm	44(23.2)
Ankle Plantar flexion 10°; mean (SD), Nm	30(17.2)
Ankle Dorsiflexion 10°; mean (SD), Nm	15.35(17.4)
Ankle Dorsiflexion 30°; mean (SD), Nm	20.65(17)
Isokinetic Dynamometry	
Hip Extension 60°/60s; mean (SD), Nm	122.8(68.1)
Hip Flexion 60°/60s; mean (SD), Nm	110.35(48.4)
Knee Extension 60°/60s; mean (SD), Nm	91.35(48.9)
Knee Flexion 60°/60s; mean (SD), Nm	46.4(24.9)
Knee Extension 120°/120s; mean (SD), Nm	66.05(43.3)
Knee Flexion 120°/120s; mean (SD), Nm	35.85(23.6)
Ankle Plantar flexion 60°/60°; mean (SD), Nm	18.5(16.6)
Ankle Dorsiflexion 60°/60°; mean (SD), Nm	16.2(7)
SD= Standard Deviation, Nm= Newton meter	

Table 53: Strength assessment of DHMN participants.

8.3.2. The Effect of Carbon Fibre Ankle Foot Orthoses

There were significant differences when walking with AFOs. Analysis showed a decrease in ankle plantar flexion angle (P=0.002) (Table 55) and an increase in knee flexion moment (P=0.0384) (Table 56). However, when using P<0.05 significance level, cut off trends were observed in proximal and distal kinematics and spatiotemporal parameters (Table 54,Table 55,Table 56) (Figure 63,Figure 64,Figure 65). Walking with AFOs showed increase in hip extension angle (P=0.0273), decrease in knee flexion angle (P=0.0283), decrease in ankle dorsiflexion angle (P=0.0115), increase in speed (P=0.0486), increase in stride length (P=0.0127) and step length (P=0.0128), decrease in double support time (P=0.0173) and percent opposite toe off (P=0.0255) (Table 54,Table 55,Table 56) (Figure 63,Figure 64,Figure 65). Hedge's G showed the effect size to be the

largest in ankle angles with (-0.94) in dorsiflexion and (1.352) in plantar flexion

Gait Spatiotemporal	NO AFO mean (SD)	AFO mean (SD)	Diff mean (SD)	Р	Hedge's G	95% CI
Speed; m/s	1.07(0.23)	1.16(0.23)	0.1(0.1)	0.0486†	0.38	-0.53 to 1.29
Stride Length; m	1.25(0.24)	1.35(0.23)	0.1(0.1)	0.0127†	0.40	-0.51 to 1.31
Stride Time; s	1.18(0.06)	1.18(0.10)	0.0(0.1)	0.8326	-0.05	-0.95 to 0.85
Strides Per Minute	51.05(2.52)	51.40(4.08)	0.4(2.7)	0.6852	0.10	-0.80 to 1.0
Step Length; m	0.62(0.12)	0.67(0.12)	0.0(0.0)	0.0128†	0.40	-0.51 to 1.31
Step Time; s	0.59(0.03)	0.59(0.05)	0.0(0.0)	0.8244	-0.05	-0.95 to 0.85
Steps Per Minute	102.09(5.04)	102.80(8.16)	0.7(5.3)	0.6852	0.10	-0.80 to 1.0
Percent Stance	0.63(0.02)	0.63(0.02)	0.0(0.0)	0.184	-0.39	-1.3 to 0.52
Single Support Time; s	0.60(0.04)	0.61(0.06)	0.0(0.0)	0.3553	0.25	-0.65 to 1.16
Double Support Time; s	0.15(0.03)	0.13(0.04)	0.0(0.0)	0.0173†	-0.64	-1.57 to 0.28
Percent Opposite Toe Off	12.83(2.40)	10.67(3.59)	-2.2(2.6)	0.0255†	-0.68	-1.61 to 0.25
Percent Opposite Foot Contact	49.85(0.77)	49.52(0.97)	-0.3(1.5)	0.4989	-0.36	-1.27 to 0.55

(Table 54, Table 55, Table 56).

Diff= difference, SD= Standard Deviation, m= meters, s= seconds, \dagger = significant with P<0.05, Hedge's G= < 0.2 minimal effect, 0.2-0.5 small effect, 0.5-0.8 moderate effect, >0.8 large effect.

Table 54: Spatiotemporal data, difference, and effect size of walking with and without AFOs in DHMN.

Gait Kinematics	NO AFO mean (SD)	AFO mean (SD)	Diff mean (SD)	Р	Hedge's G	95% CI
Pelvis X Max; °	4.83(1.58)	5.26(2.63)	0.4(1.6)	0.40	0.19	-0.7 to 1.09
Pelvis X Min; °	-5.16(1.81)	-4.72(1.53)	0.4(1.0)	0.21	0.25	-0.65 to 1.16
Pelvis Y Max; °	15.22(5.06)	14.94(3.42)	-0.3(3.7)	0.82	-0.06	-0.96 to 0.84
Pelvis Y Min; °	10.36(4.16)	10.16(2.96)	-0.2(3.0)	0.83	-0.06	-0.95 to 0.85
Pelvis Z Max; °	6.40(2.00)	7.52(2.81)	1.1(2.2)	0.15	0.44	-0.47 to 1.35
Pelvis Z Min; °	-6.32(2.13)	-5.79(4.12)	0.5(4.0)	0.63	0.15	-0.75 to 1.06
Hip Max; °	47.27(5.61)	44.69(5.62)	-2.6(6.0)	0.21	-0.44	-1.35 to 0.47
Hip Min; °	1.97(2.54)	-0.87(4.82)	-2.8(4.3)	0.0273†	-0.71	-1.64 to 0.22
Knee Max; °	75.09(2.82)	72.82(3.52)	-2.3(2.8)	0.0283†	-0.68	-1.61 to 0.25
Knee Min; °	6.77(4.44)	1.25(17.77)	-5.5(16.0)	0.49	-0.41	-1.32 to 0.5
Ankle Max; °	26.18(4.54)	21.73(4.54)	-4.5(4.5)	0.0115†	-0.94	-1.9 to 0.012
Ankle Min; °	-9.82(4.67)	-3.80(3.80)	6.0(5.2)	0.002*	1.35	0.35 to 2.36

Diff= difference, SD= Standard Deviation, $\dagger =$ significant with P<0.05, $\ast =$ significant with modified Bonferroni correction= Hedge's G= < 0.2 minimal effect, 0.2-0.5 small effect, 0.5-0.8 moderate effect, >0.8 large effect, Pelvic z Maximum= Rotation forwards, Pelvic z Minimum= Rotation backwords, Pelvic x Maximum, lateral raise, Pelvic x Minimum= lateral drop, Pelvic y Maximum= Anterior tilt, Pelvic y Minimum= Posterior tilt, Hip y Maximum= Flexion, Hip y Minimum= Extension, Knee y Maximum= Flexion, Knee y Minimum= Plantarflexion.

Table 55: Gait kinematic data, difference, and effect size of walking with and without AFOs in DHMN.

Gait Kinetics	NO AFO mean (SD)	AFO mean (SD)	Diff mean (SD)	Р	Hedge's G	95% CI
Hip Moment Y Max; Nm/Kg	1.66(0.60)	1.77(0.58)	0.1(0.4)	0.38	0.19	-0.71 to 1.090
Hip Moment Y Min; Nm/Kg	-0.53(0.16)	-0.65(0.43)	-0.1(0.4)	0.92	-0.35	-1.25 to 0.560
Knee Moment Y Max; Nm/Kg	0.82(0.35)	0.82(0.33)	0.0(0.1)	0.85	-0.02	-0.92 to 0.881
Knee Moment Y Min; Nm/Kg	-0.75(0.35)	-0.87(0.34)	-0.1(0.2)	0.0384*	-0.35	-1.25 to 0.560
Ankle Moment Y Max; Nm/Kg	1.04(0.27)	1.10(0.24)	0.1(0.1)	0.15	0.25	-0.65 to 1.158
Ankle Moment Y Min; Nm/Kg	-0.07(0.07)	-0.07(0.04)	0.0(0.1)	0.38	0.00	-0.90 to 0.897
Hip Power Max During Swing; W/kg	1.15(0.33)	1.06(0.32)	-0.1(0.3)	0.43	-0.26	-1.17 to 0.64
Hip Power Min During Stance; W/kg	-1.18(1.00)	-1.87(2.36)	-0.7(2.4)	0.63	-0.37	-1.27 to 0.54
Hip Power Max During Stance; W/kg	2.06(0.72)	2.42(1.77)	0.4(1.4)	0.85	0.26	-0.64 to 1.16
Knee Power Max During Swing; W/kg	0.43(0.22)	0.46(0.27)	0.0(0.2)	0.56	0.12	-0.78 to 1.02
Knee Power Min During Stance; W/kg	-1.85(1.00)	-1.96(1.06)	-0.1(0.9)	0.70	-0.10	-1.002 to 0.8
Knee Power Max During Stance; W/kg	2.24(1.91)	2.96(3.00)	0.7(1.9)	0.38	0.27	-0.63 to 1.18
Ankle Power Max During Swing; W/kg	0.04(0.02)	0.07(0.08)	0.0(0.1)	0.18	0.49	-0.42 to 1.41
Ankle Power Min During Stance; W/kg	-1.20(0.33)	-1.10(0.39)	0.1(0.3)	0.31	0.29	-0.62 to 1.19
Ankle Power Max During Stance; W/kg	1.09(0.40)	0.84(0.45)	-0.3(0.5)	0.11	-0.57	-1.49 to 0.35

Diff= difference, SD= Standard Deviation, Nm/Kg= Newton Meter per Kilogram, W/kg= Watts Per Kilogram, *= significant with modified Bonferroni correction= Hedge's G= < 0.2 minimal effect, 0.2-0.5 small effect, 0.5-0.8 moderate effect, >0.8 large effect= Hip Moments Maximum= Extension, Hip Moments Minimum= Flexion, Knee Moments Maximum= Extension, Knee Moments Minimum= Flexion, Ankle Moments Maximum= Plantarflexion, Ankle Moments Minimum= absorption in stance phase, Hip Power Maximum= generation in stance phase, Knee Power Maximum= absorption in stance phase, Ankle Power Maximum= generation in swing phase.

Table 56: Gait kinetics data, difference, and effect size of walking with and without AFOs in DHMN.


Figure 63:Grand average angular displacement in the sagittal plane over one gait cycle for the ankle, knee, and hip (average of left and right sides). Comparison of people with DHMN walking with and without AFO.

OTO, Opposite side tore off; OIC, Opposite side initial contact; TO, Toe off; A, improve foot clearance; $B_{,;}$ C, improve ankle support, , N=10.



Figure 64: Grand average joint moments in the sagittal plane over one gait cycle for the ankle, knee and hip (average of left and right sides). Comparison of people with DHMN walking with and without AFO.

OTO, Opposite side tore off; OIC, Opposite side initial contact; TO, Toe off, N=10.



Figure 65:Grand average joint powers in the sagittal plane over one gait cycle for the ankle, knee and hip (average of left and right sides). Comparison of people with DHMN walking with and without AFO.

OTO, *Opposite side tore off; OIC*, *Opposite side initial contact; TO*, *Toe off; A*, *absorption; B*, *generation*, *N*=10.

8.4. Discussion

This current study is aimed to explore the effect of carbon fibre ankle foot orthoses on gait kinetics and kinematics in people with DHMN. This DHMN cohort presented with plantar flexor weakness that affected the stability of the ankle in stance phase (plantar flexion failure), and dorsiflexor weakness that affected foot clearance in swing phase (foot drop), resulting in slower walking gait (chapter 5).

There were promising trends towards faster gait speed, longer strides, and shorter double support, due to the biomechanical assist of the AFOs. Where there is plantar flexion weakness, the muscle group is not capable of producing a high enough isometric force to control the ankle joint angle in stance phase, therefore, the ankle falls in passive dorsiflexion as the centre of gravity progresses forward (Brockett and Chapman, 2016). The AFO compensated for the weakness by providing enough ankle support and stability, to hold the ankle in a neutral position and allow the contralateral leg to start the swing phase sooner. This finding is consistent with studies by Dufek et al. (2014), which demonstrated that custom-made carbon fibre AFOs enabled faster walking speeds and improved step length in people with CMT (Dufek et al., 2014). However, the AFO rigidity limits the range of plantar flexion in mid to late stance. This appears to inhibit plantar flexor power generation and may transfer the motion to the knee joint, so explaining the increased knee flexor moment. The difference in ankle power generation was not significant but showed a moderate effect size so this hypothesis would need confirmation with larger sample.

Dorsiflexion weakness explains the reduction in ankle dorsiflexion angle while the leg is swinging. Therefore, knee and hip flexion is increased to clear the foot from the ground. The AFO compensated for the weakness by holding the ankle in a neutral position, which allows foot clearance and longer strides without the need to shorten the leg by excessively flexing the knee and the hip. This is in line with findings from previous research which reported that AFOs improved ankle dorsiflexion angles and reduced compensatory hip flexion especially in those with foot drop in children (Õunpuu et al., 2021), and adults (Ramdharry et al., 2012a) with CMT.

Overall, there was an improvement in walking gait with AFOs. However, there is a possibility that with restricted ankle movement, planer flexor activity might be inhibited (Figure 65). This could have long term implication for disuse weakness. Further studies are recommended with electromyography to explore changes in muscle activity. Longitudinal follow up will aid better understanding of the long term effects of wearing AFOs.

8.5. Summary

The current study explored the effect of carbon fibre AFOs on gait kinetic, kinematic, and spatiotemporal data in DHMN. The study findings are summarised in Table 57.

The Normal Gait Cycle											
% of cycle	0%	0-10%	10-30	% 3	80-50%	50-60%	60-70%	70-85%	85-100%		
Phase			Stance	phase		•		Swing phase			
Event	Initial contact	Foot Opposit (OT	flat, e toe off FO)	Heel off, Opposite initial contact		Toe off	Tibia vertical, Feet adjacent	Initial contact (next cycle)			
Period	Loading response	Mid-s	Mid-stance		ninal nce	Pre- swing	Initial swing	Mid-swing	Terminal swing		
Support	Double supportSingle leg supportDouble support					Double support	Single leg support				
DHMN Pattern in Gait and Muscle Involvement (Chapter 5)											
Muscle Involvement	Plantar f	Plantar flexion (PF) weakness affected the stability of the ankle joint in stance phase						Dorsiflexion (DF) weakness affected foot clearance in swing phase			
Spatio Temporal Variables	Increased double support time due to delayed OTO Increased stance time due to increased step and stride length					time due to and stride	Increased single support time				
v ar labits				Slov	v gait due	to the event	s timing				
Kinematics Variables	Increase	ed ankle pas	ssive dors	iflexion	Decre plan	eased ankle tar flexion	Decreased ankle dorsiflexion and increased hip flexion				
Kinetics Variables	De	ecreased pla	ntar flexi	on mome	nt and pov	wer	Decreased ankle power generation				
Gait Pattern		Pl	antar flex	ion failur	e			Foot drop			
			E	Effect of A	FO on D	HMN Gait					
Kinematics Variables	Impro	oved ankle c	ontrol an dorsifl	d decrease exion	ed ankle p	assive	Increased ank kn	e dorsiflexion a e and hip flexio	nd decreased n		
Kinetics Variables			Increased	knee flex	ion mome	ents to pick u	p the foot in pre	e-swing			
Spatio Temporal Variables	Dec	reased doub	le suppor	t time due	e to early	ΟΤΟ	faster gait du	e to longer stride	es and steps		
Comment		Imj	proved an	kle stabili	ty		Impr	oved foot cleara	nce		

Table 57: Summary of the effect of AFO on DHMN gait.

Chapter 9: The Effect of Resistance Training on Muscle Structure, Function, and Gait Patterns in Distal Hereditary Motor Neuropathy (DHMN)

9.1. Introduction

Therapeutic exercise is one of the few available treatments for peripheral neuropathies, including distal hereditary motor neuropathy (DHMN). Studies in neurological diseases suggest that exercise can improve balance, reduce fatigue, enhance physical performance, and positively affect quality of life (Roberts-Clarke et al., 2016b, Vita et al., 2016, Ramdharry et al., 2012c, Burns et al., 2017). However, there is a lack of evidence-based exercise protocols tailored to neuropathies, leading to reliance on therapist experience for program design. Existing research on Charcot-Marie-Tooth disease (CMT) has shown the potential benefits of strengthening exercises in improving muscle strength and function. However, the long-term safety and efficacy of such interventions remain underexplored (Sman et al., 2015).

This exercise trial is aimed to evaluate the effectiveness of progressive resistance training on muscle structure, using MRI, and function, using dynamometry and 3D motion analysis. It is also aimed to evaluate the safety of progressive resistance training by evaluating changes in MRI intramuscular water as a sign of active denervation.

9.2. Methods

Ten DHMN participants recruited for the natural history study were randomly assigned to "exercise group" and "no exercise group". Participants were eligible if they had lower limb muscle strength over grade 4 of the MRC scale. Both groups underwent an MRI scan, isokinetic and isometric dynamometry of the lower limb, and 3D motion analysis to capture kinetic and kinematic data of walking gait as described in chapter 4. The same measurements were repeated after 6 months and 12 months to explore the effect of resistance training on the condition.

The 6 months strength training program included resistance training of distal and proximal lower limb muscles, using graded ankle weights, performed 3-4 times per week. Resistance training of the muscles was initially prescribed at 25% of one repetition maximum power, measured using a handheld dynamometer (Citec CT 3001, CIT Technics BV, Groningen, The Netherland), at two sets of 8-12 repetitions. This was increased to 30% and 40% when subjects were easily completing 2 sets of 12 repetitions for a week (ACSM, 2021). Training session included 5 minutes warm up and cool down using low resistance exercises.

Adherence to the exercise programme was monitored throughout the study period by supplying subjects with exercise diaries. Subjects in the treatment group were planned to receive monthly visits from the PhD student to ensure safety of the program, re-assess strength and progress training. However, due to the COVID-19 pandemic physical contact restrictions, subjects were contacted via phone or video call on a weekly basis to monitor participation, pain levels, and any difficulties with the training. Exercise induced pain management involved monitoring pain levels to prevent musculoskeletal injury and ensure appropriate exercise intensity. Participants use the Visual Analogue Scale (VAS) to report pain. If the reported pain level was above 4, modifications were implemented to reduce pain while maintaining training. Pain levels above 7 during or post-exercise may indicate injury and lead to stopping the exercise for assessment and potential modification of the program. Delayed onset muscle soreness (DOMS) was anticipated as a normal early response and expected to resolve within 48 hours. Persistent or new pain required a full assessment and would lead to participant withdrawal from the trial if significant injury was found (appendix IV). Exercises instructions with photos and videos were provided for participants using Physiotools platform (appendix V).

9.2.1. Data Analysis

This DHMN cohort showed predominant distal weakness more than proximal, and in plantar flexors more than dorsiflexors (chapter 5). The natural history group (chapter 6) showed longitudinal deterioration in the posterior compartment of the lower leg more than other muscle groups. Therefore, MRI parameters at the calf posterior compartment were used primarily to explore the effectiveness and safety of exercises. The analysis plan was for paired statistical tests to compare before and after exercises, and unpaired statistical tests to compare between "exercise group" and "no exercise group". Effect sizes were to be calculated using the Hedge's G statistic due to the small sample size.

The level of adherence was measured according to the number of sessions completed and recorded in the exercise diary.

9.3. Results

9.3.1. Adherence to Exercises

The exercise group included 5 DHMN participants. The overall group adherence to the exercises program for 6 months was low. Two participants showed high adherence (case 1: 87.5%, case 2: 97.2%), one showed low adherence (case 3: 12.5%), and two showed no adherence (case 4 and case 5: 0%). It is important to note that case 1 followed the exercise program in the first 6 months, and case 2 followed the exercise program in the second 6 months of the study period. During the exercises period, pain was reported by 2 participants: case 2 reported 12 episodes with pain levels ranging from 2 to 7 on visual analogue scale. Case 3 reported 4 episodes with pain levels ranging from 5 to 8 on visual analogue scale. Factors that influenced adherence to exercise as reported by the "exercise group" participants are shown in Table 58. Those with high adherence were happy about the program and did not recommend any changes. Participant with low adherence recommended designing a less time consuming program. Those with no adherence were hoping for more interesting program to encourage engagement with the exercises.

203

	-Career requires physical fitness.							
High	-Exercise level is minimal and able to fit into a daily exercise routine.							
Adherence	-Medical background: aware of the importance of exercising in general.							
	-Have a dedicated room for exercising at home, equipped properly.							
	-knee Pain after a fall.							
Low	-Long COVID-19 recovery.							
Adherence	-Tiredness after long working days.							
	- Travel with family.							
	-Traveling abroad for long time.							
	-Couldn't fit exercise into daily routine.							
No	-Distracted with family situation.							
Adherence	-Preferred to wait for study results before committing to something							
	(exercises) that might not help.							

Table 58: Factors that influenced adherence to exercise.

9.3.2. Study Participants and Clinical Assessment

Data collected at each measurement point and the difference between measurements will be presented descriptively by group, exercise (n=5), and no exercises (n=5). However, since only 2 participants in the exercise group showed high level of adherence, this was deemed too small for a group analysis so they will be presented descriptively as individual cases. Differences between measurements points for each case has been presented in comparison to the mean change in a group of 8 DHMN participants, that includes the original "No Exercise" group, plus the three cases with no or low adherence (chapter 6). A summary of the subjects' demographics and clinical assessment at baseline is presented in Table 59 and Table 60.

Demographics and clinical assessment	Exercise	No Exercise
Numbers	5	5
Gender (M/F)	(3/2)	(2/3)
Age; mean years; range	55 (44/62)	60 (44/75)
Genetic diagnosis (HSPB1/unknown)	(3/2)	(3/2)
CMTES; mean (SD)	5.2 (5.1)	7.2 (1.9)
Foot Posture Index-6 (Pronated/Normal/Supinated)	(1/3/1)	(0/4/1)
Fall Frequency (weekly/Monthly/Yearly)	(1/1/0)	(0/0/2)
Walk-12; mean (SD)	33(13)	39(14)
Range of Motion	Exercise	No Exercise
Dorsiflexion (Limited/Normal)	(5/0)	(5/0)
Plantarflexion (Limited/Normal)	(5/0)	(5/0)
Manual Muscle Testing	Exercise	No Exercise
Hip Flexion; mode; range	5(4-5)	5(3-5)
Hip Extension; mode; range	5(3-5)	5(3-5)
Knee Flexion; mode; range	5(3-5)	5(45)
Knee Extension; mode; range	5(4-5)	5(4-5)
Dorsiflexion; mode; range	4(0-4+)	1(1-4+)
Plantarflexion; mode; range	4+(1-4+)	4+(1-4+)
M= Male, F= Female, HSPB1= Heat-shock 27-KD Pr Charcot-Marie-Tooth Disease Examination Score.	rotein 1, SD= Stand	lard Deviation, CMTES=

Table 59: Summary of the subjects' demographics and clinical assessment at baseline by group, exercises (n=5) and no exercise (n=5).

Demographics and clinical assessment	CASE 1	CASE 2	No Exercise
Numbers	1	1	8
Gender (M/F)	М	М	(3/5)
Age; mean years; range	58	61	57 (42/75)
Genetic diagnosis (HSPB1/unknown)	unknown	unknown	(6/2)
CMTES; mean (SD)	2	2	7.25(3.23)
Foot Posture Index-6 (Pronated/Normal/Supinated)	Supinated	Normal	(1/6/1)
Fall Frequency (weekly/Monthly/Yearly)	0	0	(1/1/2)
Walk-12; mean (SD)	31	30	36(12)
Range of Motion	CASE 1	CASE 2	No Exercise
Dorsiflexion; count (Limited/Normal)	Limited	Limited	(8/0)
Plantarflexion; count (Limited/Normal)	Limited	Limited	(8/0)
Manual Muscle Testing	CASE 1	CASE 2	No Exercise
Hip Flexion; mode; (range)	5	5	5(3-5)
Hip Extension; mode; (range)	5	5	5(3-5)
Knee Flexion; mode; (range)	5	5	5(4-5)
Knee Extension; mode; (range)	5	5	5(3-5)
Dorsiflexion; mode; (range)	4	4-	4+(1-4+)
Plantarflexion; mode; (range)	4	4-	4+(0-4+)
M= Male, F= Female, HSPB1= Heat-shock 27-KD Pro Charcot-Marie-Tooth Disease Examination Score.	otein 1, SD= St	tandard Deviatio	on, CMTES=

Table 60: Summary of the subjects' demographics and clinical assessment at baseline for

exercise case 1, case 2, and no exercises group (n=8).

9.3.3. Effect of Exercises on Muscle Structure (MRI Parameters)

Comparison Between Exercises Group (N=5) and No Exercises Group (N=5):

Fat Fraction (FF): The rate of change in FF varied across different muscle groups for both exercise and no-exercise groups. In the total thigh, the exercise group showed a 7.7% increase from baseline to 6 months, followed by a 2.1% increase from 6 to 12 months, resulting in a 10% total increase over 12 months. In contrast, the no-exercise group showed a 3.8% increase at 6 months and an additional 2.9% at 12 months, resulting in a 6.1% increase. At the calf level, FF remained stable in the exercise group, with no change from baseline to 6 months and a 1.6% increase by 12 months. In contrast, the no-exercise group showed a 3.8% increase group showed a 3.6% increase from baseline to 6 months and a 1.6% increase by 12 months. In contrast, the no-exercise group showed a significant 9.3% increase over 12 months, with a 3.6% increase by 6 months (Table 61).

Cross Sectional Area (CSA): At the thigh level, the exercise group generally demonstrated muscle preservation or growth, whereas the no-exercise group showed declines or minimal gains. For total thigh area, the exercise group showed a 0.7% increase at 6 months, followed by a 1.8% increase at 12 months, resulting in an increase of 2.6% over 12 months. In contrast, the no-exercise group showed a 1% decrease at 6 months and a 0.3% increase from 6 to 12 months, resulting to an overall reduction of 0.7%. At the calf level, the exercise group showed a 3.1% decrease in total calf area over 12 months, driven by declines in both plantarflexion and dorsiflexion. Plantarflexion decreased by 4.6% and dorsiflexion by 2.8%. The no-exercise group also showed declines in calf area,

with a 1.5% decrease over 12 months, but the reduction in plantarflexion was less severe (2.2%), and dorsiflexion decreased by just 0.8% (Table 62).

Remaining Muscle Area (RMA): For total thigh muscle area, the exercise group showed a 0.04% increase from baseline to 6 months, followed by a more 1.5% increase from 6 to 12 months, resulting in a gain of 1.6% over 12 months. In contrast, the no-exercise group showed a marked decline, with a 1.6% reduction at 6 months and an additional 0.3% reduction by 12 months, resulting in a total loss of 1.9%. At the calf level, the exercise group showed a 2.2% reduction in muscle area over 12 months, including a 0.7% decrease in plantarflexion and a 4.3% reduction in dorsiflexion. In comparison, the no-exercise group showed a larger 7.7% loss in total calf muscle, with a 6.3% decrease in plantarflexion and a 3.7% decline in dorsiflexion. (Table 63).

Muscle Oedema (T_{2m}) : In the total calf, the exercise group showed a 2.3% reduction in oedema from baseline to 6 months, followed by a 1.6% increase from 6 to 12 months, resulting in a small overall reduction of 0.8% after 12 months. Conversely, the no-exercise group showed a 4.7% reduction in oedema from baseline to 6 months, and a significant 9.1% increase at 12 months, resulting in a 4% increase over 12 months. Similarly, in dorsiflexion, the exercise group maintained relative stability, with a minor 0.3% increase over 12 months. In contrast, the no-exercise group showed a 3.6% increase in oedema after 12 months, following an initial 2.1% reduction at 6 months (Table 64).

FAT (%)	BAS	ELINE	6 MC	ONTHS	12 MO	NTHS
FRACTION Mean (SD)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)
Total Thigh	13 (18.2)	13.2 (6)	14.2 (20.6)	13.6 (7.0)	14.5 (20.5)	14.0 (8.0)
Knee extension	11 (17.0)	12.6 (11)	12.6 (17.9)	13.0 (12.1)	12.6 (18.0)	13.0 (12.9)
Knee flexion	15 (18.4)	13.6 (5)	17.6 (22.6)	14.1 (6.6)	17.6 (22.6)	14.2 (6.1)
Total Calf	31 (19.3)	37.8 (22)	31.9 (20.6)	41.4 (22.4)	32.4 (20.3)	42.2 (21.9)
Plantar flexion	42 (20.8)	41.5 (24)	41.8 (21.5)	46.3 (24.5)	42.5 (22.1)	47.0 (23.4)
Dorsiflexion	14 (22.4)	24.4 (16)	15.8 (22.8)	26.5 (17.4)	16.0 (22.6)	27.6 (18.0)
FOOT	45 (15.8)	54.1 (7)	43.8 (18.3)	56.0 (6.9)	45.3 (18.1)	57.3 (7.8)
FAT (%)	BASELINE	E-6 MONTHS	6MONTHS	-12MONTHS	BASELINE-	2MONTHS
FRACTION Difference	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)
Total Thigh	1.0(2)	0.5 (1.1)	0.3 (1.2)	0.4 (1.1)	1.3 (2.4)	0.8 (1.3)
Knee extension	0.8(1)	0.4 (1.0)	0.0 (1.3)	0.1 (0.9)	0.7 (1.3)	0.5 (1.3)
Knee flexion	2.2(4)	0.6 (1.9)	0.1 (1.1)	0.1 (1.6)	2.2 (4.3)	0.7 (1.0)
Total Calf	0.0(2)	3.6 (1.2)	0.4 (1.5)	0.8 (1.3)	0.5 (2.2)	4.3 (1.3)
Plantar flexion	-0.6(3)	4.7 (1.8)	0.7 (3.3)	0.7 (1.9)	0.0 (3.6)	5.5 (2.1)
Dorsiflexion	1.1(1)	2.1 (2.0)	0.1 (0.9)	1.1 (1.4)	1.2 (1.0)	3.2 (2.0)
FOOT	-1.9(4)	2.0 (2.6)	1.5 (2.6)	1.3 (3.5)	-0.5 (4.9)	3.3 (1.7)
Data presente	d as: Mean (standers devia	tion).			

Table 61: Longitudinal fat fraction and differences between measurement points in exercise group (n=5) and no exercise group (n=5).

Cross	BASE	LINE	6 MO	NTHS	12 MO	NTHS
Sectional Area (mm ²)	Exercise	No Exercise	Exercise	No Exercise	Exercise	No Exercise
Total Thigh	11089 (5084)	10385 (1886)	11167 (4900)	10276 (1932)	11373 (4977)	10309 (1923)
Knee						
extension	5525 (2258)	4913 (859)	5419 (2182)	4882 (948)	5567 (2276)	4823 (857)
Knee flexion	3788 (2040)	4018 (1036)	4001 (2083)	4009 (969)	3954 (1942)	4038 (998)
Total Calf	4958 (1385)	5231 (1754)	4926 (1738)	5121 (1784)	4769 (1590)	5152 (1919)
Plantar						
flexion	3032 (938)	3566 (1346)	2990 (1199)	3440 (1355)	2894 (1138)	3504 (1435)
Dorsiflexion	1052 (431)	854 (325)	1038 (419)	843 (351)	1023 (409)	849 (361)
FOOT	476 (144)	502 (129)	460 (120)	458 (116)	458 (121)	448 (100)
Cross	BASELINE	-6 MONTHS	6MONTHS-1	12MONTHS	BASELINE-1	2MONTHS
Sectional Area (mm ²) Difference	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)
Total Thigh	78(369.9)	-108.9 (142.0)	206.3 (148.2)	32.9 (273.2)	284.3(360.4)	-76 (378.4)
Knee	-106.4					
extension	(164.6)	-30.4 (96.9)	148.4 (191.5)	-59.5 (207.2)	42.0(28.4)	-90 (188.1)
Knee flexion	213.3 (177.4)	-9.2 (121.8)	-47.7 (208.9)	29.1 (64.5)	165.7(187.1)	19.8 (115.9)
Total Calf	-31.4 (396.3)	-110.1 (219.0)	-157.7 (213.1)	30.9 (206.0)	-189.1(358.0)	-79.1 (180.3)
Plantar						
flexion	-42.2 (312.0)	-126.7 (149.2)	-96.4 (117.1)	64.5 (113.4)	-138.6(274.4)	-62.2 (116.3)
Dorsiflexion	-14.6 (48.2)	-10.9 (54.9)	-15.2 (31.0)	6.1 (39.7)	-29.8(78.6)	-4.8 (43.4)
FOOT	-15.7 (54.2)	-43.5 (56.9)	-2.4 (29.0)	-10.8 (24.7)	-18.1(64.2)	-54.3 (52.3)
mm ² = square m	illimetres, Data p	resented as: Mean	(standers deviation	on).		

Table 62: Longitudinal cross sectional area and differences between measurement points in exercise group (n=5) and no exercise group (n=5).

Remaining	BASE	LINE	6 MO	NTHS	12 MONTHS		
Muscle Area (mm ²)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Total Thigh	20511	17942	20519	17622	20834	17568	
Total Thigh	(11036)	(3019)	(11055)	(2809)	(11182)	(2671)	
Knee extension	10291 (5119)	8604 (1993)	10142 (5045)	8499 (2051)	10324 (5204)	8354 (1805)	
Knee flexion	6901 (4249)	6969 (1994)	7184 (4523)	6895 (1848)	7074 (4260)	6933 (1870)	
Total Calf	6938 (2659)	6740 (4034)	7048 (3121)	6243 (3805)	6784 (2961)	6218 (3991)	
Plantar flexion	3599 (1700)	4382 (3059)	3696 (1977)	3934 (2896)	3573 (2013)	3962 (2975)	
Dorsiflexion	1910 (1035)	1342 (644)	1862 (1021)	1302 (670)	1828 (989)	1292 (692)	
FOOT	274 (150)	236 (94)	270 (132)	203 (67)	258 (121)	195 (73)	
Remaining	BASELINE	-6 MONTHS	6MONTHS-	12MONTHS	BASELINE-	12MONTHS	
Muscle Area (mm ²) Difference	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Total Thigh	8 (337)	-320 (365)	315 (430)	-54 (258)	323 (633)	-375 (615)	
Knee extension	-150 (134)	-105 (99)	183 (177)	-145 (255)	33 (161)	-250 (223)	
Knee flexion	284 (391)	-73 (228)	-111 (492)	38 (124)	173 (299)	-35 (185)	
Total Calf	110 (484)	-497 (361)	-264 (287)	-25 (307)	-154 (390)	-522 (173)	
Plantar flexion	97 (368)	-448 (239)	-123 (223)	28 (173)	-26 (330)	-420 (153)	
Dorsiflexion	-48 (87)	-40 (107)	-34 (58)	-10 (89)	-82 (135)	-50 (69)	
FOOT	-4 (40)	-33 (35)	-13 (21)	-8 (17)	-16 (36)	-41 (34)	
mm ² = square	millimetres, D	Data presented	as: Mean (stan	ders deviation)			

Table 63: Longitudinal remaining muscle area and differences between measurement points in exercise group (n=5) and no exercise group (n=5).

Muscle	BASE	LINE	6 MO	NTHS	12 MONTHS		
Oedema T _{2m} (ms)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Total Calf	37.9 (9.7)	40.3 (4.4)	37.0 (8.2)	38.4 (6.1)	37 (8.2)	41.9 (4.9)	
Plantar flexion	38.5 (9.3)	40.9 (6.3)	37.6 (7.8)	38.6 (8.2)	37 (8.0)	41.8 (6.2)	
Dorsiflexion	36.7 (10.3)	39.6 (2.3)	36.6 (9.6)	38.8 (3.7)	37 (9.4)	42.9 (4.2)	
FOOT	44.9 (5.1)	46.7 (6.5)	47.4 (6.4)	46.1 (7.8)	49 (4.7)	50.4 (4.4)	
Muscle	BASELINE-6 MONTHS		6MONTHS-	12MONTHS	BASELINE-12MONTHS		
Oedema T _{2m} (ms) Difference	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Total Calf	-0.9 (1.8)	-1.9 (3.8)	0.6 (1.4)	3.5 (5.2)	-0.3 (1.6)	1.6 (2.6)	
Plantar flexion	-0.9 (2.0)	-2.2 (4)	0.2 (1.9)	3.1 (5.8)	-0.7 (1.7)	0.9 (2.8)	
Dorsiflexion	-0.1 (1.3)	-0.8 (2.6)	0.4 (0.5)	4.1 (3.4)	0.2 (1.5)	3.2 (2.7)	
FOOT	2.6 (5.0)	-0.6 (9)	1.5 (3.3)	4.3 (4.0)	4.1 (4.4)	3.6 (7.4)	
ms= millisec	ond, Data pres	ented as: Mean	(standers devi	ation).			

 Table 64: Longitudinal muscle oedema and differences between measurement points in

exercise group (n=5) and no exercise group (n=5).

Case 1

Fat Fraction (FF): After 6 months of training, there was a 2.0% decrease in FF for the overall calf, with a 4.7% reduction in the plantar flexors and a 0.1% reduction in the dorsiflexors. In the thigh showed an overall increase in FF post-training, although the knee flexors showed a 0.4% decrease (Table 65) (Figure 66).

Cross Sectional Area (CSA): After 6 months of training, the overall calf CSA increased by 16 mm², with a 36 mm² increase in the plantar flexors, and a 2 mm² reduction in the dorsiflexors. In the thigh CSA decreased by 257 mm², with a 117 mm² reduction in the knee extensors and a 15 mm² decrease in the knee flexors (Table 66) (Figure 66).

Remaining Muscle Area (RMA): After 6 months of training, there was an increase in RMA for the overall calf by 243 mm², with a 279 mm² increase in the plantar flexors, and a 2 mm² increase in the dorsiflexors. In the thigh RMA decreased by 564 mm², including a 306 mm² reduction in the knee extensors, while the knee flexors showed an 8 mm² increase (Table 67) (Figure 66).

Muscle Oedema (T_{2m}): After 6 months of training, Muscle oedema in the overall calf increased by 0.6 ms, with a 0.5 ms increase in the plantar flexores, and a 1 ms increase in dorsiflexors (Table 68) (Figure 66).

Case 2

Fat Fraction (FF): After 6 months of training, there was a 1.7% decrease in FF for the overall calf, with a 3.3% reduction in the plantar flexors and a 0.6% reduction in the dorsiflexors. In the thigh there was an overall decrease in FF by 1.4%, with a 1.5% reduction in the knee extensors, and a 1.3% decrease in the knee flexors (Table 65) (Figure 66).

Cross Sectional Area (CSA): After 6 months of training, the overall calf CSA decreased by 473 mm², with a 227 mm² decrease in the plantar flexors, and a 67 mm² reduction in the dorsiflexors. In the thigh CSA increased by 300 mm², with a 26 mm² increase in the knee extensors and a 66 mm² increase in the knee flexors (Table 66) (Figure 66).

Remaining Muscle Area (RMA): After 6 months of training, there was a decrease in RMA for the overall calf by 476 mm², with a 9 mm² decrease in the plantar flexors, and a 119 mm² decrease in the dorsiflexors. In the thigh RMA increased by 995 mm², including a 276 mm² increase in the knee extensors, and the knee flexors showed an 260 mm² increase (Table 67) (Figure 66).

Muscle Oedema (T_{2m}): After 6 months of training, Muscle oedema in the overall calf decreased by 0.2 ms, with a 0.7 ms decrease in the plantar flexores, and a 1 ms increase in dorsiflexors (Table 68) (Figure 66).

No Exercise Group (n=8)

Fat Fraction (FF): During the study period, there was a 3.2% increase in FF for the overall calf, with a 3.9% increase in the plantar flexors and a 2.7% increase in the dorsiflexors. In the thigh there was an overall increase in FF by 1.2%, with a 0.6% increase in the knee extensors, and a 1.7% increase in the knee flexors (Table 65) (Figure 66).

Cross Sectional Area (CSA): During the study period, the overall calf CSA decreased by 105.4 mm², with a 96.9 mm² decrease in the plantar flexors, and a 2 mm² increase in the dorsiflexors. In the thigh CSA increased by 57.1 mm², with a 39.9 mm² decrease in the knee extensors and a 80.4 mm² increase in the knee flexors (Table 66) (Figure 66).

Remaining Muscle Area (RMA): During the study period, there was a decrease in RMA for the overall calf by 359 mm², with a 287 mm² decrease in the plantar flexors, and a 34 mm² decrease in the dorsiflexors. In the thigh RMA decreased by 137 mm², including a 131 mm² decrease in the knee extensors, and the knee flexors showed an 24 mm² increase (Table 67) (Figure 66).

Muscle Oedema (T_{2m}): During the study period, Muscle oedema in the overall calf increased by 0.8 ms, with a 0.3 ms increase in the plantar flexors, and a 2 ms increase in dorsiflexors (Table 68) (Figure 66).

	BASELINE				6 MONT	THS	12 MONTHS		
FAT (%) EDACTION	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
FRACTION	1	2	Exercise [†]	1	2	Exercise [†]	1	2	Exercise†
Total Thigh	2.9	1.7	15.9	3.1	3.6	16.6	3.4	2.3	17.1
Knee extension	2.2	1.2	14.8	2.8	3.1	15.3	3.1	1.6	15.4
Knee flexion	3.4	2.3	17.4	3.0	4.2	18.9	3.3	2.9	19.1
Total Calf	26.8	29.0	36.6	24.7	28.5	39.2	27.2	26.8	39.8
Plantar flexion	53.7	44.9	40.2	49.0	43.5	43.5	54.9	40.2	44.0
Dorsiflexion	2.0	2.7	23.9	1.9	3.0	25.8	2.6	2.4	26.6
FOOT	31.9	27.3	55.0	24.1	27.3	56.0	23.3	29.9	57.5
FAT (%)	BASE	LINE-6 I	MONTHS	6MON	THS-12	MONTHS	BASE	LINE-12	MONTHS
FRACTION	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Difference	1*	2	Exercise [†]	1	2*	Exercise [†]	1	2	Exercise [†]
Total Thigh	0.3	2.0	0.6	0.3	-1.4	0.6	0.5	0.6	1.2
Knee									
extension	0.6	2.0	0.4	0.3	-1.5	0.2	0.9	0.5	0.6
Knee flexion	-0.4	2.0	1.5	0.3	-1.3	0.2	-0.1	0.7	1.7
Total Calf	-2.0	-0.5	2.6	2.5	-1.7	0.7	0.4	-2.2	3.2
Plantar									
flexion	-4.7	-1.4	3.3	5.9	-3.3	0.5	1.2	-4.7	3.9
Dorsiflexion	-0.1	0.3	1.9	0.8	-0.6	0.7	0.6	-0.3	2.7
FOOT	-7.9	0.1	1.0	-0.8	2.5	1.5	-8.6	2.6	2.5

*= after 6 months of training, No exercise group n=8, †=Mean.

Table 65: Longitudinal fat fraction and differences between measurement points in case 1, case 2, and no exercise group (n=8).

Cross		BASELINE			6 MONT	THS	12 MONTHS		
Sectional	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Area (mm ²)	1	2	Exercise [†]	1	2	Exercise [†]	1	2	Exercise [†]
Total Thigh	12928	14362	10010	12671	14800	9968	12774	15101	10067
Knee extension	6666	7455	4758	6550	7463	4686	6712	7489	4718
Knee flexion	4644	4856	3692	4629	5141	3785	4577	5207	3772
Total Calf	5400	6143	4925	5416	6311	4814	5208	5837	4820
Plantar flexion	2568	3805	3328	2604	3946	3200	2426	3719	3231
Dorsiflexion	1681	1223	829	1679	1123	825	1659	1056	831
FOOT	492	670	466	440	581	447	392	596	443
Cross	BASE	LINE-6 I	MONTHS	6MONTHS-12MONTHS			BASEI	LINE-12	MONTHS
Sectional	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Area (mm ²)	CASE 1*	CASE 2	Exercise [†]	CASE 1	CASE 2*	Exercise [†]	CASE 1	CASE 2	Exercise ⁺
Difference	1	-	Excreise	-	4	Excreise	-	-	Excreise
Total Thigh	-257	438	-42	103	300	99	-154.1	738.7	57.1
Knee									
extension	-117	8	-72	162	26	32	45.6	33.9	-39.9
Knee flexion	-15	285	94	-52	66	-13	-66.6	350.9	80.4
Total Calf	16	168	-111	-208	-473	6	-192.2	-305.7	-105.4
Plantar									
flexion	36	141	-128	-178	-227	31	-142.4	-86.7	-96.9
Dorsiflexion	-2	-100	-3	-20	-67	5	-21.8	-167.5	2.0
FOOT	-52	-89	-19	-49	15	-4	-100.6	-74.3	-23.4
*- ofter 6 mor	the of tro	ining ma	2^{2} causes m	illimate	No av	araisa araur	n-8 +-1	Maan	

*= after 6 months of training, mm^2 = square millimetres, No exercise group n=8, †=Mean.

Table 66: Longitudinal cross sectional area and differences between measurement points in case 1, case 2, and no exercise group (n=8).

Remaining		BASELINE			6 MONT	HS	12 MONTHS		
Muscle	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Area (mm ²)	1	2	Exercise [†]	1	2	Exercise [†]	1	2	Exercise [†]
Total Thigh	25108	28245	17364	24544	28526	17204	24669	29521	17227
Knee extension	13038	14735	8338	12731	14458	8251	13002	14734	8207
Knee flexion	8971	9492	6360	8980	9846	6447	8852	10107	6385
Total Calf	7909	8728	6469	8152	9025	6159	7581	8549	6110
Plantar flexion	2378	4197	4166	2657	4459	3880	2189	4449	3880
Dorsiflexion	3294	2381	1323	3295	2180	1293	3231	2061	1288
FOOT	335	488	216	334	422	201	300	418	193
Remaining	BASELINE-6 MONTHS			6MON	THS-12	MONTHS	BASEI	LINE-12N	MONTHS
Muscle Area (mm ²) Difference	CASE 1*	CASE 2	No Exercise†	CASE 1	CASE 2*	No Exercise†	CASE 1	CASE 2	No Exercise†
Total Thigh	-564	201							
17	501	281	-160	125	995	23	-439	1276	-137
extension	-306	-277	-160 -86	125 270	995 276	-45	-439 -36	1276 -1	-137 -131
extension Knee flexion	-306 8	-277 355	-160 -86 86	125 270 -128	995 276 260	23 -45 -62	-439 -36 -119	-1 615	-137 -131 24
Knee extension Knee flexion Total Calf	-306 8 243	-277 355 297	-160 -86 86 -310	125 270 -128 -571	995 276 260 -476	23 -45 -62 -49	-439 -36 -119 -328	1276 -1 615 -179	-137 -131 24 -359
Knee extension Knee flexion Total Calf Plantar	-306 8 243	-277 355 297	-160 -86 86 -310	125 270 -128 -571	995 276 260 -476	23 -45 -62 -49	-439 -36 -119 -328	1276 -1 615 -179	-137 -131 24 -359
Knee extension Knee flexion Total Calf Plantar flexion	-306 8 243 279	-277 355 297 262	-160 -86 86 -310 -287	125 270 -128 -571 -468	995 276 260 -476 -9	23 -45 -62 -49 0	-439 -36 -119 -328 -188	1276 -1 615 -179 253	-137 -131 24 -359 -287
Knee extension Knee flexion Total Calf Plantar flexion Dorsiflexion	-306 8 243 279 2	281 -277 355 297 262 -201	-160 -86 86 -310 -287 -30	125 270 -128 -571 -468 -65	995 276 260 -476 -9 -119	23 -45 -62 -49 0 -5	-439 -36 -119 -328 -188 -63	1276 -1 615 -179 253 -320	-137 -131 24 -359 -287 -34
Knee extension Knee flexion Total Calf Plantar flexion Dorsiflexion FOOT	-306 8 243 279 2 -1	281 -277 355 297 262 -201 -65	-160 -86 86 -310 -287 -30 -15	125 270 -128 -571 -468 -65 -34	995 276 260 -476 -9 -119 -4	23 -45 -62 -49 0 -5 -8	-439 -36 -119 -328 -188 -63 -35	-1 615 -179 253 -320 -70	-137 -131 24 -359 -287 -34 -23

Table 67: Longitudinal remaining muscle area and differences between measurement points

in case 1, case 2, and no exercise group (n=8).

Muscle		BASELI	NE		6 MONT	THS	1	2 MONT	THS	
Oedema	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No	
T _{2m} (ms)	1	2	Exercise [†]	1	2	Exercise [†]	1	2	Exercise [†]	
Total Calf	31.3	33.6	40.8	31.9	33.5	38.9	31.6	33.3	41.5	
Plantar flexion	32.2	35.5	41.1	32.7	35.8	39.1	30.9	35.1	41.5	
Dorsiflexion	30.4	30.7	40.1	31.4	29.5	39.5	32.1	30.5	42.1	
FOOT	40.4	44.7	46.6	51.0	42.4	46.8	49.6	42.8	50.6	
Muscle	BASE	LINE-6 I	MONTHS	6MON	THS-12	MONTHS	BASEI	BASELINE-12MONTHS		
Oedema T _{2m} (ms) Difference	CASE 1*	CASE 2	No Exercise†	CASE 1	CASE 2*	No Exercise†	CASE 1	CASE 2	No Exercise†	
Total Calf	0.6	-0.1	-1.8	-0.3	-0.2	2.6	0.3	-0.3	0.8	
Plantar										
flexion	0.5	0.3	-2.1	-1.8	-0.7	2.4	-1.3	-0.4	0.3	
Dorsiflexion	1.0	-1.2	-0.6	0.7	1.0	2.6	1.7	-0.2	2.0	
FOOT	10.6	-2.3	0.2	-1.4	0.4	3.8	9.2	-1.9	3.9	

*= after 6 months of training, ms= millisecond, No exercise group n=8, †=Mean.

 Table 68: Longitudinal muscle oedema and differences between measurement points in case

1, case 2, and no exercise group (n=8).

Thigh Fat Fraction

Calf Fat Fraction



Figure 66: Longitudinal MRI parameters in case 1, case 2, and no exercise group.

9.3.4. Effect of Exercise on Function

9.3.4.1. Clinical Assessment

Clinical assessment at base line showed better function for the exercises group (n=5) in comparison to no exercise group (n=5) (Table 69), and for case 1 and case 2 in comparison to the no exercise group (n=8) in CMTES, FPI-6, and Walk-12 questionnaire (Table 70). Follow up assessment showed improvement in case 1 and case 2 in CMTES, Walk-12 questionnaire, and manual muscle testing.

	BASE	LINE	6 MO	NTHS	12 MONTHS		
Clinical Assessment	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
CMTES; mean (SD)	5.2(5.1)	7.2(1.9)	5 (5.4)	6.2(1.6)	5 (5.4)	6.6 (2.3)	
Foot Posture Index-6 (Pronated/Normal/Supinated)	(1/3/1)	(0/4/1)	(1/3/1)	(0/1/4)	(1/3/1)	(0/1/4)	
Fall Frequency (weekly/Monthly/Yearly)	(1/1/0)	(0/0/2)	(1/1/0)	(0/0/1)	(1/1/0)	(0/0/1)	
Walk-12; mean (SD)	33.4(12.9)	39(14.1)	31(13.7)	38.2(12.4)	31.4(13.2)	40.8(12.3)	
Range of Motion (°)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Dorsiflexion (Limited/Normal)	(5/0)	(5/0)	(5/0)	(5/0)	(5/0)	(5/0)	
Plantarflexion (Limited/Normal)	(5/0)	(5/0)	(5/0)	(5/0)	(5/0)	(5/0)	
Manual Muscle Testing	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Hip Flexion	5(4-5)	5(3-5)	5 (4-5)	5 (4-5)	5 (4-5)	5 (4-5)	
Hip Extension	5(3-5)	5(3-5)	5 (3-5)	5 (3-5)	5 (3-5)	5 (3-5)	
Knee Flexion	5(3-5)	5(45)	5 (3-5)	5 (5-5)	5 (3-5)	5 (5-5)	
Knee Extension	5(4-5)	5(4-5)	5 (4-5)	5 (5-5)	5 (4-5)	5 (5-5)	
Dorsiflexion	4(0-4+)	1(1-4+)	5 (0-5)	2+(1-4+)	5 (0-5)	2+(1-4+)	
Plantarflexion	4+(1-4+)	4+(1-4+)	4+ (1-4+)	3 (0-4+)	4+ (1-4+)	2+(0-4+)	
CMTES= Charcot-Marie-Tooth	Disease Exami	ination Score, °	= degrees.				

Table 69: Longitudinal clinical assessment in exercise group (n=5) and no exercise group

(*n*=5).

		BASELIN	NE		6 MONT	HS	12 MONTHS		
Clinical Assessment	CASE 1	CASE 2	No Exercise (N=8)	CASE 1	CASE 2	No Exercise (N=8)	CASE 1	CASE 2	No Exercise (N=8)
CMTES; mean (SD)	2	2	7.25(3.2)	1	2	6.62 (3.31)	1	2	6.87 (3.5)
Foot Posture Index-6 (Pronated/Normal/Su pinated)	Supina ted	Norm al	(1/6/1)	Supina ted	Norm al	(1/3/4)	Supina ted	Norm al	(1/3/4)
Fall Frequency (weekly/Monthly/Ye arly)	0	0	(1/1/2)	0	0	(1/1/2)	0	0	(1/1/2)
Walk-12; mean (SD)	31	30	36 (12)	29	25	35 (13)	25	26	36 (13)
Range of Motion (°)	CASE 1	CASE 2	No Exercise (count)	CASE 1	CASE 2	No Exercise (count)	CASE 1	CASE 2	No Exercise (count)
Dorsiflexion (Limited/Normal)	Limite d	Limite d	(8/0)	Limite d	Limite d	(8/0)	Limite d	Limite d	(8/0)
Plantarflexion (Limited/Normal)	Limite d	Limite d	(8/0)	Limite d	Limite d	(8/0)	Limite d	Limite d	(8/0)
			No			No			No
Manual Muscle	CASE	CASE	Exercise	CASE	CASE	Exercise	CASE	CASE	Exercise
Testing	1	2	Mode (range)	1	2	Mode (range)	1	2	Mode (range)
Hip Flexion	5	5	4+(0-4+)	5	5	4+(0-5)	5	5	4+(0-5)
Hip Extension	5	5	4+(1-4+)	5	5	4+(1-4+)	5	5	4+(1-4+)
Knee Flexion	5	5	5(4-5)	5	5	5(4-5)	5	5	5(4-5)
Knee Extension	5	5	5(3-5)	5	5	5(3-5)	5	5	5(3-5)
Dorsiflexion	4	4-	5(3-5)	5	4-	5(3-5)	5	4-	5(3-5)
Plantarflexion	4	4-	5(3-5)	4+	4	5(4-5)	4+	4	5(4-5)
				0 1	ЪT	•		0	

CMTES= Charcot-Marie-Tooth Disease Examination Score, °= degrees, No exercise group n=8.

Table 70: Longitudinal clinical assessment in exercise case 1, case 2, and no exercises group

(*n*=8).

9.3.4.2. Isometric and Isokinetic Dynamometry

Comparision Between Exercises Group (N=5) and No Exercises Group (N=5):

Isometric Dynamometry: For hip extension at 45°, the exercise group showed a 24.8% decline from baseline to 6 months, followed by a 16.5% increase from 6 to 12 months, resulting in an overall 12.4% reduction over 12 months. In contrast, the no-exercise group showed a smaller 4.1% reduction at 6 months and a further 17.7% decrease by 12 months, resulting in a total 20.4% decline in strength. Knee extension at 45° showed a 13.6% decrease at 6 months in the exercise group, followed by a further 4.2% decline at 12 months, for an overall 17.3% reduction over 12 months. Conversely, the no-exercise group showed a smaller 2% increase in strength at 6 months, and a 9% decline by 12 months, resulting in a total reduction of 6.9%. The exercise group showed a 34.5% decline in ankle plantarflexion over 12 months, while the no-exercise group showed a smaller 11.1% decline. In dorsiflexion, the exercise group remained stable with a 4.3% increase, while the no-exercise group showed a larger 100% improvement in comparison to baseline (Table 71).

Isokinetic Dynamometry: For hip extension at 60°/60s, the exercise group showed a 9.2% increase from baseline to 6 months, followed by a 5.0% increase from 6 to 12 months, resulting in a 14.7% overall increase over 12 months. In contrast, the no-exercise group showed a 1.8% gain at 6 months, followed by a 5.2% decline by 12 months, resulting in a 3.5% reduction. Similarly, for hip flexion at $60^{\circ}/60$ s, the exercise group showed a 9.3% increase at 6 months and a further 11.1% increase by 12 months, resulting in a total 21.5% gain. The no-exercise group, however, showed an initial 19.8% increase by 6 months, followed by a 6.4% reduction at 12 months, resulting in a 12.1% increase over 12 months. knee flexion strength improved by 4.7% in the exercise group compared to a 2.6% increase in the no-exercise group. Knee extension strength decreased by 8.2% in the exercise group, while the no-exercise group remained stable, with only a 1.4% decline. The exercise group showed little change in plantarflexion, with a 0% increase, while the no-exercise group experienced a 38.5% gain over 12 months. Dorsiflexion strength improved by 26.9% in the exercise group, while the noexercise group showed a 64.3% increase (Table 72).

Tananatata	BASI	ELINE	6 MO	NTHS	12 MONTHS		
Dynamometry (Nm)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Hip EX 45°	129 (79)	147 (72)	97 (49)	141 (74)	113 (49)	117 (67)	
Hip FL 45°	107 (56)	103 (48)	103 (59)	113 (56)	120 (49)	104 (49)	
Knee EX 45°	110 (83)	99 (37)	95 (69)	100 (42)	91 (58)	91 (38)	
Knee EX 90°	110 (77)	99 (38)	100 (71)	102 (34)	100 (73)	90 (44)	
Knee FL 45°	65 (50)	53 (26)	54 (40)	56 (24)	57 (41)	53 (27)	
Knee FL 90°	38 (29)	33 (15)	40 (29)	44 (18)	43 (34)	39 (17)	
Ankle PF 10°	29 (20)	27 (21)	22 (13)	35 (27)	19 (12)	30 (21)	
Ankle DF 10°	23 (22)	4 (6)	24 (20)	9 (9)	24 (21)	8 (6)	
Ankle DF 30°	26 (22)	10 (8)	30 (21)	18 (12)	30 (23)	20 (8)	
Isometric	BASELINE	-6 MONTHS	6MONTHS-	-12MONTHS	BASELINE-12MONTHS		
Dynamometry (Nm)	Exercise	No Exercise	Exercise No Exercise		Exercise	No Exercise	
Difference	(N=5)	(N=5)	(N=5)	(N=5)	(N=5)	(N=5)	
Hip EX 45°	-32 (49)	-6 (18)	16 (30)	-25 (45)	-16 (42)	-30 (36)	
Hip FL 45°	-4 (23)	10 (9)	17 (33)	-9 (10)	13 (18)	1 (8)	
Knee EX 45°	-15 (27)	2 (15)	-4 (19)	-9 (16)	-19 (28)	-8 (15)	
Knee EX 90°	-10 (15)	3 (10)	0 (16)	-12 (20)	-10 (6)	-9 (16)	
Knee FL 45°	-11 (26)	4 (6)	3 (21)	-4 (14)	-8 (12)	0 (11)	
Knee FL 90°	2 (6)	11 (10)	4 (18)	-5 (10)	5 (20)	6 (11)	
Ankle PF 10°	-7 (12)	8 (9)	-4 (6)	-5 (11)	-10 (13)	2 (7)	
Ankle DF 10°	1 (4)	5 (4)	0 (3)	0 (6)	1 (3)	4 (4)	
Ankle DF 30°	4 (7)	8 (5)	0 (4)	2 (8)	3 (5)	10 (8)	

Nm= Newton meter, °= degrees, Data presented as: Mean (standers deviation).

Table 71: Longitudinal isomeric dynamometry and differences between measurement points in exercise group (n=5) and no exercise group (n=5).

T	BASE	ELINE	6 MO	NTHS	12 MONTHS		
Dynamometry (Nm)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Hip EX 60°/60s	109 (82)	113 (77)	119 (86)	115 (85)	125 (77)	109 (79)	
Hip FL 60°/60s	107 (62)	91 (45)	117 (74)	109 (59)	130 (58)	101 (39)	
Knee EX 60°/60s	85 (67)	73 (39)	77 (64)	78 (38)	78 (50)	74 (39)	
Knee FL 60°/60s	43 (36)	39 (22)	37 (30)	44 (25)	45 (32)	41 (23)	
Knee EX120°/120s	61 (57)	51 (33)	60 (52)	58 (31)	59 (39)	53 (30)	
Knee FL 120°/120s	35 (33)	28 (16)	33 (29)	34 (23)	35 (28)	30 (18)	
Ankle PF 60°/60°	21 (24)	13 (7)	23 (20)	21 (15)	21 (17)	17 (5)	
Ankle DF 60°/60°	15 (11)	14 (6)	21 (12)	18 (10)	22 (13)	23 (9)	
Isokinetic	BASELINE	-6 MONTHS	6MONTHS-	12MONTHS	BASELINE-12MONTHS		
Dynamometry (Nm)	Exercise	No Exercise	Exercise	No Exercise	Exercise	No Exercise	
Difference	(N=5)	(N=5)	(N=5)	(N=5)	(N=5)	(N=5)	
Hip EX 60°/60s	10 (19)	2 (20)	6 (22)	-6 (20)	16 (12)	-4 (8)	
Hip FL 60°/60s	10 (18)	18 (21)	13 (31)	-7 (22)	23 (16)	11 (11)	
Knee EX 60°/60s	-8 (19)	5 (9)	1 (25)	-4 (9)	-7 (20)	1 (15)	
Knee FL 60°/60s	-6 (8)	5 (7)	8 (12)	-3 (8)	2 (11)	2 (7)	
Knee EX120°/120s	-1 (6)	7 (12)	-1 (21)	-6 (5)	-2 (25)	1 (12)	
Knee FL 120°/120s	-2 (6)	6 (10)	2 (11)	-3 (10)	0 (13)	3 (9)	
Ankle PF 60°/60°	2 (8)	9 (8)	-2 (5)	-4 (10)	0 (9)	5 (3)	
Ankle DF 60°/60°	6(7)	4 (10)	1 (4)	5 (14)	7 (7)	9 (7)	
NT. NT. (0 1 D.		M (1	1			

Nm= Newton meter, $^{\circ}$ = degrees, Data presented as: Mean (standers deviation).

Table 72: Longitudinal isokinetic dynamometry and differences between measurement points in exercise group (n=5) and no exercise group (n=5).

Case 1

Isometric Dynamometry: After 6 months of training, there was a decrease in plantar flexion at 10° strength by 5 Nm and dorsiflexor at 10° strength by 4 Nm. There was an increase in dorsiflexor strength at 30° by 8 Nm. Proximally, knee flexion showed an increase in strength at 45° and 90° by 6 Nm and 10 Nm, respectively. Knee extension showed increase in strength at 90° by 10 Nm, and a decrease in strength at 45° by 49 Nm. Hip extension and flexion showed a decrease in strength by 62 Nm and 33 Nm, respectively (Table 73) (Figure 67).

Isokinetic Dynamometry: After 6 months of training, there was an incresses in strength in dorsiflexion with 8 Nm only, no change in knee extension at 120°/120s speed, and a decrese in the rest of the parameters (Table 74) (Figure 67).

Case 2

Isometric Dynamometry: After 6 months of training, there was a decrease in plantar flexion at 10° strength by 9 Nm. There was an increase in dorsiflexor strength at 30° by 6 Nm and dorsiflexor at 10° strength by 1 Nm. Proximally, knee flexion showed an increase in strength at 45° and 90° by 33 Nm and 34 Nm, respectively. Knee extension showed increase in strength at 90° by 18 Nm, and a decrease in strength at 45° by 26 Nm. Hip flexion showed a decrease in strength by 27 Nm. Hip extension showed an increase in strength by 13 Nm. (Table 73) (Figure 67).

Isokinetic Dynamometry: After 6 months of training, there was an increse in strength in dorsiflexion with 6 Nm and knee flexion at $60^{\circ}/60$ s speed with 13 Nm, no change in knee flexion at $120^{\circ}/120$ s speed, and a decrese in the rest of the parameters (Table 74) (Figure 67).

No Exercise Group (n=8)

Isometric Dynamometry: During the study period, there was no change in plantar flexion at 10° strength. There was an increase in dorsiflexor strength at 30° by 8 Nm and dorsiflexor at 10° strength by 3 Nm. Proximally, knee flexion showed an increase in strength at 90° by 2 Nm and a decrease at 45° by 3 Nm. Knee extension showed a decrease in strength at 90° and 45° by 9 Nm and 6 Nm, respectively. Hip extension showed a decrease in strength by 9 Nm. (Table 73) (Figure 67).

Isokinetic Dynamometry: During the study period, there was an incresse in strength in dorsiflexion with 9 Nm, plantar flexion with 5 Nm, Hip extension with 3 Nm, Hip flexion with 18 Nm, and a minimal increase in the rest of the parameters (Table 74) (Figure 67).

Tananatain	BASELINE				6 MONT	HS	12 MONTHS		
Dynamometry (Nm)	CASE 1	CASE 2	No Exercise†	CASE 1	CASE 2	No Exercise†	CASE 1	CASE 2	No Exercise†
Hip EX 45°	122	249	126	61	149	123	129	162	107
Hip FL 45°	121	176	94	89	199	99	150	172	100
Knee EX 45°	121	234	86	72	194	89	97	168	81
Knee EX 90°	125	223	87	135	191	85	116	209	78
Knee FL 45°	61	135	49	67	79	51	65	112	46
Knee FL 90°	54	50	31	64	54	38	59	87	33
Ankle PF 10°	29	46	26	24	34	29	14	25	25
Ankle DF 10°	36	54	6	32	55	10	35	55	9
Ankle DF 30°	36	59	11	44	54	18	42	60	18
Isometric	BASE	LINE-6 N	IONTHS	6MONTHS-12MONTHS			BASELINE-12MONTHS		
Dynamometry (Nm)	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Difference	1*	2	Exercise [†]	1	2*	Exercise [†]	1	2	Exercise [†]
Hip EX 45°	-62	-100	-3	69	13	-16	7	-87	-19
Hip FL 45°	-33	23	5	61	-27	1	29	-5	6
Knee EX 45°	-49	-40	3	25	-26	-8	-24	-66	-6
Knee EX 90°	10	-32	-2	-19	18	-7	-10	-14	-9
Knee FL 45°	6	-56	2	-2	33	-4	4	-23	-3
Knee FL 90°	10	4	6	-5	34	-4	5	37	2
Ankle PF 10°	-5	-12	3	-10	-9	-3	-15	-21	0
Ankle DF 10°	-4	1	4	3	1	-1	-1	1	3
Ankle DF 30°	8	-5	7	-2	6	1	6	1	8

*= after 6 months of training, Nm= Newton meter, $^{\circ}$ = degrees, No exercise group n=8, \dagger =Mean.

Table 73: Longitudinal isomeric dynamometry and differences between measurement points in case 1, case 2, and no exercise group (n=8).

Taul taut		BASELIN	NE		6 MONT	HS	12 MONTHS		
Isokinetic	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Dynamometry (1111)	1	2	Exercise [†]	1	2	Exercise [†]	1	2	Exercise [†]
Hip EX 60°/60s	106	231	97	99	260	102	137	239	99
Hip FL 60°/60s	108	194	86	94	231	101	130	196	104
Knee EX 60°/60s	96	186	63	55	178	68	92	147	65
Knee FL 60°/60s	42	94	34	35	75	37	58	88	35
Knee EX120°/120s	34	152	47	34	141	52	65	114	47
Knee FL 120°/120s	23	86	25	23	74	29	42	74	26
Ankle PF 60°/60°	20	63	11	16	56	19	19	48	16
Ankle DF 60°/60°	20	30	12	28	29	17	27	35	20
Isokinetic	BASE	LINE-6 N	IONTHS	6MONTHS-12MONTHS			BASELINE-12MONTHS		
Dynamometry (Nm)	CASE	CASE	No	CASE	CASE	No	CASE	CASE	No
Difference	1*	2	Exercise [†]	1	2*	Exercise [†]	1	2	Exercise [†]
Hip EX 60°/60s	-8	29	5	39	-21	-2	31	8	3
Hip FL 60°/60s	-14	38	14	36	-36	3	22	2	18
Knee EX 60°/60s	-41	-8	4	37	-32	-3	-4	-40	1
Knee FL 60°/60s	-7	-20	3	23	13	-2	16	-7	1
Knee EX120°/120s	0	-11	5	31	-28	-5	31	-39	1
Knee FL 120°/120s	-1	-12	4	20	0	-3	19	-12	1
Ankle PF 60°/60°	-4	-8	8	3	-8	-3	-1	-15	5
Ankle DF 60°/60°	8	-1	6	-1	6	3	7	5	9

*= after 6 months of training, Nm= Newton meter, $^{\circ}$ = degrees, s= seconds, No exercise group n=8, $^{+}$ =Mean. Table 74: Longitudinal isokinetic dynamometry and differences between measurement points in case 1, case 2, and no exercise group (n=8).



Figure 67: Longitudinal isometric and isokinetic strength in case 1, case 2, and no exercises group.

9.3.4.3. Three Dimensional Motion Analysis

Comparision Between Exercises Group (N=5) and No Exercises Group (N=5):

The exercise group showed consistent improvements in gait speed and stride length, whereas the no-exercise group showed minimal improvements or declines over time. For gait speed, the exercise group showed a 4.6% increase from baseline to 6 months, followed by a 1.7% increase from 6 to 12 months, resulting in an increase of 6.5% over 12 months. In contrast, the no-exercise group demonstrated a similar 6.7% increase at 6 months, but this was followed by a 6.3% reduction by 12 months, resulting in no net change. In stride length, the exercise group showed a 4.7% increase at 6 months, followed by a 1.5% increase at 12 months, for a total increase of 6.3%. The no-exercise group showed a smaller 2.8% increase at 6 months, followed by a 1.8% decline by 12 months, resulting in a total 0.9% increase (Table 75).

Gait	BASE	LINE	6 MO	NTHS	12 MONTHS		
Spatiotemporal Parameters	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	
Speed; m/s	1.08 (0.34)	0.89 (0.21)	1.13 (0.35)	0.95 (0.19)	1.15 (0.38)	0.88 (0.25)	
Stride Length; m	1.27 (0.25)	1.07 (0.21)	1.33 (0.31)	1.10 (0.18)	1.35 (0.35)	1.08 (0.21)	
Stride Time; s	1.23 (0.23)	1.22 (0.10)	1.22 (0.18)	1.18 (0.08)	1.21 (0.18)	1.26 (0.16)	
Strides Per Minute	50.17 (7.42)	49.56 (3.81)	49.9 (6.35)	51.26 (3.15)	50.27 (6.67)	48.42 (5.49)	
Step Length; m	0.63 (0.13)	0.54 (0.11)	0.66 (0.15)	0.55 (0.09)	0.67 (0.17)	0.54 (0.11)	
Step Time; s	0.62 (0.12)	0.61 (0.05)	0.61 (0.09)	0.59 (0.04)	0.61 (0.09)	0.63 (0.08)	
Steps Per Minute	100.35 (14.83)	99.13 (7.62)	99.9 (12.70)	102.52 (6.29)	100.55 (13.34)	96.84 (10.99)	
Percent Stance	0.63 (0.03)	0.64 (0.02)	0.63 (0.02)	0.64 (0.02)	0.63 (0.02)	0.65 (0.01)	
Single Support Time; s	0.61 (0.06)	0.61 (0.05)	0.63 (0.08)	0.58 (0.05)	0.63 (0.09)	0.65 (0.11)	
Double Support Time; s	0.17 (0.12)	0.17 (0.03)	0.14 (0.06)	0.17 (0.03)	0.14 (0.07)	0.17 (0.02)	

Percent Opposite Toe Off	13.36 (7.05)	13.92 (1.25)	13.4 (3.43)	14.64 (1.58)	12.47 (4.38)	13.47 (1.29)				
Percent Opposite Foot Contact	49.68 (0.77)	49.51 (0.68)	49.9 (0.72)	50.13 (0.78)	49.82 (0.37)	50.20 (1.00)				
Gait	BASELINE	6 MONTHS	6MONTHS-	12MONTHS	BASELINE-	12MONTHS				
Spatiotemporal Parameters Difference	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)				
Speed; m/s	0.05 (0)	0.06 (0.0)	0.02 (0.06)	-0.06 (0.07)	0.07 (0.09)	0.00 (0.11)				
Stride Length; m	0.06 (0)	0.03 (0.0)	0.02 (0.06)	-0.02 (0.04)	0.08 (0.14)	0.01 (0.05)				
Stride Time; s	-0. (0.07)	-0.04 (0.0)	-0.01 (0.02)	0.08 (0.08)	-0.02 (0.06)	0.04 (0.09)				
Strides Per Minute	-0.19 (2)	1.70 (1.5)	0.29 (0.84)	-2.84 (2.67)	0.10 (2.04)	-1.14 (3.44)				
Step Length; m	0.03 (0)	0.02 (0.0)	0.01 (0.03)	-0.01 (0.02)	0.04 (0.07)	0.01 (0.03)				
Step Time; s	-0. (0.03)	-0.02 (0.0)	0.00 (0.01)	0.04 (0.04)	-0.01 (0.03)	0.02 (0.04)				
Steps Per Minute	-0.38 (4)	3.39 (3.1)	0.58 (1.68)	-5.68 (5.34)	0.20 (4.07)	-2.28 (6.89)				
Percent Stance	0.00 (0)	0.01 (0.0)	0.00 (0.01)	0.01 (0.01)	0.00 (0.02)	0.01 (0.01)				
Single Support Time; s	0.0 (0.04)	-0.02 (0.0)	0.00 (0.03)	0.06 (0.07)	0.03 (0.04)	0.04 (0.07)				
Double Support Time; s	-0.03 (0)	0.00 (0.0)	0.00 (0.02)	-0.01 (0.03)	-0.04 (0.06)	0.00 (0.03)				
Percent Opposite Toe Off	0.11 (7)	0.73 (1.4)	-1.01 (6.47)	-1.18 (2.84)	-0.90 (5.75)	-0.45 (2.35)				
Percent Opposite Foot Contact	0.28 (0)	0.63 (0.7)	-0.13 (1.06)	0.07 (0.99)	0.14 (1.07)	0.70 (0.54)				
m = meters, s = s	m= meters, s= seconds, Data presented as: Mean (standers deviation).									

 Table 75: Longitudinal spatiotemporal gait parameters and differences between measurement

points in exercise group (n=5) and no exercise group (n=5).

Comparision Between Case 1, Case 2, and No Exercises Group (N=8):

Three dimensional motion analysis showed increase in speed, stride and step length, and a decrease in double support time, in comparison to no exercise group (n=8) (Table 76) (Figure 68). However, changes in kinetics and kinematics parameters were random and not consistence. Moreover, there was a technical issue when measuring gait at baseline for case 1 that did not allow the software to calculate kinetics and kinematics, but 6 months and 12 months follow ups were measures successfully. Detailed tables of the kinetic and kinematic data differences are presented in appendix VI.

Gait	ait BASELINE				6 MONTI	IS	12 MONTHS			
Spatiotemporal Parameters	CASE 1	CASE 2	No Exercise†	CASE 1	CASE 2	No Exercise†	CASE 1	CASE 2	No Exercise†	
Speed; m/s	1.34	1.28	0.90	1.37	1.44	0.95	1.50	1.43	0.90	
Stride Length; m	1.40	1.51	1.10	1.53	1.68	1.12	1.65	1.71	1.10	
Stride Time; s	1.05	1.18	1.25	1.12	1.17	1.21	1.10	1.19	1.26	
Strides Per Minute	57.60	50.77	48.79	53.72	51.40	50.14	54.47	50.35	48.58	
Step Length; m	0.70	0.76	0.55	0.76	0.84	0.56	0.82	0.85	0.55	
Step Time; s	0.52	0.59	0.63	0.56	0.58	0.61	0.55	0.60	0.63	
Steps Per Minute	115.21	101.54	97.58	107.43	102.80	100.27	108.93	100.69	97.16	
Percent Stance	0.60	0.64	0.64	0.61	0.63	0.64	0.62	0.63	0.65	
Single Support Time; s	0.55	0.67	0.61	0.56	0.66	0.61	0.60	0.68	0.64	
Double Support Time; s	0.09	0.09	0.19	0.12	0.08	0.17	0.09	0.07	0.17	
Percent Opposite Toe Off	8.24	7.44	15.09	10.89	17.75	13.99	15.45	5.58	13.58	
Percent Opposite Foot Contact	50.56	49.54	49.48	51.12	49.67	49.96	49.24	50.15	50.09	
Gait	BASE	LINE-6 M	IONTHS	6MO	6MONTHS-12MONTHS			LINE-12N	IONTHS	
Parameters Difference	CASE 1*	CASE 2	No Exercise†	CASE 1	CASE 2*	No Exercise†	CASE 1	CASE 2	No Exercise†	
Speed; m/s	0.03	0.16	0.05	0.13	0.00	-0.04	0.16	0.16	0.00	
Stride Length; m	0.13	0.16	0.02	0.12	0.03	-0.02	0.25	0.20	0.00	
---	-------	-------	-------	-------	--------	-------	-------	-------	-------	
Stride Time; s	0.07	-0.01	-0.04	-0.01	0.02	0.04	0.06	0.01	0.00	
Strides Per Minute	-3.89	0.63	1.35	0.75	-1.05	-1.55	-3.14	-0.42	-0.21	
Step Length; m	0.07	0.08	0.01	0.06	0.02	-0.01	0.12	0.10	0.00	
Step Time; s	0.04	-0.01	-0.02	-0.01	0.01	0.02	0.03	0.01	0.00	
Steps Per Minute	-7.78	1.26	2.70	1.50	-2.11	-3.11	-6.28	-0.85	-0.41	
Percent Stance	0.01	-0.01	0.00	0.01	-0.01	0.00	0.02	-0.02	0.01	
Single Support Time; s	0.01	-0.01	0.00	0.04	0.02	0.03	0.05	0.01	0.04	
Double Support Time; s	0.04	-0.01	-0.02	-0.03	-0.01	0.00	0.00	-0.02	-0.02	
Percent Opposite Toe Off	2.64	10.32	-1.10	4.56	-12.17	-0.42	7.21	-1.85	-1.51	
Percent Opposite Foot Contact	0.56	0.13	0.48	-1.88	0.48	0.14	-1.32	0.61	0.61	
*= after 6 months of training, m= meters, s= seconds, No exercise group n=8, \dagger =Mean.										

Table 76: Longitudinal spatiotemporal gait parameters and differences between measurement points in case 1, case 2, and no exercise group (n=8).



Figure 68: Longitudinal spatiotemporal parameters in case 1, case 2, and no exercises group.

9.4. Discussion

This study aimed to explore the effectiveness and safety of resistance exercise program in people with DHMN. The COVID-19 pandemic disrupted the study protocol and planned support for participants. This may have led to low adherence in the exercise group, limiting the trial's success. Nevertheless, it provided valuable preliminary data on the potential benefits of exercise and insights for future trial design.

9.4.1. Effectivness and Safety of Resistive Exercises

Changes in MRI parameters showed possible eveidance of effect and safetty of the exercise in these two cases. In the exercising cases, there were possible, positive effects on fat fraction levels at the calf and thigh, while "No Exercises" group showed an increase with the expected natural history. This might suggest that exercising could be used to slow the neuropathy progression, but would require more extensive testing.

At the thigh level case 1 and case 2 showed increase in cross sectional area (CSA) and the remaining muscle area (RMA) during the exercise period while CSA and RMA in "No Exercises" group showed no change. This finding is consistent with Chetlin et al. (2004), who found improvements in muscle fiber diameter following a home-based resistance training program in CMT patients (Chetlin et al., 2004b). At the calf level there was a trend of decrease in CSA and RMA in both exercise cases and "No Exercise" group. This suggests that when muscles

more midly affected by the neuropathy, the opportunities to gain muscle using resistive exercises could be higher, supporting the importance of early intervention.

Changes in muscle oedema was minimal in both case 1 and case 2 at the calf level, unlike the "No Exercise" group, that showed higher rates of muscle oedema. This finding may suggest that resistance exercises are safe, do not induce neurodegeneration, and are unlikely to cause overwork weakness in people with DHMN. This aligns with studies by Burns et al. (2017), which reported no adverse effects on muscle volume or signs of acute denervation following resistance exercises in children with Charcot-Marie-Tooth (CMT) disease (Burns et al., 2017).

Dynamometry showed individual improvement in strength with exercise. In both case 1 and case 2, dorsiflexor isometric and isokinetic strength slightly improved. This observation aligns with the findings of Burns et al. (2017), who reported 30% improvements in dorsiflexor strength following 6 months resistance training (Burns et al., 2017). However, plantar flexor continued to deteriorate in the exercise cases. Plantar flexors are a more affected group of muscles in DHMN, and exercising this group when they are advanced in neuropathy may not show improvement in strength. These findings are consistence with the MRI muscle area that showed a decrease in the plantar flexors muscle area even with exercising. The "No Exercise" group showed improved isometric and isokinetic

strength in dorsiflexors and plantar flexors. However, these results as discussed in chapter 6 are compromised a small sample size and high inter-group variability. Larger natural history studies are required to understand the average changes in muscles over time.

Proximally, isometric and isokinetic strength in both exercise cases showed improvements in strength, however, the pattern was not the same between cases. Proximal improvement in the case 1 and case 2 is consistence with the MRI findings. In the "No Exercise" group, isometric strength proximally showed continued deterioration, and isokinetic strength showed selective improvement in hip flexors. Improvement in hip flexors in the "No Exercise" group could be a false finding, but the compensatory function of this group is shown in chapter 5. The DHMN cohort used their hip flexors to compensate for distal weakness in pre-swing and swing phase. Inclusion of more participants and more measurement points would be recommended for future research to ascertain the effect of exercises on strength.

Exercising seemed to show a positive impact on the general function in the exercising cases. Three dimensional motion analysis showed increase in speed, stride and step length, and a decrease in double support time, in comparison to the "No Exercise" group. Moreover, clinical assessment showed improvement in exercises cases in CMTES, Walk-12 questionnaire, and manual muscle testing.

Larger studies are required to understand the impact of strength gain on general function.

9.4.2. Factors to Improve Adherence for Successful Exercise Trials

Clinical assessment at baseline showed better function in both exercise cases in comparison to the "No Exercise" group in CMTES, FPI-6, and Walk-12 questionnaire. It is possible that the more mildly affected cases were more likely to engage in exercise in comparison to those with more advanced neuropathy. This is consistent with the findings of Menotti et al. (2014), who reported that physical activity levels correlated with muscle strength in CMT patients (Menotti et al., 2014a). Motivation to exercise was influenced by occupational and environmental factors. Both cases with high adherence have physically demanding occupations and both were already engaging in other exercise. One of them had a medical background, so education in understanding the benefits could be important. Also, having an equipped room at home dedicated for exercise motivated them to keep engaged, despite pain or time limitations. This aligns with the conclusions of Anens et al. (2015), who identified personal and socioenvironmental factors as key influences on physical activity levels in individuals with CMT (Anens et al., 2015).

Available evidence on exercise benefits in peripheral neuropathy, pain, and sociopsychological factors can also influence motivation to exercises. Participants with low and no adherence reported more pain levels and time

limitation because of work or family situations, reflecting the findings of Roberts-Clarke et al. (2016), who noted that barriers such as fatigue and pain negatively impacted exercise adherence (Roberts-Clarke et al., 2016b). Some cases preferred to wait for the study findings before committing to an exercises program, indicating a need for better motivational strategies and support systems, as suggested by Buscemi et al. (Buscemi et al., 2023).

While this study provided initial insights into the effects of resistance exercises, uncertainties about their long-term effectiveness and safety remain. Future research should include a two-phase trial to address these questions comprehensively. The first phase of the study is aimed to ascertain the effectiveness and safety of progressive resistive exercises in DHMN with a supervised exercises sessions at an equipped facility to ascertain exercises application and motivate participants to engage in the program. Once the safety and efficacy are established, the second phase should explore the feasibility of home-based exercises using a program co-designed with participants, incorporating home visits, weekly check-ins, and exercise diaries to enhance adherence. Wearable monitors could be used to track engagement and progress, as recommended by Wallace et al. (Wallace et al., 2019).

The DHMN cohort identified personal beliefs and socio-environmental factors that influenced adherence to physical activity and exercise. To address these issues, approaches such as the Behaviour Change Wheel (BCW) (Figure 69)

(Michie S, 2014) could be used when designing the exercise protocol. The BCW framework can help identify key barriers and facilitators of exercise adherence, enabling the development of tailored interventions that support long-term engagement in physical activity for individuals with DHMN. This approach could enhance motivation, adherence, and ultimately, the quality of life for those affected by this condition. It encompasses three core components: the COM-B model (Capability, Opportunity, Motivation, and Behaviour), intervention functions, and policy categories. This framework is particularly valuable in designing interventions for complex behaviours like adherence to exercise programs in individuals with DHMN (Table 77).



Figure 69: The Behaviour Change Wheel (BCW)

Table 77: Applying the Behaviour Change Wheel Model to DHMN Exercise Adherence.

9.5. Summary

The current study explored the effect of resistance exercises on muscle structure and function in DHMN. The study findings are summarised below:

- In two cases, MRI indicated that resistance exercises may impact neuropathy progression, especially with mildly affected muscles. Exercise cases showed decreased intramuscular fat in calf and thigh, and increased thigh muscle area.
- Exercises were safe with no evidence of overwork weakness in two DHMN cases.
- Clinical assessments showed functional gains even with minimal strength increase. Two exercise cases had increased walking speed, stride length, and reduced double support time.
- Better function at baseline was linked to higher exercise adherence for the two exercises cases.
- Motivation to engage is influenced by occupational demands, prior exercise habits, and home equipment.
- Low adherence was due to higher pain, time constraints, and limited evidence.
- Behaviour change approaches will be important to design tailored interventions to enhance long-term exercise engagement.

Chapter 10: Overview

This thesis has focused on the relationship between muscle structure and function in individuals with Distal Hereditary Motor Neuropathies (DHMN). By examining various observational methods, including MRI, dynamometry, and 3D motion analysis, this study has provided a comprehensive overview of muscle changes over a period of one year. The primary aim was to understand the pattern and progression of muscle degeneration, with secondary aims to assess the impact of orthotics and therapeutic exercises on muscle function and gait in people with DHMN.

10.1. Muscle Structure and MRI as an Outcome Measure in DHMN

Chapters 5 and 6 discussed the use of MRI to describe the pattern of muscle involvement and the longitudinal changes in DHMN patients. Quantitative MRI analysis confirmed higher rates of fatty infiltration and signs of active denervation distally in the foot and calf muscles. Over 12 months, MRI fat fraction at the calf level was the most sensitive measure, showing significant deterioration in muscle tissue with an increase in fatty infiltration. These findings are consistent with previous studies (Esteller et al., 2023, O'Donnell et al., 2022, Morrow et al., 2016), and highlight the importance of quantitative MRI as an outcome measure in DHMN research.

One significant aspect of these findings is the differential pattern of muscle involvement observed through MRI. The distal muscles, particularly the plantar flexors, exhibited higher rates of fatty infiltration compared to proximal muscles. This pattern of distal predominance aligns with the clinical manifestations of DHMN, where distal muscle weakness and atrophy are more pronounced. The study also revealed that within the calf muscles, the posterior compartment, including the Medial Gastrocnemius and Soleus, was more affected than the anterior compartment. This specific involvement pattern underscores the importance of targeted rehabilitation strategies, focusing on these muscles to explore the possibility of slowing the progression of muscle degeneration and maintain function.

10.2. Muscle Function and Weakness Patterns in DHMN

Chapters 5 and 6 also explored muscle function using dynamometry. The results indicated predominant distal weakness, particularly in the plantar flexors. Longitudinally, muscle strength deterioration was detectable using manual muscle testing more than dynamometry. Quantitative muscle strength measures have been shown to be more sensitive in CMT (Burns et al., 2005). The DHMN cohort in the current study showed a high standard deviation (table 35, page 137) due to the small sample and high intragroup variability. Moreover, the presence of significant weakness and limited joint mobility affected the test application. This highlights the need for adapted testing protocols in future studies to accommodate the specific limitations of DHMN patients.

The decline in muscle strength over 12 months, as detected by both isometric and isokinetic dynamometry, was more pronounced in the distal muscles, consistent with the MRI findings of increased fatty infiltration. Similar relations were shown in people with peripheral neuropathies (O'Donnell et al., 2023, Morrow et al., 2016). The discrepancy between isometric and isokinetic strength in dorsiflexion suggests that muscle testing protocols must consider joint positioning and range of motion limitations in DHMN patients. For instance, the reduced isometric strength observed for ankle dorsiflexion may be partly due to the angle chosen for assessment combined with reduced range of ankle dorsiflexion. Earlier research highlighted the challenges of using dynamometry in the presence of significant weakness and limited joint mobility (Guillebastre et al., 2013, Reynaud et al., 2019). Future studies should explore alternative testing positions, such as seated rather than supine, to enhance the accuracy and reliability of strength measurements in this population. In the supine position, our cohort were able to engage their upper body to assess in pulling and pushing the foot plate by holding the side handles. This mechanism was eliminated by changing the position to 45° seated position.

10.3. Walking Gait in DHMN

Chapter 5 described the pattern of gait deviations in DHMN patients using 3D motion analysis. The findings indicated slower walking speeds, increased double support time, and compensatory hip flexion due to plantar flexion failure and foot

drop. This pattern has been described in CMT (Ramdharry et al., 2009, Don et al., 2007, Vinci and Perelli, 2002).

In foot drop pattern observed in our cohort showed a smaller ankle dorsiflexion angle in swing phase (Figure 15, A, page 116) alongside increased hip flexion angle and decreased ankle dorsiflexors power generation in swing phase. Additionally, there was an increase in plantar flexion angle at initial contact and loading response (Figure 15, B, page 116) indicating the inability to maintain the dorsiflexion angle and use the heel as a fulcrum for the first rocker. This further indicates a dropped foot pattern, although it was not statistically significant in this cohort.

A plantar flexion failure pattern was demonstrated by reduced ankle control, with increased passive dorsiflexion angle in terminal stance and decreased ankle plantar flexion in pre-swing (Figure 15, C, page 116). Kinetic data showed a decrease in plantar flexion moment and power generation in terminal stance, which confirms the presence of plantar flexion failure.

10.4. Relationships Between Intramuscular Fat Fraction, Muscle Volume, Isokinetic and Isometric Muscle Strength, and Kinetics of Gait

Chapter 7 focused on the relationships between MRI, dynamometry, and gait parameters. It was found that increased intramuscular fat fraction correlated with decreased muscle strength and altered gait kinetics. These relationships

emphasize the potential of using MRI dynamometry and gait kinetic parameters as outcome measures in clinical trials, to assess muscle changes in structure and performance.

At both distal and proximal levels, the current study showed muscle fat fraction correlating negatively with muscle area and cross sectional area, as expected with peripheral motor neuropathy. At the proximal level, increased fat fraction led to decreased muscle capacity, strength, and power during gait. On the other hand, greater muscle area correlated with higher muscle performance. Distally, higher muscle area in plantar flexors and dorsiflexors results in greater isometric ankle strength. Increased fat fraction, however, decreases dorsiflexion isometric strength. The findings align with research on hereditary peripheral neuropathies by O'Donnell et al. (2023) and Morrow et al. (2016), confirming similar relationships between muscle fat fraction, muscle area, and strength (Morrow et al., 2016, O'Donnell et al., 2023).

Increased fat fraction in dorsiflexors correlated with reduced dorsiflexor power generation during the swing phase, explaining the foot drop gait pattern. Decreased eccentric plantar flexor activity correlated positively with the reduction in plantar flexor moments in stance phase. This reduction in plantar flexor capacity likely contributed to the plantar flexor failure pattern with the inability to control ankle dorsiflexion angle in terminal stance plus delayed heel raise and toe off. Moreover, higher disease severity (CMTES) correlates with slower gait speed and shorter step and stride lengths, indicating that neuropathy affects spatial more than temporal gait parameters.

However, due to the high variability in a small sample, some significant correlations shown in this study do not make functional sense. For example, the relation between dorsiflexors strength and plantar flexors moments during gait. Future studies with larger samples and a control group are recommended to ascertain relationships with more accuracy.

10.5. The Effect of Rehabilitation Strategies in Management of DHMN

10.5.1. Effect of Carbon Fibre AFOs on Gait Parameters in DHMN

Chapter 8 further explored the effect of carbon fiber ankle foot orthoses (AFOs) on gait kinetics and kinematics, demonstrating improvements in walking gait with AFOs, but also suggesting the need for further studies to assess the long-term effects of AFOs. The use of AFOs compensated for dorsiflexion weakness by providing ankle support and stability, which allowed for a more normal ankle position at in initial contact (Figure 63, B, page 195), improved foot clearance (Figure 63, A, page 195), and longer strides without excessive hip and knee flexion in swing phase. This is in line with findings from previous research in children (Õunpuu et al., 2021), and adults (Ramdharry et al., 2012a) with CMT. However, the rigidity of the AFOs did prevent excessive forward progression of the tibia in mid to late stance phase (Figure 63, C, page 195), but also appeared to

restrict ankle movement at pre-swing, potentially inhibiting plantar flexor activity (Figure 65, page 197). The reduction in ankle power generation at pre-swing supports this hypothesis, in that the plantar flexors reduced activation, though electromyography (EMG) data was not available to confirm this. These findings suggest that while AFOs can improve gait function, their design should balance stability with the need for natural ankle movement to avoid long-term muscle disuse and atrophy.

10.5.2. Effectiveness and Safety of Resistive Exercises

Chapter 9 explored the impact of a resistive exercise program on muscle structure and function for DHMN participants. Although the COVID-19 pandemic disrupted the study protocol and led to low adherence, the preliminary data suggested potential benefits of exercise in two cases. MRI parameters showed improvement in muscle structure in the individuals who exercised.

The exercise intervention showed improvement in muscle area and a decrease in fat fraction levels, particularly in the thigh muscles, which are less affected by neuropathy, compared to the calf. These findings may indicate that early intervention with resistance exercises could potentially slow the progression of muscle degeneration. Moreover, the safety of resistance exercises was supported by the minimal increase in muscle edema, indicating that such interventions do not aggravate neurodegeneration or induce overwork weakness. These findings align with previous research in children with CMT that showed improvement in strength after completing resistive exercises program without signs of active denervation on MRI (Burns et al., 2017).

However, these findings were limited by the problems recruiting and low exercise adherence, highlighting the need for larger and more controlled trials to validate the effectiveness of exercise interventions in DHMN. The study revealed challenges in implementing exercise interventions in this population. The low adherence rate, partly due to the COVID-19 pandemic, highlights the need for strategies to improve patient engagement and adherence in future trials. Maintaining physically active lifestyles has been reported as particularly challenging for people with rare neurological conditions (Buscemi et al., 2024). Behavioural models and tailored exercise programs that consider the specific limitations and needs of DHMN patients could enhance adherence and maximize the benefits of exercise interventions (Busse and Ramdharry, 2020).

10.6. Study Limitations and Recommendations for Future Work

The results in a small and variable sample for the natural history study, and in two individual cases for the exercises trial, are insufficient to generalize to a broader DHMN population. The COVID-19 pandemic significantly impacted the study, leading to delays, low recruitment rates, and reduced exercise adherence. Methodological issues such as long visit times, extended MRI scanning periods, and limited gait analysis space also posed challenges. These factors highlight the need for improved study designs and protocols in future research. Challenges and recommendations are discussed below.

10.6.1. Impact of COVID-19 Pandemic on the Study

The COVID-19 pandemic had a significant impact on the study. Due to lockdowns and social distancing measures, there were substantial delays in the research timeline, leading to shorter study periods and low recruitment rates. Planned home visits for exercise participants were cancelled, resulting in reduced engagement and adherence to the exercise protocol. These disruptions led to a reduction in the overall strength of the study findings and affected the ability to collect data as planned. Due to the pandemic and other factors, exercise adherence was low, making it difficult to accurately assess the impact of the exercise intervention. This limited the potential to calculate meaningful statistical differences between the intervention and control groups. These challenges highlight the importance of developing tools to remotely monitor adherence, progression, and functional improvement. Development of such tools can encourage the inclusion of participants who do not have access to research facilities. Efforts to adapt to remote research have been taken to develop a virtual version of the CMTES (Prada et al., 2022). The virtual version was shown to be reproducible and reliable, however, the responsiveness for longitudinal trials haven't been explored yet (Prada et al., 2022).

10.6.2. Methodological Challenges

Several methodological issues were encountered during the study, the most important is the long visit time. Each visit is almost a full day to complete all study measurements including clinical assessments, MRI, dynamometry and motion analysis, which posed a challenge for participants. This led to two participants withdrawing from the study. Trials should consider identifying funding for accommodation to split the visit activity over two days, and vouchers to account for the participant's time.

MRI scans required extended periods, which posed a challenge for participants and led to shortening the MRI protocol, discarding scans covering thigh muscle which would provide valuable data. Splitting the scanning time over two session might be more practical to collect needed data without compromising participants convenience.

The dynamometry assessments were constrained by participants' limited ability to move due to muscle weakness or a restricted joint range of motion. This affected the accuracy of muscle strength measurements and required modifications to the testing protocol to accommodate these limitations such as changing the testing position. Future studies targeting participants with severe distal weakness and joint movement limitations could consider using alternative muscle strength quantification options, such as hand held dynamometry, as it can be triggered with minimum strength and shown to be valid and reliable in CMT

(Burns et al., 2005). However, it is limited to measuring isometric muscle strength only.

The gait analysis was conducted in a relatively small space with a short walkway, which affected the marker visibility and limited the ability to capture a full gait cycle for both sides at the same time. This led to repeating data collection captures that extended the gait assessment may have been more tiring for some participants.

10.6.3. Research and Clinical Implications

Future research should focus on extending the study period and increasing the frequency of follow-up visits to better capture the natural history of DHMN (Fridman et al., 2023). Longitudinal studies with larger sample sizes and inclusion of control participants are needed to validate the findings and provide more robust data on the progression of muscle degeneration and the effectiveness of interventions.

Calf muscle quantitative fat fraction using the Dixon method has been proposed as the most responsive outcome measure in natural history studies in CMT, IBM (Morrow et al., 2016), and in hereditary sensory neuropathy type 1 (HSN1) (Kugathasan et al., 2019). The current study showed calf muscle MRI quantitative fat fraction to have greater responsiveness (higher SRM) than dynamometry and 3D motion analysis. This finding has important implications for future trial design. Standardized Response Mean (SRM) is a measure of effect, in this context referring to DHMN natural history. A high SRM indicates that fewer participants are needed to determine if a new treatment effectively slows disease progression when the primary outcome measure of the trial is calf MRI fat fraction. On the contrary, the more modest SRM for muscle function and gait parameters suggests that larger sample sizes are necessary if the primary outcome of a trial is dynamometry or 3D motion analysis. For a hypothetical DHMN treatment trial using calf muscle MRI fat fraction as the primary outcome measure (SRM = 1.48), approximately 8 participants per group (active and placebo) are needed to detect a 50% reduction in disease progression over one year with 80% power at a significance level of p < 0.05 (Lehr, 1992).

Investigating the role of different types of exercise and rehabilitation techniques may provide insights into more effective management strategies for DHMN. If there is possibility that resistance exercise could slow progression of muscle impairment, then this will be an important recommendation in the current landscape where there are no other disease modifying therapies. Future research could include a two-phase trial to address these questions comprehensively. The first phase of the study would aim to ascertain the effectiveness and safety of progressive resistance exercises in DHMN with a supervised intensive exercises sessions, at an equipped facility, to ascertain exercises application and motivate participants to engage in the program. Once the safety and efficacy are established, the second phase should explore the feasibility of home-based exercises using program co-designed with participants, incorporating home visits, weekly check-ins, and exercise diaries to enhance adherence. Wearable monitors could be used to track engagement and progress, as recommended by Wallace et al. (Wallace et al., 2019). Additionally, applying models for behavioural changes to improve exercise adherence in trials could enhance the outcomes of exercise interventions. Using a crossover study design is suitable in rare conditions with high variability, such as DHMN, as fewer participants are needed, variability between groups is eliminated, and participants act as their own control. This study design has been used to assess the feasibility of community-based aerobic exercise training for people with CMT (Wallace et al., 2019). Although blinding participants and incorporating washout periods in exercises trials are challenging to apply, blinding the assessor to the group can improve the muscle strength assessment accuracy.

Understanding the biomechanical effect of AFOs can inform industrial research to improve design and material. Our data suggest that carbon fibre AFOs may restrict plantar flexion muscle activity, so it will be important to ensure that when they are prescribed, duration of wearing is considered, plus a specific program of plantar flexor strengthening exercise to negate the possibility of disuse atrophy. Hedge's G showed the effect size to be the largest in ankle angles; dorsiflexion and plantar flexion. This informs future research in AFO development, if they target ankle angles, a smaller sample size is required. However, if the focus is on proximal compensatory mechanisms, larger sample sizes will be necessary to determine the effectiveness of the AFO.

Overall, this thesis provides valuable insights into the natural history of DHMN and the effectiveness of rehabilitation interventions. The findings highlight the importance of improving orthotic designs, and early exercise interventions to manage DHMN. Future research should focus on larger trials to validate these findings and develop effective management strategies for this condition.

Chapter 11: Conclusion

In conclusion, this thesis has provided valuable insights into the presentation and progression of muscle degeneration and functional impairment in individuals with Distal Hereditary Motor Neuropathies (DHMN). The use of advanced observational techniques, such as MRI, dynamometry, and 3D motion analysis, has allowed for a detailed examination of muscle changes over time, the functional consequences and the impact of targeted exercise interventions.

The study has highlighted several key findings, including the significant increase in intramuscular fat and muscle atrophy, the decline in muscle strength, and the altered gait patterns in people with DHMN. These findings highlight the importance of early diagnosis and intervention to manage the progression of muscle degeneration, and maintain mobility and quality of life in affected individuals.

The impact of the COVID-19 pandemic on the study has been considerable, leading to methodological challenges and limitations in data collection and participant recruitment. Despite these obstacles, the study has demonstrated possible benefits of exercise interventions in improving muscle function and gait in people with DHMN.

Future research should build on these findings by extending the study period, increasing follow-up visits, and applying behavioural models to improve exercise

adherence. Longitudinal studies with larger sample sizes and diverse populations are needed to validate the findings and provide more comprehensive data on the natural history of DHMN and the effectiveness of rehabilitation interventions.

Overall, this thesis has contributed to a better understanding of DHMN and its impact on muscle function and mobility, providing a foundation for future research and clinical practice aimed at improving outcomes for individuals living with this condition.

References

- ACSM 2021. ACSM's Guidelines for Exercise Testing and Prescription, Philadelphia, Lippincott Williams & Wilkins.
- AKALAN, N. E., KUCHIMOV, S., TUYDAŞ, E., SENOL, L. B., ALBAYRAK, A., TEMEL, E., SERT, R., KOSEKUL, H. & NACAR, M. E. 2017. O14: Influences of Hip Adductor Muscle Weakness on Stiff Knee Gait Pattern in Healthy Individuals. *Gait & Posture*, 57, 25-26.
- ANDERSEN, H. 1996. Reliability of isokinetic measurements of ankle dorsal and plantar flexors in normal subjects and in patients with peripheral neuropathy. *Arch Phys Med Rehabil*, 77, 265-8.
- ANDERSON, F. C., GOLDBERG, S. R., PANDY, M. G. & DELP, S. L. 2004. Contributions of muscle forces and toe-off kinematics to peak knee flexion during the swing phase of normal gait: an induced position analysis. *J Biomech*, 37, 731-7.
- ANENS, E., EMTNER, M. & HELLSTRÖM, K. 2015. Exploratory Study of Physical Activity in Persons With Charcot-Marie-Tooth Disease. *Archives of Physical Medicine and Rehabilitation*, 96, 260-268.
- ARNOLD, A. S., ANDERSON, F. C., PANDY, M. G. & DELP, S. L. 2005. Muscular contributions to hip and knee extension during the single limb stance phase of normal gait: a framework for investigating the causes of crouch gait. *Journal of Biomechanics*, 38, 2181-2189.
- BALTZOPOULOS, V. & BRODIE, D. A. 1989. Isokinetic dynamometry. Applications and limitations. *Sports Med*, 8, 101-16.
- BANSAGI, B., GRIFFIN, H., WHITTAKER, R. G., ANTONIADI, T., EVANGELISTA, T., MILLER, J., GREENSLADE, M., FORESTER, N., DUFF, J., BRADSHAW, A., KLEINLE, S., BOCZONADI, V., STEELE, H., RAMESH, V., FRANKO, E., PYLE, A., LOCHMULLER, H., CHINNERY, P. F. & HORVATH, R. 2017. Genetic heterogeneity of motor neuropathies. *Neurology*, 88, 1226-1234.
- BARTONEK, A., ERIKSSON, M. & GUTIERREZ-FAREWIK, E. M. 2007. Effects of carbon fibre spring orthoses on gait in ambulatory children with motor disorders and plantarflexor weakness. *Dev Med Child Neurol*, 49, 615-20.
- BARWICK, K. E., WRIGHT, J., AL-TURKI, S., MCENTAGART, M. M., NAIR, A., CHIOZA, B., AL-MEMAR, A., MODARRES, H., REILLY, M. M., DICK, K. J., RUGGIERO, A. M., BLAKELY, R. D., HURLES, M. E. & CROSBY, A. H. 2012. Defective presynaptic choline transport underlies hereditary motor neuropathy. *Am J Hum Genet*, 91, 1103-7.
- BAS, J., OGIER, A. C., LE TROTER, A., DELMONT, E., LEPORQ, B., PINI, L., GUYE,
 M., PARLANTI, A., LEFEBVRE, M. N., BENDAHAN, D. & ATTARIAN, S. 2020.
 Fat fraction distribution in lower limb muscles of patients with CMT1A: A
 quantitative MRI study. *Neurology*.
- BEAN, J., WALSH, A. & FRONTERA, W. 2001. Brace Modification Improves Aerobic Performance in Charcot-Marie-Tooth Disease. American Journal of Physical Medicine & Rehabilitation, 80, 578-582.
- BECKMANN, N. A., WOLF, S. I., HEITZMANN, D., WALLROTH, A., MULLER, S. & DREHER, T. 2015. Cavovarus deformity in Charcot-Marie-Tooth disease: is there a hindfoot equinus deformity that needs treatment? *J Foot Ankle Res*, 8, 65.
- BEETZ, C., PIEBER, T. R., HERTEL, N., SCHABHUTTL, M., FISCHER, C., TRAJANOSKI, S., GRAF, E., KEINER, S., KURTH, I., WIELAND, T., VARGA, R.

E., TIMMERMAN, V., REILLY, M. M., STROM, T. M. & AUER-GRUMBACH, M. 2012. Exome sequencing identifies a REEP1 mutation involved in distal hereditary motor neuropathy type V. *Am J Hum Genet*, 91, 139-45.

- BENSOUSSAN, L., JOUVION, A., KERZONCUF, M., DELARQUE, A., THEODORIDOU, E., MILHE DE BOVIS, V., THEFENNE, L., ATTARIAN, S. & VITON, J. M. 2016. Orthopaedic shoes along with physical therapy was effective in
 - Charcot-Marie-Tooth patient over 10 years. Prosthet Orthot Int, 40, 636-42.
- BERTINI, A., MANGANELLI, F., FABRIZI, G. M., SCHENONE, A., SANTORO, L., CAVALLARO, T., TAGLIAPIETRA, M., GRANDIS, M., PREVITALI, S. C., FALZONE, Y. M., ALLEGRI, I., PADUA, L., PAZZAGLIA, C., TRAMACERE, I., CAVALCA, E., SAVERI, P., QUATTRONE, A., VALENTINO, P., TOZZA, S., GENTILE, L., RUSSO, M., MAZZEO, A., VITA, G., PRADA, V., ZUCCARINO, R., FERRARO, F., PISCIOTTA, C. & PAREYSON, D. 2024. Use, tolerability, benefits and side effects of orthotic devices in Charcot-Marie-Tooth disease. *J Neurol Neurosurg Psychiatry*, 95, 434-441.
- BLUMEN, S. C., ASTORD, S., ROBIN, V., VIGNAUD, L., TOUMI, N., CIESLIK, A., ACHIRON, A., CARASSO, R. L., GUREVICH, M., BRAVERMAN, I., BLUMEN, N., MUNICH, A., BARKATS, M. & VIOLLET, L. 2012. A rare recessive distal hereditary motor neuropathy with HSJ1 chaperone mutation. *Ann Neurol*, 71, 509-19.
- BROCKETT, C. L. & CHAPMAN, G. J. 2016. Biomechanics of the ankle. *Orthopaedics and Trauma*, 30, 232-238.
- BURNS, J., REDMOND, A., OUVRIER, R. & CROSBIE, J. 2005. Quantification of Muscle Strength and Imbalance in Neurogenic Pes Cavus, Compared to Health Controls, Using Hand-Held Dynamometry. *Foot & Ankle International*, 26, 540-544.
- BURNS, J., SMAN, A. D., CORNETT, K. M. D., WOJCIECHOWSKI, E., WALKER, T., MENEZES, M. P., MANDARAKAS, M. R., ROSE, K. J., BRAY, P., SAMPAIO, H., FARRAR, M., REFSHAUGE, K. M., RAYMOND, J., BURNS, J., SMAN, A. D., CORNETT, K. M. D., WOJCIECHOWSKI, E., MENEZES, M. P., MANDARAKAS, M. R., ROSE, K. J., BRAY, P., BALDWIN, J., MCKAY, M. J., REFSHAUGE, K. M., RAYMOND, J., WALKER, T., SAMPAIO, H., MUDGE, A., PURCELL, L., MILLER, C., GRAY, K., HARMAN, M., GABRAEL, N., OUVRIER, R. A. & FARRAR, M. 2017. Safety and efficacy of progressive resistance exercise for Charcot-Marie-Tooth disease in children: a randomised, double-blind, sham-controlled trial. *The Lancet Child & Adolescent Health*, 1, 106-113.
- BUSCEMI, V., BOAZ, A., DAWES, H., JAKI, T., JONES, F., MARSDEN, J., PAUL, L., PLAYLE, R., RANDELL, E., ROBLING, M., ROCHESTER, L., BUSSE, M. & RAMDHARRY, G. 2023. Exploring the issues that influence how people with rare neurological conditions engage in physical activity. *Disability and Rehabilitation*, 1-10.
- BUSCEMI, V., BOAZ, A., DAWES, H., JAKI, T., JONES, F., MARSDEN, J., PAUL, L., PLAYLE, R., RANDELL, E., ROBLING, M., ROCHESTER, L., BUSSE, M. & RAMDHARRY, G. 2024. Exploring the issues that influence how people with rare neurological conditions engage in physical activity. *Disability and Rehabilitation*, 46, 2789-2798.
- BUSSE, M. & RAMDHARRY, G. 2020. Targeting sedentary behaviour in neurological disease. *Practical Neurology*, 20, 187-188.
- CAPODAGLIO, P., GOBBI, M., DONNO, L., FUMAGALLI, A., BURATTO, C., GALLI, M. & CIMOLIN, V. 2021. Effect of Obesity on Knee and Ankle Biomechanics during Walking. *Sensors (Basel)*, 21.

- CARTER, G. T., ABRESCH, R. T., FOWLER, W. M. J., JOHNSON, E. R., KILMER, D. D.
 & MCDONALD, C. M. 1995. PROFILES OF NEUROMUSCULAR DISEASES: Hereditary Motor and Sensory Neuropathy, Types I and II. *American Journal of Physical Medicine & Rehabilitation*, 74, S150.
- CHAMBERS, H. G. & SUTHERLAND, D. H. 2002. A Practical Guide to Gait Analysis. JAAOS - Journal of the American Academy of Orthopaedic Surgeons, 10, 222-231.

CHARNWOODDYNAMICS 2004. CODA cx1 User Guide, Charnwood Dynamics Ltd.

- CHEN, Y., HAACKE, E. M. & LI, J. 2019. Peripheral nerve magnetic resonance imaging. *F1000Res*, 8.
- CHETLIN, R. D., GUTMANN, L., TARNOPOLSKY, M., ULLRICH, I. H. & YEATER, R. A. 2004a. Resistance training effectiveness in patients with Charcot-Marie-Tooth disease: recommendations for exercise prescription. *Arch Phys Med Rehabil*, 85, 1217-23.
- CHETLIN, R. D., GUTMANN, L., TARNOPOLSKY, M. A., ULLRICH, I. H. & YEATER, R. A. 2004b. Resistance training exercise and creatine in patients with Charcot-Marie-Tooth disease. *Muscle & Nerve*, 30, 69-76.
- CHETLIN, R. D., MANCINELLI, C. A. & GUTMANN, L. 2010. Self-reported follow-up post-intervention adherence to resistance exercise training in Charcot-Marie-Tooth disease patients. *Muscle & Nerve*, 42, 456-456.
- CODAMOTION 2016. ODIN User Guide.

CONOVER, W. J. 1999. Practical nonparametric statistics, New York ;, Wiley.

- DAHLQVIST, J. R., SALIM, R., THOMSEN, C. & VISSING, J. 2020. A quantitative method to assess muscle edema using short TI inversion recovery MRI. *Sci Rep*, 10, 7246.
- DE FUENMAYOR-FERNÁNDEZ DE LA HOZ, C. P., LUPO, V., BERMEJO-GUERRERO, L., MARTÍN-JIMÉNEZ, P., HERNÁNDEZ-LAÍN, A., OLIVÉ, M., GALLARDO, E., ESTEBAN-PÉREZ, J., ESPINÓS, C. & DOMÍNGUEZ-GONZÁLEZ, C. 2024. Distal hereditary motor neuronopathy as a new phenotype associated with variants in BAG3. *J Neurol*, 271, 986-994.
- DEL PORTO, L. A., NICHOLSON, G. A. & KETHESWAREN, P. 2010. Correlation between muscle atrophy on MRI and manual strength testing in hereditary neuropathies. *Journal of Clinical Neuroscience*, 17, 874-878.
- DELGADO, D. A., LAMBERT, B. S., BOUTRIS, N., MCCULLOCH, P. C., ROBBINS, A. B., MORENO, M. R. & HARRIS, J. D. 2018. Validation of Digital Visual Analog Scale Pain Scoring With a Traditional Paper-based Visual Analog Scale in Adults. J Am Acad Orthop Surg Glob Res Rev, 2, e088.
- DESCHENES, M. R. & KRAEMER, W. J. 2002. Performance and physiologic adaptations to resistance training. *Am J Phys Med Rehabil*, 81, S3-16.
- DIMITROVA, E. N., BOZINOVIKJ, I., RISTOVSKA, S., PEJCIKJ, A. H., KOLEVSKA, A. & HASANI, M. 2016. The Role of Rehabilitation in the Management of Patients with Charcot-Marie-Tooth Disease: Report of Two Cases. *Open Access Maced J Med Sci*, 4, 443-448.
- DJORDJEVIC, D., FELL, S. & BAKER, S. 2017. Effects of Self-Selected Exercise on Strength in Charcot-Marie-Tooth Disease Subtypes. *Can J Neurol Sci*, 44, 572-576.
- DON, R., SERRAO, M., VINCI, P., RANAVOLO, A., CACCHIO, A., IOPPOLO, F., PAOLONI, M., PROCACCIANTI, R., FRASCARELLI, F., DE SANTIS, F., PIERELLI, F., FRASCARELLI, M. & SANTILLI, V. 2007. Foot drop and plantar flexion failure determine different gait strategies in Charcot-Marie-Tooth patients. *Clinical biomechanics (Bristol, Avon)*, 22, 905-916.

- DROUIN, J. M., VALOVICH-MCLEOD, T. C., SHULTZ, S. J., GANSNEDER, B. M. & PERRIN, D. H. 2004. Reliability and validity of the Biodex system 3 pro isokinetic dynamometer velocity, torque and position measurements. *Eur J Appl Physiol*, 91, 22-9.
- DUDZIEC, M. M., LEE, L. E., MASSEY, C., TROPMAN, D., SKORUPINSKA, M., LAURÁ, M., REILLY, M. M. & RAMDHARRY, G. M. 2024. Home-based multisensory and proximal strengthening program to improve balance in Charcot-Marie-Tooth disease Type 1A: A proof of concept study. *Muscle Nerve*, 69, 354-361.
- DUFEK, J. S., NEUMANN, E. S., HAWKINS, M. C. & O'TOOLE, B. 2014. Functional and dynamic response characteristics of a custom composite ankle foot orthosis for Charcot–Marie–Tooth patients. *Gait & Posture*, 39, 308-313.
- DVIR, Z. 2000. Clinical Biomechanics, New York, Churchill Livingstone.
- DVIR, Z. 2004. *Isokinetics : Muscle Testing Interpretation and Clinical Applications,* Edinburgh, Churchill Livingstone.
- EL MHANDI, L., MILLET, G. Y., CALMELS, P., RICHARD, A., OULLION, R., GAUTHERON, V. & FÉASSON, L. 2008. Benefits of interval-training on fatigue and functional capacities in Charcot–Marie–Tooth disease. *Muscle & Nerve*, 37, 601-610.
- EL MHANDI, L., PICHOT, V., CALMELS, P., GAUTHERON, V., ROCHE, F. & FÉASSON, L. 2011. Exercise training improves autonomic profiles in patients with Charcot-Marie-Tooth disease. *Muscle & Nerve*, 44, 732-736.
- ESCOLAR, D. M., HENRICSON, E. K., MAYHEW, J., FLORENCE, J., LESHNER, R., PATEL, K. M. & CLEMENS, P. R. 2001. Clinical evaluator reliability for quantitative and manual muscle testing measures of strength in children. *Muscle & Nerve*, 24, 787-793.
- ESTELLER, D., MORROW, J., ALONSO-PÉREZ, J., REYES, D., CARBAYO, A.,
 BISOGNI, G., CATERUCCIA, M., MONFORTE, M., TASCA, G., ALANGARY,
 A., MARINI-BETTOLO, C., SABATELLI, M., LAURA, M., RAMDHARRY, G.,
 BOLAÑO-DÍAZ, C., TURON-SANS, J., TÖPF, A., GUGLIERI, M., ROSSOR, A.
 M., OLIVE, M., BERTINI, E., STRAUB, V., REILLY, M. M., ROJAS-GARCÍA, R.
 & DÍAZ-MANERA, J. 2023. Muscle magnetic resonance imaging of a large cohort of distal hereditary motor neuropathies reveals characteristic features useful for diagnosis. *Neuromuscul Disord*, 33, 744-753.
- ETIKAN, I., MUSA, S. A. & ALKASSIM, R. S. 2016. Comparison of Convenience Sampling and Purposive Sampling. *American Journal of Theoretical and Applied Statistics*, 5, 1-4.
- FISCHMANN, A., MORROW, J. M., SINCLAIR, C. D. J., REILLY, M. M., HANNA, M. G., YOUSRY, T. & THORNTON, J. S. 2014. Improved anatomical reproducibility in quantitative lower-limb muscle MRI. *Journal of Magnetic Resonance Imaging*, 39, 1033-1038.
- FORRESTER, N., RATTIHALLI, R., HORVATH, R., MAGGI, L., MANZUR, A., FULLER, G., GUTOWSKI, N., RANKIN, J., DICK, D., CHRISTOPHER, B., GREENSLADE, M. & MAJUMDAR, A. 2020. Clinical and Genetic Features in a Series of Eight Unrelated Patients with Neuropathy Due to Glycyl-tRNA Synthetase (GARS) Variants. *Journal of Neuromuscular Diseases*, 1-7.
- FRASQUET, M., ROJAS-GARCÍA, R., ARGENTE-ESCRIG, H., VÁZQUEZ-COSTA, J. F., MUELAS, N., VÍLCHEZ, J. J., SIVERA, R., MILLET, E., BARREIRO, M., DÍAZ-MANERA, J., TURON-SANS, J., CORTÉS-VICENTE, E., QUEROL, L., RAMÍREZ-JIMÉNEZ, L., MARTÍNEZ-RUBIO, D., SÁNCHEZ-MONTEAGUDO, A., ESPINÓS, C., SEVILLA, T. & LUPO, V. 2021. Distal hereditary motor

neuropathies: Mutation spectrum and genotype-phenotype correlation. *Eur J Neurol*, 28, 1334-1343.

- FRIDMAN, V., SILLAU, S., BOCKHORST, J., SMITH, K., MORONI, I., PAGLIANO, E., PISCIOTTA, C., PISCOSQUITO, G., LAURÁ, M., MUNTONI, F., BACON, C., FEELY, S., GRIDER, T., GUTMANN, L., SHY, R., WILCOX, J., HERRMANN, D. N., LI, J., RAMCHANDREN, S., SUMNER, C. J., LLOYD, T. E., DAY, J., SISKIND, C. E., YUM, S. W., SADJADI, R., FINKEL, R. S., SCHERER, S. S., PAREYSON, D., REILLY, M. M. & SHY, M. E. 2023. Disease Progression in Charcot-Marie-Tooth Disease Related to MPZ Mutations: A Longitudinal Study. Ann Neurol, 93, 563-576.
- FRIEDMAN, M. 1937. The Use of Ranks to Avoid the Assumption of Normality Implicit in the Analysis of Variance. *Journal of the American Statistical Association*, 32, 675-701.
- FRIEDMAN, M. 1940. A Comparison of Alternative Tests of Significance for the Problem of Rankings. *The Annals of Mathematical Statistics*, 11, 86-92, 7.
- FUKUCHI, C. A., FUKUCHI, R. K. & DUARTE, M. 2019. Effects of walking speed on gait biomechanics in healthy participants: a systematic review and meta-analysis. *Systematic Reviews*, **8**, 153.
- GLOVER, G. H. & SCHNEIDER, E. 1991. Three-point Dixon technique for true water/fat decomposition with B0 inhomogeneity correction. *Magn Reson Med*, 18, 371-83.
- GOLDBERG, S. R., ANDERSON, F. C., PANDY, M. G. & DELP, S. L. 2004. Muscles that influence knee flexion velocity in double support: implications for stiff-knee gait. *J Biomech*, 37, 1189-96.
- GRAHAM, R. C. & HUGHES, R. A. 2006. Clinimetric properties of a walking scale in peripheral neuropathy. *J Neurol Neurosurg Psychiatry*, 77, 977-9.
- GREENBAUM, L., BAREL, O., NIKITIN, V., HERSALIS-ELDAR, A., KOL, N., REZNIK-WOLF, H., DOMINISSINI, D., PRAS, E. & DORI, A. 2020. Identification of a homozygous VRK1 mutation in two patients with adult-onset distal hereditary motor neuropathy. *Muscle & Nerve*, 61, 395-400.
- GRIMM, A., MEYER, H., NICKEL, M. D., NITTKA, M., RAITHEL, E., CHAUDRY, O., FRIEDBERGER, A., UDER, M., KEMMLER, W., QUICK, H. H. & ENGELKE, K. 2018. Evaluation of 2-point, 3-point, and 6-point Dixon magnetic resonance imaging with flexible echo timing for muscle fat quantification. *European Journal of Radiology*, 103, 57-64.
- GUILLEBASTRE, B., CALMELS, P. & ROUGIER, P. 2013. Effects of muscular deficiency on postural and gait capacities in patients with Charcot-Marie-Tooth disease. *J Rehabil Med*, 45, 314-7.
- HA, M. & HAN, D. 2017. The relationship between knee joint angle and knee flexor and extensor muscle strength. *J Phys Ther Sci*, 29, 662-664.
- HACKETT, D., ROBERTS-CLARKE, D., JAIN, N., MAVROS, Y., WILSON, G. C., HALAKI, M., BURNS, J., NICHOLSON, G., FIATARONE SINGH, M. & FORNUSEK, C. 2019. Body composition and its association with physical performance, quality of life, and clinical indictors in Charcot-Marie-Tooth disease: a pilot study. *Disability and Rehabilitation*, 41, 405-412.
- HARDING, A. E. 1993. Inherited neuronal atrophy and degeneration predominantly of lower motor neurons. *In:* DYCK, P. J. (ed.) *Peripheral Neuropathy*. 3 ed. Philadelphia: Saunders.
- HARLAAR, J., ROEBROECK, M. E. & LANKHORST, G. J. 1996. Computer-assisted hand-held dynamometer: low-cost instrument for muscle function assessment in

rehabilitation medicine. *Medical and Biological Engineering and Computing*, 34, 329-335.

- HERDA, T. J. 2022. Resistance exercise training and the motor unit. *Eur J Appl Physiol*, 122, 2019-2035.
- HISLOP, H. J. & PERRINE, J. 1967. The Isokinetic Concept of Exercise. *Physical Therapy*, 47, 114-117.
- HOLLANDER, M., WOLFE, D. A., CHICKEN, E. 2013. Nonparametric Statistical *Methods*, Newark, Wiley.
- HONG, D., FANG, P., YAO, S., CHEN, J., ZHANG, X., CHEN, S., ZHANG, J., TAN, D., WANG, L., HAN, X., XIN, L., WANG, Y., LIU, M., CONG, L., ZHONG, S., OUYANG, H., GAO, X. & ZHANG, J. 2019. Variants in MME are associated with autosomal-recessive distal hereditary motor neuropathy. *Ann Clin Transl Neurol*, 6, 1728-1738.
- HUSKISSON, E. C. 1974. Measurement of pain. Lancet, 2, 1127-31.
- ISHIHARA, S., OKAMOTO, Y., TANABE, H., YOSHIMURA, A., HIGUCHI, Y., YUAN, J. H., HASHIGUCHI, A., ISHIURA, H., MITSUI, J., SUWAZONO, S., OYA, Y., SASAKI, M., NAKAGAWA, M., TSUJI, S., OHYA, Y. & TAKASHIMA, H. 2020. Clinical features of inherited neuropathy with BSCL2 mutations in Japan. *Journal of the Peripheral Nervous System*.
- JACQUIER, A., THEURIET, J., FONTAINE, F., MOSBACH, V., LACOSTE, N., RIBAULT, S., RISSON, V., CARRAS, J., COUDERT, L., SIMONET, T., LATOUR, P., STOJKOVIC, T., PIARD, J., COSSON, A., LESCA, G., BOUHOUR, F., ALLOUCHE, S., PUCCIO, H., PEGAT, A. & SCHAEFFER, L. 2023. Homozygous COQ7 mutation: a new cause of potentially treatable distal hereditary motor neuropathy. *Brain*, 146, 3470-3483.
- KANG, K. H., HAN, J. E., HONG, Y. B., NAM, S. H., CHOI, B. O. & KOH, H. 2020. Human HSPB1 mutation recapitulates features of distal hereditary motor neuropathy (dHMN) in Drosophila. *Biochem Biophys Res Commun*, 521, 220-226.
- KANNUS, P. 1994. Isokinetic evaluation of muscular performance: implications for muscle testing and rehabilitation. *Int J Sports Med*, 15 Suppl 1, S11-8.
- KEENAN, A. M., REDMOND, A. C., HORTON, M., CONAGHAN, P. G. & TENNANT, A. 2007. The Foot Posture Index: Rasch analysis of a novel, foot-specific outcome measure. *Arch Phys Med Rehabil*, 88, 88-93.
- KIM, H. S., YOON, Y. C., CHOI, B. O., JIN, W. & CHA, J. G. 2019. Muscle fat quantification using magnetic resonance imaging: case-control study of Charcot-Marie-Tooth disease patients and volunteers. *J Cachexia Sarcopenia Muscle*, 10, 574-585.
- KNAK, K. L., ANDERSEN, L. K. & VISSING, J. 2017. Aerobic anti-gravity exercise in patients with Charcot-Marie-Tooth disease types 1A and X: A pilot study. *Brain Behav*, 7, e00794.
- KNAPIK, J. J., WRIGHT, J. E., MAWDSLEY, R. H. & BRAUN, J. 1983. Isometric, isotonic, and isokinetic torque variations in four muscle groups through a range of joint motion. *Phys Ther*, 63, 938-47.
- KNUDSON, D. V. 2007. Fundamentals of biomechanics, New York, Springer.
- KUGATHASAN, U., EVANS, M. R. B., MORROW, J. M., SINCLAIR, C. D. J., THORNTON, J. S., YOUSRY, T. A., HORNEMANN, T., SURIYANARAYANAN, S., OWUSU-ANSAH, K., LAURIA, G., LOMBARDI, R., POLKE, J. M., WILSON, E., BENNETT, D. L. H., HOULDEN, H., HANNA, M. G., BLAKE, J. C., LAURA, M. & REILLY, M. M. 2019. Development of MRC Centre MRI calf muscle fat

fraction protocol as a sensitive outcome measure in Hereditary Sensory Neuropathy Type 1. *Journal of Neurology, Neurosurgery & Psychiatry*, jnnp-2018-32019.

- KUHLENBÄUMER, G., STÖGBAUER, F., RINGELSTEIN, E. B. & YOUNG, P. 2005. *Hereditary Peripheral Neuropathies*, Steinkopff-Verlag Heidelberg.
- KUO, A. D. & DONELAN, J. M. 2010. Dynamic principles of gait and their clinical implications. *Phys Ther*, 90, 157-74.
- LANDFELDT, E., EDSTRÖM, J., LINDGREN, P. & LOCHMÜLLER, H. 2017. Patient Preferences for Treatments of Neuromuscular Diseases: A Systematic Literature Review. *Journal of Neuromuscular Diseases*, 4, 285-292.
- LAURÁ, M., BARNETT, J., BENFIELD, J., RAMDHARRY, G. & WELCK, M. 2024. Foot surgery for adults with Charcot-Marie-Tooth disease. *Practical Neurology*, pn-2023-003825.
- LE-NGOC, L. & JANSSE, J. 2012. Validity and reliability of a hand-held dynamometer for dynamic muscle strength assessment. InTech.
- LEE, S. S., LEE, H. J., PARK, J. M., HONG, Y. B., PARK, K. D., YOO, J. H., KOO, H., JUNG, S. C., PARK, H. S., LEE, J. H., LEE, M. G., HYUN, Y. S., NAKHRO, K., CHUNG, K. W. & CHOI, B. O. 2013. Proximal dominant hereditary motor and sensory neuropathy with proximal dominance association with mutation in the TRKfused gene. JAMA Neurol, 70, 607-15.
- LEHR, R. 1992. Sixteen S-squared over D-squared: a relation for crude sample size estimates. *Stat Med*, 11, 1099-102.
- LENCIONI, T., PISCOSQUITO, G., RABUFFETTI, M., SIPIO, E. D., DIVERIO, M., MORONI, I., PADUA, L., PAGLIANO, E., SCHENONE, A., PAREYSON, D. & FERRARIN, M. 2018. Electromyographic and biomechanical analysis of step negotiation in Charcot Marie Tooth subjects whose level walk is not impaired. *Gait & Posture*, 62, 497-504.
- LI, R. C., JASIEWICZ, J. M., MIDDLETON, J., CONDIE, P., BARRISKILL, A., HEBNES, H. & PURCELL, B. 2006. The development, validity, and reliability of a manual muscle testing device with integrated limb position sensors. *Arch Phys Med Rehabil*, 87, 411-7.
- LINS, C. F., SALMON, C. E. G. & NOGUEIRA-BARBOSA, M. H. 2020. Applications of the Dixon technique in the evaluation of the musculoskeletal system. *Radiologia Brasileira*.
- LIU, X., DUAN, X., ZHANG, Y., SUN, A. & FAN, D. 2020. Molecular analysis and clinical diversity of distal hereditary motor neuropathy. *Eur J Neurol*, 27, 1319-1326.
- LOCHER, N., WAGNER, B., BALSIGER, F. & SCHEIDEGGER, O. 2022. Quantitative water T2 relaxometry in the early detection of neuromuscular diseases: a retrospective biopsy-controlled analysis. *European Radiology*, 32, 7910-7917.
- LUND, H., SØNDERGAARD, K., ZACHARIASSEN, T., CHRISTENSEN, R., BÜLOW,
 P., HENRIKSEN, M., BARTELS, E. M., DANNESKIOLD-SAMSØE, B. &
 BLIDDAL, H. 2005. Learning effect of isokinetic measurements in healthy subjects, and reliability and comparability of Biodex and Lido dynamometers. *Clin Physiol Funct Imaging*, 25, 75-82.
- MA, J. 2008. Dixon techniques for water and fat imaging. *Journal of Magnetic Resonance Imaging*, 28, 543-558.
- MAGGI, G., MONTI BRAGADIN, M., PADUA, L., FIORINA, E., BELLONE, E., GRANDIS, M., RENI, L., BENNICELLI, A., GROSSO, M., SAPORITI, R., SCORSONE, D., ZUCCARINO, R., CRIMI, E. & SCHENONE, A. 2011. Outcome Measures and Rehabilitation Treatment in Patients Affected by Charcot-Marie-Tooth Neuropathy. *American Journal of Physical Medicine & Rehabilitation*, 90, 628-637.

- MARGINSON, V. & ESTON, R. 2001. The relationship between torque and joint angle during knee extension in boys and men. *Journal of Sports Sciences*, 19, 875-880.
- MARTIN, H. J., YULE, V., SYDDALL, H. E., DENNISON, E. M., COOPER, C. & AIHIE SAYER, A. 2006. Is Hand-Held Dynamometry Useful for the Measurement of Quadriceps Strength in Older People? A Comparison with the Gold Standard Biodex Dynamometry. *Gerontology*, 52, 154-159.
- MCCAUGHAN, D., BOOTH, A., JACKSON, C., LALOR, S., RAMDHARRY, G., O'CONNOR, R. J., PHILLIPS, M., BOWERS, R. & MCDAID, C. 2019. Orthotic management of instability of the knee related to neuromuscular and central nervous system disorders: qualitative interview study of patient perspectives. *BMJ Open*, 9, e029313.
- MCROBBIE, D. W., MOORE, E. A., GRAVES, M. J. & PRINCE, M. R. 2017. *MRI from Picture to Proton*, Cambridge, Cambridge University Press.
- MENOTTI, F., FELICI, F., DAMIANI, A., MANGIOLA, F., VANNICELLI, R. & MACALUSO, A. 2011. Charcot-Marie-Tooth 1A patients with low level of impairment have a higher energy cost of walking than healthy individuals. *Neuromuscular Disorders*, 21, 52-57.
- MENOTTI, F., LAUDANI, L., DAMIANI, A. & MACALUSO, A. 2014a. Amount and intensity of daily living activities in Charcot-Marie-Tooth 1A patients. *Brain Behav*, 4, 14-20.
- MENOTTI, F., LAUDANI, L., DAMIANI, A., MIGNOGNA, T. & MACALUSO, A. 2014b. An anterior ankle-foot orthosis improves walking economy in Charcot-Marie-Tooth type 1A patients. *Prosthet Orthot Int*, 38, 387-92.
- MERCURI, E., CINI, C., PICHIECCHIO, A., ALLSOP, J., COUNSELL, S., ZOLKIPLI, Z., MESSINA, S., KINALI, M., BROWN, S. C., JIMENEZ, C., BROCKINGTON, M., YUVA, Y., SEWRY, C. A. & MUNTONI, F. 2003. Muscle magnetic resonance imaging in patients with congenital muscular dystrophy and Ullrich phenotype. *Neuromuscular Disorders*, 13, 554-558.
- MICHIE S, A. L., WEST R. 2014. *The Behaviour Change Wheel: A Guide to Designing Interventions,* London, Silverback Publishing.
- MITAL, A., KOPARDEKAR, P. & MOTORWALA, A. 1995. Isokinetic pull strengths in the vertical plane: effects of speed and arm angle. *Clin Biomech (Bristol, Avon)*, 10, 110-112.
- MNATSAKANIAN, A., KISSEL, J. T., TERRY, P. & KING, W. M. 2017. One clinic's experience with carbon fiber orthoses in neuromuscular disease. *Muscle & Nerve*, 55, 202-205.
- MONTES, J. & GARBER, C. E. 2017. Benefits of exercise therapy in Charcot-Marie-Tooth disease. *The Lancet Child & Adolescent Health*, 1, 82-83.
- MORROW, J. M., EVANS, M. R. B., GRIDER, T., SINCLAIR, C. D. J., THEDENS, D., SHAH, S., YOUSRY, T. A., HANNA, M. G., NOPOULOS, P., THORNTON, J. S., SHY, M. E. & REILLY, M. M. 2018. Validation of MRC Centre MRI calf muscle fat fraction protocol as an outcome measure in CMT1A. *Neurology*, 91, e1125-e1129.
- MORROW, J. M., SHAH, S., CRISTIANO, L., EVANS, M. R. B., DOHERTY, C. M., ALNAEMI, T., SAAB, A., EMIRA, A., KLICKOVIC, U., HAMMAM, A., ALTUWAIJRI, A., WASTLING, S., REILLY, M. M., HANNA, M. G., YOUSRY, T. A. & THORNTON, J. S. 2024. Development of an initial training and evaluation programme for manual lower limb muscle MRI segmentation. *Eur Radiol Exp*, 8, 85.
- MORROW, J. M., SINCLAIR, C. D., FISCHMANN, A., MACHADO, P. M., REILLY, M. M., YOUSRY, T. A., THORNTON, J. S. & HANNA, M. G. 2016. MRI biomarker

assessment of neuromuscular disease progression: a prospective observational cohort study. *Lancet Neurol*, 15, 65-77.

- MORROW, J. M., SINCLAIR, C. D., FISCHMANN, A., REILLY, M. M., HANNA, M. G., YOUSRY, T. A. & THORNTON, J. S. 2014. Reproducibility, and age, body-weight and gender dependency of candidate skeletal muscle MRI outcome measures in healthy volunteers. *Eur Radiol*, 24, 1610-20.
- MUKAKA, M. M. 2012. Statistics corner: A guide to appropriate use of correlation coefficient in medical research. *Malawi Med J*, 24, 69-71.
- MURPHY, S. M., HERRMANN, D. N., MCDERMOTT, M. P., SCHERER, S. S., SHY, M. E., REILLY, M. M. & PAREYSON, D. 2011. Reliability of the CMT neuropathy score (second version) in Charcot-Marie-Tooth disease. *J Peripher Nerv Syst*, 16, 191-8.
- NARDONE, A., CORNA, S., TURCATO, A. M. & SCHIEPPATI, M. 2014. Afferent control of walking: Are there distinct deficits associated to loss of fibres of different diameter? *Clinical Neurophysiology*, 125, 327-335.
- NEUMANN, D. A. 2010. Kinesiology of the Hip: A Focus on Muscular Actions. *Journal of Orthopaedic & Sports Physical Therapy*, 40, 82-94.
- NEWMAN, C. J., WALSH, M., O'SULLIVAN, R., JENKINSON, A., BENNETT, D., LYNCH, B. & O'BRIEN, T. 2007. The characteristics of gait in Charcot-Marie-Tooth disease types I and II. *Gait & Posture*, 26, 120-127.
- O'BRIEN, M. 2023. Aids to the examination of the peripheral nervous system: 6th edition. *Pract Neurol*, 23, 263-264.
- O'DONNELL, L. F., CORTESE, A., ROSSOR, A. M., LAURA, M., BLAKE, J.,
 SKORUPINSKA, M., LUNN, M. P., THORNTON, J. S., CURRÒ, R., MORROW, J.
 M. & REILLY, M. M. 2022. Exploratory analysis of lower limb muscle MRI in a case series of patients with SORD neuropathy. *J Neurol Neurosurg Psychiatry*.
- O'DONNELL, L. F., PIPIS, M., THORNTON, J. S., KANBER, B., WASTLING, S., MCDOWELL, A., ZAFEIROPOULOS, N., LAURA, M., SKORUPINSKA, M., RECORD, C. J., DOHERTY, C. M., HERRMANN, D. N., ZETTERBERG, H., HESLEGRAVE, A. J., LABAN, R., ROSSOR, A. M., MORROW, J. M. & REILLY, M. M. 2023. Quantitative MRI outcome measures in CMT1A using automated lower limb muscle segmentation. *J Neurol Neurosurg Psychiatry*.
- OSTERNIG, L. R. 1986. Isokinetic dynamometry: implications for muscle testing and rehabilitation. *Exerc Sport Sci Rev*, 14, 45-80.
- ÕUNPUU, S., GARIBAY, E., ACSADI, G., BRIMACOMBE, M. & PIERZ, K. 2021. The impact of orthoses on gait in children with Charcot-Marie-Tooth disease. *Gait & Posture*, 85, 198-204.
- OUNPUU, S., GARIBAY, E., SOLOMITO, M., BELL, K., PIERZ, K., THOMSON, J., ACSADI, G. & DELUCA, P. 2013. A comprehensive evaluation of the variation in ankle function during gait in children and youth with Charcot-Marie-Tooth disease. *Gait Posture*, 38, 900-6.
- PADUA, L., PAZZAGLIA, C., SCHENONE, A., FERRARO, F., BIROLI, A., ESPOSITO, C. & PAREYSONV, D. 2014. Rehabilitation for Charcot Marie tooth: a survey study of patients and familiar/caregiver perspective and perception of efficacy and needs. *Eur J Phys Rehabil Med*, 50, 25-30.
- PERRIN, D. H. 1994. Isokinetic Exercise and Assessment David H. Perrin. *Medicine and science in sports and exercise*, 26, 118.
- PERRY, J. & BURNFIELD, J. M. 1992. *Gait Analysis: Normal and Pathological Function*, USA, SLACKIncorporated.

- PHILLIPS, M., RADFORD, K. & WILLS, A. 2011. Ankle foot orthoses for people with Charcot Marie Tooth disease – views of users and orthotists on important aspects of use. *Disability and Rehabilitation: Assistive Technology*, 6, 491-499.
- PISCOSQUITO, G., REILLY, M. M., SCHENONE, A., FABRIZI, G. M., CAVALLARO, T., SANTORO, L., VITA, G., QUATTRONE, A., PADUA, L., GEMIGNANI, F., VISIOLI, F., LAURA, M., CALABRESE, D., HUGHES, R. A., RADICE, D., SOLARI, A. & PAREYSON, D. 2014. Is overwork weakness relevant in Charcot-Marie-Tooth disease? J Neurol Neurosurg Psychiatry, 85, 1354-8.
- PONS, N., FERNÁNDEZ-EULATE, G., PEGAT, A., THÉAUDIN, M., GUIEU, R., RIPELLINO, P., DEVEDJIAN, M., MACE, P., MASINGUE, M., LÉONARD-LOUIS, S., PETIOT, P., ROCHE, P., BERNARD, E., BOUHOUR, F., GOOD, J. M., VERSCHUEREN, A., GRAPPERON, A. M., SALORT, E., GROSSET, A., CHANSON, J. B., NADAJ-PAKLEZA, A., BÉDAT-MILLET, A. L., CHOUMERT, A., BARNIER, A., HAMDI, G., LESCA, G., PRIEUR, F., BRUNEEL, A., LATOUR, P., STOJKOVIC, T., ATTARIAN, S. & BONELLO-PALOT, N. 2023. SORD-related peripheral neuropathy in a French and Swiss cohort: Clinical features, genetic analyses, and sorbitol dosages. *Eur J Neurol*, 30, 2001-2011.
- PRADA, V., LAURÀ, M., ZUCCARINO, R., REILLY, M. M. & SHY, M. E. 2022. Virtual Charcot-Marie-Tooth Examination Score: A Validated Virtual Evaluation for People With Charcot-Marie-Tooth Disease. *Neurol Clin Pract*, 12, e98-e104.
- PRADA, V., MORI, L., ACCOGLI, S., RIVAROLA, M., SCHIZZI, S., HAMEDANI, M. & SCHENONE, A. 2018. Testing overwork weakness in Charcot-Marie-tooth disease: Is it true or false? *Journal of the Peripheral Nervous System*, 23, 124-128.
- RAMDHARRY, G. M., DAY, B. L., REILLY, M. M. & MARSDEN, J. F. 2009. Hip flexor fatigue limits walking in Charcot-Marie-Tooth disease. *Muscle Nerve*, 40, 103-11.
- RAMDHARRY, G. M., DAY, B. L., REILLY, M. M. & MARSDEN, J. F. 2012a. Foot drop splints improve proximal as well as distal leg control during gait in Charcot-Marie-Tooth Disease. *Muscle & Nerve*, 46, 512-519.
- RAMDHARRY, G. M., POLLARD, A., ANDERSON, C., LAURÁ, M., MURPHY, S. M., DUDZIEC, M., DEWAR, E. L., HUTTON, E., GRANT, R. & REILLY, M. M. 2014.
 A pilot study of proximal strength training in Charcot-Marie-Tooth disease. *Journal* of the Peripheral Nervous System, 19, 328-332.
- RAMDHARRY, G. M., POLLARD, A. J., MARSDEN, J. F. & REILLY, M. M. 2012b. Comparing Gait Performance of People with Charcot-Marie-Tooth Disease Who Do and Do Not Wear Ankle Foot Orthoses. *Physiotherapy Research International*, 17, 191-199.
- RAMDHARRY, G. M., REILLY-O'DONNELL, L., GRANT, R. & REILLY, M. M. 2018. Frequency and circumstances of falls for people with Charcot-Marie-Tooth disease: A cross sectional survey. *Physiotherapy Research International*, 23, e1702.
- RAMDHARRY, G. M., THORNHILL, A., MEIN, G., REILLY, M. M. & MARSDEN, J. F. 2012c. Exploring the experience of fatigue in people with Charcot–Marie–Tooth disease. *Neuromuscular Disorders*, 22, S208-S213.
- RECORD, C. J., PIPIS, M., SKORUPINSKA, M., BLAKE, J., POH, R., POLKE, J. M., EGGLETON, K., NANJI, T., ZUCHNER, S., CORTESE, A., HOULDEN, H., ROSSOR, A. M., LAURA, M. & REILLY, M. M. 2024. Whole genome sequencing increases the diagnostic rate in Charcot-Marie-Tooth disease. *Brain*, awae064.
- REDMOND, A. C., CROSBIE, J. & OUVRIER, R. A. 2006. Development and validation of a novel rating system for scoring standing foot posture: The Foot Posture Index. *Clinical Biomechanics*, 21, 89-98.

- REYNAUD, V., MOREL, C., GIVRON, P., CLAVELOU, P., CORNUT-CHAUVINC, C., PEREIRA, B., TAITHE, F. & COUDEYRE, E. 2019. Walking Speed Is Correlated With the Isokinetic Muscular Strength of the Knee in Patients With Charcot-Marie-Tooth Type 1A. American Journal of Physical Medicine & Rehabilitation, 98, 422-425.
- ROBERTS-CLARKE, D., FORNUSEK, C., FIATARONE SINGH, M. A., BURNS, J. & HACKETT, D. A. 2016a. Examining hand dominance using dynamometric grip strength testing as evidence for overwork weakness in Charcot-Marie-Tooth disease: a systematic review and meta-analysis. *Int J Rehabil Res*, 39, 189-96.
- ROBERTS-CLARKE, D., FORNUSEK, C., SAIGAL, N., HALAKI, M., BURNS, J., NICHOLSON, G., FIATARONE SINGH, M. & HACKETT, D. 2016b. Relationship between physical performance and quality of life in Charcot-Marie-Tooth disease: a pilot study. *Journal of the Peripheral Nervous System*, 21, 357-364.
- ROBERTSON DGE, GRAHAM E, CALDWELL GE, HAMILL J, KAMEN G & SN, W. 2014a. *Research Methods in Biomechanics*, USA, Human Kinetics: Champaign.
- ROBERTSON DGE, GRAHAM E, CALDWELL GE, HAMILL J, KAMEN G & SN, W. 2014b. Three-Dimensional Kinematics. *Research Methods in Biomechanics*. 2nd ed. USA: Human Kinetics.
- ROBERTSON DGE, GRAHAM E, CALDWELL GE, HAMILL J, KAMEN G & SN, W. 2014c. Three-Dimensional Kinetics. *Research Methods in Biomechanics*. 2nd ed. USA: Human Kinetics.
- ROSE, J. & GAMBLE, J. G. 2006. *Human walking*, Philidelphia, Lippincott Williams and Wilkins.
- ROSSOR, A. M., DAVIDSON, G. L., BLAKE, J., POLKE, J. M., MURPHY, S. M., HOULDEN, H., INNES, A., KALMAR, B., GREENSMITH, L. & REILLY, M. M. 2012a. A novel p.Gln175X [corrected] premature stop mutation in the C-terminal end of HSP27 is a cause of CMT2. *J Peripher Nerv Syst*, 17, 201-5.
- ROSSOR, A. M., KALMAR, B., GREENSMITH, L. & REILLY, M. M. 2012b. The distal hereditary motor neuropathies. *J Neurol Neurosurg Psychiatry*, 83, 6-14.
- ROSSOR, A. M., MORROW, J. M., POLKE, J. M., MURPHY, S. M., HOULDEN, H., LAURA, M., MANJI, H., BLAKE, J. & REILLY, M. M. 2017. Pilot phenotype and natural history study of hereditary neuropathies caused by mutations in the HSPB1 gene. *Neuromuscul Disord*, 27, 50-56.
- SHY, M. E., BLAKE, J., KRAJEWSKI, K., FUERST, D. R., LAURA, M., HAHN, A. F., LI, J., LEWIS, R. A. & REILLY, M. 2005. Reliability and validity of the CMT neuropathy score as a measure of disability. *Neurology*, 64, 1209-14.
- SIMES, R. J. 1986. An improved Bonferroni procedure for multiple tests of significance. *Biometrika*, 73, 751-754.
- SMAN, A. D., HACKETT, D., FIATARONE SINGH, M., FORNUSEK, C., MENEZES, M. P. & BURNS, J. 2015. Systematic review of exercise for Charcot-Marie-Tooth disease. *Journal of the Peripheral Nervous System*, 20, 347-362.
- SMITH, C. A., CHETLIN, R. D., GUTMANN, L., YEATER, R. A. & ALWAY, S. E. 2006. Effects of exercise and creatine on myosin heavy chain isoform composition in patients with Charcot–Marie–Tooth disease. *Muscle & Nerve*, 34, 586-594.
- STERGIOU, N. 2020. Biomechanics and gait analysis, London, Academic Press.
- SUMNER, C. J., D'YDEWALLE, C., WOOLEY, J., FAWCETT, K. A., HERNANDEZ, D., GARDINER, A. R., KALMAR, B., BALOH, R. H., GONZALEZ, M., ZÜCHNER, S., STANESCU, H. C., KLETA, R., MANKODI, A., CORNBLATH, D. R., BOYLAN, K. B., REILLY, M. M., GREENSMITH, L., SINGLETON, A. B., HARMS, M. B., ROSSOR, A. M. & HOULDEN, H. 2013. A dominant mutation in
FBXO38 causes distal spinal muscular atrophy with calf predominance. *American journal of human genetics*, 93, 976-983.

- TANABE, H., HIGUCHI, Y., YUAN, J. H., HASHIGUCHI, A., YOSHIMURA, A., ISHIHARA, S., NOZUMA, S., OKAMOTO, Y., MATSUURA, E., ISHIURA, H., MITSUI, J., TAKASHIMA, R., KOKUBUN, N., MAEDA, K., ASANO, Y., SUNAMI, Y., KONO, Y., ISHIGAKI, Y., YANAMOTO, S., FUKAE, J., KIDA, H., MORITA, M., TSUJI, S. & TAKASHIMA, H. 2018. Clinical and genetic features of Charcot-Marie-Tooth disease 2F and hereditary motor neuropathy 2B in Japan. J Peripher Nerv Syst, 23, 40-48.
- THISTLE, H. G., HISLOP, H. J., MOFFROID, M. & LOWMAN, E. W. 1967. Isokinetic contraction: a new concept of resistive exercise. *Arch Phys Med Rehabil*, 48, 279-82.
- TIFFREAU, V., LEDOUX, I., EYMARD, B., THÉVENON, A. & HOGREL, J.-Y. 2007. Isokinetic muscle testing for weak patients suffering from neuromuscular disorders: A reliability study. *Neuromuscular Disorders*, 17, 524-531.
- VALERO-CUEVAS, F. J. 2016. Fundamentals of Neuromechanics, London, UK, Springer-Verlag London.
- VAN DER LINDEN, M. H., DE KAM, D., VAN ENGELEN, B. G., HENDRICKS, H. T. & DUYSENS, J. 2009. Fast responses to stepping on an unexpected surface height depend on intact large-diameter nerve fibers: a study on Charcot-Marie-Tooth type 1A disease. *J Neurophysiol*, 102, 1684-98.
- VAN DER WILK, D., REINTS, R., POSTEMA, K., GORT, T., HARLAAR, J., HIJMANS, J. M. & VERKERKE, G. J. 2018. Development of an Ankle-Foot Orthosis That Provides Support for Flaccid Paretic Plantarflexor and Dorsiflexor Muscles. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, 26, 1036-1045.
- VAUGHAN, C. L. 1999. Dynamics of Human Gait, Human Kinetics Publishers.
- VERVERIS, A., DAJANI, R., KOUTSOU, P., ALOQAILY, A., NELSON-WILLIAMS, C., LORING, E., ARAFAT, A., MUBAIDIN, A. F., HORANY, K., BADER, M. B., AL-BAHO, Y., ALI, B., MUHTASEB, A., DESPENZA, T., JR., AL-QUDAH, A. A., MIDDLETON, L. T., ZAMBA-PAPANICOLAOU, E., LIFTON, R. & CHRISTODOULOU, K. 2019. Distal hereditary motor neuronopathy of the Jerash type is caused by a novel SIGMAR1 c.500A>T missense mutation. *J Med Genet*.
- VINCI, P. 2001. Strengthening of the proximal muscles in Charcot-Marie-Tooth disease. *Arch Phys Med Rehabil*, 82, 563.
- VINCI, P. & GARGIULO, P. 2008. Poor compliance with ankle-foot-orthoses in Charcot-Marie-Tooth disease. *Eur J Phys Rehabil Med*, 44, 27-31.
- VINCI, P. & PERELLI, S. L. 2002. Footdrop, foot rotation, and plantarflexor failure in Charcot-Marie-Tooth disease. *Arch Phys Med Rehabil*, 83, 513-6.
- VINCI, P., PERELLI, S. L. & GARGIULO, P. 2009. About the hypothesis of overwork weakness in Charcot-Marie-Tooth disease. *J Rehabil Med*, 41, 778; author reply 779.
- VITA, G., LA FORESTA, S., RUSSO, M., VITA, G. L., MESSINA, S., LUNETTA, C. & MAZZEO, A. 2016. Sport activity in Charcot-Marie-Tooth disease: A case study of a Paralympic swimmer. *Neuromuscul Disord*, 26, 614-8.
- VU, H., GOMEZ, F., CHERELLE, P., LEFEBER, D., NOWÉ, A. & VANDERBORGHT, B. 2018. ED-FNN: A New Deep Learning Algorithm to Detect Percentage of the Gait Cycle for Powered Prostheses. *Sensors*, 18, 2389.
- WALLACE, A., PIETRUSZ, A., DEWAR, E., DUDZIEC, M., JONES, K., HENNIS, P., STERR, A., BAIO, G., MACHADO, P. M., LAURA, M., SKORUPINSKA, I., SKORUPINSKA, M., BUTCHER, K., TRENELL, M., REILLY, M. M., HANNA, M. G. & RAMDHARRY, G. M. 2019. Community exercise is feasible for

neuromuscular diseases and can improve aerobic capacity. *Neurology*, 92, e1773-e1785.

- WANG, B., LI, X., HUANG, S., ZHAO, H., LIU, J., HU, Z., LIN, Z., LIU, L., XIE, Y., JIN, Q., ZHAO, H., TANG, B., NIU, Q. & ZHANG, R. 2019. A novel WARS mutation (p.Asp314Gly) identified in a Chinese distal hereditary motor neuropathy family. *Clinical Genetics*, 96, 176-182.
- WEBBER, S. C. & PORTER, M. M. 2010. Reliability of ankle isometric, isotonic, and isokinetic strength and power testing in older women. *Phys Ther*, 90, 1165-75.
- WHITTLE, M. 2007. *Gait analysis: an introduction*, University of Michigan: Butterworth-Heinemann.
- WILLIS, T. A., HOLLINGSWORTH, K. G., COOMBS, A., SVEEN, M. L., ANDERSEN, S., STOJKOVIC, T., EAGLE, M., MAYHEW, A., DE SOUSA, P. L., DEWAR, L., MORROW, J. M., SINCLAIR, C. D., THORNTON, J. S., BUSHBY, K., LOCHMÜLLER, H., HANNA, M. G., HOGREL, J. Y., CARLIER, P. G., VISSING, J. & STRAUB, V. 2013. Quantitative muscle MRI as an assessment tool for monitoring disease progression in LGMD2I: a multicentre longitudinal study. *PLoS One*, 8, e70993.
- WINTER, D. A. 2009. *Biomechanics and Motor Control of Human Movement*, Hoboken, New Jersey, USA, John Wiley & Sons, Inc.
- WOJCIECHOWSKI, E. A., CHENG, T. L., HOGAN, S. M., MUDGE, A. J., BALASSONE, D., MENEZES, M. P., LITTLE, D. G., DWAN, L. N. & BURNS, J. 2022.
 Replicating and redesigning ankle-foot orthoses with 3D printing for children with Charcot-Marie-Tooth disease. *Gait Posture*, 96, 73-80.
- WU, C., XIANG, H., CHEN, R., ZHENG, Y., ZHU, M., CHEN, S., YU, Y., PENG, Y., YU,
 Y., DENG, J., ZHOU, M. & HONG, D. 2022. Genetic spectrum in a cohort of patients with distal hereditary motor neuropathy. *Ann Clin Transl Neurol*, 9, 633-643.
- YANG, J., LEE, J., LEE, B., KIM, S., SHIN, D., LEE, Y., LEE, J., HAN, D. & CHOI, S. 2014. The effects of elbow joint angle changes on elbow flexor and extensor muscle strength and activation. *J Phys Ther Sci*, 26, 1079-82.
- YUSHKEVICH, P. A., PIVEN, J., HAZLETT, H. C., SMITH, R. G., HO, S., GEE, J. C. & GERIG, G. 2006. User-guided 3D active contour segmentation of anatomical structures: Significantly improved efficiency and reliability. *NeuroImage*, 31, 1116-1128.
- ZOU, D., HE, T., DAILEY, M., SMITH, K. E., SILVA, M. J., SINACORE, D. R., MUELLER, M. J. & HASTINGS, M. K. 2014. Experimental and computational analysis of composite ankle-foot orthosis. *J Rehabil Res Dev*, 51, 1525-36.
- ZUCCARINO, R., ANDERSON, K. M., SHY, M. E. & WILKEN, J. M. 2021. Satisfaction with ankle foot orthoses in individuals with Charcot-Marie-Tooth disease. *Muscle & Nerve*, 63, 40-45.

Appendix I: Difference Between Right and Left Side in The Primary Parameters and Application of the Modified Bonferroni Correction

Introduction

Distal hereditary motor neuropathy lays under the CMT umbrella which is characterized by its symmetrical nature. The impact of the COVID-19 pandemic significantly influenced the recruitment rate for this project, resulting in a smaller sample size that may not accurately reflect the population. Additionally, smaller sample sizes have a higher risk of deviating from the norm due to chance alone. Therefore, right and left symmetry of the primary parameters in our sample had to be ascertained before proceeding with the analysis.

Methods

To ascertain symmetry in our sample, difference between right and left side in the primary parameters was tested using unpaired T.Test for normally distributed data and Mann-Whitney test for non-normally distributed data. Consideration to correction for the multiple comparisons to decrease the chance of type I and type II error was given using modified Bonferroni procedure. Pelvic segment is not independent left to right due to the structure of the pelvic ring therefore differences at the pelvic level will not be considered. The modified Bonferroni correction (Simes, 1986) is based on the ordered Pvalues of individual tests. $T_1, ..., T_n$ is a set of statistics with corresponding p-values $P_1, ..., P_n$ for testing hypothesis $H_1, ..., H_n$. The p-values were ordered as $P_{(1)}, ..., P_{(n)}$ to test the hypothesis H0=[H(1),...H(n)]. The null hypothesis H_0 was rejected if $P_{(j)} \leq j\alpha/n$ for any j = 1, ..., n. (Simes, 1986).

Results:

The test showed no significant difference in all primary parameters with modified Bonferroni correction for multiple testing except in pelvis angle at Z axis. However, the pelvic segment is not independent left to right due to the structure of the pelvic ring. Moreover, this difference could be caused by the lost visibility due to arm swing during walking. Test results and application of modified Bonferroni correction are shown in the table below.

Test rank	Fat Fraction Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome
1	Adductor Magnus	0.6994	0.0028	ACCEPT Ho
2	Biceps Femoris	0.5619	0.0056	ACCEPT Ho
3	Femur	0.7477	0.0083	ACCEPT Ho
4	Gracilis	0.5619	0.0111	ACCEPT Ho
5	Lateral Gastrocnemius	0.7969	0.0139	ACCEPT Ho
6	Medial Gastrocnemius	0.8731	0.0167	ACCEPT Ho
7	Peroneus Longus	0.7477	0.0194	ACCEPT Ho
8	Rectus Femoris	0.9487	0.0222	ACCEPT Ho
9	Soleus	0.7231	0.0250	ACCEPT Ho
10	Sartorius	0.6522	0.0278	ACCEPT Ho
11	Semimembranosus	0.3653	0.0306	ACCEPT Ho
12	Semitendinosus	0.7477	0.0333	ACCEPT Ho
13	Tibialis Anterior	0.9487	0.0361	ACCEPT Ho
14	Tibia	0.7477	0.0389	ACCEPT Ho

15	Tibialis Posterior	0.7477	0.0417	ACCEPT Ho
16	Vastus Intermedius	0.7477	0.0444	ACCEPT Ho
17	Vastus Lateralis	0.7477	0.0472	ACCEPT Ho
18	Vastus Medialis	0.519	0.0500	ACCEPT Ho
Test rank	Volume Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome
1	Adductor Magnus	0.7917	0.0028	ACCEPT Ho
2	Biceps Femoris	0.6306	0.0056	ACCEPT Ho
3	Femur	0.3851	0.0083	ACCEPT Ho
4	Gracilis	0.8442	0.0111	ACCEPT Ho
5	Lateral Gastrocnemius	0.9487	0.0139	ACCEPT Ho
6	Medial Gastrocnemius	0.9516	0.0167	ACCEPT Ho
7	Peroneus Longus	0.914	0.0194	ACCEPT Ho
8	Rectus Femoris	0.47	0.0222	ACCEPT Ho
9	Soleus	0.9859	0.0250	ACCEPT Ho
10	Sartorius	0.4672	0.0278	ACCEPT Ho
11	Semimembranosus	0.9883	0.0306	ACCEPT Ho
12	Semitendinosus	0.8641	0.0333	ACCEPT Ho
13	Tibialis Anterior	0.9294	0.0361	ACCEPT Ho
14	Tibia	0.9218	0.0389	ACCEPT Ho
15	Tibialis Posterior	0.9563	0.0417	ACCEPT Ho
16	Vastus Intermedius	0.912	0.0444	ACCEPT Ho
17	Vastus Lateralis	0.9367	0.0472	ACCEPT Ho
18	Vastus Medialis	0.7932	0.0500	ACCEPT Ho
Test rank	Kinematic Gait Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome
1	Ankle Max	0.8205	0.0042	ACCEPT Ho
2	Ankle Min	0.9487	0.0083	ACCEPT Ho
3	Hip Max	0.7756	0.0125	ACCEPT Ho
4	Hip Min	0.6225	0.0167	ACCEPT Ho
5	Knee Max	0.875	0.0208	ACCEPT Ho
6	Knee Min	0.5888	0.0250	ACCEPT Ho
7	Pelvis X Max	0.5642	0.0292	ACCEPT Ho
8	Pelvis X Min	0.3383	0.0333	ACCEPT Ho
9	Pelvis Y Max	0.7969	0.0375	ACCEPT Ho
10	Pelvis Y Min	0.9984	0.0417	ACCEPT Ho
11	Pelvis Z Max	0.0073	0.0458	REJECT Ho
12	Pelvis Z Min	0.0192	0.0500	REJECT Ho
Test rank	Kinetic Gait Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome

1	Ankle Moment Y Max	0.4385	0.0033	ACCEPT Ho
2	Ankle Moment Y Min	0.9487	0.0067	ACCEPT Ho
3	Ankle Power Max During Stance	0.4385	0.0100	ACCEPT Ho
4	Ankle Power Max During Swing	0.4779	0.0133	ACCEPT Ho
5	Ankle Power Min During Stance	0.519	0.0167	ACCEPT Ho
6	Hip Moment Y Max	0.6882	0.0200	ACCEPT Ho
7	Hip Moment Y Min	0.7969	0.0233	ACCEPT Ho
8	Hip Power Max During Stance	0.9955	0.0267	ACCEPT Ho
9	Hip Power Max During Swing	0.8133	0.0300	ACCEPT Ho
10	Hip Power Min During Stance	0.8977	0.0333	ACCEPT Ho
11	Knee Moment Y Max	0.6063	0.0367	ACCEPT Ho
12	Knee Moment Y Min	0.4457	0.0400	ACCEPT Ho
13	Knee Power Max During Stance	0.5527	0.0433	ACCEPT Ho
14	Knee Power Max During Swing	0.6935	0.0467	ACCEPT Ho
15	Knee Power Min During Stance	0.4779	0.0500	ACCEPT Ho
Test rank	Spatiotemporal Gait Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome
1	Double Support Time	0.66	0.00417	ACCEPT Ho
2	Percent Opposite Foot Contact	0.8714	0.00833	ACCEPT Ho
3	Percent Opposite Toe Off	0.3316	0.01250	ACCEPT Ho
4	Percent Stance	0.7676	0.01667	ACCEPT Ho
5	Single Support Time	0.5113	0.02083	ACCEPT Ho
6	Speed	0.9384	0.02500	ACCEPT Ho
7	Step Length	0.9177	0.02917	ACCEPT Ho
8	Steps Per Minute	0.9487	0.03333	ACCEPT Ho
9	Step Time	0.8955	0.03750	ACCEPT Ho
10	Stride Length	0.9193	0.04167	ACCEPT Ho
11	Strides Per Minute	0.9487	0.04583	ACCEPT Ho
12	Stride Time	0.9215	0.05000	ACCEPT Ho
Test rank	Isometric dynamometry Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome
1	Ankle ISOM DF 10°	0.8935	0.00556	ACCEPT Ho
2	Ankle ISOM DF 30°	0.7479	0.01111	ACCEPT Ho
3	Ankle_ISOM_PF_10°	0.6932	0.01667	ACCEPT Ho
4	Hip ISOM EX 45°	0.9783	0.02222	ACCEPT Ho
5	Hip ISOM FL 45°	0.7289	0.02778	ACCEPT Ho
6	Knee_ISOM_EX_45°	0.975	0.03333	ACCEPT Ho
7	Knee ISOM EX 90°	0.8972	0.03889	ACCEPT Ho
8	Knee ISOM FL 45°	0.9499	0.04444	ACCEPT Ho
9	Knee_ISOM_FL_90°	0.9873	0.05000	ACCEPT Ho

Test rank	Isokinetic dynamometry Variables	P-value	Modified Bonferroni Threshold	Modified Bonferroni Outcome
1	Ankle ISOK DF 60°/60°	0.3905	0.0063	ACCEPT Ho
2	Ankle ISOK PF 60°/60°	0.6442	0.0125	ACCEPT Ho
3	Hip ISOK EX 60°/60s	0.9607	0.0188	ACCEPT Ho
4	Hip ISOK FL 60°/60s	0.8906	0.0250	ACCEPT Ho
5	Knee ISOK EX 120°/120s	0.7929	0.0313	ACCEPT Ho
6	Knee ISOK EX 60°/60s	0.886	0.0375	ACCEPT Ho
7	Knee ISOK FL 120°/120s	0.7352	0.0438	ACCEPT Ho
8	Knee ISOK FL 60°/60s	0.8236	0.0500	ACCEPT Ho

Conclusion

The DHMN sample recruited for this project showed symmetry between right and left side, therefore, the decision was made to analyse the primary and secondary parameters for each objective using the average of both sides rather than analysing both sides separately to eliminate type 1 error caused by multiple comparisons.

Appendix II: Friedman Test Manual Calculation

Introduction

The Friedman test is a non-parametric statistical test used for analysing repeated measures data. It is mainly used when the assumptions of normality and homogeneity of variances are not met, making it an alternative to repeated measures ANOVA. The analysis of variance (ANOVA) tests the extent to which the measured values of the dependent sample differ. However, The Friedman test uses ranks rather than the actual measured values which make it a valid tool for analysing ordinal data (Friedman, 1940, Friedman, 1937). The null hypothesis for Friedman test is there is no significant difference between the dependent groups. And the alternative hypothesis is there is a significant difference between the dependent groups (Conover, 1999).

Analysis of the longitudinal MRI Muscle Fat qualitative score at the foot level with Modified Mercuri's Scale of DHMN using statistical software (© 2024 MedCalc Software Ltd) showed significant difference although the pairwise analysis showed no difference across all measurement points (chapter 6). This manual calculation is aimed to confirm the false positive type I error occurred in the analysis.

Manual Calculation Procedure

<u>A. Data Set and Ranking the Values:</u> The point in time where a participant has the highest value gets rank 1, the point in time with the second highest value gets rank 2 and the point in time with the smallest value gets rank 3. To ensure correct handling of ties (when values in two or more time points are the same), the average rank is used. This process is repeated for all rows. Afterwards the ranks of each single point of time are combined.

Since all values for each participant are the same at all measurement points, they receive the average rank. For three measurements, the average rank would be:

	Baseline			6 Month	s	12 Months			
Mercuri	Rank	Average	Mercuri	Rank	Average	Mercuri	Rank	Average	
Score		rank	Score		rank	Score		rank	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
6	1	2	6	2	2	6	3	2	
Sum rank	8	16	Sum rank	16	16	Sum rank	24	16	

Average rank =
$$\frac{1+2+3}{3} = 2$$

Since each time point (k=3) in every participant (n=8) has the rank of 2, the sum of ranks for each time point (R_j) would be:

$$R_1 = 2 \times 8 = 16$$
, $R_2 = 2 \times 8 = 16$, $R_3 = 2 \times 8 = 16$

<u>B. Compute the Test Statistic:</u> The test statistic for the Friedman test is calculated using the formula:

$$x^{2} = \frac{12}{N \cdot k \cdot (k+1)} \cdot \sum R_{j}^{2} - 3 \cdot N \cdot (k+1)$$

where:

n is the number of blocks (participants), *k* is the number of treatments (time points), R_j is the sum of ranks for the *j*-th treatment.

The test statistic χ^2 is then compared to the critical value from the chi-square distribution with *k*-1degrees of freedom to determine the p-value.

Plugging in the values to calculate χ^2 :

n=8, k =3, R₁=16, R₂=16, R₃=16

$$\chi^2 = 12/8 \times 3 \times 4(16^2 + 16^2 + 16^2) - 3 \times 8 \times 4$$

 $\chi^2 = 12/96(256 + 256 + 256) - 96$
 $\chi^2 = 12/96 \times 768 - 96$
 $\chi^2 = 9216/96 - 96$
 $\chi^2 = 96 - 96$
 $\chi^2 = 0$

In this specific case, because all the values are the same across all time points and participants, the calculated Friedman test statistic (χ^2) is 0. This result indicates that there are no differences between the treatments, as expected.

<u>C. Calculate the P-Value:</u>

1.Determine the Chi-Square Test Statistic (χ^2):

Based on the previous calculation, $\chi^2 = 0$

2.Determine the Degrees of Freedom (*df*):

$$df = k - 1 = 3 - 1 = 2.$$

3.Using the Cumulative Distribution Function (CDF):

The *p*-value for a chi-square statistic is the area under the chi-square distribution curve to the right of the given test statistic. For $\chi^2=0$ with any degrees of freedom, this area is essentially the entire area under the curve since the chi-square distribution starts at 0 and is always positive.

For a chi-square statistic of 0, the cumulative probability up to 0 is 0. Therefore, the *p*-value is 1, which indicates that a test statistic as extreme as or more extreme than 0 would occur with certainty under the null hypothesis.

4. Formula and Calculation:

The cumulative distribution function (CDF) for the chi-square distribution at 0 is:

$$P(\chi^2 \leq 0)$$

Given that the chi-square distribution is non-negative, the probability that a chisquare statistic is less than or equal to 0 is 0. Thus, the p-value is:

$$p$$
-value=1- $P(\chi^2 \le 0)$ =1-0=1

For a chi-square statistic of 0 with 2 degrees of freedom, the p-value is 1. This indicates that observing a test statistic of 0 or more extreme under the null hypothesis is certain, which makes sense since a chi-square statistic of 0 indicates no deviation from the expected values.

Conclusion:

In this specific case, because all the values are the same across all time points and participants, the manually calculated Friedman test statistic is 0. This result indicates that there are no differences in Mercuri score between the time points, as expected. When all values are the same for all time points, significant *p*-value suggests an issue with the implementation or the interpretation of the test. The Friedman test statistic calculation using statistical software in this scenario might leads to an invalid result such as type 1 error. Small sample size, outliers, and inadequate handling of ties are possible explanations for type 1 error in Fridman test (Hollander, 2013, Conover, 1999).

Appendix III: Correlograms Showing the Correlation and

Significance Level for All Parameters Comparisons in Chapter 7

Relationships Between Intramuscular Fat Fraction, Muscle Volume, Isokinetic and Isometric Muscle Strength, And Kinetics of Gait

Plantar Flexors

	Ankle ISO M PF	PF CSA	PF RMA	Ankle Moment Y Max	Ankle Power Max During Stance	Ankle ISO K PF	Ankle Power Max During Swing	Ankle Power Min During Stance	Ankle Moment Y Min	PF FF
Aplelo ISOM		0.827	0.754	0.413	0.391	0.56	0.343	0.161	0.05	-0.4
PF	1	P=0.0031	P=0.0118	P=0.2359	P=0.2645	P=0.0926	P=0.3314	P=0.6564	P=0.8911	P=0.2516
		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.827		0.843	0.503	0.275	0.368	0.017	-0.001	-0.11	-0.375
PF CSA	P=0.0031	1	P=0.0022	P=0.1382	P=0.4420	P=0.2948	P=0.9620	P=0.9971	P=0.7626	P=0.2852
	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.754	0.843		0.391	0.361	0.293	0.059	-0.31	-0.108	-0.791
PF RMA	P=0.0118	P=0.0022	1	P=0.2643	P=0.3059	P=0.4112	P=0.8705	P=0.3827	P=0.7672	P=0.0065
	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10
Ankle	0.413	0.503	0.391		0.547	0.599	0.487	0.472	0.413	-0.27
Moment Y	P=0.2359	P=0.1382	P=0.2643	1	P=0.1016	P=0.0671	P=0.1529	P=0.1680	P=0.2353	P=0.4513
Max	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10
Ankle Power	0.391	0.275	0.361	0.547		0.383	0.662	0.259	0.168	-0.539
Max During	P=0.2645	P=0.4420	P=0.3059	P=0.1016	1	P=0.2745	P=0.0372	P=0.4703	P=0.6429	P=0.1077
Stance	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10
Ankle ISOK P	0.596	0.529	0.426	0.778	0.389		0.401	0.292	0.389	-0.419
F	P=0.0692	P=0.1160	P=0.2202	P=0.0080	P=0.2665	1	P=0.2505	P=0.4133	P=0.2665	P=0.2276
	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10
Ankle Power	0.343	0.017	0.059	0.487	0.662	0.41		0.64	0.323	-0.187
Max During	P=0.3314	P=0.9620	P=0.8705	P=0.1529	P=0.0372	P=0.2388	1	P=0.0464	P=0.3621	P=0.6051
Swing	n=10	n=10	n=10	n=10	n=10	n=10	0.64	n=10	n=10	n=10
Ankle Power	0.161	-0.001	-0.31	0.472	0.259	0.162 D 0.5545	0.64		0.041	0.418
Min During	P=0.6564	P=0.9971	P=0.3827	P=0.1680	P=0.4703	P=0.6545	P=0.0464	L.	P=0.9101	P=0.2297
Stance	n=10	n=10	n=10	n=10	n=10	n=10	n=10	0.067	n=10	n=10
Ankle Manuart V	-0.06/	-0.018	0.03 D 0.0229	0.43 D 0.2145	0.042 D 0.0074	0.389 D 0.2665	0.230 D 0.5100	-0.06/		-0.139 D 0.7000
Min	P=0.0340	P=0.9002	P=0.9556	P=0.2143	P=0.9074	P=0.2003	P=0.3109	P=0.0340	1	P=0.7009
	0.4	0.375	0 701	0.27	0.530	0.178	0.187	0.418	0.053	<u>II=10</u>
DF FF	P=0.7	D=0.373	P=0.0065	P=0.27	P=0.337	P=0.6236	P=0.107	P=0 2207	P=0.8845	1
	n = 10	n = 10.2052	n = 10	n = 10.4515	n = 10	n = 10.0230	n = 10	n = 10.2297	n = 10.0043	1
	Ankle ISO M PF	PF CSA	PF RMA	Ankle Moment Y Max	Ankle Power Max During Stance	Ankle ISO K PF	Ankle Power Max During Swing	Ankle Power Min During Stance	Ankle Moment Y Min	PF FF



Table 78: Correlation between baseline plantar flexors measurements.

Dorsiflexors

	DF RMA	Ankle ISO M DF	DF CSA	Ankle ISO K DF	Ankle Moment Y Max	Ankle Power Max During Swing	Ankle Power Max During Stance	Ankle Power Min During Stance	Ankle Moment Y Min	DF FF
		0.764	0.981	0.53	0.381	0.615	0.222	0.355	0.326	-0.82
DF RMA	1	P=0.0102	P<0.0001	P=0.1153	P=0.2769	P=0.0587	P=0.5373	P=0.3147	P=0.3576	P=0.0037
		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
Andria ISOM	0.891		0.733	0.612	0.77	0.673	0.382	0.479	0.418	-0.867
DF	P=0.0005	1	P=0.0158	P=0.0600	P=0.0092	P=0.0330	P=0.2763	P=0.1615	P=0.2291	P=0.0012
	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.981	0.703		0.452	0.273	0.554	0.08	0.353	0.257	-0.719
DF CSA	P<0.0001	P=0.0235	1	P=0.1896	P=0.4450	P=0.0969	P=0.8256	P=0.3176	P=0.4734	P=0.0190
	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10
Ankle ISOK	0.53	0.703	0.452		0.591	0.299	0.386	0.25	0.445	-0.553
DF	P=0.1153	P=0.0235	P=0.1896	1	P=0.0720	P=0.4013	P=0.2706	P=0.4867	P=0.1975	P=0.0976
	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10
Ankle	0.381	0.666	0.273	0.591		0.487	0.547	0.472	0.413	-0.403
Moment Y	P=0.2769	P=0.0354	P=0.4450	P=0.0720	1	P=0.1529	P=0.1016	P=0.1680	P=0.2353	P=0.2477
Max	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10
Ankle Power	0.615	0.612	0.554	0.299	0.487		0.662	0.64	0.323	-0.592
Max During	P=0.0587	P=0.0598	P=0.0969	P=0.4013	P=0.1529	1	P=0.0372	P=0.0464	P=0.3621	P=0.0716
Swing	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10
Ankle Power	0.222	0.384	0.08	0.386	0.547	0.662		0.259	0.168	-0.595
Max During	P=0.5373	P=0.2731	P=0.8256	P=0.2706	P=0.1016	P=0.0372	1	P=0.4703	P=0.6429	P=0.0697
Stance	n=10	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10
Ankle Power	0.355	0.4	0.353	0.25	0.472	0.64	0.259		0.041	-0.07
Min During	P=0.3147	P=0.2516	P=0.3176	P=0.4867	P=0.1680	P=0.0464	P=0.4703	1	P=0.9101	P=0.8466
Stance	n=10	n=10	n=10	n=10	n=10	n=10	n=10		n=10	n=10
Ankle	0.43 D 0.2147	0.418	0.2	0.345 D. 0.2202	0.43	0.236	0.042 D.0.007.1	-0.067		-0.455
Moment Y	P=0.2145	P=0.2291	P=0.5/96	P=0.3282	P=0.2145	P=0.5109	P=0.9074	P=0.8548	1	P=0.1869
Min	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	0.455	n=10
DEEE	-0.867	-0.867	-0.6/3	-0.588	-0.612	-0.6/3	-0.527	-0.406	-0.455	
DFFF	P=0.0012	P=0.0012	P=0.0330	P=0.0739	P=0.0600	P=0.0330	P=0.11/3	P=0.2443	P=0.1869	1
	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	
	DF RMA	Ankle ISO M DF	DF CSA	Ankle ISO K DF	Ankle Moment Y Max	Ankle Power Max During Swing	Ankle Power Max During Stance	Ankle Power Min During Stance	Ankle Moment Y Min	DF FF

1 0 -1

Table 79: Correlation between baseline dorsiflexors measurements.

Knee Extensors

	Knee ISO M EX	Knee Moment Y Max	Knee ISO K EX	EX CSA	EX RMA	Knee Power Max During Swing	Knee Power Max During Stance	Knee Moment Y Min	EX FF	Knee Power Min During Stance
V. ISOM		0.902	0.982	0.855	0.844	0.566	0.5	0.43	-0.592	0.752
Knee ISOM	1	P=0.0004	P<0.0001	P=0.0016	P=0.0021	P=0.0881	P=0.1409	P=0.2149	P=0.0711	P=0.0121
LA		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
Knee	0.902		0.867	0.876	0.818	0.715	0.715	0.585	-0.507	0.872
Moment Y	P=0.0004	1	P=0.0012	P=0.0009	P=0.0038	P=0.0200	P=0.0202	P=0.0755	P=0.1350	P=0.0010
Max	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
Knoo ISOV	0.982	0.867		0.785	0.761	0.492	0.422	0.43	-0.48	0.754
FX	P<0.0001	P=0.0012	1	P=0.0071	P=0.0105	P=0.1490	P=0.2250	P=0.2154	P=0.1600	P=0.0118
127X	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.855	0.876	0.785		0.977	0.629	0.659	0.595	-0.695	0.865
EX CSA	P=0.0016	P=0.0009	P=0.0071	1	P<0.0001	P=0.0512	P=0.0384	P=0.0695	P=0.0257	P=0.0012
	n=10	n=10	n=10	,	n=10	n=10	n=10	n=10	n=10	n=10
	0.844	0.818	0.761	0.977		0.596	0.657	0.55	-0.812	0.754
EX RMA	P=0.0021	P=0.0038	P=0.0105	P<0.0001	1	P=0.0692	P=0.0390	P=0.0992	P=0.0044	P=0.0117
	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10
Knee Power	0.566	0.715	0.492	0.629	0.596		0.87	0.682	-0.251	0.584
Max During	P=0.0881	P=0.0200	P=0.1490	P=0.0512	P=0.0692	1	P=0.0011	P=0.0300	P=0.4848	P=0.0762
Swing	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10
Knee Power	0.5	0.715	0.422	0.659	0.657	0.87		0.835	-0.398	0.612
Max During	P=0.1409	P=0.0202	P=0.2250	P=0.0384	P=0.0390	P=0.0011	1	P=0.0026	P=0.2549	P=0.0599
Stance	n=10	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10
Knee	0.43	0.585	0.43	0.595	0.55	0.682	0.835		-0.116	0.728
Moment Y	P=0.2149	P=0.0755	P=0.2154	P=0.0695	P=0.0992	P=0.0300	P=0.0026	1	P=0.7487	P=0.0171
Min	n=10	n=10	n=10	n=10	n=10	n=10	n=10		n=10	n=10
	-0.592	-0.507	-0.48	-0.695	-0.812	-0.251	-0.398	0.116		-0.291
EX FF	P=0.0711	P=0.1350	P=0.1600	P=0.0257	P=0.0044	P=0.4848	P=0.2549	P=0.7487	1	P=0.4154
	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10		n=10
Knee Power	0.752	0.872	0.754	0.865	0.754	0.584	0.612	0.728	-0.291	
Min During	P=0.0121	P=0.0010	P=0.0118	P=0.0012	P=0.0117	P=0.0762	P=0.0599	P=0.0171	P=0.4154	
Stance	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	
	Knee ISO M EX	Knee Moment Y Max	Knee ISO K EX	EX CSA	EX RMA	Knee Power Max During Swing	Knee Power Max During Stance	Knee Moment Y Min	EX FF	Knee Power Min During Stance

1 0 -1

Table 80: Correlation between baseline knee extensors measurements.

Knee Flexors

	Knee ISO M FL	Knee Moment Y Max	FL RMA	Knee ISO K FL	FL CSA	Knee Power Max During Swing	Knee Power Max During Stance	Knee Moment Y Min	FL FF	Knee Power Min During Stance
W ROOM		0.856	0.918	0.935	0.897	0.492	0.519	0.529	-0.71	0.81
Knee ISOM	1	P=0.0016	P=0.0002	P=0.0001	P=0.0004	P=0.1488	P=0.1243	P=0.1160	P=0.0213	P=0.0045
FL		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
Knee	0.856		0.846	0.867	0.809	0.715	0.715	0.585	-0.783	0.872
Moment Y	P=0.0016	1	P=0.0021	P=0.0012	P=0.0046	P=0.0200	P=0.0202	P=0.0755	P=0.0074	P=0.0010
Max	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.918	0.846		0.831	0.99	0.4	0.526	0.501	-0.798	0.818
FL RMA	P=0.0002	P=0.0021	1	P=0.0029	P<0.0001	P=0.2520	P=0.1181	P=0.1404	P=0.0056	P=0.0039
	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10	n=10
Vnas ISOV	0.935	0.867	0.831		0.806	0.432	0.412	0.454	-0.649	0.792
FI	P=0.0001	P=0.0012	P=0.0029	1	P=0.0049	P=0.2124	P=0.2365	P=0.1874	P=0.0421	P=0.0064
FL	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10	n=10
	0.897	0.809	0.99	0.806		0.354	0.451	0.467	-0.735	0.829
FL CSA	P=0.0004	P=0.0046	P<0.0001	P=0.0049	1	P=0.3159	P=0.1904	P=0.1740	P=0.0155	P=0.0031
	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10	n=10
Knee Power	0.492	0.715	0.4	0.432	0.354		0.87	0.682	-0.356	0.584
Max During	P=0.1488	P=0.0200	P=0.2520	P=0.2124	P=0.3159	1	P=0.0011	P=0.0300	P=0.3120	P=0.0762
Swing	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10	n=10
Knee Power	0.519	0.715	0.526	0.412	0.451	0.87		0.835	-0.535	0.612
Max During	P=0.1243	P=0.0202	P=0.1181	P=0.2365	P=0.1904	P=0.0011	1	P=0.0026	P=0.1111	P=0.0599
Stance	n=10	n=10	n=10	n=10	n=10	n=10		n=10	n=10	n=10
Knee	0.529	0.585	0.501	0.454	0.467	0.682	0.835		-0.241	0.728
Moment Y	P=0.1160	P=0.0755	P=0.1404	P=0.1874	P=0.1740	P=0.0300	P=0.0026	1	P=0.5017	P=0.0171
Min	n=10	n=10	n=10	n=10	n=10	n=10	n=10		n=10	n=10
	-0.71	-0.783	-0.798	-0.649	-0.735	-0.356	-0.535	-0.241		-0.56
FL FF	P=0.0213	P=0.0074	P=0.0056	P=0.0421	P=0.0155	P=0.3120	P=0.1111	P=0.5017	1	P=0.0925
	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10		n=10
Knee Power	0.81	0.872	0.818	0.792	0.829	0.584	0.612	0.728	-0.56	
Min During	P=0.0045	P=0.0010	P=0.0039	P=0.0064	P=0.0031	P=0.0762	P=0.0599	P=0.0171	P=0.0925	1
Stance	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	n=10	
	Knee ISO M FL	Knee Moment Y Max	FL RMA	Knee ISO K FL	FL CSA	Knee Power Max During Swing	Knee Power Max During Stance	Knee Moment Y Min	FL FF	Knee Power Min During Stance

1 0 -1

Table 81: Correlation between baseline knee flexors measurements.

Between Clinical Measurements

	CMTES	WALK-12	FPI-6
		0.671	-0.358
CMTES	1	P=0.0337	P=0.3100
		n=10	n=10
	0.671		-0.494
WALK-12	P=0.0337	1	P=0.1470
	n=10		n=10
	-0.358	-0.494	
FPI-6	P=0.3100	P=0.1470	1
	n=10	n=10	
	CMTES	WALK-12	FPI-6

Table 82: Correlation between baseline clinical measurements.

Gait and Strength Measurements

	0.753	0.753	0.748	0.575	0.527	-0.576	-0.576
Hip ISOK FL	P=0.0119	P=0.0119	P=0.0128	P=0.0821	P=0.1173	P=0.0816	P=0.0816
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.724	0.724	0.715	0.555	0.442	-0.491	-0.491
Hip ISOK EX	P=0.0179	P=0.0179	P=0.0202	P=0.0957	P=0.2004	P=0.1497	P=0.1497
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.683	0.683	0.695	0.573	0.515	-0.552	-0.552
Hip ISOM FL	P=0.0295	P=0.0295	P=0.0258	P=0.0831	P=0.1276	P=0.0984	P=0.0984
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.651	0.65	0.647	0.506	0.442	-0.491	-0.491
Hip ISOM EX	P=0.0416	P=0.0417		P=0.1354	P=0.2004	P=0.1497	P=0.1497
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.728	0.728	0.718	0.543	0.479	-0.527	-0.527
Knee ISOK FL	P=0.0170	P=0.0171	P=0.0193	P=0.1049	P=0.1615	P=0.1173	P=0.1173
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.724	0.723	0.697	0.489	0.43	-0.467	-0.467
Knee ISOK EX	P=0.0180	P=0.0181	P=0.0251	P=0.1519	P=0.2145	P=0.1739	P=0.1739
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.742	0.742	0.765	0.628	0.515	-0.564	-0.564
Knee ISOM FL	P=0.0139	P=0.0139	P=0.0100	P=0.0518	P=0.1276	P=0.0897	P=0.0897
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.74	0.74	0.731	0.549	0.515	-0.564	-0.564
Knee ISOM EX	P=0.0145	P=0.0145	P=0.0164	P=0.0999	P=0.1276	P=0.0897	P=0.0897
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.559	0.559	0.65	0.176	0.176	-0.274	-0.274
Ankle ISOK PF	P=0.0928	P=0.0928	P=0.0417	P=0.6261	P=0.6261	P=0.4444	P=0.4444
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.486	0.486	0.521	0.535	0.285	-0.382	-0.382
Ankle ISOK DF	P=0.1540	P=0.1545	P=0.1223	P=0.1112	P=0.4250	P=0.2763	P=0.2763
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	0.62	0.62	0.645	0.568	0.479	-0.503	-0.503
Ankle ISOM PF	P=0.0560	P=0.0559	P=0.0439	P=0.0865	P=0.1615	P=0.1383	P=0.1383
	n=10	n=10	n=10	n=10	n=10	n=10	n=10
And ISOM D	0.818	0.818	0.794	0.176	0.176	-0.261	-0.261
Ankie ISOM D	P=0.0038	P=0.0038	P=0.0061	P=0.6272	P=0.6272	P=0.4671	P=0.4671
Ľ	n=10	n=10	n=10	n=10	n=10	n=10	n=10
	Step Length	Stride Length	Speed	Steps Per Minute	Strides Per Minute	Stride Time	Step Time
			1 () -1			

Table 83: Correlation between baseline strength measurements and spatiotemporal parameters.

Appendix IV: Exercise Induced Pain Management

Pain levels can be an indication of musculoskeletal injury or working at an intensity that is too high for the individual's ability and level of fitness. The perception of pain is a very personal thing and that the visual analogue scale (VAS) rating is very subjective. Participants were given the relevant information and support to help them take the decision whether or not to stop rather than the researcher taking the unsolicited decision for them based on a formal criterion. It is, however, important to distinguish between a temporary increase in background pain levels during training versus new musculoskeletal pain signifying injury. The following steps were taken to ensure these points are clear to participants from the outset:

- 1. VAS level of pain was explored at the consent meeting and some discussion had about how it will be managed when doing exercise.
- 2. VAS levels of 4 and above when doing exercise triggered a discussion with the trial physiotherapist around the participant's pain levels and if they feel it is appropriate to continue.
- 3. All means of support or modification (e.g. alternative position for the equipment to see if it reduces strain and makes them more comfortable) will be explored and implemented if it helps to reduce pain levels whilst maintaining the ability to train.

- 4. Anyone reported VAS level 7 or above during training were informed that this was set as a cut-off point because we were concerned that pain levels as high as this might indicate that damage is occurring. However, if there is no evidence that this is the case and so, after discussion with their therapist, we let the participants decide if they wish to continue.
- 5. The one exception to point 4 would be where pain greater than 7/10 continues for more than a couple of hours afterwards (most people with higher levels report it decreases as soon as they stop exercising) or if it is a new, acute onset of a different pain than what they normally experience. Either of these events might indicate injury to musculoskeletal structures and would prompt a full assessment by an experienced trial physiotherapist. If no significant injury is found, the training will be stopped for two weeks and then gradually restarted starting at two-thirds of the intensity first started. Exercises were definitely stopped if musculoskeletal pain occurs again after restarting the regime (challenge/rechallenge principle).
- 6. Finally, our management of delayed onset muscle soreness (DOMS) will remain the same i.e. we expect some participants to report mild to moderate DOMS, this is a normal response to new exercise, but this should not last longer than 48 hours and should not be severe or debilitating (they should be able to carry out their normal activities).

If pain cannot be explained by DOMS, the research physiotherapist will perform a full neuro-musculoskeletal assessment. If significant musculoskeletal injury was found on assessment the participant is removed from the trial and referred to their GP or local accident and emergency department for appropriate treatment. The methods and timing for assessing, recording, and analysing safety parameters were then followed by the research physiotherapist.

Appendix V: Exercise Program

PT PHYSIOTOOLS Personal exercise program MRI and Gait in dHMN

The National Hospital for Neurology & Neurosurgery The National Hospital for Neurology & Neurosurgery Queen Square, WC1N 3BG, London, United Kingdom

Provided by Aljwhara Alangary

Warm Up



©Physiotools

Use cardiovascular and body exercises in the following order for warm up:

1) Cardiovascular exercises: 5 to 10 minutes with moderate intensity!

2) Flexibility exercises: Start stretching exercises by carefully entering the stretch position and pause at a moderate stretch intensity for about 4 breaths. Relax all muscles as much as you can and concentrate on the muscle being stretched. Carefully increase the stretch while relaxing all noninvolved muscles.

3) Resistance exercises: Always perform last! Try these exercises at least every other day.

Start light with the resistance and you can gradually increase the resistance as it gets easier.

Start with 2 sets of 8 repetitions (with a 1-2 minute rest between sets) and gradually increase the number of repetitions by 1-2 per week until you can do 3 sets of 15. If you are increasing the resistance (Weight), drop back to 2 sets of 8 and gradually increase again.







Sit on side of bench with on leg bent and one straight.

Lean forwards from pelvis to feel stretch at back of thigh. Hold for 10 sec.

Stand in a walking position with the leg to be stretched straight behind you and the other leg bent in front of you. Take support from a wall or chair.

Lean your body forwards and down until you feel the stretching in the calf of

secs. - relax. Stretch the other leg.

Repeat for the other side

©Phy 5



Sit. Put a weight on top of your foot.

the straight leg. Hold approx. 10

Bend the ankle up and down.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)

CPhysioTools

6



Put one knee on a chair. Ankle weight is wrapped around the foot.

Bend the foot back.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)

CPhysioTools

7



Sit up straight on a chair, with a weight attached to your ankle.

Fully straighten your knee. Return back to the starting position in a controlled manner.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)

CPhysiotools



Start by lying on your front with a weight around your ankle.

Tighten your buttocks, keep your hips straight. Try to bend the knee and bring your foot towards the buttocks. Return back to the starting position in a controlled manner.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)







CPhysioTools



Stand up with a weight around your ankle. Support yourself on a chair.

Tighten your buttocks, keep your hips straight. Try to bend the knee and bring your foot towards the buttocks. Return back to the starting position in a controlled manner.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)

Start by lying on your back.

Bend your knee and bring it towards your chest. Lower the leg to the starting position in a controlled manner.

Note: Keep your pelvis level and maintain neutral spine.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)

Lie face down on a table, with hips over the edge and toes on the floor. Ankle weight attached to one ankle.

Straighten your hip. Focus on using your buttocks and control the neutral position of your lower back. Return to the starting position.

2 sets of 8 repetitions (with a 1-2 minute rest between sets)



©Physiotools

Cool Down

Use cardiovascular and flexibility exercises for Cool down!

The main purpose of the cool down is to relax the increased muscle tension caused by resistance training, lower the heart rate, and reduce metabolic fatigue

Cardiovascular exercises: 5 minutes at a very low intensity level!

Flexibility exercises: Use a very comfortable position with a soft stretch. Concentrate on long exhalations and completely relax while stretching your muscles!



Dynamic Anterior and Posterior Chain Stretch

Start in half-kneeling. Bend yourself over the front leg and take support from the floor.

Shift your weight backwards and let the leg in front straighten. Feel the stretch at the back of the thigh and knee. Then shift your weight forwards, so that the hip of the rear leg straightens. Push your pelvis towards the floor and feel the stretch at the front of your hip. Continue shifting your weight back and forth.

Repeat 3 times for each side.



Pull the ankle towards your bottom, then try to straighten the knee approx. 15 secs. while resisting with your hand. Relax your leg and repeat

the exercise pulling the ankle up a bit further. Return to starting position.

Repeat 3 times for each side.

Stand in a walking position with the leg to be stretched straight behind you and the other leg bent in front of you. Take support from a wall or chair.

Lean your body forwards and down until you feel the stretching in the calf of the straight leq. Hold approx. 15 secs. - relax. Stretch the other leq.

Repeat 3 times for each side.



KINEMATICS	BASE	LINE	6 MO	NTHS	12 MO	NTHS
KINEMATICS	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)
Pelvis X Max	4.17 (1.06)	4.14 (0.72)	4.84 (1.83)	4.01 (1.32)	5.16 (2.49)	4.19 (0.92)
Pelvis X Min	-4.41 (1.39)	-4.32 (1.05)	-5.3 (2.44)	-4.01 (1.45)	-5.75 (2.24)	-4.39 (0.96)
Pelvis Y Max	16.23 (6.26)	14.36 (6.05)	15.6 (6.28)	16.64 (3.51)	19.61 (4.61)	15.83 (4.52)
Pelvis Y Min	11.71 (4.46)	9.18 (4.99)	10.0 (4.94)	11.41 (3.04)	13.23 (2.82)	11.14 (3.67)
Pelvis Z Max	7.19 (3.33)	6.51 (2.44)	7.90 (4.61)	6.46 (1.76)	9.20 (4.71)	6.23 (3.55)
Pelvis Z Min	-7.66 (3.99)	-6.37 (2.67)	-8.2 (4.52)	-5.64 (1.40)	-8.04 (4.56)	-7.21 (2.11)
Hip Max	49.27 (4.79)	44.59 (5.85)	46.1 (9.31)	46.72 (2.02)	51.39 (4.76)	47.51 (3.98)
Hip Min	2.94 (4.27)	4.26 (4.85)	-2.0 (4.61)	3.70 (4.35)	0.36 (3.08)	4.86 (2.58)
Knee Max	75.08 (4.09)	72.20 (5.22)	73.6 (3.47)	75.13 (3.34)	73.37 (3.86)	74.62 (3.84)
Knee Min	5.45 (5.58)	8.10 (5.15)	2.55 (3.92)	8.43 (5.03)	0.51 (5.61)	9.77 (5.76)
Ankle Max	26.86 (4.78)	22.76 (6.64)	24.9 (6.08)	25.51 (3.29)	27.24 (7.24)	23.64 (3.26)
Ankle Min	-10.26 (2.95)	-11.83 (4.87)	-9.6 (4.82)	-12.33 (7.47)	-9.46 (3.45)	-14.08 (6.81)
KINEMATICS	BASELINE-	6 MONTHS	6MONTHS-	12MONTHS	BASELINE-	12MONTHS
DIFFERENCE	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)
Pelvis X Max	1.5 (3.42)	-0.13 (0.7)	0.33 (0.94)	0.17 (1.03)	1.83 (4.24)	0.05 (0.84)
Pelvis X Min	-1.78 (4)	0.31 (1.3)	-0.45 (0.93)	-0.39 (1.39)	-2.22 (3.57)	-0.07 (0.34)
Pelvis Y Max	2.63 (12)	2.28 (3.6)	4.00 (5.83)	-0.82 (3.8)	6.63 (12.92)	1.46 (5.07)
Pelvis Y Min	0.66 (9)	2.23 (3.6)	3.20 (5.40)	-0.27 (4.17)	3.86 (8.99)	1.96 (4.61)
Pelvis Z Max	2.1 (2.95)	-0.05 (1.7)	1.31 (2.89)	-0.23 (2.69)	3.45 (5.26)	-0.28 (1.80)
Pelvis Z Min	-2.08 (3)	0.72 (1.6)	0.17 (0.87)	-1.57 (1.37)	-1.91 (3.81)	-0.84 (1.55)
Hip Max	6.73 (27)	2.12 (5.7)	5.24 (6.99)	0.8 (5.31)	11.98 (26.81)	2.92 (5.54)
Hip Min	-4. (5.78)	-0.56 (5.9)	2.46 (6.95)	1.17 (4.67)	-1.99 (6.16)	0.60 (6.74)
Knee Max	13.53 (36)	2.93 (4.7)	-0.22 (2.05)	-0.51(3.51)	13.31 (37.12)	2.41 (4.49)
Knee Min	-1.82 (3)	0.33 (2.6)	-2.03 (5.63)	1.34 (1.31)	-3.85 (7.26)	1.67 (3.52)
Ankle Max	3.45 (14)	2.75 (4.8)	2.30 (5.26)	-1.87 (1.5)	5.75 (11.85)	0.88 (3.96)
Ankle Min	-1. (2.03)	-0.50 (5.2)	0.21 (2.82)	-1.75 (5.99)	-1.25 (2.89)	-2.25 (2.54)
Pelvic z Maximu	m= Rotation forwards, Pelvic	z Minimum= Rotation backw	vords, Pelvic x Maximum= lat	eral raise, Pelvic x Minimum=	= lateral drop, Pelvic y Maxim	um= Anterior tilt, Pelvic y
Minimum = Post	erior tilt, Hip y Maximum= F	lexion, Hip y Minimum= Exte	ension, Knee y Maximum= Flo	exion, Knee y Minimum= Ext	ension, Ankle y Maximum= D	Oorsiflexion, Ankle y

Appendix VI : Kinematic and Kinetic Gait Measurements of the Exercising Participants

 Minimum = Posterior tilt, Hip y Maximum= Flexion, Hip y Minimum= Extension, Knee y Maximum= Flexion, Knee y Minimum= Extension, Ankle y Maximum= Dorsiflexion, Ankle y Minimum= Plantarflexion, Data presented as: Mean (standers deviation).

 Table 84: Longitudinal kinematic gait parameters and differences between measurement points in exercise group (n=5) and no exercise group

(*n*=5).

	BASELINE				6 MONTHS				12 MONTHS			
VINEMATICS			Mean No	SD No			Mean No	SD No			Mean No	SD No
KINEMATICS	Case 1	Case 2	Exercise	Exercise	Case 1	Case 2	Exercise	Exercise	Case 1	Case 2	Exercise	Exercise
			Group	Group			Group	Group			Group	Group
Pelvis X Max		4.72	4.08	0.85	7.52	5.64	3.89	1.11	9.12	6.13	3.94	0.79
Pelvis X Min		-5.18	-4.26	1.16	-8.71	-6.80	-3.88	1.22	-7.96	-8.34	-4.30	0.82
Pelvis Y Max		10.73	15.75	5.97	21.47	19.94	14.99	4.72	27.81	16.72	16.58	3.58
Pelvis Y Min		6.98	10.72	4.80	13.60	15.16	9.80	3.87	17.25	11.97	11.58	3.11
Pelvis Z Max		4.97	7.04	2.78	6.19	4.80	7.60	3.67	12.00	7.34	7.23	4.49
Pelvis Z Min		-3.87	-7.33	3.15	-7.08	-5.33	-7.11	3.84	-7.82	-5.79	-7.83	3.76
Hip Max		45.58	46.81	6.01	53.21	55.22	44.49	5.48	58.91	52.93	47.84	3.15
Hip Min		-0.38	4.18	4.38	0.19	2.94	0.61	5.84	3.08	-2.37	3.18	3.54
Knee Max		75.44	73.24	4.97	78.68	73.42	73.94	3.33	79.51	69.82	73.83	3.26
Knee Min		10.58	6.47	5.35	2.81	8.29	5.47	5.78	1.59	6.51	5.41	8.09
Ankle Max		28.56	24.09	6.15	29.47	26.55	24.52	4.90	25.23	36.60	24.07	4.55
Ankle Min		-6.67	-11.69	3.86	-3.73	-7.54	-12.35	6.03	-5.62	-5.82	-13.29	5.28
	BASELINE-6 MONTHS				6MONTHS-12MONTHS				BASELINE-12MONTHS			
KINEMATICS			Mean No	SD No			Mean No	SD No			Mean No	SD No
DIFFERENCE	Case 1	Case 2	Exercise	Exercise	Case 1	Case 2	Exercise	Exercise	Case 1	Case 2	Exercise	Exercise
			Group	Group			Group	Group			Group	Group
Pelvis X Max	7.52	0.92	-0.20	0.49	1.60	0.48	0.05	0.78	9.12	1.40	-0.14	0.99
Pelvis X Min	-8.71	-1.62	0.38	0.65	0.75	-1.53	-0.42	0.61	-7.96	-3.16	-0.05	0.71
Pelvis Y Max	21.47	9.21	-0.77	5.46	6.34	-3.22	1.60	5.82	27.81	5.99	0.83	5.32
Pelvis Y Min	13.60	8.18	-0.92	5.51	3.65	-3.19	1.78	5.52	17.25	4.99	0.86	4.83
Pelvis Z Max	6.19	-0.18	0.56	1.97	5.81	2.55	-0.37	2.16	12.00	2.37	0.19	2.33
Pelvis Z Min	-7.08	-1.45	0.22	1.88	-0.73	-0.46	-0.72	1.62	-7.82	-1.92	-0.50	1.93
Hip Max	53.21	9.64	-2.32	8.21	5.70	-2.29	3.35	7.32	58.91	7.35	1.03	5.89
Hip Min	0.19	3.32	-3.57	6.57	2.89	-5.32	2.57	6.07	3.08	-2.00	-1.00	7.29
Knee Max	78.68	-2.02	0.70	5.17	0.82	-3.60	-0.11	1.61	79.51	-5.61	0.59	4.97
Knee Min	2.81	-2.29	-0.99	3.34	-1.22	-1.78	-0.06	5.13	1.59	-4.07	-1.05	7.31
Ankle Max	29.47	-2.01	0.44	5.91	-4.23	10.06	-0.46	2.67	25.23	8.05	-0.02	3.88
Ankle Min	-3.73	-0.87	-0.66	4.01	-1.89	1.72	-0.94	4.64	-5.62	0.85	-1.60	2.41
Pelvic z Maximum= Rotation forwards, Pelvic z Minimum= Rotation backwords, Pelvic x Maximum= lateral raise, Pelvic x Minimum= lateral drop, Pelvic y Maximum= Anterior tilt, Pelvic y												
Minimum = Post	Minimum = Posterior tilt, Hip y Maximum= Flexion, Hip y Minimum= Extension, Knee y Maximum= Flexion, Knee y Minimum= Extension, Ankle y Maximum= Dorsiflexion, Ankle y											

Minimum= Plantarflexion, No Exercise Group (n=8).

Table 85: Longitudinal kinematic gait parameters and differences between measurement points in case 1, case 2, and no exercise group.

KINETICS	BAS	ELINE	6 MO	NTHS	12 MONTHS					
KINETICS	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)				
Hip Moment Y Max	1.57 (0.63)	1.25 (0.36)	1.74 (0.72)	1.23 (0.37)	1.85 (0.88)	1.24 (0.62)				
Hip Moment Y Min	-0.51 (0.24)	-0.47 (0.14)	-0.5 (0.19)	-0.99 (1.06)	-0.69 (0.34)	-0.53 (0.18)				
Knee Moment Y Max	0.67 (0.45)	0.64 (0.23)	0.78 (0.42)	1.35 (1.67)	0.80 (0.42)	0.79 (0.24)				
Knee Moment Y Min	-0.84 (0.21)	-0.47 (0.19)	-0.8 (0.29)	-0.52 (0.22)	-0.89 (0.35)	-0.54 (0.27)				
Ankle Moment Y Max	1.09 (0.28)	0.95 (0.28)	1.13 (0.27)	0.76 (0.48)	1.13 (0.27)	0.99 (0.29)				
Ankle Moment Y Min	-0.12 (0.10)	-0.03 (0.02)	-0.1 (0.08)	-0.72 (1.55)	-0.13 (0.10)	-0.02 (0.01)				
Hip Power Max Swing	0.95 (0.32)	0.95 (0.21)	1.08 (0.46)	1.09 (0.23)	1.00 (0.30)	1.00 (0.36)				
Hip Power Min Stance	-1.21 (0.89)	-0.53 (0.32)	-1.3 (0.78)	-1.53 (1.93)	-1.99 (1.21)	-0.63 (0.43)				
Hip Power Max Stance	1.84 (0.83)	1.59 (0.53)	2.06 (1.03)	2.04 (0.83)	2.62 (1.29)	1.63 (0.76)				
Knee Power Max Swing	0.37 (0.11)	0.30 (0.08)	0.48 (0.27)	0.48 (0.41)	0.56 (0.38)	0.35 (0.13)				
Knee Power Min Stance	-1.68 (1.19)	-1.32 (0.82)	-1.7 (0.93)	-3.57 (5.32)	-2.04 (1.14)	-1.81 (1.10)				
Knee Power Max Stance	1.90 (0.93)	0.88 (0.44)	2.55 (1.56)	1.65 (1.47)	2.65 (1.70)	1.08 (0.78)				
Ankle Power Max Swing	0.05 (0.03)	0.03 (0.01)	0.08 (0.04)	0.04 (0.02)	0.13 (0.10)	0.03 (0.02)				
Ankle Power Min Stance	-1.18 (0.25)	-1.03 (0.42)	-1.2 (0.32)	-1.85 (1.84)	-1.35 (0.30)	-1.04 (0.38)				
Ankle Power Max Stance	1.29 (0.63)	0.88 (0.11)	1.17 (0.53)	2.69 (3.41)	1.22 (0.54)	1.11 (0.13)				
KINETICS DIFFERENCE	BASELINE	E-6 MONTHS	6MONTHS-	12MONTHS	BASELINE-12MONTHS					
	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)	Exercise (N=5)	No Exercise (N=5)				
Hip Moment Y Max	0.4 (1.06)	-0.03 (0.2)	0.10 (0.23)	0.01 (0.61)	0.59 (1.29)	-0.01 (0.34)				
Hip Moment Y Min	-0. (0.23)	-0.52 (1.0)	-0.10 (0.24)	0.46 (0.96)	-0.28 (0.24)	-0.06 (0.08)				
Knee Moment Y Max	0.24 (0)	0.71 (1.5)	0.02 (0.14)	-0.56 (1.58)	0.26 (0.39)	0.15 (0.24)				
Knee Moment Y Min	-0. (0.50)	-0.04 (0.1)	-0.01 (0.08)	-0.02 (0.22)	-0.22 (0.54)	-0.06 (0.19)				
Ankle Moment Y Max	0.2 (0.51)	-0.19 (0.5)	0.00 (0.09)	0.24 (0.59)	0.26 (0.55)	0.04 (0.12)				
Ankle Moment Y Min	-0. (0.06)	-0.68 (1.5)	-0.02 (0.02)	0.70 (1.54)	-0.03 (0.06)	0.01 (0.02)				
Hip Power Max Swing	0.32 (0)	0.14 (0.2)	-0.08 (0.28)	-0.09 (0.27)	0.24 (0.55)	0.05 (0.17)				
Hip Power Min Stance	-0. (0.35)	-1.00 (2.0)	-0.67 (0.88)	0.90 (1.91)	-1.02 (1.17)	-0.09 (0.31)				
Hip Power Max Stance	0.59(1)	0.46 (0.3)	0.56 (0.83)	-0.41 (0.33)	1.15 (1.55)	0.05 (0.26)				
Knee Power Max Swing	0.18 (0)	0.18 (0.4)	0.08 (0.30)	-0.14 (0.34)	0.26 (0.53)	0.04 (0.12)				
Knee Power Min Stance	-0. (0.91)	-2.26 (5.2)	-0.32 (0.43)	1.77 (5.04)	-0.70 (0.62)	-0.49 (0.80)				
Knee Power Max Stance	1.03 (2)	0.78 (1.2)	0.10 (0.58)	-0.57 (0.79)	1.13 (2.26)	0.20 (0.54)				
Ankle Power Max Swing	0.04 (0)	0.02 (0.0)	0.05 (0.12)	-0.01 (0.02)	0.09 (0.11)	0.00 (0.01)				
Ankle Power Min Stance	-0. (0.80)	-0.82 (1.9)	-0.07 (0.18)	0.80 (1.79)	-0.41 (0.67)	-0.02 (0.20)				
Ankle Power Max Stance	0.14 (0)	1.80 (3.4)	0.05 (0.16)	-1.58 (3.43)	0.19 (0.56)	0.22 (0.17)				
Hip Moments Maximum= Extension, Hip Moments Minimum= Flexion, Knee Moments Maximum= Extension, Knee Moments Minimum= Flexion, Ankle Moments Maximum= Plantarflexion, Ankle Moments										
Minimum= Dorsiflexion, Hip Power Maximum = generation in swing phase, Hip Power Minimum= absorption in stance phase, Hip Power Maximum= generation in stance phase, Knee Power Maximum= generation in										
stance phase, Knee Power Maximum= generation in swing phase, Knee Power Minimum = absorption in stance phase, Ankle Power Maximum= generation in stance phase, Ankle Power Minimum= absorption in stance										

phase, Ankle Power Maximum= generation in swing phase, Data presented as: Mean (standers deviation). Table 86: Longitudinal kinetic gait parameters and differences between measurement points in exercise group (n=5) and no exercise group

(*n*=5).

		BASI	ELINE		6 MONTHS				12 MONTHS			
KINETICS			Mean No	SD No			Mean No	SD No			Mean No	SD No
	Case 1	Case 2	Exercise	Exercise	Case 1	Case 2	Exercise	Exercise	Case 1	Case 2	Exercise	Exercise
			Group	Group			Group	Group			Group	Group
Hip Moment Y Max		2.23	1.29	0.41	2.37	2.23	1.28	0.48	2.88	2.28	1.28	0.61
Hip Moment Y Min		-0.76	-0.45	0.16	-0.54	-0.91	-0.81	0.84	-0.62	-1.01	-0.56	0.26
Knee Moment Y Max		1.21	0.58	0.26	1.00	1.18	1.06	1.34	0.89	1.22	0.73	0.32
Knee Moment Y Min		-0.94	-0.60	0.26	-1.09	-0.91	-0.62	0.30	-1.14	-0.84	-0.65	0.35
Ankle Moment Y Max		1.33	0.97	0.26	1.17	1.40	0.86	0.42	1.22	1.45	1.00	0.26
Ankle Moment Y Min		-0.26	-0.05	0.04	-0.10	-0.25	-0.47	1.22	-0.10	-0.29	-0.04	0.05
Hip Power Max Swing		1.37	0.90	0.20	1.66	1.41	0.97	0.27	1.17	1.19	0.95	0.33
Hip Power Min Stance		-1.58	-0.74	0.67	-0.91	-1.77	-1.44	1.56	-2.93	-2.82	-0.91	0.85
Hip Power Max Stance		2.39	1.61	0.63	3.10	3.02	1.80	0.80	3.72	2.84	1.84	1.05
Knee Power Max Swing		0.53	0.31	0.06	0.64	0.86	0.41	0.33	1.20	0.60	0.34	0.11
Knee Power Min Stance		-2.64	-1.33	0.91	-1.97	-2.37	-2.77	4.22	-1.71	-3.10	-1.80	1.11
Knee Power Max Stance		2.87	1.14	0.66	4.71	3.16	1.64	1.24	4.97	2.44	1.41	1.10
Ankle Power Max Swing		0.06	0.04	0.02	0.06	0.07	0.06	0.04	0.06	0.06	0.09	0.10
Ankle Power Min Stance		-1.24	-1.07	0.36	-1.75	-1.27	-1.58	1.45	-1.57	-1.57	-1.11	0.35
Ankle Power Max		1 43	1.02	0.46	0.76	1.65	2 11	2 72	0.99	1 74	1 1 1	0.35
Stance		1.15	1.02	0.10	0.70	1.65	2.11	2.72	0.77	1.7 1	1.11	0.55
	BASELINE-6 MONTHS				6MONTHS-12MONTHS					BASELINE-	12MONTHS	
KINETICS DIFFERENCE	<u> </u>	Case 2	Mean No	SD No	Case 1		Mean No	SD No	Case 1	~ •	Mean No	SD No
	Case 1		Exercise	Exercise		Case 2	Exercise	Exercise		Case 2	Exercise	Exercise
II'm Managet V Mar	2.27	0.01	Group	Group	0.51	0.04	Group	Group	2.00	0.05	Group	Group
Hip Moment Y Max	2.37	0.01	-0.01	0.14	0.51	0.04	0.00	0.24	2.88	0.05	-0.01	0.19
Hip Moment Y Min	-0.54	-0.15	-0.36	0.08	-0.08	-0.10	0.25	0.21	-0.62	-0.25	-0.11	0.15
Knee Moment Y Max	1.00	-0.03	0.47	0.11	-0.11	0.04	-0.33	0.15	0.89	0.01	0.14	0.23
Knee Moment Y Min	-1.09	0.02	-0.03	0.13	-0.04	0.08	-0.02	0.11	-1.14	0.10	-0.05	0.13
Ankle Moment Y Max	1.17	0.07	-0.11	0.09	0.05	0.05	0.13	0.09	1.22	0.12	0.03	0.14
Ankle Moment Y Min	-0.10	0.01	-0.42	0.03	0.00	-0.04	0.43	0.02	-0.10	-0.03	0.01	0.04
Hip Power Max Swing	1.66	0.04	0.08	0.21	-0.49	-0.21	-0.02	0.23	1.17	-0.18	0.06	0.13
Hip Power Min Stance	-0.91	-0.19	-0.70	0.27	-2.02	-1.05	0.53	0.30	-2.93	-1.24	-0.17	0.32
Hip Power Max Stance	3.10	0.64	0.19	0.65	0.63	-0.19	0.03	0.92	3.72	0.45	0.23	0.58
Knee Power Max Swing	0.64	0.33	0.10	0.09	0.56	-0.26	-0.07	0.06	1.20	0.07	0.03	0.10
Knee Power Min Stance	-1.97	0.28	-1.43	0.38	0.26	-0.74	0.96	0.53	-1.71	-0.46	-0.47	0.66
Knee Power Max Stance	4.71	0.30	0.50	0.30	0.26	-0.73	-0.24	0.47	4.97	-0.43	0.27	0.54
Ankle Power Max Swing	0.06	0.01	0.03	0.05	0.00	-0.01	0.03	0.10	0.06	0.00	0.05	0.10
Ankle Power Min Stance	-1.75	-0.03	-0.50	0.31	0.18	-0.29	0.47	0.24	-1.57	-0.32	-0.03	0.17
Ankle Power Max Stance	0.76	0.22	1.09	0.28	0.23	0.09	-1.00	0.10	0.99	0.31	0.09	0.32
										D1 <i>d</i>		

Hip Moments Maximum= Extension, Hip Moments Minimum= Flexion, Knee Moments Maximum= Extension, Knee Moments Minimum= Flexion, Ankle Moments Maximum= Plantarflexion, Ankle Moments Minimum= Dorsiflexion, Hip Power Maximum= generation in swing phase, Hip Power Minimum= absorption in stance phase, Hip Power Maximum= generation in stance phase, Knee Power Maximum= generation in stance phase, Knee Power Maximum= generation in stance phase, Ankle Power Maximum= generation in stance phase, No Exercise Group (n=8).

Table 87: Longitudinal kinetic gait parameters and differences between measurement points in case 1, case 2, and no exercise group.

	,	Visit1: Basel	line	V	visit 2: 6 Mo	nths	Visit 3: 12 Months			
	MRI	3D Motion Analysis	Dynamom etry	MRI	3D Motion Analysis	Dynamo metry	MRI	3D Motion Analysis	Dynamo metry	
GAIT_101	>	 ✓ 	~	✓	\checkmark	\checkmark	~	\checkmark	 	
GAIT_102	>	 ✓ 	~	 ✓ 	\checkmark	\checkmark	~	 	 	
GAIT_103	<	 ✓ 	\checkmark	 	\checkmark	\checkmark	~	 	 	
GAIT_104	<	 ✓ 	\checkmark	 	\checkmark	\checkmark	\checkmark	 	 	
GAIT_105	>	×	×	×	×	×	×	×	×	
GAIT_106	>	\checkmark	\checkmark	~	\checkmark	\checkmark	>	\checkmark	\checkmark	
GAIT_107	<	 ✓ 	\checkmark	 	 	 ✓ 	~	 	\checkmark	
GAIT_108	<	 ✓ 	\checkmark	 	 	 ✓ 	~	 	\checkmark	
GAIT_109	 	 ✓ 	\checkmark	 	 	 ✓ 	~	 	\checkmark	
GAIT_110	 	 ✓ 	\checkmark	 	 	 ✓ 	~	 	\checkmark	
GAIT_111	 	 ✓ 	\checkmark	 	 	 ✓ 	×	×	×	
GAIT_112	×	 ✓ 	\checkmark	×	×	×	×	×	×	
GAIT_201	>	\checkmark	\checkmark	×	×	×	×	×	×	
GAIT_202	<	 ✓ 	\checkmark	×	×	×	×	×	×	
GAIT_203	>	\checkmark	\checkmark	×	×	×	×	×	×	
GAIT_204	>	\checkmark	\checkmark	×	×	×	×	×	×	
GAIT_205	<	 ✓ 	\checkmark	×	×	×	×	×	×	
GAIT_206	<	 ✓ 	\checkmark	×	×	×	×	×	×	
GAIT_207	~	 ✓ 	✓	×	×	×	×	×	×	
GAIT_208	>	 ✓ 	\checkmark	×	X	X	X	X	×	
GAIT_209	\checkmark	 ✓ 	 ✓ 	×	×	×	X	×	×	
GAIT 100= I	OHMN. GAI	$\Gamma 200 = Control$	ol. \checkmark = Complete	ed. $\mathbf{X}_{= \text{Not}}$	t completed.					

Appendix VII: Participants Study Allocation and Visits Activities



Appendix VIII: Published Abstracts Related to This PhD

Gait Kinetics and Kinematics in Distal Hereditary Motor Neuropathy, Longitudinal study, (2023), PNS Abstracts 2023. J Peripher Nerv Syst, 28: S3-S254. https://doi.org/10.1111/jns.12585

Muscle Structure. Function. and Gait Patterns Distal Hereditary in Motor Neuropathy and The Effect of Carbon Fiber Ankle Foot Orthosis on Gait, (2022), PNS 2022 Abstract Supplement. Peripher Nerv J Syst, 27: S4-S183. https://doi.org/10.1111/jns.12506

Muscle Structure, Function, and Gait Patterns in Distal Hereditary Motor Neuropathy and the Effect of Carbon fiber Ankle Foot Orthosis on Gait, Authors, C. (2022) "NMSG 2022 Abstracts and Meeting Information", RRNMF Neuromuscular Journal, 3(3). doi: <u>https://doi.org/10.17161/rrnmf.v3i3.19210</u>

Effect of Distal Hereditary Motor Neuropathy on Muscle Structure, Function, and Gait Patterns: Two Case Reports, (2021), 2021 Peripheral Nerve Society virtual event. J Peripher Nerv Syst, 26: 307-440. https://doi.org/10.1111/jns.12460

Effect of Distal Hereditary Motor Neuropathy on Muscle Structure, Function, and Gait Patterns: Two Case Reports, Russo Paulk, L. (2021) "MSG 2021 Meeting Information and Abstracts ", RRNMF Neuromuscular Journal, 2(4), pp. 45–98. doi: <u>https://doi.org/10.17161/rrnmf.v2i4.15849</u>

Natural History and the Effect of Rehabilitation Interventions in Motor Neuropathy (Protocol), (2020), 2020 Peripheral Nerve Society Virtual Event. J Peripher Nerv Syst, 25: 438-576. <u>https://doi.org/10.1111/jns.12416</u>

Exploring Muscle Structure, Function, and Gait Patterns in People With Distal Hereditary Motor Neuropathy: Natural History and the Effect of Rehabilitation Interventions, Study Protocol. Barohn, R. J. (2020) "Abstracts from the 2020 Muscle Study Group Annual Scientific Meeting", RRNMF Neuromuscular Journal, 1(4), pp. 35–77. doi: <u>https://doi.org/10.17161/rrnmf.v1i4.14646</u>